

Japan's Instrumentalisation of Taiwan for Its Security and Power-Political Relations vis-à-vis the PRC in International Society

Michinori Wakamatsu

A thesis submitted in partial fulfilment of the requirements for the degree of Doctor of Philosophy

The University of Sheffield Faculty of Social Sciences School of East Asian Studies

Date: 20/10/2020

Abstract

This thesis investigates Japan's utilisation of Taiwan as a foreign policy instrument for its security vis-à-vis the PRC. It considers the non-recognition and sanctioning of Taiwan within international society and examines how Japan incrementally treats Taiwan as if it is a part of international security efforts despite its non-recognition of the latter in its efforts to navigate its One-China policy. Therefore, Japan-Taiwan relations can be regarded as largely hierarchical. Chapter 1 discusses Japan One China policy and Taiwan, and how these frameworks condition Japan's engagement with Taiwan in the context of the former's One China policy. This relationship requires the innovative theoretical approach that turns to the English School of International Relations, wherein Japan is the central protagonist in the grand narrative, which gives Japan an opportunity to be the regional accommodator of the marginal entity like Taiwan in international society. Chapter 2 then focuses on the English School of International Relations and discusses how Japan has been narrated as a good student of Eurocentric international society, while also examining how the non-sovereign political communities have been engaged by the great Eurocentric great powers. These attempts elucidate how Japan's accommodation of Taiwan into international society has not received the scholastic attention it deserves. Chapter 3 then engages with the literature on Japan–Taiwan relations, focusing on Japan's foreign policy instruments to elucidate how the nation determines the boundaries of an acceptable Japan-Taiwan security cooperation while pacifying the PRC. Here, we conceptualise how Japan socialises Taiwan through international security norms as if it is a member of the multilateral security forums without affording it formal membership. Chapter 4 then examines how Taiwan fits into the Japanese concept of CSD and assesses what Japan is willing to do

in terms of the Japan–Taiwan security cooperation while providing Taiwan with deterrence against the PRC. Chapter 5 focuses on the ICAO to assess what Japan can do to ensure Taiwan behaves as if it is a member of the ICAO in view of Japan's security. Chapter 6 then uses the Japan–Taiwan fisheries agreement to examine how Japan made Taiwan a quasi-equal stakeholder on a par with both Japan and the PRC while dividing the potential cross-strait united front against Japan. It also examines the normative dynamics behind the enhancement of conflict alleviation despite the tensions between the nations with reference to the ESIR perspectives. Taking these empirical investigations into account, the conclusion considers the importance of further research on Japan's accommodation of 'internationally ostracised entities' within international society in terms of the ESIR paradigm.

Contents

Abstract	1
Acknowledgements	7
Note on Conventions	8
Glossary of Main Terms and Abbreviations	9
Chapter 1. Introduction	13
1.1 Background and Research Questions 1.1.1 Research questions	
1.2 Japan's One-China Policy and an Unrecognised Taiwan 1.2.1 Taiwan's unrecognised status	
1.2.2 Japan's utilisation of Taiwan for its security and power politics	20
1.2.3 Taiwan as a regional tension-generator	
1.2.4 Japan's cautious approach toward Taiwan in One China Policy	
1.2.5 Taiwan as an unrecognised state: A question of diplomatic limbo	25
1.3 Rejecting Social Constructivism and Neorealism in Favour of the English School of International	l
Relations	
1.3.1 Constructivism	
1.3.2 Taiwan's identity and Japan's re-conceptualisation of Taiwan's membership in international security	
efforts	30
1.3.3 Norm constructivism in relation to Japan	
1.3.4 Japanese identity and its othering	
1.3.5 Rejection of Neorealism	44
1.4 Sources: Tensions between Formal and Informal Sources	47
1.4.1 Japanese language	
1.4.2 Rejection of the statistical data as an information source	
1.4.3 Limited use of interviews and extensive use of elite debates	53
1.4.4 Secondary sources	55
1.5 Multiple case studies	56
1.5.1 Case study on defence security	
1.5.2 Case study on aviation security	
1.5.3 Case study on maritime security	
1.4.9 Summary	63
1.5 Significance of the Study	64
1.6 The Relevance of Japan	66
Chapter 2. Literature Review on the English School and De Facto States	68
2.1 Introduction	68
2.2 What is the English School of International Polations?	69

2.2.1 The pluralist and solidarist debate	
2.3 De Facto States in International Society	81
2.3.1 The issue of legitimacy and illegitimacy of de facto states in international society	
2.3.2 Preference for territorial integrity and a bias against de facto states in international society	
2.3.3 Engagement without recognition	
2.4 How to Engage with the De Facto States	
2.4.1 Overview: Engagement with the de facto states – three approaches	
2.4.2 Active containment of the de facto states	
2.4.3 Ignoring or having a limited engagement with the de facto state: Can de facto states reach out	
international society?	
2.4.4 The EU's low-key engagement with de facto states	97
2.5 Summary	99
Chapter 3. Japan–Taiwan Relations, Japan's Foreign Policy Instruments, and Japan's Socialisatio Taiwan through International Security Norms	
· ·	
3.1 Introduction	102
3.2 The Hierarchical Relationship Between Japan and Taiwan	102
3.2.1 Japan–US alliance and Taiwan	106
3.2.2 Taiwan's attempt to upgrade its place in Japan's foreign policy initiatives	
3.2.3 Taiwan's domestic policy analysis of its relationship with Japan	
3.2.4 Vulnerable Taiwan within the context of Sino-Japanese relations	
3.2.5 The Taiwan issue in Sino-Japanese relations.	
3.2.6 Taiwan as a security asset for Japan	
3.2.7 Japan's quiet and creative diplomacy vis-à-vis its relationship with Taiwan	
3.2.8 Semi-institutionalised Japan—Taiwan proxy communication channels and Sino-Japanese grea	
power management	
3.3 Literature on Japan's Foreign Policy Instruments	126
3.3.1 Yoshida Doctrine or Abe Doctrine?	
3.3.2 Japan's utilisation of international norms and rules as foreign policy instruments	
3.3.3 Japan's non-kinetic use of the SDF for peace diplomacy	
3.3.4 Japan's instrumentalisation of the security partnership to enhance defence interoperability and normative power	ıd
3.3.5 Japan's utilisation of non-traditional security efforts in accordance with international norms.	
3.3.6 Japan's diversification of security partners	
3.3.7 Japan's pacifist identity	
3.4 Taiwan's Diplomatic Exclusion and Japan's Socialisation of Taiwan through International Security Norms	144
3.4.1 Taiwan's diplomatic limbo: its under-utilised security resources and non-accessibility to updated	
information on international security	145
3.4.2 Japan can enhance Taiwan's voluntary compliance with international security norms	
3.4.3 Japan's higher position to reward Taiwan for normative security behaviours in the hierarchic Japan-Taiwan relations	al
3.4.4 Engagement with Taiwan is a litmus test for Japan's creative diplomacy	
3.5 Summary	
Chapter 4. Collective Self-Defence	152
4.1 Introduction	
4.2.1 Japanese pacifism	
4.2.2 Consequence of the tensions between domestic law and international law	
•	
4.3. Taiwan's Ambiguous Status in the Japanese Concept of CSD	
4.3.1 Japan's less committed yet incrementally significant security policy for Taiwan	
4.3.2 Japan's non-engagement in arms sales to Taiwan	171
4.3.3 Japan's CSD in Asia Pacific and Taiwan	
4.3.4 Political risks of high-profile defence cooperation	
4.3.5 Japan remains ambiguous in its defence commitment to Taiwan while providing it with greate	
deterrence	
4.3.6 Okinotorishima	178

	4.4 Taiwan's Strategic Need to be an Asset for Japan	179
	4.4.1 Taiwan's defence contribution to Japan's security	
	4.4.2 Taiwan as an intelligence provider for Japan' military security	
	4.4.3 Taiwan as a potential regional cybersecurity contributor for Japan	
	4.4.4 Japan's renegotiation of Taiwan as a security dialogue partner	187
	4.5 Japan-Taiwan Non-Traditional Security Cooperation: Japan-Taiwan Coast Guard Relations	192
	4.5.1 Japanese constabulary asset sales to Taiwan	
	4.5.2 Potential for Japan–Taiwan coastguard relations	
	4.5.3 Concrete Japan–Taiwan coastguard relations	196
	4.5.4 Taiwan's misinterpretation of Japan's maritime security ties to the Philippines	197
	4.6 Japan-Taiwan Non-Traditional Security Cooperation in the HADR	100
	4.6.1 Taiwan's HADR capabilities and the PRC's sanctions	
	4.6.2 Japan's instrumentalisation of the HADR to enhance its security ties with Taiwan	
	4.6.3 Taiwan's HADR contribution to Japan's regional soft security leadership and soft power	201
	projection in international society	202
	4.6.4 HADR cooperation with Taiwan as a strategic shield for Japan's normative power in the Sino-U	
	rivalry	204
	4.7 Summary	205
	4.7 Summary	205
_	hapter 5. International Civil Aviation Organization	207
_	•	
	5.1 Introduction	207
	5.2 What is the ICAO and what does it mean for Taiwan?	208
	5.2.1 The main agenda of the ICAO	
	5.2.2 Taiwan as a marginalised entity in the ICAO	
	5.2.3 Japan's accommodation of Taiwan into international aviation security	
	5.2.4 Taiwan as a missing link in international aviation security and its implications for Japan's	
	security	213
	5.2.5 Taiwan as an aviation security asset for Japan's security and beyond	
	5.2 Januaria Januaria di sation of Tainan ar an Ariation Committe Accetin milation to the Sina	
	5.3 Japan's Instrumentalisation of Taiwan as an Aviation Security Asset in relation to the Sino- Japanese Balance of Power	216
	5.3.1 Risks posed by Taiwan's inability to receive aviation intelligence and information: Implications	
	for Japan	
	5.3.2 Japan's socialisation of Taiwan through international aviation security norms for Taiwan's as if	
	membership	
	5.3.3 Japan's utilisation of the ICAO's regulations to improve Taiwan's aviation security	
	contributions	219
	5.4 The implications of Japan's enhancement of aviation security ties with Taiwan in the context of the	1e
	ICAO	
	5.4.1 Japan's civil aviation security strategy and Taiwan	
	5.4.2 Japan-Taiwan aviation-related law enforcement cooperation and its geopolitical implications	
	5.4.3 Taiwan's exclusion: its negative implications for aviation security, and Taiwan's potential role in	
	aviation security for Japan	
	5.4.4 Japan's position on Taiwan's restricted international space	
	5.4.5 Japan's low-key stance vis-à-vis Taiwan in 2013 and high-profile stance vis-à-vis Taiwan in 2010	
	in the ICAO	227
	5.5 The Bilateral Japan-Taiwan Open Skies Agreement	228
	5.5.1 What is the Open Skies agreement?	
	5.5.2 The Open Skies agreement's geopolitical implication from a historical perspective	
	5.5.3 Japan-Taiwan Open Skies agreement: Geopolitical Implications for Taiwan's airspace	230
	5.5.4 Economic balance of power and Japan's accommodation of Taiwan into the regional commercia	
	aviation regime	
	5.5.5 The Open Skies agreement: Increased Japan-Taiwan tourism	
	5.5.6 The PRC's instrumentalisation of economic coercion against Japan, Taiwan, and others	
	5.5.7 The Japan-Taiwan Open Skies agreement: sustainable Japan-Taiwan tourism and commercial	
	security vis-à-vis the PRC's economic coercion	
	5.6 The PRC's Non-Normative use of the ADIZ	
	5.6.1 What is the ADIZ for?	
	5.6.2 Japan's defence minister's view of the PRC's non-normative ADIZ	
	5.6.3 The tension between the PRC's ADIZ and normative ADIZ	239

5.6.4 The PLA's coercive military aviation behaviours	
5.7 Japan–Taiwan Anti-International Terrorism and Anti-Transnational Crime Cooperation	. 247
5.7.1 Geopolitical importance of airport security	
5.7.2 Japan–Taiwan airport security cooperation	
5.7.3 The issues with airport security in Taiwan: Implications for Japan's security	. 251
5.7.4 Japan-Taiwan airport and port security intelligence sharing and multilateral efforts	
5.7.5 Travel documents, border security, and identity management	
5.7.6 The airport security cooperation between Japan and Taiwan: Implications for the contingencies	
in Taiwan and the East China Sea5.7.7 Taiwan's as if membership in international security and the hierarchical Japan—Taiwan relation	
5.8 Summary	. 260
Chapter 6. The Fisheries Agreement	. 262
6.1 Introduction	. 262
6.2 What is the Traditional Fisheries Policy in Japan? Japan's Comprehensive Security and Food	
Security Policy	. 263
6.3 Japan's Instrumentalisation of the Fisheries Agreement as a Security Asset	. 264
6.3.1 Japan's geopolitical instrumentalisation of the fisheries agreement with Taiwan vis-à-vis the	264
PRC	
6.3.3 Taiwan and the East China Sea	
6.3.4 Implications for the PRC's claims over Taiwan and the Senkaku Islands	
6.3.5 Paramilitary balance of power	
6.4 The Fisheries Agreement with Taiwan as Japan's Comprehensive Security Policy for Law Enforcement and Economic Security	. 284
6.4.1 Law enforcement	
6.4.2 Economic security for Taiwan and economic insecurity for Okinawa: Implications for power politics in Japan–Taiwan–China triangle	. 290
6.5 Normative Factors that Facilitated the Japan–Taiwan Fisheries Agreement despite the Tensions in	
the East China Sea	
6.5.1 Different historical interpretations of Japanese imperialism	
6.5.2 Cultural dynamics	
6.5.3 Democratic identity and democratic peace theory	
0.5.4 Political dimension	. 306
6.6 Summary	. 309
Chapter 7. Conclusion	. 311
7.1 Overview	. 311
7.2 Outline of the Argument	. 312
7.3 Contribution to Japanese Studies	. 315
7.4 Contribution to the ESIR Paradigm	. 319
7.5 Contribution to Taiwan Studies	. 321
7.6 Future Research on Japan's Accommodation of Internationally Isolated Actors for International Security	/ 322
Bibliography	. 325

Acknowledgements

First, I would like to express my sincerest gratitude to my supervisor, Professor Hugo Dobson, and Glenn Hook for their continuous support during my PhD research and for their patience, attention, and immense knowledge. Indeed, during the most difficult period of my PhD, Professor Dobson intervened to guide me toward a breakthrough, passionately reviewed my numerous drafts and helped me with my academic English while showing an extraordinary patience that deeply humbles me. I could not have imagined having to complete my PhD studies without his generous interventions.

I would like to thank the former minister of education and science, Hase Hiroshi, who actually took his precious time to meet me in person. He also connected me with the former state minister, Furuya Keiji, and former Vice Admiral Kaneda for their participation for e-mail interviews. I am also grateful to other officials who gave me interesting insights.

My sincere thanks also go to my family, especially Johanna Cristine Dy, Jollibee and my parents, as well as my friends, particularly Gao Bingyu, who constantly supported me with understanding and forbearance. My gratitude also extends to Lisa in the SEAS Office and numerous others.

Note on Conventions

The bibliography is presented in the Harvard referencing style in accordance with the University of Sheffield's regulations. Romanisation is used for the Japanese words and Pinyin for the Chinese words included in the bibliography. In terms of the transliteration of Japanese names, these are given in their traditional order (i.e. the surname first and the given name second). This also applies to any Taiwanese and Chinese names, except where the author of the source in question uses his or her English name without disclosing their Chinese name. British spelling is used for North American sources throughout the dissertation.

Glossary of Main Terms and Abbreviations

ACSA Acquisition and Cross Servicing Agreement

ADIZ Air Defence Identification Zone

AIM missile Sidewinder Short-range Air-to-Air Missile

APEC Asia Pacific Economic Cooperation

ASDF Air Self-Defence Force

ASEAN Association of Southeast Asian Nations

ASW Anti-Submarine Warfare

CAA Civil Aviation Authority

CCG Chinese Coast Guard

CCP Chinese Communist Party

CGA Coast Guard Administration (Taiwan)

CLB Cabinet Legislation Bureau

CSD Collective Self-Defence

CSIS Centre for Strategic and International Studies

DPJ Democratic Party of Japan

DPP Democratic Progressive Party

EEZ Exclusive Economic Zone

ES English School

ESIR English School of International Relations

EU European Union

EWA Electronic Warfare Associates, Inc

FAA Federal Aviation Administration

FIR Flight Information Region

GDP Gross Domestic Product

GHQ General Headquarter

GSDF Ground Self-Defence Force

GSOMIA General Security of Military Information Agreement

HADR Humanitarian Assistance and Disaster Relief

IATA International Air Transport Association
ICAO International Civil Aviation Organisation

ICJ International Court of Justice

IIPS International Institute for Policy Studies

IPC International Peace Cooperation act

ISR Intelligence Surveillance Reconnaissance

JCA Japan Coast Guard

JICA Japan International Cooperation Agency

KMT Chinese Nationalist Party or *Kuomintang*

LDP Liberal Democratic Party

MEP Members in European Parliament

MOD Ministry of Defence

MOFA Ministry of Foreign Affairs

MOI Ministry of Interior

MOIT Ministry of Industry and Trade

MOU Memorandum of Understanding

MSDF Maritime Self-Defence Force

NATO North Atlantic Treaty Organisation

NGO Non-Governmental Organisation

NHK Nippon Hōsō Kyōkai

NIDS National Institute of Defence Studies

ODA Official Development Assistance

PACOM United States Pacific Command

PKO Peace-Keeping Operation

PLA People's Liberation Army

PLAAF People's Liberation Army Air Force

PLAN People's Liberation Army Navy

PRC People's Republic of China

QDR Quadrennial Defence Review

R2P Responsibility to Protect

RAND Research and Development

RCEP Regional Comprehensive Economic Partnership

ReCAAP Regional Cooperation Agreement on Combating Piracy

and Armed Robbery against Ships in Asia

RIMPAC Rim of the Pacific Exercise

ROC Republic of China

ROCMND Republic of China Ministry of National Defence

RUSI Royal United Service Institute

SARs Severe Acute Respiratory Syndrome

SDF Self-Defence force

TPP Transpacific Partnership Agreement

TRA Taiwan Relations Act

UN United Nations

UNCLOS United Nations Convention on the Law of the Sea

UNCTED United Nations Security Council Counter-Terrorism

Committee

UNDOC United Nations Office on Drugs and Crime

UNPKO United Nations Peace-Keeping Operation

UNSC United National Security Council

VOA Voice of America

WHA World Health Agency

WHO World Health Organisation

WMD Weapons of Mass Destruction

WTO World Trade Organisation

Chapter 1. Introduction

1.1 Background and Research Questions

In the post-Cold War era, despite the non-visibility of Taiwan as a regional security actor, cross-strait tensions have significantly affected Japan's national security strategy, albeit with a comparative lack of domestic and international attention. In fact, from this period on, the presence of Taiwan has been a recurrent issue for Japan in terms of Sino-Japanese relations, creating certain tensions that are, in turn, affecting Japan's security. Furthermore, if Taiwan remains an internationally isolated entity, a 'missing linkage' could emerge within the international security efforts that require 'statehood' to be a member. However, if Japan's security experts give serious thought to how to socialise Taiwan as a foreign policy instrument for regional peace and stability, Taiwan could contribute to Japan's security – especially in terms of the rising threat of the PRC – albeit while remaining largely underappreciated by Japan. In other words, Taiwan, which behaves in accordance with international security norms by Japan's support, helps consolidate the normative developments for peace and stability in the regional order amid the rising threat of the PRC.

1.1.1 Research questions

This study attempts to answer the following questions. How does the *de facto* state of Taiwan fit into Japan's regional security strategy vis-a-vis the PRC? Specifically, if Japan and Taiwan cannot openly engage in security relations in high-profile manners

_

¹ During the missile crisis in the Taiwan Strait in the mid-1990s, Japan was reminded of the geopolitical implications of potential military conflict in the Taiwan Strait, which is geographically located next to Japan's southern flank. During the Chen administration from 2000–2008, the Taiwan Strait was often considered a significant source of geopolitical tensions in both Sino-US relations and Sino-Japanese relations. Currently, under the Trump administration, the issue of Taiwan has returned to the geopolitical frontline of the Sino-US tensions, which may affect Japan's security again.

in the absence of a Japanese recognition of Taiwan due to Japan's One-China policy and the PRC's political sanctions directed at Taiwan, how can Japan engage Taiwan in terms of security? Given Japan-Taiwan security relations inevitably cause geopolitical risk of tension with the PRC, what can Japan do to make it difficult for the PRC to sanction this Japan-Taiwan security ties despite Japan's One China policy? Why is Taiwan important for Japan's security in managing its competitive relations with the PRC? Given that Taiwan has geographical advantages and abundant security resources that are underutilised and underappreciated internationally due to the nation's unrecognised status within international society, how can Japan maximise these advantages and resources for its own security without causing excessive tensions with the PRC? As with other de facto states in the international community, if Taiwan is regarded as a loophole in regional security affairs, how can Japan socialise Taiwan in terms of international security norms to alleviate the issue? Finally, can Taiwan increase its contributions to Japan's regional security strategy? Through addressing these questions, this thesis attempts to shed some light on Japan's underappreciated but unique international security behaviours, which are demonstrated through its engagement with an unrecognised Taiwan amid the rise of the PRC.

1.2 Japan's One-China Policy and an Unrecognised Taiwan

In the field of studies on Japan's security policy, one of the least scrutinised aspects of the nation's changing international security role in the post-Cold War era is how it engages in its security policy on Taiwan. This is largely due to the fact that Taiwan's status remains disputed. Nonetheless, Taiwan's unrecognised status creates space for Japan to use it as part of its unique security contributions, as will be examined. In this section, we examine how Japan–Taiwan relations function at an informal level within the framework created through Japan's One-China policy. Within this framework,

Japan socialises Taiwan as a foreign policy instrument in accordance with international security norms vis-à-vis the PRC. Here, Japan influences Taiwan's 'state behaviour' by socialising Taiwan through international security norms generated based on statecentric international law and international institutions for the former's security. Japan also induces Taiwan's voluntary compliance with international rules and norms for peace and stability for the latter's security contributions to Japan via signing the relevant agreements as rewards. This Japanese utilisation of Taiwan as a foreign policy instrument plays out without the recognition of the latter's statehood, despite the enhancement of Japan's power-political relations vis-à-vis the PRC. Thus, Japan has tacitly and informally included Taiwan in its regional strategy. Provided one of Taiwan's national security agendas remains the expansion of its international space, the nation can receive rewards from Japan via the signing of agreements if its state behaviour is deemed as beneficial to Japan's security. However, in view of its One-China policy, Japan continues to deny the recognition of Taiwan's statehood. In fact, in order to avoid excessively breaching this policy, Japan enhances its informal security ties with Taiwan in purely incremental terms. Clearly then, Japan's insistence that it follows a non-strategic Taiwan policy needs to be reconsidered.

1.2.1 Taiwan's unrecognised status

According to Article 1 of the Montevideo Convention (1933), the criteria regarding what qualifies a political entity as a state are as follows. First, the entity must have a permanent population; second, it must have a defined territory; third, a centralised government must be in control; and fourth, the entity must have the capacity to enter into relations with other states. However, the fourth criterion is somewhat ambiguous since what constitutes the capacity to enter into relations with other states is difficult to define. As such, since Japan does not recognise Taiwan as a state and all the legal and

political engagements between the nations are conducted on informal or quasi-official levels, the fourth criterion here remains contested. Furthermore, the PRC's White Paper (2004) states that there is only one China and that the PRC represents all China and does not allow 'One China, One Taiwan' or 'Two-China' concepts. While the PRC is ready to engage with the Taiwan authorities led by the KMT on an equal footing in the cross-strait talks, the government in Beijing is the central government, while the authority in Taiwan remains a purely local government. According to the White Paper, the ROC ceased to exist when the PRC was established in 1949, and the ROC government that fled to Taiwan remains an illegitimate government. In addition, since the PRC does not allow the recognition of the Two-Chinas concept, and given the emergence of the so-called 'Theory of the Undetermined Status of Taiwan', Taiwan's international relations with other states are strictly limited to the informal level.

Japan also essentially follows this formula through its One-China policy established in 1972, which restricts it interactions with Taiwan to economic and cultural dimensions in order to appease the PRC. Upon Japan's normalisation of the diplomatic ties with the PRC in 1972, the Joint Communique stated the following:

[t]he Government of the People's Republic of China reiterates that Taiwan is an inalienable part of the territory of the People's Republic of China. The Government of Japan fully understands and respects this stand of the Government of the People's Republic of China, and it firmly maintains its stand under Article 8 of the Potsdam Proclamation (Joint Communique of the Government of Japan and the Government of the People's Republic of China, MOFA, 29 September 1972).

Given that the Japanese government 'fully understands and respects' the PRC's position on Taiwan, Taiwan's status remains ambiguous within the framework of Japan's One-

China policy while rendering Japan–Taiwan relations unofficial in nature. As such, Japan can continue to regard its relations with Taiwan as informal, while Taiwan has no authority to challenge this. Clearly then, Japan–Taiwan relations are largely hierarchical in nature.

Meanwhile, Cohen (1971) argued that given the US's normalisation of the ties with the PRC, the most difficult issue remained how to handle the status of Taiwan without completely disregarding US-Taiwan ties. Cohen (1971) also pointed out that Japan faced the same issue, noting that with its One-China policy, the nation simultaneously benefits from its official ties with the PRC while maintaining informal ties with Taiwan. However, on closer examination, it is clear that Japan is incrementally navigating the complexity of its One-China policy to enhance the nation's security by utilising Taiwan in acceptable ways while simultaneously appeasing the PRC. Elsewhere, Chen (2014) used the constructivist theory of argumentative persuasion to theorise how the PRC's interpretation of the One-China principle has evolved over the years along with the PRC's understanding of the Taiwan Strait issues and the evolution of identities and interests in Taiwan. Here, Chen (2014) argued that the PRC came to more tacitly accept the existence of the ROC since it has some Chinese roots, which may indicate some equality between the mainland and Taiwan and suggests a shift from the relationship between the central government that represents the whole of China and the Chiang Clique. While the PRC is highly unlikely to recognise the ROC, it continues to have international sovereignty over Taiwan while not having the capacity to exert control over Taiwan's domestic Westphalian sovereignty. Thus, there would appear to be some flexibility on the part of the PRC regarding how they are re-conceptualising the One-China principle. It would also appear that Japan has gradually increased its flexibility in navigating how it approaches its One-China policy to ensure more comprehensive

ties with Taiwan, which includes low-key yet comprehensive informal security interactions between the two nations.

If Japan frames Japan–Taiwan relations according to its One-China policy, its political decisions will advance the nations' bilateral political and security ties rather than those of Taiwan. In other words, the Taiwanese political input into Japan's security policy on Taiwan has a minimal influence, if any. In short, Japan does not treat Taiwan as an equal partner in its international security efforts within international society. Precisely because Taiwan is a non-recognised entity that has no independent foreign policy leverage vis-à-vis Taiwan's relations with Japan, Japan holds unique opportunities to exploit Taiwan by using it as a foreign policy instrument for its own security. Regarding this point, Wilkins (2012) argued that the combination of Taiwan's anomalous status and Sino-US relations restricts Japan-Taiwan relations to informal ties that are conducted within the framework of economic and cultural ties. Here, Wilkins (2012) advocated for low-key security relations between Japan and Taiwan in the nontraditional security sphere. Nonetheless, while Taiwan does not enjoy international recognition, meaning it is often neglected and forgotten, it remains a regional economic and military power and is a stable democratic entity to be reckoned with. Indeed, the sheer amount of security resources Taiwan offers are deemed to be beneficial for Japan's security – provided the nation is appropriately accommodated – especially visà-vis Japan's complicated relations with the PRC.

However, Taiwan is not a state since it lacks international recognition, and the nation continues to suffer from the PRC's political sanctions. Indeed, the nation has long been expelled by the UN and the relevant agencies as well as by any official multilateral

security forums that require 'statehood' for membership. Here, Munro (1994) argued that Taiwan's aspiration for some form of representation in the UN in the early 1990s was a product of the nation's democratic development. However, the PRC increasingly gained a geopolitical advantage to block Taiwan's formal international activities. Eventually, it is up to Beijing to decide if Taiwan can be represented in the UN at any level other than under the One-China policy, effectively subordinating the nation to the PRC. Thus, even an incremental inclusion of Taiwan within the UN is not realistically achievable. This aspect of the One-China policy, which relates to Taiwan's international status vis-à-vis the international organisations that require its members to be nation states, is the most relevant aspect for this research since it is pertinent to how Japan socialises Taiwan through international security norms in the international security efforts that lie outside of the international organisations. For his part, Chen (2017) pointed out Taiwan's adherence to the One-China principle, which satisfies the PRC's requirement for Taiwan's membership within the WHA and other international organisations. In fact, Chen (2017) argued that under the Tsai administration – which, unlike the Ma administration, does not follow the 92 consensus that claimed that One China means the ROC – Taiwan is unlikely to be able to satisfy the PRC's preconditions for representation in the WHA or other similar organisations. Meanwhile, the former president, Lee Teng-hui (1999), argued that while the relationship between Taiwan and mainland China has always been a divided nation, Taiwan has a right to be part of international society as a democratic member and is willing to play a role in contributing to global peace despite the intimidations imposed by the PRC. In fact, Taiwan is willing to contribute to peace and stability if the surrounding international security environment allows it to do so, which points to the restricted international space afforded to Taiwan due to the PRC's sanctions and the absence of international recognition.

1.2.2 Japan's utilisation of Taiwan for its security and power politics

How then does Japan specifically influence Taiwan's state behaviour in ways that are beneficial to its own security? In reality, even though Japan does not recognise Taiwan, the two nations can still informally discuss regional security or strategic issues in one form or another albeit in low-key manners. In addition, Japan's non-recognition of Taiwan's statehood – which renders Taiwan a 'non-state actor' – does not mean the former does not consider the latter in terms of its regional balance of power. For example, as Thomas and Williams (2017) pointed out, both Japan and Taiwan often invite MOFA or MOI officials or other proxies to engage in non-state-centric lowpolitical diplomatic discussions to influence high-political diplomacy, including through the de facto embassies that are deployed in both Tokyo and Taipei.² Notably, the fisheries negotiation also took place through such informal communication channels in the presence of MOFA and coastguard officials (Thomas and Williams, 2017). Clearly then, Taiwan deploys significant, high-profile figures who represent the central government via informal channels to communicate with Japan, which means that Japan can engage in 'unofficial' political dialogues with Taiwan provided it invests some political capital, even given its One-China policy. Certainly, as Kawashima (2013) pointed out, the signing of the fisheries agreement between Japan and Taiwan had huge strategic implications and reduced Japan's concerns over the cross-strait united front, albeit that it may have provoked the PRC. Here, it should be noted that the fisheries agreement was also signed with reference to state-centric international law, namely,

² Thomas and Williams (2017) argued that precisely because Japan appeases the PRC through its One-China policy, Taiwan must resort to informal diplomacy or para-diplomacy, which seemingly is non-state-centric in that MOFA-led non-state actors such as municipal or local entities are unimportant in its exchanges with Japan.

UNCLOS legislation, meaning Japan can indeed influence Taiwan's state behaviour in certain ways, which will be further explored in a later chapter. In other words, Japan can use state-centric international law to influence Taiwan's state behaviour in ways that are deemed beneficial to its own security, acting in accordance with international security norms without having to officially recognise Taiwan.

Furthermore, Japan has a significant influence in deciding whether or not Taiwan is indeed a tacit and informal stakeholder in the former's regional or transregional security strategy. For example, as Scott (2019) pointed out, Taiwan considers its Indo-Pacific strategy as a way to consolidate its identity as a maritime member of the Indo-Pacific region and to redirect its geopolitical focus away from the PRC. However, the nation requires support from both the US and Japan if its Indo-Pacific strategy is to be successful (Scott, 2019). Indeed, during his inauguration speech of 20 May 2020, the Taiwanese president, Tsai Ing-wen (2020), expressed the nation's strategic intensions to enhance its ties with the US, Japan and Europe, as well as with other like-minded countries, in view of increasing the nation's international space, and indicated how the nation is willing to cooperate with the regional states to enhance prosperity, peace and stability in the Indo-Pacific region. For certain, depending on who is in power, Taiwan's domestic policy vis-a-vis Japan may change; however, it is worth noting that even under the Ma administration, Japan and Taiwan signed the fisheries agreement, which again demonstrates Japan's significant influence in shaping Taiwan's regional security role, even within the constraints of its One-China policy.

Meanwhile, Samuels and Heginbotham (2018) argued that the long-held Japanese military strategy of forward defence – which involves keeping the adversaries at the

Japanese borders at bay – no longer makes strategic sense due to the rapid increase of the Chinese offensive capabilities. In fact, they argued the case for Japan's active denial to engage in protracted warfare against the Chinese military, which will likely incur unacceptable damage and prevent the adversary from gaining the decisive victory. Here, the authors note that Japan will gain resiliency by dispersion, deception and concealment by diversifying the targets the PLA must attack and through a greater use of more multiplied infrastructures, especially in terms of protracted aerial and naval warfare against the Chinese offensive. However, the authors failed to address the notion that the mere existence of Taiwan, which functions as a military buffer for Japan vis-àvis the PLA, diverts the significant Chinese military resources away from Japan to Taiwan. In fact, the White Paper published in 2020 by the Ministry of Defence in Tokyo provided an analysis dedicated to Taiwan's military capabilities and the cross-strait military balance of power, including within the context of the US's Indo-Pacific strategy (MOD White Paper 2020, pp. 86–89). While the White Paper does admit that the military balance of power is heavily tilted in favour of the PRC to the detriment of Taiwan, the MOD discusses Taiwan's self-defence capabilities, while the attendant cross-strait power politics indicate Japan's significant interest in the military role of Taiwan. Thus, the quiet alignment of Taiwan with Japan's security strategy is mitigating the crisis of the latter's vulnerability and enhancing its resiliency. Therefore, while Japan is reluctant to acknowledge Taiwan's indirect defence role due to the One-China policy, this 'non-state actor' is undoubtedly supplementing Japan's powerpolitical relations with the PRC, as will be further examined in a later chapter.

Nonetheless, Soeya (2001) argued that Japan maintained a de facto 'Two-China' policy by default in the post-war period and insisted that the nation lost any legitimate claim

to be a strategic actor, including on the issue of China–Taiwan relations, due to its history of imperialism. In fact, the author even argued that the Taiwan factor has not convinced Japan about the explicit function of the Japan–US alliance in relation to China. However, as will be demonstrated in this dissertation, Japan is clearly instrumentalising Taiwan for its power relations with the PRC within the context of the One-China policy, regardless of the historical origins of its policy on its relationships with China and Taiwan. Furthermore, Japan's incremental and somewhat surreptitious accommodation of Taiwan within the regional security environment differs from how the US utilises Taiwan as a diplomatic card in a more high-profile manner in Sino-US confrontations. As such, Soeya's (2001, 2011) insistence on Japan's non-strategic Taiwan policy and his view that Japan continues to follow the Japan–US alliance as a middle power requires more nuanced analysis.

1.2.3 Taiwan as a regional tension-generator

Meanwhile, it is true to say that Taiwan's disputed status generates tensions within the context of the Sino-US rivalry and Sino-Japanese relations. Swine (2004) pointed out that the Chen administration's actions — which included holding the defensive referendum in 2004, where the Taiwanese people were asked how they viewed the PRC's missiles and the use of force against Taiwan — triggered significant anxiety within the PRC over Taiwan's status. The US is obliged to provide sufficient support to Taiwan vis-à-vis the PRC while simultaneously reassuring the PRC that it does not support Taiwan's independence (Swine, 2004). Thus, the One-China policy does not contradict the US's moral and strategic obligation regarding Taiwan since the cross-strait war is causing devastation to the Taiwanese people that far outweighs the

importance of maintaining the status quo (Swine, 2004). Furthermore, as Lieberthal (2005) noted, while the fundamental baseline for Taiwan is to ensure that the PRC does not use force, for the PRC, the priority is to make sure that Taiwan is not recognised as a sovereign state, while for the US, it remains maintaining the peace and stability in the Taiwan Strait. In fact, Lieberthal (2005) goes as far as suggesting that, for the PRC, the existence of the Anti-Secession Law correlates with how the US considers the Taiwan Relations Act as a geopolitical means of reducing the risk of Taiwan independence, with the law largely aimed at increasing the PRC's capacity to manage the Taiwan Issue. In short, the PRC would never accept the international recognition of Taiwan nor the latter's sense of a separate 'statehood' or its declaration of independence. Here, Ross (2006) argued that Taiwan's diminishing independence movement reduced the unnecessary risk of a major war between the US and the PRC and noted that while the way that the PRC treats Taiwan in military terms is an important indicator of how the nation treats other states in the region, there is little regional sympathy for Taiwan independence. In fact, in various nations, including Australia and South Korea, the US's commitment to the security of Taiwan is a significantly divisive political issue in alliance politics (Ross, 2006).

1.2.4 Japan's cautious approach toward Taiwan in One China Policy

Elsewhere, Hara (2015) argued that the issue of legal status vis-à-vis the Senkaku Islands, Taiwan and Okinawa is conceptualised through the shared historical context of the geopolitical tensions and the rapprochement of the US's allies and adversaries during the Cold War era. Given these geopolitical concerns, Japan cannot completely ignore the Taiwan issue and lay all the responsibility of the security of Taiwan on the US alone, however much it would like to do so. Meanwhile, in terms of Japan's relations with Taiwan, this requires an especially cautious stance by the Japanese

government due to the PRC's sensitivity regarding Japan's historical relations with Taiwan and its imperial history in China. Here, Johnson (1972) argued that the mutual negative perceptions of Japan and China in both nations are largely structural and grounded in historical events of the nineteenth and twentieth centuries that were fuelled by imperial expansion. Meanwhile, the two nations also tend to harbour mutual misunderstandings, including in terms of Japan's revived defence security and its relations with Taiwan (Johnson, 1972). While the two nations may see the short-term rapprochement, they are perhaps destined for long-term rivalry. Thus, the US–Japan approach to the Taiwan issue is conducted within the context of the One-China policy in view of minimising the risk of unnecessary conflict. This is why Japan must cautiously navigate its One-China policy, framing what the nation is willing to do and what it is not willing to do in terms of its informal security ties with Taiwan without unnecessarily antagonising the PRC. In fact, as Scott (2019) noted, Taiwan's direct involvement in the Quad³, which involves Japan, Australia, India and the US, is deemed as crossing the red line by the PRC.

1.2.5 Taiwan as an unrecognised state: A question of diplomatic limbo

Fewer and fewer countries in the world are recognising Taiwan in diplomatic terms, with only 17 states affording the nation full international status (*The New York Times*, 2019). In fact, this geopolitical context surrounding Taiwan gives it a unique status as a de facto state within international society. In other words, this status of an unrecognised actor is keeping Taiwan in a state of diplomatic limbo, a nation deprived of the privileges afforded to recognised sovereign states. Japan's position is stronger,

_

³ The emerging informal Indo-Pacific security groups that consists of Japan, Australia, India and the US.

and its One-China policy means that Taiwan's interactions with Japan are determined under the politico-legal framework set out by the latter. In short, Japan's political interest lies in balancing its ties with both the PRC and Taiwan, and within this context, Japan can semi-legitimise or even delegitimise Taiwan if it wishes to do so.

As such, a realist approach does not apply since Taiwan is not a state, while it is an entity that enhances the balance of power in Sino-Japanese relations. Moreover, a social constructivist approach cannot be applied since the existing identity-based research does not sufficiently explain how Japan utilises Taiwan as a foreign policy instrument for Japan's security vis-à-vis the PRC. Therefore, the research must turn to an alternative theoretical framework that allows for a conceptualisation of Japan's instrumentalisation of Taiwan as such an instrument. This will be examined in the next section.

1.3 Rejecting Social Constructivism and Neorealism in Favour of the English School of International Relations

This section attempts to elucidate why this study rejected a constructivist and Neorealist approaches in favour of the approach devised by the English School of International Relations (ESIR). As noted above, identity-based research cannot effectively conceptualise how and why Japan instrumentalises Taiwan as a foreign policy tool for its security vis-à-vis the PRC. Also, since Neorealism understates the roles of international norms and non-state actors like Taiwan, and because of the excessive emphasis on the role of the military in Neorealist debates on Taiwan, a Neorealist

approach is not applicable in this research. In addition, since an unrecognised Taiwan lacks 'membership' in international security efforts that require statehood, we must conceptualise how Taiwan's membership can be accommodated within the international society of states. This aspect can be more appropriately theorised using the ESIR approach, which will be examined in Chapter 2. First, we must examine the constructivist approach more closely in relation to Taiwan's identity and Japan's reconceptualisation of the former's place in international security efforts. In the process, we look at the two major strands in the constructivist literature related to Japan: the norm constructivist approach and the approach that focuses on Japan's identity and its so-called 'otherings'. The general discussion of constructivist literature effectively renders Taiwan's norms and perspectives non-applicable for scholastic examinations related to Japan's foreign policy. Reviewing the constructivist literature on Japan will help to elucidate how the nation's norms of domestic pacifism continue to define the acceptable behaviours for its defence security policy and allow the nation to evolve in an incrementalistic manner. This school of thought also argues that Japan would not dramatically shift its defence security policy without significant material changes in the international system surrounding the nation. Meanwhile, investigating the other constructivism-based school of thought regarding Japan – which is exemplified in the notions of Japanese identity and its 'otherings' - helps to illuminate how Japan conceptualises the 'othering' of its regional neighbours, with this 'otherness' significantly contributing to any changes to Japan's defence security policy. The review of the two major schools of constructivist thought generates the argument that such approaches do not allow for effectively conceptualising how Japan socialises Taiwan as if it is a member of the regional security environment vis-à-vis the threat of the PRC.

Also, it will elucidate how the Neorealist approach, which restricts debates on Taiwan in the context of binary understandings between whether Taiwan will be abandoned or engaged, thus conceptualising Taiwan merely as a geopolitical pawn, is not conceptually appropriate for Japan's instrumentalization of Taiwan in international society.

Therefore, based on these above-discussed concepts, we consider a research approach involving qualitative methods, specifically case studies, for this thesis on why and how Japan utilises Taiwan as a foreign policy instrument in international society while navigating the ambiguities of its One-China policy. This thesis also discusses how the politico-legal tensions resulting from the formal/informal contradiction in Japan—Taiwan relations marked by Japan's non-recognition of Taiwan have certain implications for this research methodology. The tension related to the officiality/informality nature of Japan—Taiwan relations is blurring the line between official and informal sources, which means this research requires the extensive use of multiple sources.

1.3.1 Constructivism

Constructivists such as Wendt (1995) have argued that social factors are pertinent to international relations and that international relations are not determined by material factors alone. As Wendt (1992, 391) famously stated, '[a]narchy is what states make of it', meaning that the international structure is not given; rather, it is socially constructed. As such, the mechanistic understanding of the international system that forces state actors to behave in deterministic ways as espoused by neorealists such as Waltz (1979) is questionable. In fact, the social structure within international politics, be it

cooperative or conflictual, comprises the shared knowledge that each state possesses. In other words, for Wendt (1999), constructivism is an approach that broadly encompasses the ideational factors and the conventional methodological and epistemological study of international relations. Meanwhile, for the constructivists, the concept of security is negotiated and re-negotiated by domestic actors in the state until they determine what constitutes the most serious threat to their security.

For Wendt (1999), the state's security and identity are negotiated and renegotiated in terms of international relations. As Zalewski and Enloe stated (1995, pp. 282–283), 'identity determines how you are treated, what is expected of you, what you expect of yourself'. Therefore, constructivists such as Macdonald (2013) have argued that the function of the relationship between security and identity is to investigate why state actors act the way they do. For example, Dobson (1998) discussed Japan's UN-led peace-keeping operations (PKO) to examine the constitutive norms and regulative norms that facilitated and restrained Japan's PKO contribution to international and national security. McDonald (2013) went on to ask why the possession of nuclear weapons by the P5 states, as well as Israel, India, and Pakistan, is not regarded as a threat to international security while North Korea and Iran are regarded as grave threats to international security in terms of the role of norms, identity, and representation. Another important concept of constructivism that relates to the current research is the concept of 'othering'. The concept of securitisation requires the social construction of a specific threat that justifies 'actions outside the normal bounds of political procedure' (Buzan et al., 1998, pp. 23–24). Based on the above review of constructivism, we can now examine why we rejected a constructivist approach in favour of the ESIR approach.

1.3.2 Taiwan's identity and Japan's re-conceptualisation of Taiwan's membership in international security efforts

As discussed above, the central points in constructivism are the role of identity and the ideational factors of the state actors within international society. Simply put, the role of Taiwan's national identity and the development of ideational factors in the nation are largely nonrelevant to Japan's defence security policy, as will be examined below. Thus, the constructivist approach is not applicable to the study of Japan's utilisation of Taiwan as a foreign policy instrument vis-à-vis the PRC in international society. While the role of national identity is central in a constructivist-based analysis of international relations, the ESIR regards the role of international society as the central force. For example, Lebow (2016) argued that since the regional and international societies are 'thinner', they have less effect on the development of the multiple identifications of the state, while national identities affect the norms of regional and international society to a far greater extent. However, this constructivist conception of the role of national identities within the regional and international societies fails to take into account that the existence of an anomalous state such as Taiwan has no influence in terms of shaping the norms of international society. However, in the ESIR approach, the role of international society is more significant in affecting the role of the member states, as will be examined in the Chapter 2. If Japan has a regionally privileged status in determining which nation has the right to membership of international society within a regional context, Taiwan's membership must be accommodated by Japan. 4 This

hierarchical relationship can be more appropriately conceptualised using the ESIR approach than using the constructivist approach.

Furthermore, as Reus-Smit (2009, p. 70) noted, one of the foundations of international society is mutual recognition; thus, what is essential is who is afforded legitimate statehood and full sovereignty. While Japan has full membership within international society due to its recognised statehood and regional status of a great power, unrecognised Taiwan enjoys no such privileges. Here, Rich (2020) examined Taiwanese public perceptions of the nation's diplomatic ties with the recognised states and argued that Taiwan can provide alternative 'creative unofficial relations' for the other non-recognised states within the international community while enhancing its more important informal ties with the US and other major actors. Nonetheless, it should be stated that, realistically speaking, Taiwan's 'creative unofficial relations' cannot function without substantial political backing from major powers such as Japan. In other words, Taiwan requires the support of a regional great power such as Japan to fully realise its diplomatic endeavours within international society of states if it acts like a state-actor (Wendt, 1995).

Thus, this dissertation will examine Japan's approach to the re-conceptualisation of Taiwan's non-official membership within international security through the lens of the ESIR approach. With regard to Japan's approach to membership in terms of regional multilateralism, Hamanaka (2010) observed how the nation has considered arguably the world's most powerful state, the US, in terms of when, why and how Japan concludes that it is in its own interests to include the US in any Japan-initiated regional frameworks, or in regional frameworks initiated by other regional states. Hamanaka

(2010) used potential regional leader state theory to conceptualise how a potentially powerful state such as Japan is incentivised to include or exclude the world's most powerful state, the US. Nonetheless, Hamanaka's (2010) study fails to address how and why Japan proceeds with Taiwan's pseudo membership in the regional environment within the context of balancing its relations with the PRC and Taiwan by inducing the latter to incrementally contribute to Japan's security while continuing to appease the former in terms of the Taiwan issue. In other words, if Japan affords Taiwan explicit membership to a multilateral security forum that requires statehood for membership, this will likely be regarded as the former's recognition of the latter's statehood, which will cause tensions and potential conflict with the PRC, especially given the Anti-Secession Law (2005) that clearly runs against Japan's security enhancement. However, since Taiwan does exist as an informal state-like entity, neglecting the nation completely - especially its competent capability of contributing to Japan's physical security – amounts to a geopolitical waste and may result in Taiwan threatening Japan's physical security if it feels excessively isolated or abandoned within the context of international security efforts related to Sino-Japanese relations.

Taiwan's exclusion in terms of membership within the international society of states makes it a unique case in the international security efforts. While there is ongoing debate over why North Korea and Iran are considered to be grave threats to international security given their potential for possessing nuclear weapons while others are not (McDonald, 2013), both nations continue to be recognised as members of international society, or at least their statehood is not denied. Indeed, both North Korea and Iran are members of the UN, while Taiwan does not even have a seat, despite the fact that the nation is a liberal democratic regime based on human rights and the rule of law,

possesses a highly developed economy with its own competitive currency, and has its own substantial military. Therefore, Taiwan's social actors and norm entrepreneurs – and even the 'state behaviour based on Taiwan's domestic norms' – have minimal, if any, influence on the nation's international security efforts. Indeed, Taiwan's admiration for Japan may entertain the conservative revisionists' desire to beautify Japan's past intra-regional imperialism and colonialism since it provides an idealistic alternative to the PRC's (and South Korea's) historical demonisation of Japan.

Nonetheless, it is extremely difficult to find any linkage between how Japan develops its security policy and the evolution of Taiwan's national identity, even if the debate on Taiwan's national identity may incorporate the nation's preference for having ties with Japan rather than China, which is largely due to Taiwan's identity (He, 2014) and its admiration for the 'already modernised Japan' (Serjrup, 2012a). Meanwhile, Li and Zhang (2017) agreed with Yang's (2016) argument that Taiwanese people have not yet abandoned the ethno-cultural Chinese identity, in addition to adding that the identity convergence between the two sides of the Taiwan Strait is, if difficult, still possible if China embraces the concept of a democratic regime. Nonetheless, regardless of whether Taiwan is Taiwanese or Chinese, policymakers in Tokyo are highly unlikely to consider Taiwan's national identity in terms of its foreign policymaking, except perhaps to confirm the degree of the Taiwan–China separation. This is especially pertinent given that Taiwan remains materially powerless in the face of the regional great power rivalry between Japan and China (He, 2014). Furthermore, the PRC is unlikely to embrace the liberal democratic values and norms that are embraced by the Taiwanese (Hsueh, 2014; Link, 2015; Li and Zhang, 2017; Krejsa and Cho, 2017). Meanwhile, as He (2014) noted, Taiwan's identity politics are grounded in the concept of democracy, which

means the Taiwanese government cannot drastically transform the nation's foreign policy in a way that may contradict the mainstream self-image of Taiwan. Thus, the strong Taiwanese identity based on liberal democratic values is reinforcing the cross-strait political separation. This benefits Japan's regional maritime balance of power in Japan–China–Taiwan relations. However, merely studying Taiwan's national identity cannot help us elucidate why and how Japan has been instrumentalising Taiwan as a foreign policy instrument to enhance its security vis-à-vis the rise of China since Japan's foreign policy decision-making process is completely out of the control of Taiwan, irrespective of the national identity Taiwan adopts. Taiwan is indeed a component of the regional power balance for Japan vis-à-vis the PRC in terms of the issue of the East China Sea. Nonetheless, the more Japan engages with Taiwan as an independent actor, the more it damages the legitimate narrative of the CCP regarding Taiwan being a sacred part of China. As such, the legitimacy of the CCP is questioned, which requires a careful handling on Japan's part of its engagement with Taiwan if the nation intends to manoeuvre the ambiguity of its One-China policy.

If Japan's national security interests regarding Taiwan relate to maintaining the political separation between the PRC and Taiwan in view of the regional maritime balance of power, Taiwan's debates on national identity and security do appear to have some scholastic value for the examination of Japan's security. Wan (2007) argued that while Japan's understanding of the Taiwan issue from the 1980s to the Missile Crisis of the mid-1990s was largely political, it gradually incorporated the security dimension of its understanding of China–Taiwan tensions between the late 1990s and the 2000s. This was due to the change in the international security environment surrounding Japan, which included the change in the dynamics within the triangular relationship involving

normalising Japan, non-democratic China, and democratising Taiwan. Here, Wan (2007) pointed out that Japan has significant interest in the future cross-strait relations due to its geopolitical linkage with the East and South China Sea issues, including in terms of securing the Japanese sea lines of communication. Nonetheless, as Wan (2007) suggested, even the president of Taiwan does not have political leverage to influence Japan's Taiwan policy since Japan's realpolitik interest regarding the latter is linked to its alliance with the US, while the Japanese people remain content with the constraints imposed on Japan's foreign policy, which merely evolves incrementally. While Japan will likely continue to balance its relations with China and Taiwan, Taiwan cannot affect Japan's Taiwan policy in any sense, despite the fact that Japanese public opinion may increasingly express some enthusiasm for a more democratic and Japan-friendly Taiwan as well as some resentment toward a repressive and hostile China (Wan, 2007). In addition, the former Japanese de facto ambassador in Taiwan, Uchida Yoshihisa (2006), noted that Taiwan wants Japan to recognise its role as a military buffer for its neighbour, especially given that the missiles directed at Taiwan from southern China can easily be redeployed against Japan and that Taiwan's defence airspace management is linked to Japan's air security.

In other words, while Japan reiterates the 'non-solution' to the issue of the Taiwan—China separation for political reasons, it has been instrumentalising Taiwan as a security maximisation asset via imparting international security norms generated in international society in which Taiwan does not have a membership in order to manipulate Taiwan's normative behaviours for its own physical security. Thus, Japan can enable Taiwan to behave as if it is, in fact, a member of international society within the regional setting. For these reasons, given that the ESIR places an emphasis on who has the privilege to

be a member of international society based on legitimate statehood, it holds a theoretical advantage in conceptualising how Japan utilises Taiwan as a foreign policy instrument for its own security. As will be further explained in Chapter 3, Japan, as a regional great power, has the power to grant Taiwan semi-membership to international society.

Nonetheless, a number of analysts continue to focus on the debate over domestic politics related to Taiwan's national security under the assumption that such internal debates can enhance Taiwan's future membership within multilateral security forums that require statehood for membership. For example, Babones (2016) argued that Taiwan should seek behavioural change rather than political change and should normalise its status as a de facto state without declaring its independence if it wishes to be taken more seriously within the international community. Elsewhere, Greer (2019) claimed that there are reasons why Taiwan prioritises the purchase of high-end military assets from the US and attempts to develop high-profile indigenous defence equipment rather than use cheap, expendable and mass-producible anti-access assets. This includes the fact that Taiwan believes there is greater political value in its military relations with the US and the fact that raising the morale among the Taiwanese population in terms of its military power is more politically important in symbolic terms within the context of hyper-partisan Taiwanese domestic politics than the nation's actual defence needs (Greer, 2019). As such, Greer (2019) suggested that Taiwan must be invited into highprofile US-led multinational military exercises to raise the nation's morale and increase its defence capabilities. However, neither Taiwan's attempt at behavioural change nor its domestic political interest in how to construct its geopolitical role as a de facto buffer for Japan against the PRC's military pressure (Uchida, 2006) can, in any sense, affect the Japanese elites' debate over Japan's security policy, including in terms of its

conception of the CSD. Furthermore, Japan will never allow Taiwan to play an 'official' role in the US-led mini-lateralism, in which Japan is undoubtedly the linchpin. Clearly, it is up to Japan's political decision-makers to define the acceptable boundaries of Japan—Taiwan security ties within the context of its One-China policy. In short, what is debated in Taiwan has no place in Japan's security policy on Taiwan.

1.3.3 Norm constructivism in relation to Japan

Here, we must clarify why norm constructivism does not allow for theorising how Japan utilises Taiwan as a foreign policy instrument for its security vis-à-vis the PRC. First, in the norm constructivist school of thought, Japan is largely described as having strong pacifist domestic norms that strictly regulate the nation's evolution of its strategic security policy in an incrementalistic manner. Regarding this point, Berger (2011) argued that the military-political culture is influenced by the domestic norms, values and beliefs, and that each country has a different reaction to the same event due to its specific military political culture, which is manifested in varying degrees of military power and the use of force. From a constructivist point of view, all the states seek legitimation from both domestic and international audiences in order to mobilise support for the policy in question (Goddard and Krebs, 2018). State behaviour that does not resonate with the dominant domestic norms soon loses its domestic support, while even if the behaviour does resonate with the domestic norm, if it deviates from the international norm, it is rendered illegitimate within international society. Norm constructivists generally agree that Japan has been and will continue to be normalising, even though the resiliency of Japan's pacifist identity restricts, conditions and retains the evolution of the nation's defence security policy on an incremental basis. However,

Kallender and Hughes (2019) argued that Japan is heading toward a radical shift in its defence security policy, while also conceding that the nation has undergone some normalisation in recent years in terms of having more self-defence capabilities and accepting a greater defence security role both independently and within the context of the Japan–US alliance. Therefore, it can be argued that the domestic pacifist norms in Japan are no longer considered as a constraint on Japan's defence security policy. Nonetheless, this remains largely nonrelevant to Japan's security policy on Taiwan since, even if it has the capability, it simply wants to use Taiwan as a non-traditional security instrument and as a more useful geopolitical buffer against the PRC within the context of the One-China policy.

On reviewing the literature on norm constructivism in relation to Japan, it is clear that the nation seeks the legitimation of the evolution of its security policy by way of adjusting its defence policy within the boundaries of the legal constraints and its domestic norm of pacifism. As both Hagström and Gustafsson (2015) and Oros (2015) noted, even a conservative revisionist leader such as Abe, who wishes to radically change the pacifist constitution, must adjust the evolution of Japan's defence security within an acceptable framework of Japan's pacifist identity. Within the international arena, Funabashi (1991/1992) advocated for Japan's identity transformation into a 'global civilian power' following the nation's traumatic experience of the Gulf War. Elsewhere, Easley (2017) argued that the future trajectory of Japan is "Normal Japan" which may attempt to incrementally maximise its material power, reputational status, and political influence in line with its national strength, economic scale, and the compliance with international norms within the context of the current international system. Nonetheless, Japan will likely adhere to its pacifist identity, with the pacifist

nature of its institutional and domestic constraints continuing to restrain the evolution of its security policy within the parameters of the pacifist regime (Easley, 2017). Furthermore, it should be noted that many norm constructivists (e.g. Gady, 2015b; Oros, 2015; Hornung, 2016; Easley, 2017) have agreed that, without significant material changes in the international system that will affect Japan's security, the nation is unlikely to deviate from the incrementalistic development of its defence security policy due to the constraints of the nation's pacifist identity.

Kersten (2016) argued that the inherent tension between pacifist and democratic legal norms, Japan's strategic autonomy and its dependency on its alliance with the US may cause Japan's new security identity to fluctuate. Meanwhile, Greenstein and Tensley (2016) pointed out that the major reason why the US and Japan find it difficult to construct a shared collective memory of the atrocities both nations committed during WWII – including the nuclear bombing of Hiroshima – relates to the fact that both have dual identities as victim and victimiser. Nonetheless, historical impunity generally comes at the expense of reconciliation (Greenstein and Tensley, 2016). Kersten (2016) concluded that this is where the normative role of Australia comes into play as an alleviator of the aforementioned tensions. Alongside the US, Australia is a major security partner for Japan and the historical fact that Australia was a victim of WWII gives credence to Japan's new security identity as a future-oriented nation focused on historical reconciliation and shared democratic values (Kersten, 2016). Elsewhere, Lind (2009) pointed out that both the international community and Japan should ask why Germany was able to engage in more dedicated self-reflections and sincere apologies than Japan. Here, Lind (2009) argued that Japan should seek the middle-ground, which lies in acknowledging its imperial misdeeds and the subsequent suffering of its

neighbours as victims of Japanese aggression while giving the Japanese youth ample opportunity to be proud of post-war accomplishments such as economic success, cultural creativity and the forging of a liberal democratic path, which are ensuring Japan is more future-oriented based on Adenauer's model of 1950s West Germany. This would help Japan achieve some reconciliation with its neighbours while containing the political backlash from the Japanese conservatives (Lind, 2009).

However, the following question remains: If the resilience of Japan's domestic norms of pacifism is tangible, how does this affect the nation's security policy on Taiwan? Japan is a regional great power, one that does not formally recognise Taiwan, a nation that has, at best, a minimal influence on the development of Japan's pacifist-based national identity. Furthermore, it is argued that Japan's legitimation of its security policy through harmonising its defence policy development with its domestic norms of pacifism and legal constraints cannot help us theorise Japan's incremental instrumentalisation of Taiwan vis-à-vis the PRC. This is because Japan's unofficial ties with Taiwan are rooted in the former's One-China policy, and the PRC considers Japan's adherence to the One-China principle to be an essential premise of the healthy development of Sino-Japanese relations. There is always the risk of Taiwan emboldening the Japanese conservatives' historical revisionism (He, 2014; Serjrup, 2012b), which would be detrimental to Japan's physical security if it leads to historical tensions with its neighbours. Certainly, Japan and Taiwan share certain democratic norms, and their historical ties continue to bind the two actors. However, unlike Australia, which has an unquestionable role in facilitating Japan's new future-oriented security identity (Kersten, 2016), it is unclear whether Taiwan has such a role in alleviating the aforementioned tensions, especially in terms of the Japan-led debate over

CSD. It is pointed out here that while Japan and Taiwan indeed share historical tensions, including the issue of 'comfort women', the Taiwanese authority tends to keep these non-high-profile issues out of the strategic considerations for Taiwan–Japan relations for its own survival. Furthermore, as will be examined in Chapter 3, Japan–Taiwan relations benefit from the nations' historical ties, which includes Taiwan's more unique interpretation of the Japanese colonial rule. Therefore, it remains unclear as to whether either Japan or Taiwan are interested in enhancing a high-profile historical acknowledgement of the former's aggression within the context of the bilateral Japan–Taiwan relations to push Japan forward as a future-oriented country, an approach espoused by Lind (2009). Overall then, norm constructivism does not present an appropriate theory for conceptualising how Japan utilises Taiwan as a foreign policy instrument.

1.3.4 Japanese identity and its othering

In this section, we engage with the literature related to the concept of Japanese identity and 'othering'. Here, it is argued that the identity-based research grounded in constructivism is ineffective for theorising Japan's instrumentalisation of Taiwan visà-vis the PRC, largely because we cannot find a linkage between Japanese identity and the othering of Taiwan and Japan's physical security. Buzan et al. (1998, p. 25) defined securitisation as 'constituted by the intersubjective establishment of an existential threat with a saliency sufficient to have substantial political effects'. As pointed out by Hagström and Gustafsson (2015), it would appear that numerous authors (e.g. Gustafsson, 2015; Suzuki, 2015; Hagström and Hanssen, 2015; Tamaki, 2015, Bukh, 2015; Suzuki, 2019) engage in identity-based research on Japan and its othering of its

neighbours to counter the argument that Japan's pacifist identity is fragile in the face of the material changes within the international system surrounding Japan. They clearly observed that Japan is in a normatively higher position in international society vis-à-vis Japan's Northeast Asian 'inferior others' such as China, South Korea and North Korea, while Southeast Asia is defined more as an opportunity and is thus more positively identified. In these studies, China and North and South Korea are largely depicted as 'securitised others' that demonise Japan as a historical aggressor while denying its postwar pacifist national identity. Meanwhile, Japan's othering of its neighbours as 'inferior and immoral entities' may amplify the nation's normalisation of its defence security policy even further while eroding the nation's pacifist constraints. Nonetheless, simply put, these debates cannot help us to conceptualise how Japan utilises Taiwan for its own physical security.

Jamieson (2018, p. 157) argued that in order to measure whether or not the securitisation of others is successful, the following four criteria must be met: (i) identification of the security threat; (ii) articulation of the security threat; (iii) an intersubjective agreement about the security threat; and (iv) demonstrated political effects that cannot be easily reversed. Elsewhere, Arrington and Yeo (2019) pointed out that despite the numerous pledges made by Japan and South Korea to enhance their future-oriented ties, a resurgence of the domestic norms of historical resentment toward Japan has emerged in Korea, exacerbated by Japan's mishandling of its historical contrition towards its imperial history, including the instrumentalisation of economic embargo. Here, Japan's historical tensions with its neighbours may derail the development of its new security identity as a future-oriented nation. Meanwhile, the most complex case is Taiwan (He, 2014; Serjrup, 2012b), which is othered by Japan as an inferior but positive entity and

may thus provide an alternative conception of Japan's imperial history to the demonisation presented by China and Korea. Taiwan also often 'others' Japan, albeit as a 'positive other', in ways that suit its own political needs. This bilateral Japan—Taiwan relationship reinforces Japan's position as the more civilised state while subordinating Taiwan to a position of learning from Japan's past within the hierarchical order. Furthermore, Japan requires the decision-making of strong statesmanship if it is to enhance its historical reconciliations with South Korea and China while persuading the Japanese public that these reconciliations are important to the issue of historical memory. Meanwhile, Japan's historical reconciliation with Taiwan is arguably less crucial to enhancing stable political ties. Certainly, Japan—Taiwan relations involve some degree of historical contention, but this does not affect the bilateral relations as much as in the case of South Korea and China. Nonetheless, Japan does require a different form of political decision-making to enhance its strategic ties with Taiwan, which strongly relates to manoeuvring the ambiguity of its One-China policy.

Meanwhile, as will be examined in Chapter 3, Taiwan can provide an alternative historical view of Japanese intra-regional imperialism to that offered by China and North and South Korea. Thus, the existence of Taiwan may reinforce the ontological security of the conservative revisionists in Japan. Nonetheless, it needs to be stated here that the Japanese conservative revisionist sympathy or the othering of Taiwan as an 'inferior' but 'benign' entity within Asia's hierarchical order will not, in any sense, help shape Japan's security policy on Taiwan. Here, the answer is straightforward. In short, Japan engages with Taiwan only under its One-China policy and will not enhance its defence ties with Taiwan to appease the PRC for its own security. As such, Japan-related identity-based research, including that related to the nation's othering, is not

applicable to how it instrumentalises Taiwan vis-à-vis the PRC. Thus, defining Taiwan as a modernising other in contrast to the already modernised Japan cannot help us to theorise how the former contributes to the latter's physical security. Furthermore, since Taiwan is an unrecognised state, this severely limits its foreign policy leverage vis-à-vis Japan. In short, Taiwan cannot be a security asset for Japan if it is not incrementally accommodated into the regional security environment by Japan while manoeuvring its One-China policy. Therefore, it is difficult to find a connection between notionally defining Taiwan as a 'positive other' for Japan and affording Taiwan membership to the regional security environment. Thus, the identity-based research grounded in constructivism can be rejected as an approach to theorising Japan's instrumentalisation of Taiwan vis-à-vis the PRC for its own physical security.

1.3.5 Rejection of Neorealism

Waltz (Waltz 1979, 113) argued that 'in international politics [military] force serves not only as the *ultimate ratio*, but also as the first and constant one.' Antunes and Camisão (2018) pointed out while defensive realists such as Waltz (1979) perceives states to be security maximising-oriented. In their view, states try to seek sufficient power to maintain security. Meanwhile, offensive realists such as Mearsheimer (2001) advocates for power maximising-oriented state behaviour among great powers in order to ensure their survival in anarchic system. Both defensive and offensive realists consider states to be unitary actors and give little credits to non-state actors, which is not suitable for state-like non-state actors such as Taiwan. Furthermore, as noted above, both defensive and offensive realists give excessive credits to military security, while ignoring the role of international norms based on international law, which inevitably

militarises the debates on Taiwan's international security utility. This has had outsized implications for debates on US-Taiwan relations. Certainly, Taiwan does play a role in Japan's considerations in the regional geopolitical balance of power as a de facto defence buffer for Japan against the PRC (Uchida 2006), yet Japan's security interests regarding Taiwan is not just merely limited to military security but also reaches into non-military security spheres based on international security norms, as will be examined in Chapter 3 and the three case study chapters. Then, if debates on regional security issues involving Taiwan in the field of neorealism is excessively relying upon military security, this is not conceptually appropriate for theorizing Japan-Taiwan security interactions.

For example, one factor that Japan cannot ignore is the argument for the eventual abandonment of Taiwan in the US's grand strategy debates. For examples, Gries and Wang (2019) argued that all three sides in the US-China-Taiwan relations are mired in wishful thinking. Under the Trump administration, the US prioritises America first, while China believes the US can use Taiwan as a bargaining chip and is largely unwilling to defend Taiwan, and Taiwan has convinced itself that the PRC will not attack its shores. Elsewhere, Lynch (2018) argued that, with the advent of the Taiwan Travel Act, the Trump administration may use the 'Taiwan card' as a bargaining chip in its confrontation with the PRC and urged Taiwan not to be a pawn for the Trump administration amid Sino-US tensions, especially given that Trump's domestic political base is unlikely to support Taiwan if cross-strait military conflict arises. Meanwhile, realists such as Mearsheimer (2014) posited that as the PRC becomes ever more powerful in the future, it will become increasingly more difficult for the US to commit to the security of Taiwan. Furthermore, Tang (2014) argued that the multiple factors

including the rapid military build-up of the PLA, Taiwan's domestic political wrangling that delayed the purchase of the US defence assets and reduced the US's trust on Taiwan's eagerness to defend itself, and the US's unwillingness to antagonize the PRC over the Taiwan question all contributed to the change of the cross-strait military balance of power in favour of the PRC. In addition, Taiwan's international isolation and the lack of Taiwan's strategic importance for the US make Taiwan a strategic liability (Tang 2014). Even Gilley (2010), maintained that Taiwan's 'Finlandisation' vis-à-vis the PRC would both secure Taiwan's autonomy and help the US's grand strategy. Here, Gilley (2010) argued for the gradual reduction of the US's security commitment to Taiwan amid the cross-strait détente under the former Ma administration, stating that 'Taiwan's Finlandisation should be seen not as a necessary sacrifice to a rising China but rather as an alternative strategy for pacifying China'.

Some others advocate for US's greater engagement with Taiwan including military ties. For examples, Tucker and Glaser (2011) argued the US should not abandon Taiwan but instead should strengthen ties with Taiwan including the continuous arm sales to Taiwan in order to make sure Taiwan can confidently negotiate with the Mainland China, retain the regional trust of the US among the US allies in Asia, seek for US's expanded economic ties with Taiwan, and utilize Taiwan's democratic model for the region including the PRC's autocracy. Tucker and Glaser (2011) pointed out that if Taiwan is captured by the PRC and is used as a military springboard for the PLA, the PLA can prevent Japan from securing the sea lanes of communication that connects the East and South China Seas. Meanwhile, Haass and Sacks (2020) questioned the wisdom of "strategic ambiguity" for peace and stability in the Taiwan Strait via the Taiwan Relations Act and instead argued for "strategic clarity" without the change of the US's stance on Taiwan's status. In their view (2020), US's such a move will make the PRC

think twice about resorting to the use of force against Taiwan, deters the cross-strait kinetic conflict and signals the US's credibility to defend its "regional allies" other than Taiwan.

Clearly, abandonment/engagement debates on Taiwan in neorealism neglect international security norms that Japan utilises to socialise Taiwan in international society. As will be examined in literatures on de facto states and international society in chapter 2 and literature on Japan-Taiwan relations in chapter 3, non-recognised states including Taiwan is vulnerable to international recognition/derecognition. This binary understanding leads to abandonment/engagement debates on Taiwan, which conceptualise Taiwan as a geopolitical pawn. This is especially the case if the debates on Taiwan are militarised like neorealism. Thus, this research requires an alternative theory, which can socialise Taiwan through international security norms for Japan's security as a foreign policy instrument vis-à-vis the PRC.

1.4 Sources: Tensions between Formal and Informal Sources

According to Navari (2009, p. 12), students of ESIR 'spend their time in archives getting their hands dirty', are 'immersed in diplomatic archives, memoires, and newspapers', and 'spend time in international institutions, listening to what international civil servants say and observing what they are doing'. However, while the first point raised by Navari (2009) is highly relevant to this research, his allusion to 'immersion' in certain areas does not necessarily apply to the data gathering and substantiation of the present research. Matsuda (2013) noted the opaque nature of the Japanese government's decision-making process related to the policy on Taiwan and the subsequent inaccessibility. In addition, the tension between the officiality and the

informality of sources complicate the corroboration of the findings within the context of research on relations between the sovereign state of Japan and unrecognised Taiwan (Matsuda, 2013). Certainly, merely investigating archives and surfing the internet will not enable us to verify the validity of the gathered data. However, what is unique about Japan-Taiwan relations is that Japan's official Taiwan policies often do not surface from governmental sources but from more informal ones. Grix (2019) pointed out that the over-reliance on media sources may indicate a lack of sources to back up the dissertation's scholastic argument. However, this is precisely the reason why the importance of informal sources needs to be elucidated within the context of informal Japan–Taiwan relations. As noted by Deans (2001), informal Japan–Taiwan relations are supported by informal channels formed to enhance the informal political ties between the two actors. On this point, Wilkins (2012, p. 115) pointed out that '[d]ue to the sensitivities attending Taiwan's anomalous diplomatic status' both 'official documentation of and scholarly inquiries into the bilateral relationship are relatively limited'. Indeed, this lack of official documentation and scholarly inquiries also had implications for the methodology utilised in this research. Furthermore, from this research, we can observe how informal sources are documenting, depicting, and promoting Japan's informal security engagement with Taiwan within the context of Japan's One-China policy. Thus, what appears to be informal sources and what appears to be secondary sources – such as academic sources, Japanese newspaper interviews with Taiwan's high-profile figures (e.g. the current and past presidents of Taiwan, and Taiwanese ambassadors in Tokyo), think tank reports, comments made by Japanese elite figures in the unofficial think tank diplomacy involving Taiwan - can be considered as 'official messages' in the form of 'unofficial sources'. Therefore, the fact that Japan is socialising Taiwan as a foreign policy instrument for its own security visa-vis the PRC creates a novel situation, where what appears to be secondary informal sources may actually turn out to be 'semi-official'. This complicated tension between official and unofficial relations creates the unique difficulty of confirming the official attitude of the government of Japan, which in turn makes corroborating the evidence somewhat problematic.

Thus, multiple sources were combined to create an overall picture of what is happening in Japan's security interactions with Taiwan. Due to the nature of the informal relations between Japan and Taiwan, the governmental security agencies in Japan are not willing to publish official data, reports, or other documents that are considered outside of what is acceptable in Japan's One-China policy. Here, how the current author engaged in data gathering to substantiate the creation of knowledge both in empirical and theoretical terms needs to be clarified in order to illuminate how the main argument of this study was initially enhanced before overall claims, evidence, and conclusions were formulated. The unique point in Japan-Taiwan relations relates to the absence of Japan's recognition of Taiwan's statehood, and it is often difficult for readers to discern 'official' sources from 'non-official' sources. For example, when the National Institute for Defence Studies (NIDS) published a large report on the security situation surrounding Taiwan in 2017, it raised tensions between Japan and China, and the PRC government subsequently admonished Japan; however, in the words of Monma (2017a), the report is 'academic' in nature, which indicates 'informality'. Two anonymous officials at the Ministry of Defence in Tokyo (who do not want to be identified due to the sensitive nature of the issue) 'informally' reiterated how the unofficial report of 2017 published by NIDS caused significant tensions with the PRC. Furthermore, even the high-profile fisheries agreement signed between Japan and Taiwan (2013) is

officially a 'torikime', meaning it is a private-based pact as opposed to a 'kyoutei' or 'jyouyaku' (an international agreement or treaty), thus blurring the demarcation of what is considered as official or informal.

Furthermore, the UN-based agencies do not provide official statistics on Taiwan. One may be reminded of the fact that the US State Department (2020) even made an 'official' press statement denouncing the ICAO for blocking Twitter users merely for referencing Taiwan in order to silence the 'informal' Twitter debates over Taiwan's exclusion, especially in relation to the outbreak of COVID-19. During a private interview with the right-wing Sankei Shimbun on 26 February 2020, the current deputy president of Taiwan, Chen Jianren, subsequently expressed his gratitude for Japan's political support in affording Taiwan semi-representation in WHO's technical meeting related to COVID-19 despite the PRC's ostracisation and Taiwan's lack of recognised statehood (Sankei Shimbun, 2020). Thus, it is clear that Taiwan's informal grassroots debates on Twitter can be a source of official geopolitical debate, and that the informal Japanese sources such as Sankei Shimbun can be an essential avenue for Taiwan to express its official diplomatic message to the Government of Japan. Moreover, due to the nature of the non-official bilateral relationship between Japan and Taiwan, the two actors cannot engage in the official security dialogue that may involve military affairs. Therefore, in terms of Taiwan's informal relations with Japan in the absence of the Japanese government's recognition of Taiwan's statehood, the recourse to informal avenues such as think tank diplomacy also functions as a 'quasi-official' replacement for security-based dialogue with Japan. For example, the Security and Strategy Research Institute of Japan and the Taiwan Strategy Research Association, both of which are private and informal institutes, signed the MOU cooperation agreement to

enhance the institutionalised security dialogue (Security and Strategy Institute, 2019), which is unprecedented. Clearly, the conceptual demarcation between what constitutes 'official' sources and what constitutes 'informal' sources is highly ambiguous when it comes to Japan—Taiwan security relations. Therefore, we must extensively utilise both to better elucidate the Japanese elites' perspectives of Taiwan.

Both primary and non-official sources, and a variety of other sources, were extensively used to improve the validity of the narrative on Taiwan's position as Japan's foreign policy instrument for improving its security within the context of its One-China policy. Secondary sources were also crucial to enhancing the validity of the findings since Japan's ties with Taiwan remain unofficial, meaning that official documents are not as readily available as in the case of Japan's other security partner states. However, the de facto Japanese embassy in Taipei, the former interchange association (now the Japan-Taiwan Exchange Association), publishes monthly reports and writings on various topics related to Taiwan – including culture, history, foreign policy, domestic politics, security affairs, economy and Taiwan-Japan relations – as 'non-official' texts. In addition, non-official Japanese academic sources like Toua⁵ are often used as an avenue for the former high-profile Japanese figures that were deployed in Japan-Taiwan Exchange Association to express their views on Taiwan, including Japan-Taiwan security ties. This includes the former director; whose previous role was the director of the Joint Staff Office in the MOD. The use of these sources can enhance the validity of the findings. Furthermore, the unofficial memoir of the former ambassador, Uchida (2006), was helpful in confirming the existence of intelligence-sharing in terms of situational analysis and intelligence analysis cooperation between Japan and Taiwan.

_

⁵ A Japanese Academic Publication on East Asia

Nonetheless, the extensive use of non-official texts does not mean official texts related to a Japan–Taiwan security cooperation do not exist. Indeed, records of parliamentary debates on how Taiwan fits into the Japanese concept of CSD, agreements between Japan and Taiwan in the field of non-traditional aviation and maritime security, and the high-profile quasi-official fisheries agreement do exist. These were also extensively utilised in this research for data gathering and enhancing the validity of the findings. In addition, the ICAO's official documents helped conceptualise Taiwan's 'as if membership' of the ICAO that Japan was seeking to achieve based on international aviation security norms generated in the ICAO. Both formal and informal texts on bilateral Japan–Taiwan aviation security cooperation helped confirm the validity of these concepts.

1.4.1 Japanese language

A number of officials in Taiwan, including the former Japanese de facto ambassador, Ikeda Tadashi, and several academics (e.g. Ikeda, 2010; Fukuda, 2018) have expressed their concerns regarding the lack of knowledge about Japan, including the limited linguistic abilities in Japanese among the Taiwanese elites. Due to Japanese colonial rule, the elderly generations of Taiwan may have spoken Japanese, but these generations have since passed away or are no longer active in the policy-making bodies or academia in Taiwan. When the Japanese language is lost, so is the access to broader knowledge on Japan within Taiwan. This gave the author, a native Japanese speaker, a significant advantage in accessing the broader scope of knowledge contained in the Japanese primary and secondary sources, which is simply not available to Taiwanese researchers. In addition, the author can read Chinese text, which helped in supplementing data gathering.

1.4.2 Rejection of the statistical data as an information source

The prominent international surveys conducted by UN-based security agencies do not include Taiwan due to the political consideration of the PRC's sensitivity and Taiwan's lack of internationally recognised statehood. Furthermore, due to the political inaccessibility of the Japanese government's decision-making body responsible for its Taiwan policy (Matsuda, 2013), creating a statistical survey is virtually impossible. Thus, using a quantitative method for this research was not considered. While Taiwan's governmental and private entities may conduct many interesting quantitative surveys on Taiwan's domestic political affairs, the nation's statistical input has no influence on Japan's Taiwan policy. As discussed above, Taiwan's domestic norms or identity-based debate simply do not matter to Japan's security policy on its neighbour. Meanwhile, while Taiwan's surveys on the PLA's movements around the East China Sea and the Bashi Channel may be of interest to Japan's defence establishments, they are conceptually closer to the intelligence analysis and defence analysis of situational awareness (which are herein discussed in the case studies related to CSD and the ICAO). Nonetheless, this intelligence sharing between Japan and Taiwan does not equate to a quantitative-type methodology. Therefore, a qualitative-type method was deemed to be the better option.

1.4.3 Limited use of interviews and extensive use of elite debates

Interviews can be essential to complimenting archival and less-official sources.

Nonetheless, in this research, interviews were kept to minimum and were used only as

part of a supplementary method for extracting the data or confirming the validity of the evidence. The difficulty of accessing the non-transparent decision-making body responsible for Japan's Taiwan policy (Matsuda, 2013) became a significant barrier to engaging in interview-based research. Therefore, only interviews via written correspondence with a very limited number of high-profile figures were possible. Specifically, interviews with a high-profile Japanese politician (2019) and a highprofile Japanese participant of the track 2 security dialogue between Japan and Taiwan (2019) were conducted through written correspondence to supplement the argument, if not to corroborate the detailed points raised in the case study chapters. However, as a limitation, politicians may often produce a somewhat biased account of events. Nonetheless, non-politicians, such as the former vice admiral, provided the author with a detailed account of how Taiwan cancelled the track 2 security dialogue between the Okazaki Research Centre and the representative of the Ma administration to refocus on a track 2 Japan-US-Taiwan security dialogue. He added that the security dialogue between his organisation and the counterpart in Taiwan was reconvened under the Tsai administration. This somewhat helped supplement this study's argument that, regardless of who is in power in Taiwan, Japan–Taiwan security relations continue to improve incrementally since it is up to Japan's political decision-makers to determine the boundaries of the overall bilateral security interactions, while simultaneously refuting Kawashima's argument (2016) that if the cross-strait relations are stable, Japan–Taiwan relations are more likely to improve.

Since government-to-government interactions are restricted in Japan—Taiwan relations, especially in the field of security, informal exchanges and the subsequently produced informal sources play an important role in elucidating how Japan's security engagement

with Taiwan is conducted. As will be examined in Chapter 3, Taiwan extensively instrumentalises track 2 security dialogues via think tank diplomacy to supplement its diplomacy in the absence of robust official communication channels due to the international non-recognition of Taiwan. Thus, the current author visited numerous think tanks, including Chatham House, the Royal United Service Institute in London, the Sasakawa Peace Foundation in Tokyo, and the Legislative Yuan in Taipei, for relevant data gathering and deeper insight. As for the US-based think tanks – which include the CSIS, the Brooking Institute, and Project 2049 – due to the issue of physical inaccessibility, YouTube was utilised to observe the recorded elite political debates. In addition, YouTube channels dedicated to the Tokyo Foreign Press Club and the Japan National Press Club were examined, when high-profile political figures such as former president Lee Teng-Hui (2015) and Premier William Lai (2017) from Taiwan conducted press conferences on Japan—Taiwan relations in Tokyo. Also, the recorded official comments made by high-profile UN figures such as Ban Ki Moon were utilised from the ICAO's official channel in YouTube.

1.4.4 Secondary sources

Finally, while the extensive use of documentary evidence proves the existence of debates among Taiwanese intellectuals, academics, policy-makers and journalists related to Taiwan–Japan relations, the case studies confirm the minimal influence that Taiwan has on its security relations with Japan within the hierarchical order. Again, this documentary evidence supports the argument that it is up to Japan's political decision-makers to determine the acceptable boundaries of the security interactions between Japan and Taiwan within the context of Japan's One-China policy. According to *Ziyou*

Shibao (2019), the previous de facto Japanese ambassador, Numata Mikio, stated that upon his first meeting with the former president, Lee, the latter characterised Japan—Taiwan Relations as '####' three times, which literally means 'one-sided love'. In other words, while there are many things that Taiwan wants Japan to do, Japan has no intension to respond to their wishes. Certainly, it is erroneous to assume that the extensive data gathering from multiple textual sources and open debates is accurate or valid, especially within the context of the unique tension between what is considered as official and what is considered as informal in Japan's security engagement with Taiwan as well as the difficulty in accessing the opaque decision-making body responsible for Japan's Taiwan Policy. However, this does help to better consolidate the argument on how Japan instrumentalises Taiwan as a foreign policy instrument for its own security within the context of its One-China policy while ignoring what Taiwan wants Japan to do in return, except when Japan chooses to accommodate Taiwan for its own benefit.

1.5 Multiple case studies

Regarding the debate on what is considered a case study, Robson (1993, p. 146) stated that it is '[a] strategy for doing research which involves an empirical investigation of a particular contemporary phenomenon within its real-life context using multiple sources of evidence'. Here, the case study method is used to examine how Japan is utilising Taiwan as a foreign policy instrument for its security vis-à-vis the PRC, with the case studies designed to demonstrate the validity of this narrative. Maoz (2002) pointed out that when a researcher adopts the case study method, the theory needs to be generalisable in accordance with theoretical consistency to better justify it and to delve deeper into the details of the phenomenon. Therefore, theory-testing, theory-building and theory-refuting approaches are required. For this purpose, it was important to

triangulate the findings on Japan's state behaviour vis-à-vis Taiwan. Regarding this triangulation, Eisenhardt (1989, p. 538) pointed out that a 'triangulation made possible by multiple data collection methods provides stronger substantiation of the constructs and hypotheses'. Thus, this dissertation adopts three case studies. The major advantage of using multiple case studies lies in the capacity for comparing different sets of data, which can lead to a more established cause and effect mechanism and a better theorisation of the social phenomenon this research is attempting to elucidate.

Certainly, in the case study selection, a researcher needs to recognise the selection bias, and some theoretical inconsistency (Maoz, 2002). This dissertation also has a minor theoretical anomaly, which we discuss in the section 1.4.8 on case study on maritime security. Nonetheless, the advantages that stems from conducting case studies are quite significant in terms of delving deeper into the cause and effect mechanism in broader and more explicit fashion and testing the validity of the theory in question. With three sets of case studies from the domains of defence, aviation, and maritime security, we can establish a more consolidated theoretical framework for the social phenomenon in international society in the regional setting, where Japan accommodates Taiwan and induces or rewards Taiwan's specific state behaviours in relation to Japan's security vis-à-vis the threat of the PRC via state-centric international agreements based on international law. Furthermore, given that Taiwan is important for Japan's security due to its role as a military buffer and Japan's secured access to sea and air lines of communications, the case studies could be easily identified within the military, aerial, and maritime security domains. In addition, the three domains of national security can be explained through reference to the state-centric international agreements that Japan signed with Taiwan (or aim to sign) to induce or reward Taiwan's specific state

behaviours in relation to Japan's security. These amount to defence security actions that are debated within the context of the CSD issue in the UN Charter, aviation security within the context of the Chicago Convention of the ICAO, and the fisheries agreement within the context of UNCLOS. If the research can elucidate on how and why Japan socialises Taiwan with international security norms in the three dimensions of defence and aerial and maritime security, it can make a convincing case for the conceptual and empirical validity of the narrative. Furthermore, precisely because this research relies heavily on informal sources due to the informal nature of the relations between Japan and Taiwan, the use of multiple case studies will increase the validity of the narrative. In short, using multiple case studies allows us to better elucidate the low-key but comprehensive nature of Japan's security engagement with unrecognised Taiwan, even within the context of Japan's navigation of its One-China policy.

1.5.1 Case study on defence security

In terms of the issue of CSD, Taiwan can never play an official part in Japan's CSD actions in the absence of the latter's recognition of the former and its One-China policy. It is essential to state here that the very concept of Japanese CSD is disputed, and while Taiwan remains a contested entity. Therefore, the status of Taiwan in terms of whether the Japanese CSD applies to it remains ambiguous at best. Nonetheless, Japan can indirectly encourage Taiwan to voluntarily align its defence policy with Japan's defence security policy to supplement its defence strategy as if Taiwan is indeed a member while increasing the defence role of Taiwan as a military buffer for Japan. Thus, this amounts to an extensive case study aimed at elucidating how Taiwan's 'as if membership' is conceptualised in Japan's interpretation of the tensions between its domestic pacifist

constitution and international law on CSD, and how this legal debate affects Japan's defence security policy on Taiwan. It will allow us to observe how unrecognised Taiwan cannot attain official membership to Japanese domestic debates over the tension between the domestic and international law related to the Japanese concept of CSD, where Japan utilises Taiwan as a defence buffer while incrementally enhancing the latter's contribution to Japan's security via the non-traditional security cooperation within the context of Japan's One-China policy.

Thus, the more Japan increases its defence capabilities around the southeast Western Islands near Taiwan, the more Taiwan's independent defence capabilities become important as a military buffer for Japan's military security vis-à-vis the PRC in the East China Sea. Nonetheless, Japan has no intension of officially committing to Taiwan's physical security vis-à-vis the PRC's aggression, regardless of the fact that Japan's CSD and improved defence security capabilities increase deterrence for Taiwan in the face of the threat of the PRC. In short, Japan is maintaining the ambiguity of its security commitment to Taiwan while it continues to encourage Taiwan to improve its contribution to Japan's security via the enhancement of non-traditional security cooperation through Japan—Taiwan non-military HADR and coastguard cooperation. The case study will also reveal how Taiwan's quiet defence alignment with Japan's defence security through this non-traditional security cooperation with Japan also improves both nations' crisis management capabilities.

1.5.2 Case study on aviation security

In terms of the chapter on the ICAO, this involves a case study aimed at examining the civil aviation security agency in the UN, of which Taiwan is not a member in the absence of recognised statehood and due to the PRC's sanctions. Nonetheless, Japan socialises Taiwan with international aviation security norms created in the ICAO as a foreign policy instrument to better handle the threat of the PRC in terms of air security. Japan has signed numerous aviation-related agreements with Taiwan or has encouraged Taiwan to take specific actions in terms of aviation security in accordance with the security rules and norms based on the Chicago Convention of the ICAO. As such, the case study engages in an extensive empirical investigation on the social phenomenon of how internationally excluded Taiwan has been accommodated into international aviation security efforts by Japan even though it does not officially grant Taiwan explicit membership. Clearly, Taiwan has enjoyed no membership of the International Civil Aviation Organisation in the absence of recognised statehood and the PRC's political sanctions, with the exception of a brief period under the Ma administration. However, the nation's airspace management is closely linked to the stable management of Japan's air lines of communication in the region due to the airspace connectivity between the two nations. While Taiwan is excluded from the ICAO, since its air security is linked with Japan's, the enhanced Taiwanese air security management improves Japan's, provided the former behaves as if it is a member of the ICAO and acts in compliance with the international norms of civil and defence aviation security. Thus, this case study will reveal how Japan has been filling the void resulting from Taiwan's exclusion in the international aviation security efforts. This ICAO-related chapter suggests an alternative membership for an unrecognised state such as Taiwan in multilateral security forums including the UN-related agencies. An 'as if membership' of Taiwan in international security efforts granted through Japan's political power is

conceptualised here, which requires Taiwan to behave according to international security norms as Japan's foreign policy instrument.

1.5.3 Case study on maritime security

Maoz (2002) pointed out that the researcher needs to recognise that each theory has some inconsistency and various anomalies. The current researcher admits that the chapter on the fisheries agreement involves a minor theoretical inconsistency in relation to how it heeded various Taiwanese perspectives. Nonetheless, that is because Japan semi-officially recognised Taiwan as a semi-independent political actor via the signing of the fisheries agreement with Taiwan based on the UNCLOS, which ensured that Taiwan was a semi-equal actor with Japan and China in the East China Sea issue. This meant the third case study involved an anomaly since Japan and Taiwan signed the fisheries agreement, meaning that while it was signed 'informally', it was clearly a de facto 'semi-official' acknowledgement of Taiwan as a stakeholder. As such, Taiwan was granted semi-legitimate membership by Japan. Therefore, this case study is different in nature to the other two case studies, wherein "as if" membership was conceptualised as opposed to a semi-legitimate membership in the marine issues related to the East China Sea. Accordingly, this presents an extensive investigation into the major maritime security crisis on how Japan enhanced peace and stability by granting Taiwan a near equal membership in the East China Sea issue by sharing the marine resources via the fisheries agreement. This phenomenon also removed the possibility of the Taiwan–PRC united front against Japan. As will be examined in the chapter on the fisheries agreement, Japan acknowledged Taiwan as a de facto autonomous political actor that operated independently from the PRC in the marine-related East China Sea issue. This created the need to examine Taiwanese perspectives since Japan acknowledged Taiwan's semi-legitimate membership alongside Japan and China. As such, from the PRC's perspective, Japan violated its One-China policy in the context of Sino-Japanese regional great power management. Thus, the PRC retaliated against Japan, as was manifested in China's informal refutation of Japan's sovereignty over Okinawa (*The New York Times*, 2013). As was argued earlier in our rejection of a constructivist approach for this research, the more Japan engages with Taiwan as a semi-independent actor that is almost equal to Japan and the PRC, the more it affects the PRC's ontological insecurity and the legitimacy issue of the CCP.

Nonetheless, since Japan treated Taiwan as a semi-equal participant in the maritime issue related to the East China Sea, Taiwan was accorded semi-official membership. This made Taiwan a 'semi-official member' whose perspectives are worthy of attention, as a negotiating partner in regional security issues for Japan as opposed to merely being Japan's foreign policy instrument. It is because the bilateral fisheries agreement signed between the two nations is 'Japan-Taiwan bilateral relations' based on understanding the two nations are semi-equal only on this particular marine issue in the East China Sea. This runs contrary to the CSD and the ICAO, where Japanese policymakers deal with Taiwan as 'a foreign policy issue' in Sino-Japanese relations, wherein Taiwan has no political power in influencing Japan's foreign policy. Thus, the need to examine bilateral Japan-Taiwan relations in terms of Taiwanese perspectives surfaced. Therefore, unlike the chapters on CSD and the ICAO, which did not regard Taiwan as a 'legitimate actor' that is equal to Japan or the PRC, the chapter on the fisheries agreement includes a discussion on the dynamics of how Japan-Taiwan relations affected Japan's decision to sign the agreement with Taiwan despite the fact that this

was considered a violation of Japan's One-China policy and the PRC's claim of sovereignty over Taiwan.

Nonetheless, as will be explained, the hierarchical relations between a regional great power that is Japan and an unrecognised Taiwan remain solid within the regional setting. Therefore, Japan's socialisation of Taiwan as if it is a member of international society has theoretical consistency, regularity, and continuity. Thus, the case study on the fisheries agreement, while somewhat inconsistent, sharpens and reinforces the theoretical validity of the proposed theory in this research. Japan's accommodation of the internationally isolated entity, Taiwan, is theoretically generalisable to how Japan accommodates other internationally isolated actors, such as the Philippines under the Duterte administration, as will be discussed in the concluding chapter.

1.4.9 Summary

The constructivist-based literature on Taiwan, the norm constructivism related to Japan, and the work related to Japan's identity and othering cannot allow us to conceptualise why and how Japan accommodates Taiwan as a foreign policy instrument via socialising it as a part of the international security efforts vis-à-vis the threat of the PRC. In addition, Japan's non-recognition of Taiwan also creates a unique situation where there is an inherent tension between Japan's 'formal' engagement with Taiwan and its 'informal' engagement, which has certain implications for this research. Thus, this study is aimed at reconceptualising how Japan accommodates Taiwan's membership to the international society of states via a different theoretical approach. The next chapter presents a literature review related to the ESIR and to de facto states in general. This

will help elucidate how existing studies have conceptualised the way in which the Eurocentric powers of the West have been accommodating the non-sovereign political communities, including non-recognised states. The review will also address whether the ESIR has examined Japan's accommodation of Taiwan into international society for its own regional security in terms of whether or not this presents a weakness. For certain, this perspective deserves more scholastic attention from the ESIR.

1.5 Significance of the Study

The PRC has been pressuring Japan not to engage in any military or strategic relationship with Taiwan in the context of Japan–China–Taiwan relations. However, as will be explained, Japan has not been meekly following its One-China policy. Indeed, this study will outline how Japan has gradually improved its indirect security ties with Taiwan even within the context of its One-China policy, a point that is highly significant. Despite the prevalent characterisation of Japan as a reactive state (Funabashi, 1991/1992; Calder 1988, 2003), the nation has been playing an under-appreciated security role through its engagement with Taiwan in view of managing its relationship with the PRC. First, Japan has been incrementally accommodating the non-state entity Taiwan in the regional setting by affording it quasi-membership despite the absence of statehood. Therefore, Japan's incremental accommodation of Taiwan functions to create new international security norms and rules within international society in the regional setting. Second, Japan has been incrementally instrumentalising unrecognised Taiwan as an international security asset for its own security, despite its One-China policy. Third, if Taiwan presents a loophole in the regional security environment (which

dictates that it has no access to the multilateral international security efforts that require statehood for membership), Japan has been accommodating Taiwan to mitigate such a missing linkage in non-traditional international security efforts through its unique security behaviour. That is, it has enabled Taiwan to behave as if it is a part of the non-traditional security efforts. As such, Japan has been innovating the new international security norms related to how to engage with de facto states within international society. If it can be demonstrated that Japan is indeed utilising Taiwan as a foreign policy instrument, this will highlight Japan's underappreciated but unique security role within international society in the regional setting. This will, in turn, contribute to our understanding of Japan's unique conception of security, which is different from that of the US.

If, despite the geographical proximity (and the obvious importance of the sea and air as lines of communication for Japan's security due to Taiwan's strategic location), Japan is less committed to the security of Taiwan, a number of questions emerge. First, if Japan has kept the Taiwan issue low-key in Sino-Japanese relations – which is in direct contrast to the way the Taiwan issue occupies a central position in US–China relations – why is Japan incrementally accommodating Taiwan for security purposes vis-à-vis the PRC? Second, what kind of security gains has Japan received by accommodating Taiwan? Third, how has Japan exploited Taiwan for its own security while not treating the latter as an equal security partner in the absence of recognition? Within this context, Japan can be a provider of diplomatic space to enable Taiwan to behave as if it is a part of the international security efforts, even if it cannot officially participate in the attendant multilateral forums given its lack of statehood.

1.6 The Relevance of Japan

The primary locus of this study in terms of inclusion/exclusion is Japan. There exist a vast number of studies on how the West has been including or excluding the non-Western powers, including the works of Bull and Buzan (1979, 2014), while there are numerous works related to how the non-Western powers have been responding to the expansion of international society. However, studies on how Japan has been accommodating an excluded member of international society (Taiwan) are virtually non-existent.

Understanding the role of Japan's accommodation of a non-sovereign political community such as Taiwan within the regional security environment is essential. Japan assumes a security leadership role as the most important ally of the US in the region, while, until fairly recently, it was the second largest economic superpower in the world. Taiwan is certainly a less visible issue in Japan's foreign policy due to the former's unique status. However, Taiwan remains an important neighbour for Japan due to its geopolitical importance. The informal nature of the virtual diplomacy conducted within Japan–Taiwan relations is neglected in studies in the field of international relations, despite its significance to the East Asian region (Deans, 2001). Japan remains the second largest democratic economic entity in the world, and its role as a regional security provider is emerging as it continues to normalise its international security role. Therefore, Japan's engagement with Taiwan is crucial to understanding the regional security environment. If Japan has been accommodating the de facto state of Taiwan within international society, this will shed new light on the underappreciated security

role of Japan within the context of the nexus of inclusion/exclusion in regional security studies and in the ESIR.

Chapter 2. Literature Review on the English School and De

Facto States

2.1 Introduction

This chapter reviews the existing literature on the ESIR and de facto states in general to answer the following questions:

- What is the current state of the English School of International Relations?
- How does the English School accommodate excluded outsiders in international society?
- What is a de facto state, and why is international society in favour of territorial integrity and biased against de facto states?
- How has international society engaged with de facto states without recognition?

These questions will be used to identify the strengths and weaknesses of the current literature to generate a new question: How can a regional great power Japan accommodate the de facto state of Taiwan in international society in terms of its security while balancing its ties both with Taiwan and the PRC? In other words, this review considers how the existing literature on the ESIR and the de facto states has not considered the perspective of how the non-European regional great power Japan has been engaged in the inclusion of Taiwan in the international society of states in order to enhance the regional order.

2.2 What is the English School of International Relations?

Early contributors to the ESIR, such as Hedley Bull (1977), argued that the international system is simply a component of international society, that is, the states, sovereignties, political communities, and even human elements that constitute the rules and norms that determine order. Bull (1977) conceptualised such an international society in terms of the need to understand the sociology of European history to understand international society, which has spread throughout the globe. Here, Bull (1977) argued that even under conditions of disorderly anarchy, this does not mean chaos ensues, as was argued by the Hobbesian view of the world, and that the political communities that exist still constitute the basic rules and norms of stable order. For this order to exist, rules and norms are essential, even if they are subject to normative changes. Bull (1977) identified international relations as social relationships and argued that the three pillars of social functions – security against violence, observance of agreements, and property rights – are essential to forming a society. As such, early English School contributors, such as Bull (1977), focused on how to maintain order rather than the enhancement of justice within international society. In other words, the early English School emphasised the coexistence of a society of states. Meanwhile, modern English School theorists such as like Buzan (2014) went further in maintaining that the society of states is essential while focusing more on the rules and norms involved in state-centric solidarism.

Early prominent English School scholars, such as Watson (1984) and Bull (1977), emphasised the importance of the primary institutions that constitute the essential components of international society, such as the balance of power, international law, diplomacy, war, and the great power management. Their grand narrative appears to follow the primary line that the rules and norms that regulate the contrived power

balance, the use of international law to regulate relations among states, the diplomatic missions to communicate with each other, the control of damage through the rules of just war, and the privileged positions of the great powers. Therefore, according to Brown (2009), international society is a norm-governed society wherein at least the minimum rules are followed, and wherein the member states of the society accept certain responsibilities in view of forming a society of states. Regarding the importance of the primary institutions, Buzan (2014, p. 17) pointed out that social changes to such institutions occur and stated that the primary institutions are

constitutive of both states and international society in that they define not only the basic character of states but also their patterns of legitimate behaviour in relations to each other, as well as the criteria for membership of international society.

For example, regarding the changing nature of the balance of power and the great power management, Bull (1977) highlighted that while the balance of power in nineteenth-century Europe was based on the concept of the 'Concert of Europe', the balance of power during the Cold War involved two superpowers with different ideologies that also had nuclear weapons. It can be stated that the role of the great powers that make the rules for lesser states has not changed in terms of being constitutive elements of the primary institutions. As such, Buzan (2014) argued that the social forms of the balance of power are conceptually different from the Hobbesian worldview of anarchy or the neorealist understanding of the balance of power (Waltz, 1979). In short, the mechanistic understanding of the balance of power does not have prominent status in the English School. However, Buzan (2014) pointed out that the English School considers the socially constructed balance of power to be both contrived and accidental.

Meanwhile, other ESIR scholars, such as Wilson (2016), have questioned when the concepts of the institutions became primary institutions.

Elsewhere, Schouenborg (2012) argued that the primary core institutions that constitute international society are socially constructed entities and are thus malleable. While many of the examples listed in this study are often considered to be revolutionary events – such as the French Revolution, international fascism, international communism, and the EU supranational state – alternative historical narratives of these events can recount the distinct possibilities of creating unique core institutions that may challenge the supremacy of the international society of the Westphalian sovereign states. Some of the core institutions originate from the regionally differentiated conditions in regional international society, but legacies can still exist beyond the transregional outreach in a regionally unrestrained manner. The French Revolution inspired a concept known as 'popular will' or 'popular sovereignty', which is necessary for the legitimacy of a sovereign state. Meanwhile, international fascism discredited the legitimacy of colonialism, international communism discredited the Soviet Union as an empire of sorts, and the EU created new conditions for the legitimacy of internal structures, such as the abolishment of warfare inside the EU.

2.2.1 The pluralist and solidarist debate

One of the debates related to pluralism and solidarism can essentially be identified as the tensions between human rights and sovereign equality. Bull (1977) preferred pluralism and order over justice based on positivist law and mutual respect of sovereignty in international society, at least within a European context, while he acknowledged the importance of a shared culture to bind states. Within the contemporary context, one of the most controversial debates related to pluralism and solidarism is identified as relating to humanitarian intervention, the responsibility to protect (R2P), and sovereign equality. Therefore, international society has a moral obligation to intervene and violate the sovereignty of a state if the state in question is violating the human rights of its citizens or the state authority cannot stop a massacre. Buzan (2014) argued that the pluralist/solidarist debate is not mutually exclusive but mutually complementary, describing it as yin and yang. Meanwhile, Buzan (2014), Bull (1977), and Watson (1984) denied the possibility of the creation of a world society based on human elements as it is unrealistic, while Buzan (2014) argued for the emerging convergence of more interests and values among states to form more state-centric solidarist societies of states. Elsewhere, Knudsen (2016) appeared to support the idea of a normalised R2P and humanitarian intervention as a fact of life within the contemporary setting.

2.2.2 Membership of non-European entities in international society

The English School is often criticised as being Eurocentric, and rightly so. In the English School grand narrative (Wight, 1977, 1952; Bull and Watson, 1984; Watson, 1992, 1990), when the expansion of the European international society began, there were many regional societies that were separated from each other, with each regional society having its own distinct civilisation-specific rules and norms for mutual interaction. Any contact between the civilised European states and the non-European political communities in different civilisations could not be conducted based on similar

rights and equality, which were accorded to sovereign states with similar civilisational traits in Europe. In the absence of the shared rules and norms between the civilised European states in the West and the barbarians and savages in the rest of the world, equality did not apply. Thus, the civilised Europeans engaged on equal terms with likeminded states in Europe while enhancing the civilisation mission elsewhere. The civilised peoples in Europe needed to define the standard of civilisation to accommodate non-European uncivilised peoples if the West was to enhance its contact with the barbarians and savages in the rest of the non-European uncivilised world.

It is important to note that traditional ESIR theorists such as Bull (1977) considered the expansion of international society as corrosive since non-Western societies bring more cultural diversity and undermine the rules and norms of the Eurocentric international society. The civilised peoples in European states constructed the shared rules and norms of respecting each other as equals based on the practice of recognition, reciprocity, and respecting mutual sovereignty, which enhanced international law, diplomacy, and trade in the society of states in Europe, while the barbarians and savages among the non-European peoples were excluded from such fair treatment (Ringmar, 2014, pp. 448– 452). Ringmar (2014) argued that the primary reason why Europeans engaged in brutal acts against non-Europeans in the non-Western world (while paying close attention to conforming to the rules and norms in diplomacy, trade, and warfare in the European society of states) was because the former were able to confirm their identity as civilised Europeans. Therefore, the brutal treatment of uncivilised non-Europeans, who were incapable of following international norms, elevated the higher social status of the Europeans. The only way for the inferior non-Europeans to be treated as equals by the superior Europeans was to behave like accultured Europeans within global international society. Therefore, the European recognition for equal treatment was essential for the admittance of non-Europeans into European international society and for their survival.

For Bull (1977) and Watson (1984), international society expanded to the rest of the world from Europe through global decolonisation and the creation of newly independent states, whose local elites were immersed in European cultures. Elsewhere, Keene (2002) pointed out that the early English School theorists ended up being accused of sanitising its accounts both by eliding the coercive nature of Eurocentric expansion of international society and defining the 'civilisation standard' to legitimise the subordinate positions of the non-European peoples below the civilised peoples in Europe within the framework of the expansion of international society. In criticising such a Eurocentric account, Little (2015) argued that while both the early English School theorists, Bull and Watson, and the modern English School theorist, Buzan, emphasised the creation of a European society of states, upon historical examination, it is clear that international societies existed elsewhere before they emerged in nineteenth-century Europe, and that these non-European societies talk about 're-admission into the international society'.

So much of the literature on the English School has not changed its Eurocentric attitude in defining the West as a metropolitan core and the rest of the world as periphery. Here, Stivachtis (2015) argued that the Eurocentric civilisation standard for inclusion in and exclusion from international society observed historical continuities, even though the criteria for the forms of civilisation standard may have changed. Stivachtis (2015) also observed that only the civilised states were admitted into international society, which required the members of international society to engage in external relations that were

based not only on the proper codes of conduct but also on civilised domestic norms, such as good compliance with the norms of human rights and democracy based on international law. Furthermore, Buzan (2014) argued that even after decolonisation, the concept of the civilisation standard is well and alive and still practised in international society in the form of conditionality and human rights. From Buzan's (2014) perspective, while economic aid is considered normative practice for first-world countries in the contemporary context, the concept of the standard of civilisation is still used in practice, if not in rhetoric, in terms of exercising moral authority and power over third-world countries. In fact, Buzan (2014) noted that membership of certain clubs of international organisations and in society retain Eurocentric standards of measurement, such as human rights, democracy, capitalism, development, and ecologism. Indeed, Buzan (2014) argued that as the diffusion of international society was facilitated, the tension between universal sovereign equality, non-intervention, and the Eurocentric global order may take shape, but the West still has the upper hand in determining who is civilised or uncivilised within the context of broader international society if the insider/outsider of the Western civilisation applies.

While international society has conceptual flexibility to accommodate non-European entities as members of international society if they meet the standards of civilisation, these international bodies are not generally recognised as equal. For example, Suzuki (2005) observed how Japan, as a semi-civilised state, had to encounter the Janus-faced European international society. Furthermore, while agreeing with Buzan's insistence on the English School's theoretical flexibility, Englehart (2010) discussed how Siam used British high culture and class-based elitism to downplay the Eurocentric racial prejudice against the people of Siam by establishing stronger personal ties with British

elites and royalty to maintain the sovereign independence of Siam and to avoid Western colonisation. The non-European entities had to internalise their humiliation in view of being treated as equals if they were to be admitted into the Eurocentric international society.

Meanwhile, in the English School accounts, many of the 'non-civilised peoples' that were excluded from international society attempted to upgrade their international status by conforming to the institutions discussed earlier, while others did not. For example, Zielonka (2013) argued that while the EU could be regarded as a normative power that shapes the external environment through the promotion of liberal democratic values, human rights, good governance, and the rule of law, the core normative components generated are Western European values. Here, Zielonka (2013) noted that a close examination of the EU's core values through discursive analysis reveals the imperial nature of dictating 'outsiders' through the normative impositions that resemble the mission of civilisation. Central and Eastern European states were accommodated into the club if they embraced the EU's core values, while Arab states are unlikely to accept such imperial values for religious reasons and due to the experience of European colonialism (Zielonka, 2013).

Meanwhile, it is also undeniable that the non-Western states that avoided direct Western colonisation had to accommodate Western superiority to seek equality and respect from the Western core. The most successful case is arguably Japan, which defeated two regional empires – China and Russia – and became the first non-European empire in its own right. For example, Lai (2014) argued that in nineteenth-century power politics, the vague concept of civilisation using positivist international law became a vital

instrument to measure the compliance of outsiders with civilisation. Here, Lai (2014) observed that Japan fully grasped the importance of international law and enhanced its reforms and improved the European narratives of Japan's foreign policy behaviours. Meanwhile, the author also noted that China proved itself to be incapable of comprehending concepts, such as international law, during the first Sino-Japanese War, and that Japan was accommodated in the elite club of international society, while China was defined as barbarous after the war. Furthermore, Wirth (2015) argued that the significant reason why Japan enhanced its alliance with the US, even after the demise of the Soviet Union, is that the former instrumentalised its alliance with the latter as a normative link to define itself as a member of Western international society. As such, Japan was able to maintain a superior civilisational position in ideational and social terms regarding the rise of China and the Asian Century (Wirth, 2015). After the Meiji restoration, Japan attempted to become a member of the Eurocentric international society while placing Asia in a subordinate position.

It is entirely possible that the conception of great power is malleable within the framework of the English School. In fact, Zhang (2011) argued that the concept of China was invented and reinvented in the English School's concepts of international society from the early stages of the theory's academic development. In Zhang's (2011) view, China was first conceptualised as having a similar society of states or entities in the pre-modern 'Warring States' era that resembled Western international society. However, China was subsequently regarded as an uncivilised or barbarous state following the Opium Wars, revolutionary China was essentially seen as a self-imposed outsider during the Cultural Revolution while leading the third-world's revolt against the West before it was transformed into one of the would-be great powers as

conceptualised by Bull (1977) following its reform and opening-up policy. According to Zhang (2011), China is now regarded as both a facilitator and a challenger of the Western-dominated world order, concluding that pluralists tend to emphasise cultural diversity – including in terms of non-intervention and regime diversity – and thus retain more favourable views of China, while the solidarists – who assert that human rights and democracy are the new standards of civilisation – tend to view China as a challenger to the Eurocentric international society.

Elsewhere, Suzuki (2008) described Japan and China as 'frustrated great powers' suffering from ontological insecurities because of 'recognition games' aimed at gaining entry to the elite club of international society. Importantly, Suzuki (2008) observed that both nations appear to be interested in impressing the privileged and established Western elite states, which means the traditional Western privileged great powers' standing can be consolidated. As such, as Suzuki (2008) noted that the privileged Western great powers are able to consolidate their standing if frustrated great powers such as Japan and China attempt to conform to the norms created by the West in the process of attaining the same privileges. In other words, in Suzuki's view (2008), Japan and China are inadvertently reproducing the normative structures of the current international society, which is led by the West, as opposed to overthrowing it. Here, it is essential to note that if non-Europeans attempt to conform to Western standards of civilisation, non-European states can also inadvertently further consolidate the West's global rule-making authority and normative power in the process of acculturing Western rules and norms.

Upon examination, both Japan and China try to shoulder international responsibility and comply with international norms while continuing to respect their historical norms. In the English School's understanding of Japan and China, both would-be great powers attempt to use their normative self-identifications to enhance their international standing within the modern context. For example, Black and Hwang (2012) argued that both nations tackled the issue of piracy in the Gulf of Aden based on their respective self-identification through the culturalist and post-colonialist approaches. According to these authors, China's foreign policy elites based their state behaviour regarding the issue of piracy on its self-identification of *Tianxia* and considered the Somali pirates to be inferior, while they also noted that China also gave the role of *Tianxia* to the UN for China's legitimacy and authority and acted based on its own non-interference policy following the Somalis' call for help. Meanwhile, Japan enhanced its anti-piracy operations in Somalia based on its self-identification as a cultural marker lying somewhere between the members of international society and those of the noninternational society, where Japan's domestic anti-militarist norms - such as dispatching the crews of the JCG along with the MSDF – and the normative sensitivity of respecting local cultures – which stemmed from its successful anti-piracy policy in Southeast Asia – proved to be transferable to the Gulf of Aden (Black and Hwang, 2012).

Elsewhere, Zarakol (2010) argued that the outsiders of the Eurocentric international society that suffered from the imposed ontological insecurities had to be incorporated into the European elite club of international society, but that these outsiders were forced to internalise their identities of inferiority and subordination and were defined as barbarous, uncivilised, less human and less enlightened than their European counterparts. Here, Zarakol (2010) pointed out that these oppressions contributed to the

hardened stance toward the acceptance of past crimes committed by non-European states such as Japan and Turkey. By accepting the international historical narratives of their past wrongdoings, non-European countries are admitting their morally inferior positions regarding Europeans within the hierarchical international society.

Following this logic, the existing literature has focused on how semi-civilised states, including Japan, are accommodated in the elite club of Eurocentric international society. In other words, the Eurocentric international society in the West accommodates non-Western entities, such as Japan, in paternalistic-like narratives. The exception here is Black (2012, 2014), who, in challenging such a Eurocentric narrative, argued that through a non-military maritime security policy, Japan has also become a regional and global norm entrepreneur in Southeast Asia and the Gulf of Aden. Zhang (2015) argued that while the globalisation of international society penetrated the regional society of East Asia, the localisation of the primary institutions, such as sovereignty and the developmental states, needs to be recognised while the regional differentiation of international society in the region also affects the configuration of the global international society. However, Zhang (2015) concluded that whether the existence of regional international society in East Asia can be proved is, at best, debatable.

Interestingly, what has not been discussed at all in the existing English School literature is how Japan's accommodation of regional outsiders such as Taiwan relates to the grand regional narrative. In other words, Japan has been treated and researched as an outsider itself and as a latecomer to global international society and has received little academic attention in the existing literature of the English School. In fact, Japan is largely treated as a non-European recipient or a good student of Western civilisation and is rarely

regarded in terms of being a regional early comer that accommodates regional outsiders such as internationally unrecognised Taiwan. One critique the current author would like to direct to the current international society theory is that while this theory is flexible enough to accommodate all political communities, including non-sovereignty-based entities, there are hardly any studies on how the regional great power of Japan can accommodate de facto state entities such as Taiwan. This critique entails how such an accommodation shapes new rules and norms for primary institutions in the evolution of international society. Therefore, the grand narrative of this research on the ESIR is how Japan has been expanding international society to accommodate a non-sovereign political community, Taiwan, for its security in regional settings. Japan's relations with Taiwan affects Japan's contrived balance of power with China, its diplomacy with China and Taiwan, its use of international law in relation to Taiwan, its potential involvement in a war with China over Taiwan, and its great power management with China. Therefore, there is room for studying the use of international society theory to elucidate Japan's accommodation of Taiwan while maintaining a relationship with China in view of securing the regional order.

2.3 De Facto States in International Society

Now we can move onto the discussion related to how international society's engagement with the de facto states has been understood in the existing literature.

2.3.1 The issue of legitimacy and illegitimacy of de facto states in international society

If the politico-legal conditions discussed in the previous chapter qualifies an entity in question as a state, what makes other political communities de facto states in international society? It is commonly agreed that a de facto state is a political reality in which a state-like entity functions quite well as a domestic sovereign entity but suffers from a lack of international recognition of statehood within international society and from the prominent international institutions (Pegg, 1998; Harvey, 2010). While there is a dearth of in-depth literature on de facto states, Scott Pegg is a pioneer on the studies of unrecognised states, and his paper 'International Society and the De Facto State' (1998b) can be seen as one of the most influential works on the issue. Unlike Robert Jackson's work entitled 'Quasi-States: Sovereignty, International Relations and the Third World' (1993), which used the term 'quasi-state' to categorise a non-recognised state-like entity, Pegg uses the term 'de facto state' to distinguish unrecognised state-like entities from quasi-states that may be internationally recognised but do not have an effective domestic governance. On this point, Pegg (2000, p. 91) stated the following:

[n]o matter how long or how effective its territorial control of a given area has been, that control is neither recognised nor is it considered legitimate. The quasistate is legitimate no matter how ineffective it is. Conversely, the de facto state is illegitimate no matter how effective it is.

Meanwhile, Fabry (2010) argued that without state recognition by the international society of states, a political community in question – such as a de facto state – cannot join the society of states and insisted that recognition and self-determination are two sides of the same coin. Furthermore, which de facto states have the right to obtain an international legal personality remains highly contested within international society. For example, Jeffrey's (2016) observation on Somaliland enhances Pegg's (1998a)

conceptualisation of a de facto state. Here, Jeffrey (2016) argued that despite the presence of relatively good governance, democracy, and a nascent economy that requires no international aid, in the absence of international recognition of statehood, Somaliland remains illegitimate in the eyes of the international community as a de facto state. The African Union is also reluctant to afford Somaliland recognition due to the potential rise in demand from other breakaway or independent movements (Jeffrey, 2016). This lack of international recognition also subsequently galvanised the internal coherence between the different sectarian clans in Somaliland under the common goal of achieving international recognition, and if this is indeed achieved, this may have huge regional repercussions (Jeffrey, 2016). It is fair to say that however well Somaliland is functioning domestically, it continues to be treated as a de facto state without international recognition and is regarded as illegitimate on the international stage.

Furthermore, Pegg and Kolstø (2015) argued that while the internationally illegitimate Somaliland enjoys a high degree of internal legitimacy among its population, the internationally recognised Somalia functions as a failed state. According to the authors, the people of Somaliland most desire security and peace from its government, which the internationally illegitimate authority in Somaliland has been able to deliver since internationally legitimate Somalia committed brutality against its people. Here, Pegg and Kolstø (2015) pointed out that the parent state of Somalia has no such capabilities at all to deliver this to people in Somaliland, while the people of Somaliland have no desire for unification with such a failed parent state. However, the authors concluded that the lack of international aid for Somaliland (due to of its non-international recognition) hampered its economic growth, while the strong internal legitimacy has

not brought about international recognition for the nation, adding that this situation will likely continue for many years to come. It is fair to say that while the international recognition gave the failed state of Somalia a legitimacy in international society, the de facto state of Somaliland is unwilling to be associated with the de jure state of Somalia even at the expense of potential material benefits, such as international aid, that comes from state recognition via unification.

Meanwhile, to classify the state-like entity as a de facto state, the entity must have the capability to establish ties with other countries (Bartmann, 2008). However, this remains a grey area. Indeed, the capability to enter into a relationship with other countries remains a contentious issue in international relations since there are no clearly defined rules and regulations in international law regarding the number of recognised states that are legally required to gain the necessary capacity for entering into relations with other states (James, 1986). There is also no substantive evidence to demonstrate that a de facto state needs to obtain a recognition of statehood from any major global powers, including from the five privileged permanent members of the United Nations Security Council. For example, while South Ossetia is recognised and is militarily, economically, and politically protected by Russia, it is still regarded internationally as a de facto state. Therefore, what constitutes the capacity for entering into relations with other countries is ambiguous. Furthermore, as Harvey (2010, p. 19) stated,

[the] lack of agreement among scholars over how best to represent unrecognised entities as an analytical concept, and the contentious issue of which examples should and should not be included as case studies, has become a defining feature of literature and debate on this subject.

On this point, Berg and Toomla (2009, p. x) argued that it is wrong to regard the de facto states universally as 'internationally unrecognised territories' since the level of recognition of these entities in international society varies significantly, and the normalisation of the relations with these entities is a better option than encouraging the re-integration of these entities into the parent states or encouraging a form of power-sharing. Furthermore, Berg and Kuusk (2010) argued that sovereignty is an ambiguous concept and whether the entity in question has control over its external and internal sovereignty is more of a political question, even though the legal debate on sovereignty is always prioritised in international society, including within the context of the English School. Regarding this point, Berg and Kuusk (2010, p. 47) stated the following:

[n]eedless to say, this society was made up of countries which were ready to recognise each other's sovereignty on the condition of reciprocity. But while having that sovereignty, legal power for mutual recognition does not necessarily mean that authorities are able to exercise full control over their jurisdictions in one or the other way.

However, it is essential to point out that such a debate may contribute to solidifying unequal relationships in the hierarchical nature of international society between the sovereign states – especially the great powers – and the de facto states. Such an unequal relationship gives great powers unilateral authority to recognise or even further delegitimise the de facto states according to the former's national interests, while the opposite does not apply for the de facto states. In short, reciprocity does not exist in the relations between the great powers and the de facto states, and the former's privileged status places the latter in a subordinate and politically inferior position within international society. In critiquing this unequal relationship, both Berg and Kuusk (2010) and Pegg (1998) pointed out that it is time to admit that sovereignty is simply an

artificial concept. For certain, a de facto state's undetermined status does significantly reduce its foreign policy leverage regarding its relations with great powers in an unequal manner. Therefore, as noted above, which political unit has the international legal personality depends on the will of international society, especially among the great powers.

2.3.2 Preference for territorial integrity and a bias against de facto states in international society

Why does international society have to keep de facto states illegitimate? Here, it can be argued that there is a strong tendency to maintain the territorial and sovereign status quo and that there exists a strong bias against the secessionist entities in international society following global decolonisation. This means that after independence from the former colonial powers was achieved by the newly established states, their defined sovereign territories became somewhat sacrosanct. For this reason, it is challenging for the aspiring states to achieve independence, and such bias against de facto states has become one of the significant constraints that inhibit such entities' potential for external behaviours.

Caspersen (2018) argued that when the de facto state attempts to engage with the broader world, it always creates a tension between self-determination and territorial integrity in varying degrees, and that when the engagement is conceived, both the parent state and the de facto state have very different concepts of what constitutes engagement. Here, Caspersen (2018) held that, for the parent state, the engagement means re-

integration of the breakaway territories or the hierarchical relationship that defines the parent state as a gatekeeper from which the de facto state must gain the approval to engage with recognised states or international institutions. Meanwhile, for the de facto state, it means engagement with the broader world to consolidate its de facto separation or hope for eventual recognition (Caspersen, 2018). For example, Dalay (2017) observed that the global powers – the USA, Russia, and the UK – and the regional actors – Turkey, Iraq, and Iran – all disagreed with the Kurdish aspiration for statehood in a referendum in 2017 and instead recommended that Kurdistan compromised by obtaining more constitutional rights within the context of the Iraqi federation or the confederation.

2.3.3 Engagement without recognition

The preference for territorial integrity and the bias against de facto states within the international society of states creates the need for international society to engage with de facto states in ways that have little effect on the status quo. Engagement without recognition (of statehood) is becoming an increasingly important topic in international relations. Here, Ker-Lindsay (2013) was probably the first to conceptualise engagement without recognition for third-party states to interact with the contested state bilaterally, with the author noting that the international community has a high degree of latitude in terms of direct engagement with an unrecognised entity (with implicit recognition) in question while maintaining the explicit intention of not recognising it as a sovereign de jure state. The author then argued that engagement with non-recognition takes multiple forms, ranging from participation in less formal conferences, establishing a non-official quasi-embassy, and making a less clear-cut declaration of non-recognition, while if the

recognising state wants to have extensive relationships with the contested state, it can do so as much as it wants provided it maintains the position of non-recognition. In other words, as Ker-Lindsay (2015) explained, this can constitute a non-recognition in all but name, in that the explicit and implicit recognition exists in a political sense with such intentions. Both bilateral and multilateral engagements with the contested state are possible but always create confusion and misinterpretations (Ker-Lindsay, 2015). Furthermore, while the bilateral interactions between the recognising state and the contested state create confusion, the multilateral engagements require even greater caution due to the higher collective authority and the legitimacy accorded (Ker-Lindsay, 2015).

Meanwhile, Ker-Lindsay and Berg (2018) argued that as the international understanding on engagement without recognition with the contested states grows among the policymakers and academics within the international community, there are two patterns among the conceptions of what constitutes engagement without recognition, the first of which relates to the levels of willingness on the part of international society to engage with the contested states, while the second relates to the role of the engagement of supranational actors such as the EU with the contested states. Here, Ker-Lindsay and Berg (2018) claimed that a better understanding of the concept of engagement without recognition would help mitigate the stigmatisation of the contested states for conflict management and settlement mechanisms in international society. However, the academic strand that focuses on the de facto states does not appear to agree on how international society should engage with the de facto states. On this point, Pegg and Berg (2016) maintained that the traditional narrative on how the sovereign states deal with de facto states needs to be revisited, with the authors

observing that the international environment surrounding the de facto states is not as hostile as has been said, calling into question the claim that the international community has no principled method of engagement with de facto states. Here, Pegg and Berg (2016) confirmed the existing literature's validity of the critical emphasis on great power politics using the de facto states and international society's liberal bias toward more democratised de facto states in their preference for engagements, while the authors also found the method of how to address the linguistic representations of the officials in the de facto states as irrelevant and rejected the importance of foreign diaspora's domestic political influence on the great powers' foreign policy on the de facto states in question.

2.4 How to Engage with the De Facto States

Here, we move on to the analysis of the literature related to the engagement with the de facto states in general.

2.4.1 Overview: Engagement with the de facto states – three approaches

Previous studies on international engagement with de facto states highlight that there are different conventional approaches to this aspect within international society, which include the active containment of the de facto state, ignoring or having a limited engagement with the de facto state and the imposition of constraints on de facto states in reaching out to others, and the EU's attempt to engage with its regional de facto states in a low-key manner. The existing literature largely addresses the promotion of democracy and human rights due to the bias towards addressing selections between Western states in the context of EU policy (e.g. Popescu, 2007; Hock, 2011; McNamara, 2014). This approach may appear to have some degree of academic validity in terms of

assessing the strategic intentions of Western powers. However, Ker-Lindsay (2013) argued that while engagement without recognition is becoming increasingly important in international politics, there is a wide variety of international engagement with unrecognised states due to concerns over how much the state can engage with de facto states.

These three approaches have been considered depending on the strategic calculations of the members of international society. In fact, it is clear that the existing literature tends to regard the de facto states as the generators of regional insecurities and turmoil within the existing international system. These three main approaches are discussed in detail below.

2.4.2 Active containment of the de facto states

This containment strategy, which involves the active opposition of de facto states through embargoes and sanctions, is made possible when the third-party states consider de facto states to present a grave threat to their national security. The concerned third-party states are compelled to mobilise the political and economic measures at their disposal to contain the perceived threat posed by the de facto states. The continuation of this containment strategy may potentially backfire in the mid- to long-term future in two ways: the increase of international criminal activities from and in the de facto states and the increasing sense of diplomatic and political isolation, which will lead to potentially more aggressive behaviours by the contained de facto states.

In terms of international criminal activities, Northern Cyprus and the Republic of Dniester were examined by Scott Pegg, Pal Kolstø, Charles King, and Julian Duplain. During the 1990s, Northern Cyprus turned into an active loophole for international criminal gangs exploiting the gross lack of extradition and taxation agreements (Pegg, 1998a). In another example, Kolstø (2006), King (2001), and Duplain (1995) examined the case of the Dniester Republic, with Kolstø (2006, p. 728) stating that '[t]he Dniester Republic has a thoroughly criminalised economy, based on smuggling and the fake brand industry' and that '[t]he Republic has also been accused of being a free haven for fugitive gangsters and former KGB officials in hiding'. Elsewhere, King (2001) asked the pertinent question of why these de facto states could rationalise the use of the criminalised economy based on patronage, corruption, and illegal trades. Meanwhile, Kolosove and O'Loughlin (1998, p. 1) pointed out that, in the post-Soviet space, 'the elites of de facto states have strong criminal backgrounds and specialise in the illegal transit of weapons, drugs and money laundering'.

It could be argued that such a containment policy may be one of the most effective strategies if the strategic intention of external powers is to substantially undermine the very existence of the targeted de facto state. However, it can also create unintentional negative repercussions for international society, also having substantial negative implications for international and regional security, especially if they block the targeted de facto state entirely.

For example, Heydarian (2017a) pointed out that Marawi, a Muslim majority region in Southern Mindanao with a specific independence movement, attracted ISIS fighters from the Middle East and Caucasus regions and engaged in urban warfare that

prolonged the internal war in Mindanao, which also has numerous porous borders that allow easy access for the Jihadists into Malaysia and Indonesia. According to Heydarian (2017a), the US, to which Duterte is said to express frequent hostility, pledged counterterrorism support, specifically for urban warfare, where the armed forces of the Philippines lack experience, while Australia dispatched two AP-3C Orion surveillance aircraft for the intelligence, surveillance, and reconnaissance activities. In another example, Heydarian (2017b) argued that there is a significant need for reconstruction and re-integration of the 600,000 displaced residents in post-conflict Marawi, lest the discontent and radicalisation of the Muslim population may follow.

One crucial aspect of King's 2001 article 'The Benefits of Ethnic War: Understanding Eurasia's Unrecognised States' is the author's rationale behind why these international criminal gangs have succeeded in developing state-like entities with some effective governance, even if they are gross violators of human rights or are entirely against the international norms of democracy upheld in these Caucasus and Eurasian regions. The connection between organised criminal gangs and domestic political leaders was also noted by both Collier and Hoeffler (2002) and Hock (2011). What these four authors have in common is that they attribute the theory of greed and grievance in civil wars to this connection between organised crime and domestic politics in these de facto states. According to this theory, domestic leaders in politically fragile states often preserve such organised crime deliberately in the absence of effective internal control mechanisms and attempt to maximise the economic benefits of the criminalised economy (Hock, 2011).

2.4.3 Ignoring or having a limited engagement with the de facto state: Can de facto states reach out to international society?

In terms of generally ignoring the de facto states and having limited dealings with them, this often takes the form of strict non-access to economic aid from the international organisations or existing states. For the impoverished de facto states, not being able to obtain external aid to build the economy is harmful. James (1986) critiqued this non-access to approach by arguing that the sovereignty of a nation-state should amount to the right to deliver material benefits to the de facto state in economic terms, not merely in terms of providing territorial or geographical border definitions.

As Pegg (1998a, p. 6) stated, '[m]ore costly to the de facto state is the general inability of most intergovernmental organisations and non-governmental aid agencies to deal with non-sovereign entities' while in examining the case of Eritrea, Pool (1993, p. 402) stated that '[a]t independence, Eritrea carries with it a heavy burden of prolonged warfare and famine, enters a world economy in great difficulty and at a time when the guiding orthodoxy of lending is highly restrictive'.

It must be noted here that behind this economic neglect of the de facto states by the international organisations or recognised states, there exists a strategic trade-off by the third-party states regarding the parent state as patronage for these unrecognised entities. In exchange for undermining the de jure independence movements in the de facto states, the recognised states provide the de facto states with economic rewards by granting them access to economic aid from the international organisations, and therefore the

foreign direct investment toward the de facto states is enhanced. However, Hock (2011) argued that the reluctance on the part of the recognised states and the international organisations in terms of enhancing their engagement with the de facto states derives from the long-standing importance attached to the territorial integrity and inviolability of state borders. Regarding this point, Jackson (1987, p. 531) stated the following:

[o]nce sovereignty is acquired by virtue of independence from colonial rule, then extensive civil strife or breakdown of order or governmental immobility or any other failures are not considered to detract from it.

However, some proactive and creative economic initiatives commenced by certain de facto states must be scrutinised. For example, despite the trade embargoes and the travel sanctions imposed on it, Northern Cyprus has made its higher education industry into a lucrative international business that generates handsome economic benefits. This presents a viable alternative economic activity that attracts international students from across the world into the reputable local universities (*The New York Times*, 2014).

Regarding this new development, Koldas et al. (2018) argued that the case of Near East University in Northern Cyprus could be a unique case in terms of how a private non-state actor in a de facto state can reach out to international society. Here, the authors discussed the benefits of internationalisation and branding without established state support even in the absence of international recognition of Northern Cyprus, emphasising the use of the women's sports teams to enhance the international reputation of the university by attracting more international students and enhancing the community-building. This is most notably because universities in Northern Cyprus are excluded from international networks such as Erasmus due to the non-recognition of the nation.

The problem is that for certain extremely impoverished de facto states, such as Eritrea, it is fair to say that it is virtually impossible to expect these entities to produce creative economic measures such as those adopted by Northern Cyprus for economic selfsufficiency and survival. In short, internal conflicts, the interplay of criminality and domestic leaders, and economic weakness in these vulnerable de facto states make legitimate economic activity extremely difficult. In addition, for international society, the political- and security-related costs of ignoring and having limited dealings with the de facto states are only recognised in the long term. Realistically speaking, international society does not appear to be required to pay significant political or economic prices by neglecting the small, impoverished de facto states of today. After all, Eritrea, Somaliland, Chechnya, or the Islamic areas in the Mindanao island of the Philippines are often considered to be of minimum economic and political importance for international society. However, it must be noted that this non-access to international aid for de facto states (due to the limited damage inflicted on regional and international security) is overly simplistic. Granted, it may be possible to ignore the problem in the short term, but it may create a significant problem in the long term.

Fazal (2018) argued that the secessionists tend to be more attuned to international humanitarian law to gain more international recognition, which was the case in the Catalan independence movement as well as in Somaliland and Iraqi Kurdistan. However, Fazal (2018) also pointed out that these secessionists may also conclude that, if their good behaviour in the international society of states is not adequately rewarded, they may turn to further violence to gain their independence. Here, the authors noted the case of South Sudan, which ignored international law and the norms of non-violence

against civilians and still achieved legitimate independence. In fact, Fazal (2018) suggested that an alternative way to alleviate the problem is giving these secessionists access to soft recognition from international economic or financial organisations in the wider world.

Therefore, in the long term, this approach may be counterproductive to the geopolitical stability of the region and, potentially, the globe. In addition, as Pegg (2000) argued, for small, impoverished de facto states, having access to outside help when developing economic activities may make a huge difference to their long-term 'national' security and survival. Furthermore, the recent brutal internal military conflict that occurred in the de facto state of Marawi in Mindanao and which devastated the whole city in 2017 should not be ignored, given that this was due to the general neglect of this de facto state. In fact, Mindanao, including Marawi, is now regarded as having the potential to be transformed into another transregional base for radicalised religious-based terrorism. In the Philippines, under the Duterte administration that began in 2016, one of the security cooperation agendas between the nation and Japan has revolved around the counterterrorism related to the issue of Mindanao (Parameswaran, 2017), which indicates that Japan cannot afford to ignore the de facto state in its region. However, Iwami (2018) argued that the human securitisation of the ODA under the legal reforms of the Japanese security policy is observable and that, in theory, the geographical scope of the Japanese PKOs has been expanded, including in terms of the participation of the non-UN authorised missions and in the more flexible use of force in certain limited conditions. Nonetheless, Iwami (2018) concluded that Japan remains reluctant to instrumentalise the SDF for peacebuilding operations, including in terms of the Mindanao issue, with the nation likely to continue to instrumentalise the ODA to contribute to peace and stability in the post-conflict resolutions in Marawi (Iwami, 2018).

The de facto states have very limited capabilities to reach out to international society unless others accommodate them within this context. However, with the exception of Northern Cyprus, which, as noted above, demonstrated various unique ways to reach out to the international society in a low-key manner, international society is unwilling to give much diplomatic space to the de facto states.

2.4.4 The EU's low-key engagement with de facto states

The current literature on the EU's engagement with de facto states suggests that the EU is employing soft power approaches within its geographical periphery surrounding the former Soviet Union. 'Soft power' means making others do what we want through the attractiveness of the actor (Nye, 2004; Hock, 2011). In other words, the soft power approach of a state actor involves the use of cultural, social, diplomatic, political, educational, or ideational instruments in a non-coercive manner to encourage other countries to behave in a way that resonates with its national interests.

International society's limited engagement generally takes place in the low-key form of humanitarian support, financial support, or the enhancement of democracy or civil society in these unrecognised states to avoid the sensitive issue of secession. Popescu (2007, p. 14) defined the EU's engagements with Abkhazia and South Ossetia as follows:

(i) to decrease the (financial) dependence of the secessionist entities on Russia and to give them an opportunity to diversify their options, ii) to create links between the secessionists and Tbilisi and promote reconciliation, and iii) to promote knowledge about Europe and its values.

Whether or not these approaches are regarded as non-political, they can certainly be questionable. Indeed, it is essential to note that the very existence of an unrecognised state is disputed, which means that any engagement with the disputed entity, such as a de facto state, will inevitably be contested or politicised.

Behind this limited engagement, there lies the intention to depoliticise the engagement as far as possible. Regarding this point, Pegg (1998, p. 8) stated: '[f]or international society, the greatest potential cost to this approach is angering the sovereign state on whose territory the de facto leadership operates'. Meanwhile, Hock (2011) studied the case of the EU's limited engagement with Abkhazia and suggested that while the EU began to provide a humanitarian form of financial aid in 1997 to this war-torn de facto state, the support undoubtedly lacked an organised and coordinated plan and sufficient financial aid was not delivered due to the lack of strategic importance, the geographic distance, and the highly dangerous situation of Abkhazia for the EU. The soft power approach may appear to assist the grassroots development of Abkhazia in terms of, for example, increasing employment and job opportunities for the locals, providing financial and humanitarian assistance for refugees and nutrition programmes for the locals and diaspora groups (Hock, 2011). However, it is clear that the EU does not want to raise political tensions by touching upon such sensitive issues. Nonetheless, it needs to be stated here that the problem is that engagement with any entity can be politically disputed even if the EU intends to keep it apolitical.

In fact, Hock (2011) argued that soft power approaches are only successful in gaining some support from politically active civil society and democratic activists in these de facto states; however, these democratic values, based on the Lisbon treaty, have not necessarily received the wider support of the general population in Abkhazia. If soft power is defined as persuasive power (Nye, 2004), this limited engagement by the EU with Abkhazia requires re-adjustments to encourage the latter to behave in a way that aligns with the former's interests. According to Popescu (2007), the opposition party in Abkhazia often claims that the NGOs dispatched from the EU are often associated with spies and agents working for the Western powers, and that these Western NGOs carry out espionage operations for the Georgian government in Abkhazia. It would appear that regardless of what the EU's intention may be, a degree of politicisation is inevitable in its dealings with Abkhazia.

It is clear that the EU is undertaking a rather low-profile engagement with the de facto states located on its geographical periphery. Since these de facto states still heavily rely on Russia for their strategic survival (Toal and O'Loughlin, 2014), the limited acceptance and acknowledgement of de facto states does not appear to incentivise these states to behave in accordance with the EU's interests. It is highly unlikely that the EU's low-key soft power approaches in engaging with unrecognised states will transform these entities within the EU's periphery into international and regional security assets for the EU.

2.5 Summary

This literature review highlighted the strengths and weaknesses of the existing literature related to the ESIR and to de facto states in general. The strength of the current literature lies in how it has proven the flexibility of its theoretical structure in terms of how international society can accommodate non-sovereign political communities, including de facto states, within its grand narrative, despite its Eurocentric nature.

Meanwhile, relevant studies share the common perspective that de facto states are singularly regarded as regional security issues or the generators of problems within international society. Furthermore, since the recognition of de facto states is likely to create more problems, engagement without recognition is becoming a topic of increasing interest in the current literature on de facto states in international society.

However, none of the relevant studies consider de facto states as potential peacemakers, if they are given opportunities to contribute to regional peace and stability by others. A further weakness relates to the fact that Japan, a non-European regional great power that accommodates outsiders such as Taiwan within international society, has not been addressed by the existing studies in the body of literature related to the ESIR and de facto states.

Therefore, the contribution of the current thesis will be to construct a grand narrative on how Japan has been the main protagonist in expanding international society to incrementally include an excluded entity (Taiwan) in a non-Eurocentric manner. As such, the study will treat Japan as a core member of the institutions in international society, one that regards Taiwan as a subordinate peripheral outsider in a paternalistic way. This new grand narrative is conceptually different to how Japan is included in the Eurocentric international society as a non-European outsider by the Western powers within the context of conventional non-Western approaches to the English School. In fact, as yet, there has not emerged a grand narrative that regards Japan as an essential

insider of the core institutions of international society, one that creates the rules and norms for the regional order through its engagement with an outsider of the society of states (Taiwan).

As will be examined in the next chapter, Japan's accommodation of Taiwan is unique in the sense that it contributes to shaping the rules and norms for the regional order. It will be argued that Japan is socialising Taiwan as a foreign policy instrument to enhance its own regional security interests in accordance with its own standards of civilisation for international society without fully recognising Taiwan and while still officially following the One-China policy within the context of the regional security environment.

Chapter 3. Japan–Taiwan Relations, Japan's Foreign Policy Instruments, and Japan's Socialisation of Taiwan through International Security Norms

3.1 Introduction

The previous chapter elucidated on how the ESIR is theoretically flexible enough to deal with the accommodation of non-sovereign political entities into international society, including de facto states, by the European great powers, even though the entities in question cannot attain fully-fledged membership. This chapter builds on this by explicating how the non-European regional great power Japan accommodates Taiwan into the international society of states for the sake of its own security, without recognising Taiwan within the context of its One-China policy.

First, we review the literature related to Japan–Taiwan relations to build the argument that it is up to Japan to decide what kind of security interactions the two nations will engage in within the hierarchical relations in the absence of Japan's recognition of Taiwan in international society, and that Japan navigates the ambiguities of its One-China policy to incrementally utilise Taiwan as a security asset in ways that do not destabilise Japan–China–Taiwan relations. Second, we review the literature related to Japanese foreign policy instruments to ascertain the security contribution Japan has made to international society, and to elucidate on what the nation is willing to do and what it is unwilling to do in its security engagements with Taiwan in the context of its One-China policy. In addition, we attempt to clarify how Japan's security interactions with Taiwan will remain incremental, regardless of whether it adheres to the Yoshida Doctrine or shifts toward the more radical Abe Doctrine. Third, we conceptualise how Japan has been encouraging Taiwan to behave as if it is a member of international

security efforts based on international security norms for Japan's security, – despite Taiwan's non-access to the multilateral security forums in the absence of any statehood – while pacifying its One-China policy.

3.2 The Hierarchical Relationship Between Japan and Taiwan

In the words of Morris (2017, p.3), 'this is a compelling association in today's Taiwan, where real and imagined traces of "Japaneseness" are treasured for their making Taiwan as separate from "China" proper'.

However, it may be favourable for the ontological security of the Taiwanese identity for the nation's people to regard themselves as at least partially Japanese, albeit that such an incorporation of 'Japaneseness' into what defines Taiwanese identity further reinforces the Taiwanese geopolitical subordination to Japan within international society. As noted in the previous chapter, if accommodated actors such as Japan, China or Turkey – which have historically been incorporated by the Eurocentric great powers into international society – attempt to demonstrate normative compliance with the Western norms, this inadvertently consolidates the Western unilateral moral authority of defining who is civilised and who is not. The same social structure applies to the contemporary relations between the non-European regional great power Japan and subordinated Taiwan. Therefore, since Japan engaged in the civilising mission of the 'barbarians' in Taiwan during the Japanese colonial period, the social structure within which Japan is currently civilising Taiwan in a unilateral manner has changed little. As such, Japan continues to hold the moral authority to judge whether Taiwan is behaving in accordance with the Japanese concept of the standards of modern civilisation for

Japan's security within the context of its One-China policy in the regional security environment. Following this logic, for Japan, Taiwan's interpretation of its history under Japan's colonial rule also functions as a foreign policy instrument.

On this point, Sejrup (2012) argued that Taiwan is always regarded as a pupil of Japan's historical modernity and progress as a metropolitan core, positioning Taiwan as a subordinate periphery in the former Japanese empire. This defines Japan as progressive and modern while defining Taiwan as progressing and modernising. Here, Sejrup (2012) pointed out that Taiwan attempts to learn from Taiwanese essentialised and simplified interpretations of its Japanese past to overcome the present socio-political problems in the nation, and added that when Taiwan finds Japan committing transgressions in its territory – such as exploiting the impoverished local women for prostitution tourism – it simply attempts to justify it by attributing it to the transregional hierarchical economic order and conceding that Taiwan also exploits local women for prostitution in the lesser-developed Southeast Asian region and in China, thus rendering the controversy non-problematic. Sejrup (2012) went on to point out that Japan also notes its colonial accomplishments in Taiwan to highlight the contemporary improvements there in order to raise its status vis-à-vis Taiwan. Clearly, the social structure wherein Taiwan learns from Japan's past to improve its own ongoing progress inadvertently consolidates the normative superiority of Japan over Taiwan in a hierarchical manner. Meanwhile, Japan's misconduct is justified by subordinated Taiwan, which attempts to imagine the transregional hierarchical society by placing Japan at the top of the ladder as the most civilised while conceptualising Taiwan as currently progressing in the middle of the hierarchy and placing the rest of the region as more normatively inferior (Sejrup, 2012). In addition, as Sejrup (2012) stated, conservative politicians in Japan utilise this

Taiwanese imagination of the hierarchical society of states as an instrument to justify the conservative or right-wing interpretations of Japan's past imperial history.

Furthermore, Lim (2018) argued that there exist two major dimensional reconciliations because of the selective remembering and forgetting in Taiwan in terms of its historical reconciliations with Japan. Here, Lim (2018) observed that, on an external level, Taiwan largely achieved reconciliation with Japan even under the KMT regime, and that while Japan and Taiwan have also had various historical disputes, the government in Taiwan is unwilling to jeopardise its relations with Japan for political reasons, unlike with the Koreas and the PRC. Meanwhile, at a domestic level, things are more complicated, and the two different historical memories of the Taiwanese who lived under Japanese colonial rule as Japanese imperial subjects and those who fought the war of resistance against the Japanese in mainland China still foster lingering tensions (Lim, 2018). However, the cultural affinity toward Japan is observable in transgenerational manners, including among those who experienced anti-Japanese education in Taiwan during the KMT authoritarian era, with Taiwanese people becoming fonder of Japanese culture, which demonstrates the Taiwanese flexibility visà-vis Japan. Lim (2018) concluded that the simplistic and binary classifications of the perpetrators and victims should be avoided in the historical reconciliation. It is important to state here that even the KMT under Chiang-Kai-Shek established traditional ties with the LDP under pro-Taiwan figures such as Kishi Nobusuke in the post-war period for the ROC's political survival as a small country, and that Taiwan's political leverage vis-à-vis Japan became even more weakened and limited after the Japanese de-recognition of the ROC in favour of the PRC. Therefore, whether Taiwan likes it or not, it needs to engage with Japan in a peaceful manner for its own survival

in the absence of Japanese recognition. It can be argued that one of the reasons why Taiwan is not interested in bringing up the historical disputes with Japan for political reasons is that, unlike with South Korea and the PRC – which are recognised as sovereign states by Japan – Taiwan remains a de facto state and its subordinate position below Japan significantly reduces its foreign policy leverage vis-à-vis Japan, while a reciprocity of recognition exists between Japan, South Korea and the PRC in a legally equal sense. Given this regional political environment, Taiwan functions merely as an object to be accommodated by Japan in the context of its One-China policy.

3.2.1 Japan-US alliance and Taiwan

Security-related studies on Japan have frequently discussed Taiwan within the context of how it is embraced in the Japan–US alliance as a consumer of extended security deterrence (Kotani and Auer, 2005; McDevitte, 2005; Yang, 2005). Japan is traditionally believed to have taken a low-key approach to the issue of the Taiwan Strait. Matsuda (2018) discussed the implication of the issue of the Taiwan Strait for the Japan–US alliance, including from the perspective of CSD, and argued that while Japan takes lesser roles and missions in relation to the US, it continues to play a substantial role in the security of Taiwan in the context of the Japan–US alliance. Meanwhile, those who argued for Japan's middle power status tend to downplay the nation's role in providing security for Taiwan, even in the context of the Japan–US alliance. For example, Soeya (2011) argued that the Japan–US alliance guideline of the 1990s was meant to deter North Korea, despite Chinese suspicions of Japanese involvement in the issue of the Taiwan Strait.

Due to the lack of international recognition of Taiwan, Japan cannot address its relationship with Taiwan in the same way as it does with the 'official' sovereign states that stand on an equal footing in international society. Moreover, Japan's official adherence to its One-China policy is premised on the government's acceptance of engaging with Taiwan in only socio-economic and civilian terms, not in strategic security terms, let alone in terms of defence ties. Regarding this point, Soeya (2005) argued that unlike the widely held belief that Japan-China-Taiwan relations are becoming more pro-Taiwan and more anti-China, Japan's policy toward Taiwan remains pro-China, while the issue of the Taiwan Strait remains deeply embedded in the Japan-US alliance, meaning Japan is unable to create its own independent strategy on Taiwan (Soeya, 2005). Matsuda (2012, 2013) provided a more nuanced analysis of the security role of Japan as a 'minor' and a 'passive balancer'. Overall, the existing security-related studies involving Japan tend to regard Taiwan's existence as a low-key issue despite Taiwan's geostrategic importance for Japan.

3.2.2 Taiwan's attempt to upgrade its place in Japan's foreign policy initiatives

Lin (2014) rightly pointed out the geostrategic importance for Japan as a trading nation of securing the sea lanes of communication by leaving Taiwan in friendly hands, geographically connecting Taiwan with strategic chokepoints between the East and South China Seas in the first island chains. However, Lin (2014) also non-critically introduced Richard Samuel's argument that while the Japan-US alliance is important for Taiwan's security, Taiwan is more important for the structure of the Japan-US alliance due to the potential risk of 'alliance collapse' without Japanese assistance during the cross-strait conflict due to the strategic linkages between the US's

commitment to the defence of Taiwan and the Japan–US alliance. However, it must be stated here that the Japan–US relationship functions as a global security provider, as exemplified by the Free and Open Indo-Pacific strategy, while the issue of the Taiwan Strait is a region-specific issue. Therefore, it is incorrect to say that Taiwan is more important for the Japan–US alliance than the other way round. However, this does not mean Taiwan has no place in Japan's transregional strategies such as the Free and Open Indo-Pacific strategy. On this point, Matsuda (2019) argued that while Japan is unlikely to mention Taiwan explicitly in its Indo-Pacific strategy, this does not mean that Taiwan has no role, with Japan more likely to utilise Taiwan as a 'hidden asset' in the initiative.

Furthermore, the hierarchical bilateral relations between Japan and Taiwan also help to create diplomatic environments wherein Taiwan's political input into Japan's security policy on Taiwan has a minimum influence, if any. One symbolic example that substantiates this point relates to Taiwan's lively debates on Japan's potential creation of the Taiwan Relations Act (TRA). Here, it is essential to note that the potential enactment of the Japanese version of the TRA is a domestic Japanese law, over which Taiwan has zero control in terms of any final decisions, even though it can be surmised that Taiwan can at least engage in informal or 'behind-the-scenes' lobbying activities. On this point, Taiwan's current de facto ambassador in Tokyo, Frank Hsieh, admitted at the legislative Yuan that it is very difficult for Taiwan to convince Japan to adopt the Japanese version of the TRA since the latter abruptly terminated diplomatic ties with the former in 1972 without creating such a legal document (*Taipei Times*, 2017). Furthermore, in an interview conducted by Tzou Jiing-wen, Kishi Nobuo, the then minister for foreign affairs and the brother of Prime Minister Abe, stated that 'Japan is not currently making any moves toward implementation of a TRA-like policy, but the

National Diet has done research in this area' (*Taipei Times*, 2016). Meanwhile, according to Rira Monma (2018), the head of China Research at the Ministry of Defence in Tokyo, as the current deputy foreign minister, Kishi Nobuo stated that there is the possibility that the Japanese version of the TRA may see progress in the next two to three years, without stipulating a clause for Japanese security involvement in Taiwan, such as the provision of weapons, at the trilateral security convention involving the US, Japan and Taiwan that was held in Taipei in December 2017, where Taiwan was described as an 'important partner'. Therefore, while Taiwan may wish to secure Japan's more reliable security commitment to peace and stability in the Taiwan Strait, it is highly unlikely that Taiwan will be able to obtain such a security commitment from the government of Japan. Overall then, the hierarchical nature of Japan—Taiwan relations is clear, and it is up to Japan to lead the relations in strategic security terms given that it can exercise important political authority to incrementally accommodate Taiwan in the international society of states.

3.2.3 Taiwan's domestic policy analysis of its relationship with Japan

From this perspective, even though scholars working in the field of Taiwan studies analyse the nation's relations with Japan through the perspective of domestic policy analysis, their understanding of Japan is unlikely to be heeded by the government of Japan. For example, the English-language literature on Japan—China—Taiwan relations, such as the work of He (2014), has discussed Taiwan's 'national' historiography vis-àvis its self-image in relation to Japan and China as historical 'significant others' in the context of identity politics in Taiwan. Here, He (2014) argued that such identity politics vis-a-vis the Japanese—Chinese nexus affects Taiwan's foreign policy preference in terms of its relationships with Japan and China. Elsewhere, Chen and Hwang (2015)

analysed the sentiments of Taiwan's younger generations toward Japan and China in relation to the issue of the Senkaku Islands through examining online debates, concluding that even though Taiwan has incorporated Japan largely positively and China somewhat negatively into its national identity construction, this does not necessarily mean that Taiwanese netizens would side with Japan against China in the issue of maritime sovereignty over the East China Sea.

However, as examined earlier, without official recognition of Taiwan by the government of Japan, it remains questionable as to whether Taiwanese identity politics would afford the nation any foreign policy leverage to affect Japan's security policy on Taiwan. Meanwhile, Dreyer (2016) elaborated on the Japan-related policy of each of the Taiwan administrations in chronological order, with reactions from both Japan and China. Here, the author briefly illuminated the Tokyo–Taipei relations under the authoritarian KMT era as an analytical template, specifically elucidating on Japan–Taiwan relations after Taiwan's democratisation, from Lee Teng Hui to the last Ma administration. Dreyer (2016) rightly concluded that Japan–Taiwan relations are likely to improve under the current Tsai administration and that Japan is unlikely to concede to the PRC's demands over Taiwan.

Meanwhile, Deans (2000) examined how the opposition political forces and nationalists in the PRC, Japan and Taiwan exploit the issue of the Senkaku Islands to embarrass the legitimacy of the ruling elites while the ruling elites attempt to downplay this issue. Therefore, for Deans (2000), the symbolic importance of the islands in question within the context of internal politics among these three actors is more important than how the three nationalisms affect the regional political environment.

However, the studies based on Taiwan's domestic policy analysis vis-à-vis its relations with Japan would be unlikely to illuminate 'Japan's security policy' on Taiwan, largely due to Taiwan's unrecognised status, which significantly reduces its diplomatic influence on its more powerful neighbour. Taiwan's 'domestic sovereignty' may grant itself an internal space for lively democratic discussions on its domestic policy on Japan within the context of internal politics, but it must be stated here that such internal debates in the de facto state are not the solution to the nation's international predicament of being in diplomatic limbo, including in terms of its subordinate relations with Japan in international society.

3.2.4 Vulnerable Taiwan within the context of Sino-Japanese relations

Taiwan appears to be concerned about the potential enhancement of the Sino-Japanese relationship and the attendant strategic implications for Japan–Taiwan relations. This was especially true in 2017, since that year marked the 45th anniversary of the normalisation of official Sino-Japanese diplomatic ties, as expressed by the chief secretary, Zhang Shuling, at the Taiwan–Japan Relations Association. As Shuling stated, Taiwan hopes it does not become a 'zero sum game' (Radio Taiwan International Japanese, 6 June 2017). Certainly, Japan's strategic understanding of Taiwan also includes the realist understanding of the balance of power. For example, regarding Japan's strategic thinking on power politics vis-à-vis its relationship with Taiwan and the PRC, Hornung stated the following:

[p]oor relations between Beijing and Tokyo provide Japan with the latitude to seek improved ties with Taipei to hit back on Beijing. In other words, Japan can play a 'Taiwan card' against the PRC. It is not a coincidence that most of the improvements in Japan—Taiwan ties since Abe came into office came against a backdrop of spiralling Beijing-Tokyo relations (Brookings Institute Homepage, 2018).

As examined, Japan has a far greater influence in this bilateral relationship. Notably, this was the case even under the tenure of President Ma Ying Jeou, who has a reputation for being anti-Japanese, when Taiwan signed the fisheries agreement with Japan (Ogasawara, 2015). Here, Takeuchi Takayuki (2012) questioned the political sincerity of the Ma administration's East China Sea Peace Initiative by pointing out that Ma had used provocative language such as the 'non-removal of the use of force' (against Japan) when he was mayor of Taipei. Regarding Ma's record of anti-Japanese activism during his early tenure in 2008, the then chief executive of the Ma administration, Liu Chaoshiuan, even cited 'the use of force [against Japan] if necessary' when referring to the maritime tensions with Japan over the East China Sea that ensued following the collision between the Japanese coastguard patrol ship and the Taiwanese fisheries boat near the Senkaku Islands (Kiyomizu, 2010). Meanwhile, Matsuda (2013) elaborated on how Ma expended significant political effort to negate the Japanese image of Ma as an anti-Japanese figure following this maritime incident, since it was not in Taiwan's national interests to have Ma labelled as anti-Japanese. In other words, there is an established mechanism in Taiwan, which functions to enhance ties with Japan for its own strategic survival within the context of the hierarchical Japan-Taiwan relations, regardless of how the domestic political debates on Japan evolve in Taiwan.

Furthermore, it is essential to state here that Taiwan can maintain its de facto independence only by securing Japan and the US as strategic supporters vis-à-vis its relationship with China. On this point, Leng and Liao (2016) analysed the hedging strategy of the Ma administration in the context of Sino-Japanese relations and argued that the administration tried to maximise Taiwan's autonomy and security interests by enhancing its relations with the US, Japan and China at the same time. However, the point that is not adequately discussed in this study is that despite the suggestion that Taiwan's hedging strategy can be a model for other small and middle-sized regional powers in the face of emerging China, Taiwan will remain a de facto state given the absence of international recognition and the existence of the PRC's One-China principle. As such, Taiwan lacks the foreign policy leverage to influence the dynamics of the Japan-US-China great power relations in any meaningful way and continues to be on the receiving end of this triangulated power relationship. Finally, unlike this study's suggestion of Taiwan potentially being a model for the other small and mediumsized regional powers in terms of its hedging strategy, we must address the fact that since it is an unrecognised state, Taiwan cannot be compared with Japan's other regional security partners. As Leng and Liao (2016) also confirmed, Japan is now a major lynchpin in the regional hedging strategy within the context of Sino-US relations. However, Japan has no intention at all to recognise Taiwan, which is a necessary process if Japan were to upgrade its security ties with Taiwan in a high-profile manner. In fact, Japan will likely continue to improve these ties in an incremental and low-key manner in the absence of recognition of Taiwan, which would ensure that Taiwan does not significantly disrupt Sino-Japanese relations.

3.2.5 The Taiwan issue in Sino-Japanese relations

As discussed in Chapter 2, the defacto states are often regarded as issues to be managed, diffused, and ultimately solved as if these unrecognised entities were regional troublemakers. However, if Taiwan is a source of tension in Sino-Japanese relations, there is a clear strategic need for Japan to discipline the state behaviours of Taiwan to ensure the nation contributes to regional peace and stability. Bush (2019) cautioned the potentially more liberalised use of Taiwan's domestic referendum, including in terms of the issue of Taiwan's legal status, since the PRC might interpret such a development as an act of secession and consequently trigger the use of force. Bush (2019) further questioned whether the US's security commitment to Taiwan was solid enough under the Trump administration in the case of an outbreak of war, alluding to the developments of domestic politics in Taiwan, especially pertinent if Taiwan does not obtain prior approval for such acts from the US. As conceptualised in the ESIR, if the great powers can unilaterally impose the rules and norms on the small states (Bull, 1977; Buzan, 2014), it is possible that Japan would also interfere, disagree, or at least express non-agreeable views regarding the developments in Taiwan. This would be especially pertinent if Japan incrementally expands the use of Taiwan as a foreign policy instrument vis-à-vis the PRC. One needs to be reminded of the fact that the former de facto Japanese ambassador in Taiwan, Uchida Katsuhisa, expressed some political disagreement regarding Taiwan's domestic referendum of 2004, which led to Japan's concern over the potential increase in cross-strait tensions and the subsequent adverse effects on Japan's security and national interests (Uchida, 2006, pp.186–194).

Elsewhere, Wang (2000) argued that while Japan enormously benefitted from the 1972 system, where it simultaneously enhanced its economic and political relationships with China and its economic relationship with Taiwan, this geopolitical arrangement has subsequently eroded, largely due to the Taiwan factor in the revised Japan–US alliance guidelines of 1997. Wang (2000) concluded that while Japan's strategic preference remains a peaceful solution of the cross-strait issue involving China and Taiwan, the strengthened Japan–US alliance locked Japan further into the possible future cross-strait military contingency, albeit reluctantly. Therefore, from this geopolitical perspective, the Taiwan issue is a major cause for concern in terms of Sino-Japanese relations. While Wang (2000) appeared to erroneously believe that Japan recognised China's claim over Taiwan, he also wrongly suggested that this interpretation means Japan's Taiwan policy differs from that of the US. Here, it is important to stress that while Wang's misinterpretation of Japan's position on Taiwan must be corrected, this study focuses on advancing the view that Japan's Taiwan policy is indeed unique and creative in terms of its security measures, as will be examined later.

Meanwhile, other analysts approached the Taiwan issue from the perspective of Japan's past imperial history. For example, Zha (2001) argued that while Taiwan's inclusion in the Japan–US alliance is troublesome for the PRC, the Beijing government tends to hold that the Taiwan issue is, in strategic terms, more of an issue that must be dealt within the context of Sino-US relations. Here, Zha (2001) suggested that the greater concern regarding the Taiwan issue in Sino-Japanese relations is how the Taiwan-independence advocates in Taiwan and the Chinese Communist Party in Beijing disagree in terms of their respective interpretations of the historical legacy of the Japanese Empire, while the two sides remain sensitive to the Japanese governmental

stance on Japanese imperial history, albeit in different terms. As Zha (2001) noted, this has created both historical and cultural divisions on the two sides of the Taiwan Strait, even though the Taiwan issue is more of an 'irritant' than a 'destroyer' in Sino-Japanese relations.

It is noteworthy that analysts such as Lynch (2018) are expressing their concerns regarding the Trump administration's use of the Taiwan card vis-à-vis China, while other analysts, such as Mazza (2018), are more concerned with the Taiwan–China tensions, with their views seemingly favouring President Tsai's more restrained approach to the cross-strait tension over President Xi's more nationalistic approach, be it real or perceived. Meanwhile, Gries and Wang (2019) argued that Taiwan's underestimation of the PRC's resolve to wage the cross-strait war and the Taiwanese distrust of its military's self-defence capabilities, the PRC's underestimation of the US's political will to defend Taiwan, and the unreliability of the Trump administration may all contribute to miscalculations within the potential cross-strait conflict. If this hypothesis is true, Japan's prudent behaviour vis-à-vis Taiwan is, in conjunction with the Japan–US alliance, even more critical for the regional order. In fact, as Kawashima (2019) pointed out, the Sino-US confrontation, and the US's more visible treatment of Taiwan under the Trump administration in relation to the PRC, have not had a direct impact on Japan's relations with Taiwan.

3.2.6 Taiwan as a security asset for Japan

As examined earlier, Taiwan has no foreign policy leverage to influence Japan's Taiwan-related policy. However, this does not mean that Taiwan cannot be a security asset for Japan. In fact, as Chow (2016) noted, while Taiwan may have only limited access to international economic forums such as the WTO, the APEC, and the Asian Development Bank, it remains a developed economic entity to be reckoned with and is still a crucial global supply chain. Furthermore, Taiwan aspires to being a member of the TPP and the RCEP in the future, despite the PRC's political obstruction. In fact, in this sense, Taiwan needs to work on its internal reforms regarding its regulatory regimes if it is to substantially improve its economic efficiency and transparency in compliance with global standards. Here, Chow (2016) pointed out that even though Taiwan's participation in the TPP may have limited economic dividends for both Japan and the US, the positive net effects for the social well-being in these two countries are potentially huge. It is essential to state here that this means Japan is a significant player given its significant economic status (the largest) in the TPP and the RCEP (the second largest after China), especially after the withdrawal of the US from the TPP's negotiations under the Trump administration. Taiwan is a regional medium-sized developed economic power, and economic power remains an important component in the balance of power for the ESIR (Bull, 1977). Therefore, if Japan invests political capital in its relationship with Taiwan, the latter can still be a security asset for the former, provided Japan can assist in granting Taiwan membership to the trans-regional economic blocks.

Nevertheless, deLisle (2016) argued that Taiwan must adopt more creative approaches if it is to meaningfully participate in international organisations or become a stakeholder in the regional security affairs, such as the issues surrounding the East and South China

Seas. Here, deLisle (2016) largely discussed the role of US-Taiwan relations in improving Taiwan's international space in addition to the role of Taiwan's relationship with Japan as a supplementary instrument for enhancing the former's presence in the diplomatic endeavours. However, if Taiwan wishes to be recognised as a military buffer for Japan, as the former Japanese diplomat in Taiwan Uchida Katsuhisa (2006) noted, deeper investigations into how Taiwan can be a security asset for Japan are required. Furthermore, if Taiwan is regarded merely as a military buffer for Japan, Japan does not need to creatively reconceptualise how Taiwan can be a security asset beyond the military defence of its southern flank. To maximise Taiwan's underappreciated security resources as well as its economic and technological power, democratic norms, and geographical importance for Japan's security – even in the absence of international recognition – Japan's engagement with Taiwan requires more nuanced investigations, as will be undertaken herein.

Meanwhile, other academics, such as Fukuda (2018), have noted the importance of Japan simultaneously balancing its relationships with both China and Taiwan such that it can gain maximum benefits from both sides. Here, Fukuda (2018) argued that while maintaining peace and stability is the most important factor for Japan's national interests regarding the issue of the Taiwan Strait, given the PRC's more active security policy vis-à-vis the East and South China Seas, Japan's relationship with Taiwan is essential due to Taiwan's geostrategic position, even though the complex dynamics of Japan–China–Taiwan relations render Japan–Taiwan security cooperation highly problematic. However, even Fukuda (2018) admitted the existence of military information sharing between Japan and Taiwan through the deployment of the retired SDF officials and the active officials of the Japanese coastguard in the Japan–Taiwan

Exchange Association in Taipei, the track 2 and track 1.5 security talks between Japan and Taiwan, and so on. This means that Japan has been navigating the complexity of the One-China policy to incrementally improve its security cooperation with Taiwan in ways that would not destabilise Japan—China—Taiwan relations. In other words, if Japan uses its security relationship with Taiwan more creatively, it can utilise Taiwan as a security asset for its own regional security interests. Overall then, low-key but creative security interactions between Japan and Taiwan are essential.

3.2.7 Japan's quiet and creative diplomacy vis-à-vis its relationship with Taiwan

The hierarchical relations between Japan and Taiwan, and the existence of the China factor, result in the political need for Japan to resort to creative diplomacy to navigate the ambiguities of the One-China policy. The positive aspect of creative diplomacy is that it helps conceptualise Taiwan as a quasi-dialogue partner or even as a low-key and hidden but cooperative partner for regional security affairs, albeit incrementally.

Kawashima (2016) argued that the stable cross-strait relations between China and Taiwan under the Ma administration led to the PRC taking a non-interference stance vis-à-vis Japan—Taiwan relations, and advanced the argument that Japan and Taiwan were able to sign high-profile agreements, including the fisheries agreement, without a significant negative reaction from the PRC. Furthermore, Kawashima (2016) concluded that Japan—Taiwan relations are closely structured in relation to the 1972 system and the Japan—US alliance. Certainly, as was discussed above, the China factor continues to affect the development of Japan—Taiwan relations, and Japan's security policy

remains centred on the Japan–US alliance. However, such a mechanistic understanding understates the nuances of Japan's creative diplomacy, which incrementally expands Japan's security engagement with Taiwan, even under the framework of the 1972 system and the Japan-US alliance. Despite the recent cross-strait tensions under the Abe and Tsai administrations, it is noteworthy that Japan and Taiwan enhanced their law enforcement cooperation in the form of the anti-smuggling agreement, the expansion of intelligence sharing, and the coastguard exchanges, including the conclusion of the bilateral maritime rescue and search agreement, as will be examined in the chapters on the ICAO and CSD.

Meanwhile, Thomas and Williams (2017) argued that in terms of Japan–Taiwan relations, Taiwan's para-diplomacy conducted via subnational interactions with its counterparts in Japan enhances the attendant relations, and the hybridity of high-politics and low-politics can be blurred in this context since the low-politics in Taiwan often affect the enhancement of the high-politics. This development was manifested in how Taiwan's massive civilian financial donations provided to Japan further improved the Japanese people's positive sentiments toward Taiwan, and in how, as a window for negotiations, the Japanese non-official proxy authority also responded to Taiwan's overtures for the conclusion of the fisheries agreement, as will be examined in the chapter on this agreement. Elsewhere, Thomas and Williams (2017) analysed how certain subnational actors in Taiwan often play the proxy role of the central authority, including in the negotiations over the fisheries agreement with Japan, and discussed how the central authority in China is sensitive toward the subnational ties among local authorities in Japan and Taiwan that bypass the bilateral government-to-government ties in the context of the One-China principle. It is certainly true to say that Taiwan is

fairly adept at creatively enhancing its ties with Japan – perhaps with the exception of the MOFA-based central authority diplomacy – in the absence of Japan's governmental recognition, and there is certainly room for further expansion of such subnational ties between the non-state actors from Japan and Taiwan in economic, cultural, social or even occasional political terms. Meanwhile, as argued by Deans (2001), while there are a number of de facto institutionalised bilateral channels between Japan and Taiwan regarding routine tasks such as issuing visas – which may function in the role of de facto embassies due to the nature of informal politics – Taiwan has to rely on informal channels to enhance its ties with Japan, channels that present clear limits to what is possible, especially in terms of high-politics. Specifically, Deans (2001) focused on the role of Nikkakon, the group of Japanese lawmakers that engages in informal parliamentary diplomacy with Taiwan in the absence of official ties. However, as Deans (2001) pointed out, such informal ties are somewhat fragile since they are significantly influenced by the overall directions of Japanese domestic politics, and the Taiwanese reliance on Japanese individual politicians can be personality-based.

However, these studies failed to address how, in terms of security and military ties, the role of the Japanese state is essential since the Japanese central authority maintains a grip on the SDF, the coastguard, and the national police agency for national security, while it also failed to address how it is up to the government of Japan to decide whether Japan is interested in enhancing its security ties with Taiwan. As will be examined later, Japanese informal diplomacy does play a role in enhancing the security ties with Taiwan, especially since Japan's One-China policy is premised on non-military, non-security and/or non-political interactions with its neighbour. In other words, it must be pointed out that the government of Japan uses proxy diplomacy in creative ways to

communicate regarding the strategic, military, and security issues with Taiwan through the track 2 and track 1.5 security dialogues via the retired officials and active academics operating as proxies, including the participation of former prime ministers such as Abe (before he became the prime minister again), the parliamentary diplomacy, or the informal window of the communication channels with Taiwan in the absence of official ties, as will be elaborated on in the three case study chapters.

Importantly, in contrast to the times within which Deans (2001) discussed the limited roles of the institutionalised informal communication channels that were not capable of handling the high-politics in 2001, it would appear that Japan has incrementally expanded the roles of such informal communication channels with Taiwan, including within the security dimension. Yeh et al. (2015) argued that Japan and Taiwan concluded the fisheries agreement by shelving the issue of sovereign ownership of the Senkaku Islands and by allowing the fishing vessels of both sides to fish without being interfered with by the respective maritime law enforcements within the de facto EEZs through the non-official (yet officially empowered) proxy agencies in 2013. Nevertheless, Japan continues to consider the China factor due to the PRC's claim over Taiwan, and this special treatment only applies to Japan and Taiwan within the context of the fisheries agreement over the East China Sea. Importantly, Yeh et al. (2015) pointed out that, in fact, Japan considered Taiwan to be a claimant separate from the PRC in relation to the issue of the East China Sea, and effectively eliminated its strategic concerns over the cross-strait united front, which could have placed Japan under significant geopolitical pressure regarding the defence of the East China Sea. In fact, the two sides utilised an amicable bilateral relationship that differs in nature to Japan's other bilateral ties in Northeast Asia. Yeh et al.'s (2015) collaborative work on

the Japan-Taiwan fisheries agreement helps to illuminate Japan's strategic calculation that resulted in this creative diplomacy, that is, Japan's utilisation of the aforementioned semi-official proxy agencies such as the former Interchange Association, which engaged in negotiations with Taiwan, albeit with the attendance of officials from governmental agencies from both sides. Here, even in the absence of the Japanese government's lack of recognition of Taiwan, Taiwan was tacitly recognised as an 'semiofficial actor' in the issue of the East China Sea, while the bilateral fisheries committee was established such that the follow-up issues could be handled in a bilateral context. While Yeh et al.'s (2015) assertation that the creation of a regional fisheries management organisation that includes Japan, China, South Korea and Taiwan in the East China Sea will not be seriously considered due to Taiwan's lack of international recognition and its lack of foreign policy leverage, it can be stated here that this does indicate that a relatively low-key and minor issue, such as the conclusion of the fisheries agreement with Taiwan, represents significant geopolitical behaviour in terms of Japan's creative diplomacy, while utilising Taiwan as a security asset, as will be elaborated in the chapter dedicated to the fisheries agreement.

3.2.8 Semi-institutionalised Japan—Taiwan proxy communication channels and Sino-Japanese great power management

If Japan is to utilise Taiwan as a security asset, the former must communicate with the latter regarding security issues while navigating the ambiguities of the One-China policy. Here, Japan must engage through creative means. In other words, it is clear that the incremental semi-institutionalisation of the informal communication channels between Japan and Taiwan that define the latter as 'a dialogue partner' or 'a de facto

stakeholder' are gradually becoming semi-normalised, which is helping to encourage Taiwan to behave in accordance with Japan's security preference or its standard of civilisation within the international liberal order, while not excessively disrupting Japan's relations with the PRC in the context of its One-China policy for regional great power management.. Therefore, while Taiwan suffers from its international exclusion, Japan has been providing help to allow its neighbour to enhance its links with international society through the alternative means of quasi-formal communication channels to incrementally accommodate Taiwan into the regional security environment in ways that do not require Taiwan's statehood for membership.

Specifically, for the case of Taiwan, Bartmann (2008, pp.117–118) stated that 'Taiwan's unique position in the international system is that it is able to follow established diplomatic protocols in some situations', adding that it 'must resort to paradiplomacy in most of its critical relationships with other states'. Elsewhere, Abb and Yang (2018) argued that while Taiwan gives greater autonomy to the smaller and medium-sized think tanks in terms of creating ideational diversity under the democratic regime, in comparison with the highly centralised think tanks that are politically loyal to the CCP in the PRC, Taiwan has skilfully used the established major think tanks for the track 2 diplomacy and has nurtured personal ties with its counterparts in the region amid its international isolation and the PRC's sanctions. Furthermore, as Abb and Yang (2018) noted, Taiwan's major think tanks are coordinating with the Kyoto University's prestigious Southeast Asian Studies programme to enhance President Tsai's top foreign policies, such as the New Southbound Policy, to explore collaborations with their counterparts in Southeast Asia. Regarding this point, the former de facto Japanese director in Japan-Taiwan Exchange Association Ogata Makoto (2016, p. 20) pointed

out that while the PRC is indicating its stance in attempting to constrain the Tsai administration's foreign policy behaviour, Japan and Taiwan need to strengthen the track 2 diplomacy and the civilian exchanges between the two nations, especially since Taiwan has numerous potential high-profile officials in waiting in Taiwanese think tanks, who may play a central role in the decision-making body of the government if there is a change of administration in Taiwan.

Here, it is essential to point out that Japan's dual relations with Taiwan and the PRC still operate under the framework of its One-China policy, and that Japan will not destabilise the triangulated relations for the sake of enhancing its ties with Taiwan, even though the nation is utilising Taiwan as a foreign policy instrument for its own security and that of the region in general. In this process, while Japan may incrementally semiofficialise its informal communication channels with Taiwan, the management of the inevitable tensions between Japan's One-China policy and the skilful use of semiinstitutionalised informal communication channels in a way that does not incur the wrath of the PRC are essential for Sino-Japanese regional great power management. Regarding this point, President Tsai (Office of the President ROC, 2 March 2019) asked Japan to begin bilateral security talks through the respective communication channels in ways that respect Japan's preferred form and prioritise the substance of security communication, while she also expressed her understanding of the legal difficulty that stems from the absence of Japan's legal recognition of Taiwan, subsequently requesting Japan to overcome such obstacles in an interview with Sankei Shimbun. However, it is clear that Japan is not interested in Taiwan's plea. Meanwhile, Kingston (2018) argued that Japan has become more resolute in enhancing its ties with Taiwan in a way that is less attentive to the PRC's sensitivity on Taiwan in recent years, especially under Abe.

In the light of this regional strategic environment described by Kingston (2018), the current research will observe how Japan has incrementally expanded the scope of such informal proxy communication channels with Taiwan. If Japan concludes that Taiwan is improving the former's security, it may reward Taiwan by widening the opportunities for a greater range of foreign policy exchanges in semi-official communication channels. Furthermore, as was discussed above, the role of Nikkakon in the political communication with Taiwan is becoming ever more critical in Japan—Taiwan relations, and other forms of 'quasi-official' informal communication channels are emerging to facilitate the bilateral discussions and negotiations of the relevant regional security issues, as will be examined in the case study chapters.

3.3 Literature on Japan's Foreign Policy Instruments

This section reviews the literature on the recent developments surrounding Japan's foreign policy instruments to elucidate what the nation is willing to do and what it is not willing to do in terms of its security relationship with Taiwan within the context of its One-China policy. As noted in the introductory chapter, it is up to Japan's political decision-makers to determine the acceptable framework, meaning Japan has the overwhelming authority to decide what is possible and what is not possible in its security interactions with Taiwan in view of appeasing the PRC. Here, Japan carefully manoeuvres the ambiguity of its One-China policy so as not to breach the framework too much while incrementally maximising Taiwan's potential for enhancing Japan's security in the region. The section also aims to elucidate on how Japan's utilisation of Taiwan as a foreign policy instrument for its own security remains more incrementalistic in comparison with Japan's other security partners vis-à-vis its

complicated relationship with the PRC, regardless of whether Japan adheres to the Yoshida Doctrine or shifts to the more radical Abe Doctrine. Importantly, if Japan's security interactions with Taiwan are deemed to enhance normative regional order via non-kinetic and non-traditional security cooperation, Japan will be more willing to utilise Taiwan as a foreign policy instrument within the context of its One-China policy. It is noted here that Japan's non-traditional security cooperation with Taiwan must not involve high-profile kinetic military components, and must be regarded as normative, humanitarian and non-coercive from the perspective of international security norms and Japan's domestic pacifism. Thus, Japan-Taiwan non-traditional security ties will not breach its One China policy.

3.3.1 Yoshida Doctrine or Abe Doctrine?

Singh (2016) argued that Japan's security identity that was shared among the Japanese institutions during the Cold War, was the peace-state security identity that shaped and constrained Japan into being a minimalist, economic-based, and reactive state in terms of its security, which favoured its subordinate position in the Japan–US alliance that was built on the Yoshida Doctrine. In the post-Cold War period, this was replaced by the internationalist identity that shaped Japan's security policy into being more internationalist and more proactive in giving Japan a more balanced role both within the context of the Japan–US alliance and outside of it. Furthermore, as Singh (2016) noted, Japan's peace-state identity was built on the civilian bureaucrats who prioritised the MOFA and the MOIT in Japan's security policy-making body, reducing the MOD to a weaker position in the process, while the change in the balance of power during the post-Cold War period in Japanese domestic institutional politics allowed the

internationalist security identity to flourish, which increased the role of the SDF and the MOD in the framework of the Japanese security policy-making body. Meanwhile, the 'presidentialisation' that afforded the prime ministers a greater role allowed for swifter and more powerful security decision-making actions, which was manifested in the refuelling missions in the Indian Ocean and the dispatch of the SDF in Iraq.

Meanwhile, Pehlivantürk (2016) argued that Japan has been shifting from being a pacifist or peace state, wherein the concept of the national role stemmed from the Yoshida Doctrine, to an international peacekeeping state, utilising human security and comprehensive security in greater degrees through the more active use of the PKO for international security in the post-Cold War era. This is largely due to international pressure, the changes in domestic leadership and the social norms, and Japan's stronger desire for international prestige, which includes its aspirations to be a permanent member of the UNSC. As such, Japan's national conception role expanded the territorial conception of its national security from one of homeland defence to one of world security through peacekeeping. Here, it is clear that the evolution of Japan's security identity in the post-Cold War era also enhanced its more proactive stance in its engagements with Taiwan while being less attentive to the PRC's hypersensitivity on Taiwan, as will be discussed in the chapter on the fisheries agreement.

Meanwhile, Lande (2018) claimed that in examining the reassurance, arms, and alliance building, Japan's security policy vis-à-vis China continues to be explained in terms of defensive realism. In Lande's view (2018), even under Abe, Japan has not deviated from the Yoshida Doctrine, and insists that Japan has certainly increased its capabilities both independently and in its role in the Japan–US alliance, while it continues to focus

on homeland defence. It is essential to state here that Taiwan is Japan's de facto defence proxy or at least a less-recognised military buffer. In short, Taiwan functions as a deflective barrier in the face of the PLA's significant military resources, which could otherwise be directed at Japan. Indeed, the mere existence of Taiwan as a separate entity to the PRC reduces the Chinese military pressure on Japan's southern flank. Furthermore, Japan's recent increased military presence in the East China Sea is encouraging Taiwan to accept its neighbour's defence strategy, albeit indirectly, which is exemplified by the way that Taiwan has been paying close attention to Japanese deployments of land-based anti-ship missiles in the Nansei Islands, as will be examined in the chapter on CSD.

Meanwhile, other authors, including Hughes (2017), have asserted that Japan is shifting to the Abe Doctrine, which means it is more likely to resort to the use of force in order to assist the US's war efforts due to international structures such as the Japan–US alliance, the indivisibility of the US's national security interests with those of Japan, and the weakened domestic hurdles for restraining Japan's domestic decision-making bodies. However, it is clear that regardless of whether Japan adheres to the Yoshida Doctrine and continues to pursue international security incrementally or whether it shifts to the Abe Doctrine and seeks a more a radical shift in national security policy, its security interactions with Taiwan remain unquestionably incremental. As will be examined in the chapter on CSD, Taiwan's ambiguous status in the Japanese conception of CSD in the absence of international recognition, Taiwan's role in the Sino-Japanese regional great power management issue, and the absence of Japan's recognition of Taiwan's statehood render Japan's security interactions with Taiwan gradualist at best. Therefore, Japan's security interactions with Taiwan offer the least researched case to

demonstrate how the former's foreign policy is not determined by foreign pressures alone. Precisely because Japan's security interactions with Taiwan face numerous hurdles, this offers a unique case study for elucidating on how Japan creatively utilises Taiwan as a foreign policy instrument to enhance its own security vis-à-vis the PRC in ways that will not excessively antagonise its powerful neighbour.

3.3.2 Japan's utilisation of international norms and rules as foreign policy instruments

The current foreign minister, Kono (2018), emphasised that the key aspect for the Indo-Pacific strategy is Japan's strength in enhancing the 'rule of law' and its 'high-quality agreements' while cooperating with the various stakeholders, including the PRC, and stated that Japan's strategic vision does not go against the PRC's One Belt One Road; rather, it aims to ensure Japan's contribution will complement the PRC's vision. Therefore, it must be stated here that Japan is not necessarily interested in confronting the PRC.

Nonetheless, Asplund (2018) argued that Japan is only interested in linking the provision of the ODA to the ASEAN states as a collective unit as a way of 'othering' China to enhance the strategic upper hand in the context of Sino-Japanese competition in the region. Here, Asplund (2018) pointed out that Japan is not interested in bringing up the issue of universal values such as human rights or democracy in providing the ODA in the context of bilateral ties with the individual ASEAN member states for political reasons, such as Japan's history of regional imperialism, the risks of pushing the ASEAN member states further into the geopolitical orbit of China, and the non-

subscribing attitudes of the illiberal regimes among the ASEAN member states, except for the enhancement of the maritime rule of law through the capacity building of the maritime law enforcements for the littoral states in the South China Sea. For these reasons, Asplund (2018) argued that Japan is probably not interested in being a normative power in the region, except for the relatively successful case of Japanese normative diffusions of enhancing the maritime rule of law as a foreign policy instrument to achieve the upper hand over China in the region. Meanwhile, Dreyer (2019) claimed that Japan and Taiwan will resort to their mutual geostrategic importance, their shared democratic values, and the generally positive perceptions of the Japanese colonial era in Taiwan to improve the bilateral ties, while neither side is interested in antagonising the PRC, even under Abe and Tsai.

As such, it can be stated here that it would appear that Japan is utilising additional foreign policy instruments when it comes to Japan—Taiwan relations, such as the shared democratic values and its colonial ties with Taiwan, as opposed to its individual bilateral ties with the ASEAN member states, where these normative and ideational factors are retained as marginal priorities on bilateral levels, despite the geopolitical fact that Japan utilises its strategic ties with both Taiwan and the ASEAN member states to enhance its power balance in relation to the PRC. In other words, Japan can utilise universal values such as human rights and democracy and its past imperial history to enhance its ties with Taiwan in positive ways, which is not the case for its relationships with its individual partners in South East Asia.

Elsewhere, Foot (2017) stated that contrary to the power transition theory, the US has used great power management techniques as a foreign policy approach to stabilise its

power relationships with Japan and the PRC. According to Foot (2017), the US instrumentalised its alliance with Japan from the 1990s on to encourage the latter to increase its security role within the Japan–US alliance for regional and global security purposes. As such, Japan would not be an independent strategic power capable of challenging the US's primacy. In addition, the US initially considered the PRC to present a weak counterbalance vis-à-vis the Soviet Union, and encouraged the normative alignment by helping the PRC to join the WTO with the caveat that it would encourage the PRC to liberalise its political regime in the future, such that the PRC and the US would converge in mutual national interests. Here, Foot (2017) pointed out that the problem is that while the US and the PRC achieved a degree of convergence in their shared understanding of global governance, the PRC ended up being a military challenger to US dominance in the region while Japan did not develop into such a comprehensive challenger due to the Japan-US alliance. Foot (2017) went on to highlight that in terms of Sino-Japanese relations, while the PRC is unwilling to give Japan the great power status that Japan feels it deserves, Japan is enhancing its internal balancing and external balancing, including in terms of the capacity building of the Philippines and Vietnam and forging greater political and strategic ties with India and Australia. It is important to state here that, at present, Japan is not showing any significant sign of potentially violating its One-China policy in its relations with Taiwan, with the exception of the case of the fisheries agreement, which can be interpreted as a sign of the violation of great power management if adherence to the One-China policy is defined as a rule and the code of conduct in Sino-Japanese relations.

3.3.3 Japan's non-kinetic use of the SDF for peace diplomacy

Patalano (2015) argued that while the coercive nature of the naval forces and the attendant regional developments may lead to regional security competition, in terms of Japan, the nation has also instrumentalised the MSDF for trust-building through the HADR for Japan's foreign policy. Specifically, Patalano (2015) pointed out that natural disasters are becoming increasingly significant regional risks to human security, particularly in the Asia Pacific region, and since Japan has a complex military history in view of its past imperialism, the existence of Japanese domestic pacifist norms – and the fact that Japan is a maritime state itself – the role of naval diplomacy for peaceful purposes in the context of the HADR will continue to be essential for Japan's regional security initiatives. It is essential to state here that Japan's peaceful utilisation of its MSDF is manifested through the enhancement of security relations with other security partners.

For example, as Patalano (2016) argued, despite the fact that Japan–NATO security relations have undergone gradual improvements, it was the anti-piracy operations by the navies of Japan and NATO that enhanced the naval cooperation, interoperability, and operational coordination. As such, Japan uses its MSDF as a foreign policy instrument for naval cooperation to enhance global common interests, such as the sea lines of communication, maritime governance and stability, and to combat any international non-traditional security threats (Patalano, 2016). Meanwhile, Patalano (2016) admitted that the return of sea control and traditional naval deterrence cannot be dismissed due to the rise of China and a resurgent Russia. However, Patalano (2016) also pointed out that even China and Russia joined the anti-piracy operations in the Gulf of Aden, which means the era of global naval cooperation continues to hold sway, including in terms of Japan's instrumentalisation of its MSDF for peaceful purposes.

Clearly, since Japan's One-China policy disallows it from engaging in explicit defence ties with Taiwan, a high-profile defence diplomacy with Taiwan is not an option, except in terms of some form of ad hoc intelligence sharing. However, Taiwan's autonomous self-defence capability would help to contribute to the stable development of the crucial sea lines of communication around Japan's southern flank, albeit indirectly, provided Taiwan remains a de facto state. In addition, Japan has been indirectly involved in the defence of the sea lines of communication near Taiwan since the later part of the Cold War, as exemplified in the 1,000 nautical miles of defence, as will be examined in the chapter on CSD.

3.3.4 Japan's instrumentalisation of the security partnership to enhance defence interoperability and normative power

Bacon and Burton (2018) argued that although it was NATO that initiated the concept of natural partnerships and strategic parallelism as strategic narratives, Japan was a good recipient of such strategic narratives, and Abe instrumentalised such strategic narratives to enhance the security and political relations with NATO by conceptually linking the instability of Eastern and Central Europe (e.g. Ukraine) and the issues of the East and South China Seas for Japan. Furthermore, NATO's concept of collective security was also instrumentalised by Japan to enhance military interoperability, including in terms of the issue of piracy in the Gulf of Aden and in the horn of Africa. While Japan does not regard NATO as a major participant in the hard security issues of the Asia Pacific region, especially in terms of the issue of the East China Sea, it still views it as a normative and political power, one that shares the norms of rules-based

order and the rule of law in the international liberal order. For certain, Japan is still somewhat displeased with how NATO mishandled the issue of Crimea, and with the general expansion of NATO and the EU. Japan made an overture to Russia, and both sides are still committed to the development of these strategic narratives in enhancing ties between the nations.

Similarly, it can be stated here that Japan also uses shared liberal democracy, human rights, and the notion of an 'important partnership' as strategic narratives vis-à-vis its relationship with Taiwan. Much like how Japan and NATO diverge in terms of their respective interests with Russia, Japan is not necessarily willing to commit itself to Taiwan's security explicitly due to the concerns over antagonising the PRC. Nevertheless, the mere existence of Japan as a regional power does provide the geostrategic balancing effects for the survival of Taiwan vis-à-vis the PRC.

Vosse (2018) argued that Japan instrumentalised its anti-piracy operations to enhance its naval cooperation with international actors such as NATO, the EU, and other independent deployers in the Gulf of Aden. Here, in obtaining significant operational experience from working with non-US partners, Japan proved to both its own people and the international community that it has the capabilities to deploy its naval and aerial forces in the distant military bases in Djibouti, utilising its own successful regional anti-piracy contributions in Southeast Asia – such as ReCAAP and the Information Sharing Centre – to diffuse such international security norms in the Gulf of Aden. The nation thus gained an opportunity to contribute a Japanese commander to the multinational force, and somewhat influenced its own international security policies, such as Abe's CSD policy. Furthermore, as Vosse (2018) concluded, in terms of de-centring from the

US, Japan indeed increased its own foreign policy leverage and independent international security contributions from this experience. It is essential to state here that as Japan gains more independent international security instruments besides the Japan–US alliance, this also influences its autonomous foreign policy leverages vis-à-vis the cross-strait issues. However, Japan will not engage in high-profile security partnerships with Taiwan akin to Japan's security ties with NATO, the EU, and other international actors due to its One-China policy. Elsewhere, Wu (2018) pointed out that Taiwan cannot even participate in multilateral security forums, including ReCAAP.

3.3.5 Japan's utilisation of non-traditional security efforts in accordance with international norms

Ishizuka (2013) argued that Japan's instrumentalisation of the PKO, the anti-terrorism units, and the HADR for international security would raise its status and prestige in international society, would improve its security vis-à-vis non-traditional security threats and would enhance its prospects of being a permanent member of the UNSC. However, Japan still needs to improve its overall crisis management capabilities including in terms of its rapid deployments, logistical capabilities, and a more flexible use of force by amending its constitutional restraints (Ishizuka, 2013). In addition, Ishizuka (2013) pointed out that while Japan's instrumentalisation of the PKO and the anti-terrorism units is political, the HADR is more normative, which means the latter is the least controversial approach, while the capabilities required for non-traditional security issues are all interconnected and cannot be developed independently.

Elsewhere, Iwami (2018) argued that contrary to the realist arguments made by Hughes (2015), which overstate the security reform initiated by Abe as revolutionary and radical, Japan's security reform appears to be more incremental, as is argued by constructivists such as Oros (2015, 2017a, 2017b) and Singh (2013a). Here, Iwami (2018) claimed that even though the human securitisation of the ODA under the legal reforms of the Japanese security policy is observable and, in theory, the geographical scope of the Japanese PKOs has been expanded, including in terms of the participation of the non-UN authorised missions and the more flexible use of force in certain conditions, Japan remains reluctant to instrumentalise the SDF for peacebuilding operations, including in the issue of Mindanao. Iwami (2018) concluded that Japan is likely to continue to instrumentalise the ODA to contribute to peace and stability in the post-conflict resolutions in Marawi.

Meanwhile, Matsuda (2012) pointed out Japan would not yield to its One China policy if Japan's cooperation with Taiwan in global security is humanitarian in nature or may affect Japan's public health issues without Taiwan's participation. Japan has enhanced the bilateral non-traditional security cooperation with Taiwan by increasing the bilateral coastguard search and rescue cooperation, as well as by increasing its anti-terrorism activities, its anti-transnational organised crime cooperation, and its economic security cooperation with airspace and aviation security implications – such as the bilateral Open Skies Agreement, and the civilian HADR cooperation – in ways that do not conflict with its One-China policy in view of enhancing its security, as will be examined in the chapters of CSD and the ICAO. Since Taiwan is geographically located in a geostrategic chokepoint that connects Japan, South East Asia and the PRC via the interconnected seas and airspace, the role of non-traditional security in enhancing the

stable border security between Japan and Taiwan is crucial to minimising the risks of maritime, aerial, and transnational organised crime as well as terrorism. These non-traditional security threats may originate from Japan's southern neighbours, including both Taiwan and South East Asia, and there is a lesser-known non-traditional security role of Taiwan in law enforcement that contributes to Japan's security, as will be examined in the ICAO chapter.

Meanwhile, Heng (2015) argued that Japan has been utilising its smart power, which involves integrating the hard and soft powers of the military and civilian assets such as JICA, JCG, and NGOs in the arenas of the Gulf of Aden, Iraq and the South Pacific in terms of anti-piracy operations, humanitarian reconstruction efforts, and HADR for its international security efforts. While the military may be threatening and coercive in nature, depending on how it is utilised, it can also generate the combination of hard and soft power, which amounts to smart power for Japan, which is essential if the SDF is to establish some legitimacy, trust and coordination among the target countries and international society in general. Finally, Heng (2015) pointed out that although the use of smart power may be regarded as taking away the necessary resources from the SDF for traditional tasks such as countering the threat of North Korea and China, the use of smart power through the SDF for international security will increase Japan's allies and friends within international society. It can be stated here that Japan similarly utilises Taiwan as a foreign policy instrument to expand its regional security contribution by utilising the latter's relatively affluent security resources, especially in Southeast Asia and in South Pacific Region, in the field of non-traditional security in ways that do not violate Japan's One-China policy.

Meanwhile, Cohen (2018) argued that the current US foreign policy that appears to reduce its internationalist security commitment is not of Trump's making but signals a more foundational shift in the US, which was mainstream prior to WW2, especially among the older generation of the Americans including the elites that understood America's importance as a vanguard of international security vis-à-vis the rise of fascism. With the disappearance of Communism, the US is likely to be more transactional in its approach to international security. Arguably, the government of Japan is concerned about the absence of US leadership in global security and governance. Therefore, if the US's general willingness to provide for global security turns out to be uncertain in the long term, including in the Asia Pacific region, it is possible that Japan will have no choice but to shoulder more responsibility, albeit unwillingly, for the stable development of the regional order, including in terms of the security of the sea lines of communication in the East and South China Seas. Within this geopolitical context, Japan may become incrementally more interested in utilising Taiwan's security resources indirectly for its own security in the context of its One-China policy.

3.3.6 Japan's diversification of security partners

Fatton (2018) argued that the main reason why Japan is enhancing its strategic autonomy from the US and is increasing its hard balancing actions against the PRC relates to how it perceives the growing unreliability of the US as a security guarantor for Japan's national security and the PRC's increasingly coercive behaviour vis-à-vis the issues of the East and South China Seas. Japan is using internal balancing and a diversification of its security partners in the region as instruments to bolster its autonomy as a hedge in the context of the Japan–US alliance and to counterbalance the

rise of China. Meanwhile, Hughes (2018) examined the Japanese diversifications of arms transfers and arms exports and imports as part of a case study and concluded that Japan has been hedging against abandonment and entrapment in the context of the Japan–US alliance 'within' this bilateral alliance, not 'outside' of the alliance, which is central to Japan's security policy. Here, Hughes (2018) pointed out that Japan may also attempt to reduce the expensive arms costs by joining an international collaboration in terms of arms developments, while all the security partners that Japan is pursing to enhance its security ties for the arms transfer strategy are also allies or partners of the US.

In contrast, Auslin (2016) argued that Japan is attempting to increase its security contributions through a more robust security policy, that is, a strengthened alliance with the US and the security partnerships with like-minded countries such as India, Australia, and the ASEAN states, commensurate with its economic power to ensure that the international liberal order in the Asia Pacific is maintained, and that no single power – especially the authoritarian state of China – dominates the region. As such, Japan is embracing realism in its foreign policy while the population remains pacifist.

Gronning (2017) claimed that the fact that Japan has been enhancing its bilateral security ties with the Philippines and Vietnam at the expense of the Cold War-style security dependence on the US can be explained by multiple factors, both domestic and international. These include the sudden shift in the geopolitical balance of power in the region with the rise of China, and the existential challenge to Japan's national security interests in the maritime domain, the domestic reforms of the security policy and the attendant stance in Japan, Japanese nationalism, and the nation's attempt to mitigate its concern about the uncertainty of the Japan–US alliance, developing its own foreign

policy leverage to keep the US committed to the security of Japan within the context of the Japan–US alliance. However, Gronning (2017) also pointed out that Japanese security ties with the Philippines and Vietnam have domestic and international constraints that keep the development of the bilateral security relations incremental, namely, the latent power disparity between Japan and China, Japan's significant edge in terms of the overall power disparity between the two regional security partners, Japanese domestic norms of pacifism, and Japan's preoccupation with the security of the East China Sea in relation to the South China Sea.

It can be stated here that Japan cannot explicitly engage in a security partnership with Taiwan in the absence of Japan's recognition of Taiwan, as it does with other security partners such as India, Australia, and the ASEAN member states. The reason for this is straightforward: if Japan enhances its military ties with Taiwan, this will breach its One-China policy and will induce the risk of destroying the relationship with the PRC. Thus, Taiwan may declare itself an important partner for Japan, while this creates a strategic need for Taiwan to demonstrate its worth as a valuable partner and to regularly remind Japan of its often-forgotten existence in the hierarchical Japan—Taiwan relations in international society. Nonetheless, Taiwan will never be officially designated as Japan's defence partner and Japan will not officially express its commitment to the defence of Taiwan. However, Taiwan may still be useful for Japan in terms of its power politics with the PRC even if Taiwan remains a non-state actor, provided Japan utilises Taiwan strategically as a foreign policy instrument in the context of its One-China policy.

3.3.7 Japan's pacifist identity

Gibson (2016) argued that although Japan's pacifism remains tentative, and there is a disturbing trend of a nationalistic shift in Japanese society, the significant number of peace actors in Japanese domestic politics continue to push for peace education. In his words, '[c]onflict, as peace education teaches, is inevitable; the means to resolve conflict need not be, however, a default setting of violence' (Gibson, 2016, p.106). In contrast, Patey (2017) maintained that the pacifist national identity in Japan remains highly robust, to the point, in fact, that even a minor global security contribution – such as that of the PKO in South Sudan – may trigger a large backlash at home if handled inappropriately and may potentially result in the downfall of the Abe administration, given the former defence minister Inada's resignation over the issue of the cover-up of the military log in the MOD.

Other scholars of Japanese pacifism have analysed Japan's pacifist norms within a historical context. Here, Anno (2018) observed that there have always been some tensions between Japan's national identity and the notion of democracy, noting that in the later part of the Edo period, with the arrival of the black ships, as the shogunate was regarded as incapable of achieving modernisation, the larger political actors were invited into the creation of the political community in Japan, which led to the nascent form of political liberalisation. In Anno's (2018) view, during the Meiji period in Japan, as the sovereign ruler, the emperor allowed Japan to adopt Western modernisation and various domestic political reforms in the context of imperial rule to join the elite Eurocentric international society. Meanwhile, Japan responded to the nascent form of domestic democratic experiments and attempted to enhance its more liberal international norms in international society in accordance with the Wilsonian ideals (Anno, 2018). In the post-war period, Japan's national identity was firmly rooted in the

domestic pacifist norms, which infused the Japanese with a unilateral pacifist narrative as opposed to making democratic development a key component of its national identity. Therefore, as Anno (2018) argued, Japan has been more interested in peace diplomacy than democracy promotion in its foreign policy agenda. Even in the contemporary context, Japan has not yet found a way to reconcile the tension between its unilateral pacifism – which also involves somewhat revisionist historical narratives – and its democratic development that has some elements of parochialism - all of which complicate Japan's relationships with its immediate neighbours. Meanwhile, Curtis (2013) argued that Japan's foreign policy since the Meiji Restoration has maximised its standing in the prevalent international system as opposed to making it in its own image, and stated that Japan is likely to adhere to the Yoshida Doctrine even under Abe. Curtis (2013) concluded that despite Abe's nationalist tendency, the radical departure from the previous foreign policy – manifested in its autonomous foreign policy with the fullyfledged military – is unlikely, if Japan continues to believe that the security relationship with the US in the context of the US's regional dominance is maintained and that the economic relationship with China benefits Japan.

As discussed in the previous chapter, there exists a general bias against the secession movements, with unrecognised entities labelled as troublemakers in international society, and the PRC is obliged to resort to non-peaceful means including the use of force by the anti-secession law if Taiwan becomes a de jure state. Therefore, Japan must consider its domestic pacifism with reference to these geopolitical factors. Some may argue that such a domestic pacifist norm has been weakened, but are the Japanese people willing to be involved in a cross-strait kinetic conflict? Whether Japan remains pacifist or not, if the Taiwan Strait is a regional flashpoint, Japan is likely to utilise

Taiwan in ways that contribute to the alleviation of regional security issues. Thus, Japan will utilise Taiwan to enhance the stable development of the regional order based on Japan's pacifist norms in ways that would not provoke the PRC excessively in the context of Japan's One-China policy, even though it will continue to provide deterrence for Taiwan vis-à-vis the PRC through the Japan–US alliance and through its own unique initiative. Therefore, in order to make Japan-Taiwan security interactions normative, humanitarian and less coercive, they will be kept low-key but comprehensive non-traditional security cooperation in ways that do not violate Japan's One China policy excessively. All these factors will likely ensure Japan's security interactions with Taiwan remain incrementalistic, regardless of whether Japan adheres to the Yoshida Doctrine or shifts to the more radical Abe Doctrine, while utilising Taiwan to enhance normative developments in the region.

3.4 Taiwan's Diplomatic Exclusion and Japan's Socialisation of Taiwan through International Security Norms

This section reviews the existing literature on Taiwan's diplomatic handicap, which conceptualises how Taiwan cannot participate in the multilateral security forums that require 'statehood' for membership. It can be stated here that precisely because Japan is restricted in its hard security relationship with Taiwan by its own One-China policy, it needs to instrumentalise Taiwan as a quasi-autonomous non-traditional security actor for Japan's security and the wider region, in contrast to the PRC, which has, allegedly, been causing tensions in the East and South China Seas in recent years. It is essential to note that, as a de facto state, Taiwan is isolated from international non-traditional security efforts, including anti-terrorism, anti-transnational organised crime in the

maritime, aerial, and cyber domains, international law enforcement, and even medical security operations such as the global fight against epidemics. However, this means Taiwan's under-utilised security resources and roles should be better-utilised such that it is not be a generator of international security issues and contributes to the stable development of regional order. The argument here is that Japan has a unique role to play in terms of improving Taiwan's voluntary compliance with international security norms to mitigate the regional security loophole that stems from Taiwan's non-access to the multilateral security efforts. The discussion here will help illuminate how Japan aims to enhance Taiwan's voluntary alignment with its own national security interests and its voluntary compliance with international security norms.

3.4.1 Taiwan's diplomatic limbo: its under-utilised security resources and non-accessibility to updated information on international security

The existing literature on the international utility of Taiwan focuses on its diplomatic efforts in expanding international space after effectively losing an official seat in the UN (Wang and Ching-Hsin, 2001). Glaser (2015) discussed Taiwan's limited international security roles, specifically in terms of the field of non-traditional security and the role of the US in helping Taiwan to expand its international security contributions and quasi-representations in the international organisations amid its exclusion from the international organisations that require 'statehood'. Glaser's work (2015) rightly considered Taiwan's non-representation in international organisations to be a loophole in international security bodies since, by preventing Taiwan from accessing updated knowledge on international security and from making its own international security contributions, it prevents the nation from utilising its own

knowledge and experiences. However, Glaser's work does not specifically address how Taiwan can be an international or regional security asset for regional great powers such as Japan. In other words, if Japan can connect Taiwan with the updated knowledge on the international security regulations and rules that are produced in multilateral security forums and is able to provide Taiwan with avenues to contribute its own knowledge and experiences to manage the international security risks to Japan and others, this would serve Japan in terms of its security and that of the region in general. Therefore, it is noted here that Japan's security engagement with Taiwan offers the former an opportunity to develop its creative bilateral and multilateral diplomacy through its engagements with the latter.

3.4.2 Japan can enhance Taiwan's voluntary compliance with international security norms

Since Taiwan is an immediate neighbour, Japan's security is affected if Taiwan remains a loophole or a missing link in international non-traditional security in terms of the efforts against international terrorism, transnational organised crime, or even in the fight against epidemics, since these non-traditional security issues know no borders. On this point, the current de facto Taiwanese ambassador in Tokyo, Frank Hsieh (2018a), argued that Taiwan's security is directly linked to Japan's security, and as a stakeholder, Taiwan must not be ignored for the peaceful solution of the ongoing airspace and aviation security issue in the Taiwan Strait. Here, the ambassador called on Japan's various sectors to help Taiwan in mitigating the civilian aviation tensions with the PRC. As will be discussed in the ICAO chapter, Japan's approach to engaging with Taiwan will socialise Taiwan in accordance with international security norms, which include the enhancement of the stable management of air security in the East China Sea and

Taiwan's neighbouring seas and airspaces for Japan's security. Certainly, since the Koizumi administration, while Japan has continued to officially follow the One-China policy, its engagements with Taiwan have gradually started to take form in terms of a de facto 'One China, One Taiwan policy' (Yang, 2009).

3.4.3 Japan's higher position to reward Taiwan for normative security behaviours in the hierarchical Japan–Taiwan relations

In addition to this geopolitical development, as discussed earlier, Taiwan may not be able to formally participate in the multilateral security frameworks, including the UNCLOS, in the absence of 'statehood'. However, if Japan invests significant political capital here, it can still instrumentalise Taiwan for international security, and can still enhance Taiwan's regional security profile to strengthen its own standing in the region despite its One-China policy and Taiwan's international isolation. Furthermore, as deLisle (2011) argued, while although Taiwan is not a member of the UN or the UNrelated security agencies where membership requires statehood, Japan can still enhance Taiwan's pseudo participation 'as if' it is a member of such international organisations. Here, deLisle (2011) pointed out that Taiwan's unilateral compliance may increase its hope for future entry into international organisations by enhancing its compliance with international law, thus increasing its benefits from international society. In other words, it needs to be stated here that even though the international organisations and institutions may ignore unrecognised states and may reject their membership, this does not mean Taiwan is prohibited from enhancing its voluntary compliance with the international regulations and agreements made in such international organisations. Furthermore, it also needs to be stated here that Japan is free to help Taiwan behave in

accordance with international law to enhance the international normative order for Japan's regional security interests. Finally, since Taiwan's ontological security relies on Japan as a positive other as discussed earlier, the latter functions as a normatively higher entity in the hierarchical Japan–Taiwan relations, which gives Japan an influential authority to grant semi-recognition to Taiwan as reward for Taiwan's good security behaviour, including by way of signing bilateral security-relevant agreements to elevate Taiwan's prestige in international society, even within the context of Japan's One-China policy. Japan's security behaviour vis-à-vis Taiwan is in sharp contrast to how the PRC has been sanctioning Taiwan's aspiration to be represented in international security bodies that require 'statehood' for membership, as will be examined in the ICAO chapter. Therefore, Japan could aim to make Taiwan a *normative* asset in multinational security efforts even though Taiwan is internationally excluded through the PRC's sanctions.

As such, it needs to be stated here that if Taiwan can participate in Japan-led international and regional security efforts, despite the international exclusion, this would incrementally fill the void in regional security efforts. On this point, the current de facto Taiwanese ambassador in Tokyo, Frank Hsieh (2018b), argued that Japan and Taiwan can still cooperate with each other even though Taiwan suffers from various harassments from the PRC in the form of diplomatic sanctions, the cross-strait aviation tensions in the Taiwan Strait, the economic sanctions against Taiwanese business in the PRC, and the exclusion of Taiwan from the WHO. The ambassador then went on to argue that this is especially the case considering both actors have already established bilateral non-traditional security cooperation, such as the bilateral anti-natural disaster efforts, and can work together to mitigate the ongoing Sino-US trade tensions, including

Taiwan's aspiration to be a member of the Japan-led TPP. Thus, if Japan incrementally uses Taiwan's security contributions to fill the power vacuum that stems from Taiwan's absence in multilateral security efforts, it can turn Taiwan into an instrument for its power politics with the PRC. Certainly, Taiwan may resist some of the political demands from Japan in the context of Taiwan's domestic politics, but it will ultimately have to gradually satisfy Japan's strategic needs in the context of hierarchical Japan—Taiwan relations due to the absence of Japan's recognition and the PRC's threat of using force against the nation. Here, it is essential to state that it is ultimately up to Japan—Taiwan security relationship if it adheres to its One-China policy.

3.4.4 Engagement with Taiwan is a litmus test for Japan's creative diplomacy

Meanwhile, Drifte (1998) argued that Japan's commitment to bilateralism, especially within the context of its alliance relationship with the US and others, has rendered its multilateral diplomacy less effective and less imaginative, including in terms of its aspiration to become a permanent member of the UNSC. Taiwan is not a member of the UN, and its non-participation in the UN-relevant security agencies could affect Japan's security interests. Therefore, it can be argued that Japan could renegotiate the membership of Taiwan in international security efforts in informal terms, meaning Taiwan can be a security asset 'as if' it were a member of the UN, while Japan could prove its creative diplomacy on a global level. Similarly, Babones (2016) argued for a normalisation of the de facto state of Taiwan as opposed to making it a de jure state, which would ensure it is taken seriously in international society. In Babones' (2016) view, a behavioural change rather than a political change in Taiwan is essential to

ensuring the world prevents the PRC from dictating to Taiwan on the PRC's terms. In other words, it can be stated here that Japan's socialisation of Taiwan in terms of improved compliance with international norms would help Taiwan to be more of an autonomous security actor while pacifying the PRC's sensitivity on the Taiwan issue in the context of Japan's One-China policy. Meanwhile, Drifte (1998) argued that Japan lacks debates on why it has to be a permanent member of the UNSC and what it plans to do if this were the case, other than budgetary and procedural contributions, while its permanent membership in the UNSC may increase the fear of inequality among the UN members in the developing countries due to the global perception of Japan siding with the US too closely. If Drifte's view (1998) is correct, it can be stated here that Japan's re-conceptualisation of the membership in multilateral diplomacy operations, including the UN, through its informal engagements with Taiwan in international security efforts may provide an alternative view of Japan's creative diplomacy in international society and may stimulate more discussions about what kind of international security contribution it will provide as a normalised great power.

3.5 Summary

Reviewing the literature related to Japan—Taiwan relations has demonstrated that Japan has an influential authority to accommodate Taiwan into the international society of states for its own security. The analysis of Japan's foreign policy instruments elucidated what Japan is willing to do and what it is unwilling to do in its security relationships with Taiwan within the context of its One-China policy. Japan is willing to enhance non-traditional security cooperation with Taiwan for the former's security if this bilateral cooperation is normative, humanitarian and non-coercive in nature without breaching its One China policy. In addition, regardless of whether Japan adheres to the

Yoshida Doctrine or shifts toward the Abe Doctrine, its security interactions with Taiwan remain, at best, incremental. Meanwhile, the assessment of the literature on Taiwan's diplomatic exclusion has explicated how Japan has a significant role to play in encouraging Taiwan to voluntarily behave in accordance with international security norms in the multilateral security forums that require statehood for membership, despite Taiwan's non-access to multilateral security efforts. With all these points in mind, the following case study chapters will elucidate how Japan can accommodate Taiwan into the international society of states for Japan's security within the context of its One-China policy in empirical terms.

Chapter 4. Collective Self-Defence

4.1 Introduction

This chapter is dedicated to the concept of CSD and elucidates, with reference to ESIR theory, how Japan has incrementally accommodated Taiwan in the areas of indirect defence and non-traditional security. As was argued in the previous chapters, it is up to Japan's political decision-makers to decide what the nation is willing to do and what it is unwilling to do in terms of Japan-Taiwan security interactions. As such, how Taiwan fits into the Japanese concept of CSD also reflects Japan's decision-making, while Taiwan's political input regarding Japan's security policy has no place in Japanese debates on its CSD due to the hierarchical relations. In short, Japan uses Taiwan as a foreign policy instrument to maintain the regional balance of power with the PRC, but Japan also navigates its One-China policy in its security engagement with Taiwan to pacify the PRC for regional great power management. Furthermore, Japan's socialisation of Taiwan through international security norms indicates how Japan incrementally instrumentalises Taiwan as a normative security asset for its own security as well as that of the region, while Taiwan cannot be a full member or earn full status in Japan's concept of CSD in relation to pacifying the PRC. Therefore, Taiwan needs to demonstrate its worth as a security asset in the hierarchical Japan-Taiwan relations in international society, with this mechanism consolidating Japan's strong position against Taiwan in normative terms. Also, if Japan-Taiwan security cooperation were to progress, it would be conducted in the field of non-traditional security, which is regarded as normative and less-coercive in international society, since the bilateral normative security cooperation will make it difficult for the PRC to sanction it despite Japan's One China policy. Finally, this chapter clarifies how Japan's security interactions with Taiwan remain incrementalistic, regardless of whether the nation

sticks to the Yoshida Doctrine or shifts toward the Abe Doctrine, since Japan continues to navigate the ambiguities of its One-China policy to seek acceptable boundaries regarding how far the nation is willing to enhance its security cooperation with Taiwan.

First, the Japanese concept of CSD is addressed before we discuss Taiwan's ambiguous status within this concept to elucidate how Taiwan cannot be a full-fledged member here in the absence of Japan's official recognition in the hierarchical Japan-Taiwan relations, albeit that Taiwan continues to benefit from the greater deterrence Japan offers through CSD policies. Following this, we examine how Taiwan regularly reminds Japan of its security contribution to Japan, and how it consolidates Japan's power politics with the PRC, even in the context of its One-China policy. In short, Taiwan's indirect defence alignment with Japan supplements Japan's defence security against the PRC. Due to its One-China policy, Japan does not engage in kinetic military ties with Taiwan, while it does engage in ad hoc intelligence exchanges, especially those related to the development of the PLA, and the two actors continue to engage in semiinstitutionalised security talks while Japan navigates its One-China policy. Next, we examine Japan-Taiwan coastguard relations to illuminate the extent to which the bilateral constabulary cooperation is acceptable to Japan within the context of its One-China policy, demonstrating what Japan is willing to do and what it is not willing to do in its security interactions with Taiwan in relation to the Japanese concept of CSD. We also examine Japan-Taiwan humanitarian assistance and disaster relief cooperation to illuminate how acceptable the bilateral HADR is to Japan within the context of its One-China policy and how it remains difficult for the PRC to oppose this since it is more normative than political. This demonstrates how the low-key HADR cooperation is incrementally indicating what Japan is willing to do in its security interactions with Taiwan in relation to the Japanese concept of CSD.

4.2 What is Collective Self-Defence in the Japanese Context? Political Tensions between Domestic Law and International Law

First, we examine the political tension between the evolution of the Japanese government's interpretations of its domestic pacifist constitution and international law to elucidate the concept of CSD within the Japanese context. In the case of Japan, the contradictions in its post-war pacifist constitution and the UN Charter have resulted in tensions in Japan's domestic political context regarding the rights of the use of force as a sovereign state. Discussing these tensions between domestic law and international law is essential to explaining why Japan was legally able to defend only itself, despite the international legal permissibility provided by the UN Charter. It is also essential to discuss the evolution of Japan's defence policy, which has incrementally allowed the nation to engage in self-defence despite its pacifist constitution. This discussion is essential since, even after the reinterpretation of the pacifist constitution, Japan's CSD is severely constrained in terms of its international and regional security contributions due to the lingering tensions between domestic and international law in Japan.

In early post-war Japan, amid strong anti-militarist sentiment, Japan enacted a pacifist constitution that renounced war 'and the threat or use of force as a means of settling international disputes' (The Constitution of Japan, 3 November 1946). In fact, Japan, as a sovereign state, literally renounced the use of force or any kind of war potential by stating the following:

[a]spiring sincerely to an international peace based on justice and order, the Japanese people forever renounce war as a sovereign right of the nation and the threat or use of force as a means of settling international disputes. In order

to accomplish the aim of the preceding paragraph, land, sea, and air forces, as well as other war potential, will never be maintained. The right of belligerency of the state will not be recognised (Article 9, The Constitution of Japan, 3 November 1946).

However, on 20 May 1957, the National Defense Council and the cabinet initiated the Basic Policy on National Defense (Ministry of Defense [MOD], 1957). According to this policy, the government of Japan's interpretation of the renunciation of war potentially referred to military aggression, and it did not deny Japan's right to defend itself against threats to its own existence (MOD, 1957). In addition, Japan's interpretation of its pacifist constitution allowed it to possess the minimum required defence capabilities (MOD, 1957), with the government of Japan allowing an exclusively defence-oriented policy with the minimum necessary level of self-defence (MOD, 1957). In other words, Japan unilaterally regarded its right to self-defence as permissible in the domestic legal context, but voluntarily considered the use of force to settle international disputes as unconstitutional while embracing such domestic legal arrangements.

International law does not prohibit the use of force by sovereign states as a means of individual defence or CSD. Regarding the permissibility of these actions, Article 51 of the UN Charter (1945) states the following:

[n]othing in the present Charter shall impair the inherent right of individual or collective self-defence if an armed attack occurs against a Member of the United Nations, until the Security Council has taken measures necessary to maintain international peace and security. Measures taken by Members in the exercise of this right of self-defence shall be immediately reported to the

Security Council and shall not in any way affect the authority and responsibility of the Security Council under the present Charter to take at any time such action as it deems necessary in order to maintain or restore international peace and security.

Therefore, UN member states have the inherent right to possess the military means to defend themselves or to conduct CSD. In accordance with this right, using the 1957 National Defence Policy, Japan internalised the international legal norms regarding individual defence despite the explicit prohibition of war potential and the use of the rights of belligerency by its pacifist constitution. As noted above, nation states are legally allowed to cooperate to counter threats posed by a foreign country through CSD according to Article 51 of the UN charter. However, under its exclusively defence-oriented policy, Japan did not internalise its international legal norms on CSD and officially decided to maintain a self-imposed prohibition on its use. Regarding the Japanese government's stance here, the Annual White Paper on the Defence of Japan (2013) stated the following:

[i]nternational law permits a state to have the right of collective self-defense, which is the right to use force to stop an armed attack on a foreign country with which the state has close relations, even if the state itself is not under direct attack. Since Japan is a sovereign state, it naturally has the right of collective self-defence under international law. Nevertheless, the Japanese Government believes that the exercise of the right of collective self-defence exceeds the minimum necessary level of self-defence authorised under Article 9 of the Constitution and is not permissible (Section 2 in the Constitution and the Right of Self-Defence in the defence of Japan Annual White Paper, 2013).

Under its exclusively defence-oriented policy, Japan voluntarily relinquished its international legal right to defend other countries. The Japanese government considered CSD to be beyond the minimum necessary level of self-defence allowed by its pacifist constitution. However, regional security mechanisms such as NATO require member states to be able to exercise CSD to gain membership according to Article 5 of the North Atlantic Treaty (4 April 1949). In principle, Japan considered CSD to be beyond its constitutional limits until the reinterpretation of its pacifist constitution under the Abe administration in 2014, which, in theory, enabled Japan to engage in such activity (MOD, 2013). Before this reinterpretation, CSD was officially regarded as unconstitutional since the CLB's definition of the minimum-required defence regarded CSD to go beyond this in military terms (Green and Hornung, 2014). Regarding Japan's long-held interpretation of its pacifist constitution and the highly restrictive defence behaviours that stemmed from it, Singh (2013c, p. 79) stated the following:

[s]uch an interpretation overshadowed the legal provisions of UN Charter accrued to Japan as a member of the organisation, namely Article 51 that permits all member states to carry out individual defence activities and participate in collective security and collective self-defence activities that involve military action.

Even though international law applies to all sovereign member states equally regardless of their might, size, and wealth, it by no means implies that every sovereign state must accept it. For example, according to Wakefield (2016), nations can always waive their rights even if these rights are defined by international law. According to this legal logic, while international law permits nation states such as Japan to resort to CSD, this does not mean that Japan is obliged to internalise such an international legal norm. As such,

Japan is allowed to abandon such rights voluntarily if deemed to be in the interests of the nation. Accordingly, domestic law and international law often conflict in terms of how international law is understood within the domestic legal-political context.

4.2.1 Japanese pacifism

Mori (2015) stated that Japan internalised a unilateral pacifist mindset in the post-war period. Behind the Japanese people's strong resistance to the reinterpretation of Japan's pacifist constitution lies widespread unilateral pacifist norms. The Japanese believe that Japan alone can maintain regional peace as long as the nation adheres to the pacifist constitution. In relation to this, Patey (2017) argued that if Japanese people do indeed adhere to their pacifist identity, even minor global security issues such as Japan's contribution to the UNPKO in South Sudan may endanger the security reforms initiated under Abe. Furthermore, Japan has long used its domestic pacifism, including its pacifist constitution, as a shield to deflect political demand from the US that Japan shoulders more security responsibility in the post-war period and thereafter. Finally, Japan's main goal in facilitating CSD is to uphold the liberal regional order in international society for its own security through its contribution to regional peace and stability, including in terms of the Taiwan Strait. Here, Japan's national interest lies in not becoming involved in the cross-strait war. Therefore, calls from Chen Shui-bian and Lee Teng-hui for Japan's greater regional security leadership, and Ma Ying-Jeou's endorsement of the Japan–US alliance have no effect on Japan's regional security policy in the face of its domestic pacifist norms and its lack of recognition of Taiwan.

The people of Japan continue to strongly resist the amendment of the pacifist clause of Article 9 of the constitution. For example, according to a public opinion survey on public attitudes toward the constitution conducted by the NHK in April 2016, support

for the amendment of Article 9 is relatively low, with only 22.15% of those surveyed stating that the amendment was necessary, while 39.2% stated the opposite (NHK, 2016). In addition, even the LDP and the Koumei Party's recommended constitutional scholars argued that CSD would be unconstitutional without amending the pacifist clause of Article 9 (Nikkei Shimbun, 2015).

Nevertheless, Prime Minister Shinzo Abe argued that an overly strict reading of the pacifist constitution would inhibit the effective deterrence of a Japan–US alliance at a critical juncture in matters of regional security issues if Japan did not contribute more to the alliance (Prime Minister of Japan and His Cabinet, 2014). At the same time, what actually constitutes a minimum use of force remains a controversial subject since the advancements in military technology and various other aspects have drastically changed this concept. However, the Japanese government still clearly maintains that the defence capabilities that the constitution allows Japan to possess must be within the scope of minimum defence. In this sense, weapons that are regarded as violating the Japanese pacifist constitution, such as intercontinental ballistic missiles, long-range bombers, or attack aircraft carriers, are considered to be strictly unconstitutional (MOD, 2014). Even the largest helicopter carriers constructed in the post-war period, which include Izumo and Kaga, were designed for anti-submarine warfare to ensure they remain defensive in nature (Reuters, 2017). In short, in line with Japan's defensive capabilities as a whole, such military resources remained within the legal boundary of the pacifist constitution, as interpreted by respective LDP governments. Furthermore, a helicopter carrier used its airlift capabilities to enhance search-and-rescue operations and to provide food and medical support in remote regions during HADR missions in the Philippines for peaceful and humanitarian purposes (Miller, 2014).

4.2.2 Consequence of the tensions between domestic law and international law

The tension between domestic law and international law has manifested itself in terms of obstacles to Japan's execution of CSD. In fact, among the world's primary state actors, Japan's military power is the most legally constrained (Institute for International Policy Studies [IIPS] 2017). For example, according to the International Peace Cooperation Act of 1992, Japan needs to adhere to the five following principles to participate in international peacekeeping operations:

- 1. The parties to the conflict must have reached a ceasefire agreement.
- 2. The parties to the conflict must consent to Japan's participation.
- 3. Operations must maintain strict neutrality.
- 4. Japan can withdraw its troops if any of the basic requirements are not met.
- 5. The use of force must be limited to the minimum needed to protect the lives and safety of personnel.

However, the Japanese concept of CSD does test situational constraints by allowing previously banned missions such as police-like missions (*Kaketsuke keigo*). However, these missions continue to face severe legal constraints. Since Japan's international security contributions are still severely restricted by its domestic legal constraints, its SDF personnel face more severe risks in any given conflict. In South Sudan, for example, even though the Japanese SDF personnel in the Peacekeeping Operation Unit were engineering soldiers and were not specially trained for battlefield situations, they were given police-like missions, including rescuing its own nationals or defending other peacekeeping personnel from other countries (Isezaki, 2016). Regarding how the

domestic legal constraints impeded Japan's international security efforts in South Sudan in operational terms, Isezaki (2016) stated the following:

[t]he UN force's senior officers are well aware that the SDF are not a constitutionally sanctioned military organisation, and that Japan is not equipped with a legal system that allows it to pursue criminal responsibility for military transgressions overseas. This is precisely the kind of scenario the United Nations is most anxious to avoid. Nor is the UN force's top command likely to assign an engineering unit to engage in armed rescue operations.

In addition, Tatsumi (2017) noted Japan's strong domestic aversion to the SDF engaging in any situation resembling actual combat in South Sudan. For example, according to *Mainichi Shimbun*, approximately 300 local soldiers were killed in combat by either government forces or the rebels in Juba in South Sudan in July 2016. In fact, SDF engineering units had been operating there even though the UN had issued a warning regarding the potential breakout of genocide (*Mainichi Shimbun* Editorial, 2017). Furthermore, many UNPKO activities are conducted in the most high-risk regions of the world. Here, referring to the United Kingdom's international peacekeeping efforts, a high-profile panellist stated at RUSI's conference on the United Kingdom's PKO that 'more than 45%–55% of PKO activities are classified as high risk' (RUSI, 2015). Despite the Japanese government's insistence that the SDF engineering units have 'reached a juncture' in their decision to withdraw from South Sudan, it is clear that the strongly pacifist streak in Japanese public opinion has influenced the Japanese government's decision to withdraw its SDF from South Sudan (Tatsumi, 2017).

Certainly, if the logic discussed above is strictly followed, it could be rightly reasoned that the strategic interests of Japan, such as the enhancement of the Japan–US alliance,

trump the importance of domestic law. However, this would have a significantly negative effect on the rule of law and the division of powers, which are essential for the development of liberal democracy in Japan. Therefore, even after the reinterpretation of its pacifist constitution, Japan needs to develop its own defence and security policy in accordance with the strict restrictions. For example, while a pre-emptive strike by cruise missiles launched by the SDF to minimise the risk of North Korean missile launches against Japan is not necessarily unconstitutional according to the traditional interpretation of the constitution, Japan does not currently have such offensive capabilities (Miller, 2017). Despite Japan's reinterpretation of its pacifist constitution, the nation's defence and security policy remains defensive in nature due to the domestic legal constraints imposed by the country's minimum defence capabilities. Nevertheless, the debate on pre-emptive strikes is intensifying, as manifested in the LDP's policy proposal, which explicitly calls for the possession of independent pre-emptive strike capabilities (LDP, 30 March 2017). In addition, Japan decided to acquire joint strike missiles from Norway in 2019 (Defense News, 2019), while the date of delivery remains unknown.

With the international security environment deteriorating for Japan due to the military rise of China, the instability in the Korean Peninsula, and the proliferation of international terrorism, the nation does need to shoulder its share of regional security responsibility commensurate with its own national strength, albeit incrementally. Japan also needs to enhance the international liberal order, which has benefited the country in the post-war era. Elsewhere, the US and the UK would appear to be gradually shifting their central focus away from the traditional foreign and security policies of the post-war international liberal order, as indicated by the election of US President Donald Trump, the decision on Brexit, and Trump's virtual lack of concern for universal

concepts such as human rights. Therefore, Japan must determine what kind of security and foreign policy it views as appropriate in the context of the Japan–US alliance and the enhancement of its own defence and security capabilities for regional peace and stability in the Asia Pacific region and beyond (IIPS, 2017).

4.3. Taiwan's Ambiguous Status in the Japanese Concept of CSD

As discussed in the previous chapters, Taiwan cannot attain full membership status in international society even though Japan accommodates it for its own security. It is up to Japan to decide what the nation is willing to do and what it is unwilling to do regarding Japan—Taiwan security interactions within the hierarchical order, including how Taiwan fits into the Japanese concept of CSD. To understand how this plays out, we must first examine the ambiguous status of Taiwan in relation to this concept. In fact, Taiwan's unrecognised status in the hierarchical Japan—Taiwan relations means its explicit membership in the Japanese concept of CSD is not part of the equation. For example, regarding the government-to-government Japan—Taiwan security talks that President Tsai requested in 2019, the incumbent foreign minister, Kono Taro, stated 'Japan has been maintaining non-governmental substantive relations with Taiwan, so our government will respond to the statements properly based on the above-mentioned foundation' (MOFA, 2019).

However, to enhance Japan's security, it remains possible to partially renegotiate Taiwan's position in CSD operations in legal terms. Neither the tension between domestic law and international law in Japan's legalistic politics nor Taiwan's unique status in international law can help us to clarify how Taiwan is conceptualised within the Japanese concept of CSD. However, Japan can still instrumentalise Taiwan for its own security within the context of Japanese CSD, while Taiwan's security can still be enhanced through the country's nebulous status in the Japanese CSD operations. For

certain Taiwan cannot influence Japan's approach to CSD; however, the nation will have opportunities to remind Japan of its standing within the Japanese concept of CSD if it contributes to Japan's security and regional security in general.

Abe argued that a more proactive stance by Japan on regional and global security would enhance the Japan–US alliance. However, while the alliance is bound by international treaty and the concept of CSD is legally permissible under the UN Charter, neither is necessarily above Japan's domestic law, as was discussed earlier. This fact complicates Japan's potential utilisation of CSD. Furthermore, an entity that is unrecognised by international law, as is the case with Taiwan, will potentially complicate Japan's concept of CSD. Taiwan does not have a defined status in international law, and, unlike the US, Japan is not legally bound to commit to Taiwan's security in the absence of a domestic law such as a Japanese version of the Taiwan Relations Act. Taiwan's atypical status raises one crucial question: is it legally permissible for Japan to defend Taiwan by activating the Japanese concept of CSD? Clearly, to resolve the issue of defending Taiwan in military terms within the context of both international law and domestic law, the Japanese government will need to further debate the legality of defending its neighbour via CSD.

The former foreign minister, Kishida Fumio, stated during the parliamentary debate of 15 June 2015 at the House of Representatives that Japan's concept of CSD includes 'divided' and 'unrecognised' states. However, Japan has relinquished all rights, titles, and claims to Taiwan, which means it is not in a position to determine the legal status of Taiwan under the San Francisco Peace Treaty (House of Representatives, 2015).

Even after the security bill was passed in 2015, Japan did not engage in high-profile global military support. This involves maintaining collective security by providing

logistical or indirect military support, including refuelling, missile and air defence, or sea mine disposals in multinational military operations against ISIS in the Middle East. Oros (2017a) pointed out that despite the concept of global security responsibility espoused by former prime minister, Koizumi Junichiro, with Japan's regional security situation severely deteriorating in recent years, Japan's top national security priorities are now related to its immediate neighbours. This means that Japan can choose when to exercise CSD and how it engages in military action within the context of a Japan—US alliance and its relations with other security partners during a regional or global security crisis, especially if the crisis does not directly threaten Japan's national security.

If Japan's provision of security and deterrence for Taiwan vis-à-vis the PRC through the application of the concept of CSD encourages Taiwan to seek de jure independence, this will be detrimental to Japan's security. Japan's interests vis-à-vis Taiwan are to incrementally integrate Taiwan into international society in view of allowing the latter to contribute to the former's security and that of the region as a foreign policy instrument. However, regionally specific security issues such as the Senkaku Islands issue have, to some extent, complicated the development of stronger Japan–Taiwan relations and have obscured the 'Taiwan factor' within the context of the Japan–US alliance, as will be discussed in the chapter on the fisheries agreement.

To determine the potential applicability of CSD vis-à-vis Taiwan, the conditions imposed by the Government of Japan regarding the use of CSD must be examined. The security bill provides 'three new conditions' for the use of force: (i) when an armed attack against Japan has occurred, or when an armed attack against a foreign country that is in a close relationship with Japan occurs and as a result threatens Japan's survival and poses a clear danger to fundamentally overturn people's right to life, liberty, and

pursuit of happiness; (ii) when there is no appropriate means available to repel the attack and ensure Japan's survival and protect its people; (iii) the use of force limited to the minimum extent necessary should be interpreted to be permitted under the Constitution as measures for self-defence in accordance with the basic logic of the government's view to date (The Guideline on Peace and Security Legislation, 2015). While the reinterpretation of the constitution did expand Japan's international security contributions in geographical, situational, and operational terms, the scope of the SDF's military activities continues to be strictly restricted.

Japan's defence and security policy must be exclusively defence oriented, and the primary role of the SDF remains to consolidate Japan's alliance with the US (Green and Hornung, 2014). In theory, Japan's operational and situational scope for providing logistical support and conducting defensive operations, such as providing missile defence or conducting anti-submarine warfare, has increased following the reinterpretation (Green and Hornung, 2014). However, Japan's current defence system is restricted by the Japan–US alliance. Therefore, it is highly unlikely that Japan will conduct independent military intervention on behalf of Taiwan. At the same time, if the US decides to conduct military intervention in Taiwan in the face of PRC aggression, Japan may be able to provide support. At present, Japan is maintaining some strategic ambiguity regarding this matter (Matsuda, 2017).

Japan's CSD has, in theory, increased its deterrence capabilities vis-à-vis the PRC's potential invasion of Taiwan and can provide greater naval warfare support, including in terms of the previously banned missile defence. It has also comparatively enhanced the MSDF's anti-submarine and anti-sea mine warfare capabilities to better defend a US Navy intervention during any Taiwan contingency. In addition, Japan has the third

largest global economy and highly robust defensive military capabilities, and it is the only regional security actor with ample strategic clout in the Asia Pacific region that is willing to forge stronger ties with Taiwan in any meaningful way, regardless of PRC opposition. Meanwhile, Watanabe (2015) pointed out that the US's more high-profile treatment of Taiwan was also facilitated by Japan's CSD since the aforementioned security bill strengthened the Japan–US alliance and reduced the security burden imposed on the US in the Asia Pacific region. Here, it can be inferred that these developments mitigate the political constraints imposed on Japan's engagement with Taiwan.

Nonetheless, Japan must meet the above three conditions and must convince its general population that Taiwan warrants the activation of CSD operations. Furthermore, the former foreign minister, Kishida, has avoided explicitly answering the question of whether Taiwan meets the following criterion: 'an armed attack against a foreign country that is in a close relationship with Japan occurs and as a result threatens Japan's survival'. During a debate on the security bill, Kishida further stated that unrecognised or divided states are included in the Japanese concept of CSD, but the Government of Japan is not in a position to determine Taiwan's status (House of Representatives, 2015). Japan probably has arguments to ensure Taiwan meets the criteria for the potential applicability of CSD in order to defend it. For example, Beijing's One-China principle – which proposes the 'one country, two systems' method as an ultimate resolution of Taiwan's status – would negatively affect Taiwan's 'internal popular sovereignty', which stems from Taiwan's vibrant democracy (Stokes, 2017).

Taiwan's belief in the universal values of human rights, democracy, and popular sovereignty may help justify Japan's use of CSD as a deterrent vis-à-vis the autocratic

PRC. Indeed, the nation as whole has favourable sentiments toward Taiwan due to the latter's vibrant liberal democracy and good record on human rights, while a coercive China is seen as posing a threat to both countries' regional maritime security environments and the maintenance of the inclusive and liberal regional security order. Meanwhile, Taiwan has been contributing to Japan's human security in a broader sense, as indicated by the massive donations it made to Japan during the triple disasters in Tohoku in 2011, while the nation has also been contributing to regional soft security, such as in the case of the non-military HADR.

It is open to debate which country meets the criteria for the potential applicability of Japanese CSD. Even South Korea, which will be a frontline state actor in the occurrence of any contingency on the Korean Peninsula, requires Japan to obtain consent from the ROK government for the SDF to land in ROK territory (National Diet Library Research and Rippou KousaKyoku, 2016 Kikuchi). Furthermore, the Japanese Diet needs to approve the use of CSD. However, Taiwan has made huge Japanese investments and enjoys significant trade ties with its neighbour, while it remains an essential choke point for both air and sea lines of communication. If Taiwan is left unattended and falls into the hands of the PRC, this will pose a major military threat to the southern flank of Japan, and the military takeover of the Senkaku Islands will be a realistic possibility.

While Japan is not powerful enough to unilaterally shape the future course of the issue of the Taiwan Strait, it can prevent China from dominating Taiwan and making unreasonable unilateral changes to the status quo without the US's participation. In fact, Japan is the only significant regional military and security actor that can allow the incremental inclusion of Taiwan into the regional security environment. In addition, if the US refuses to be bound by the One-China policy and seeks to reinterpret its own

national interests under the Trump administration (Stokes, 2017), Japan's security contribution in terms of deterrence for Taiwan against the PRC's potential aggression will be even more crucial for achieving stable regional security. From this perspective, it is telling that the former secretary general of the National Security Council under the Ma administration, Su Chi, suspected Japan's intentions in deploying the SDF on Yonaguni Island – a non-disputed Japanese island that is geographically close to Taiwan – when a cross-strait rapprochement was ongoing.

Japan engages with Taiwan by partially accommodating the nation into its non-kinetic military cooperation strategy (e.g. through intelligence sharing) or into its non-military maritime security alignment through the concept of CSD. Japan's regional security efforts with Taiwan also help to create a stable and peaceful external environment for the latter, despite the military and non-military threats from the PRC. This, in turn, is strategically beneficial for the defence of Japan's southern flank given Taiwan's role as a military buffer.

Furthermore, Japan has taken symbolic political actions with strategic implications, including partaking in the bilateral fisheries agreement with Taiwan. This is especially pertinent if Japan's security interests are perceived to be violated within the context of China–Taiwan relations. Taiwan has been perceived by various international observers to be behaving, at least to some extent, in concert with the PRC against Japan vis-à-vis the issue of the East China Sea or the Japan–US alliance following the nationalisation of the Senkaku Islands by the Government of Japan in 2012. Meanwhile, Japan has made a political move that is considered favourable for the strategic alignment of Taiwan in accordance with the Japan–US alliance, as will be elaborated in the chapter on the fisheries agreement. In addition, the recent official change in the name of the de

facto Japanese embassy in Taiwan from the Interchange Association to the Japan–Taiwan Exchange Association in 2016 is an indication of Japan's interest in strengthening its ties with Taiwan. In fact, this change in name was the first since Japan's derecognition of the ROC. Moreover, according to Ishihara (2013), an expert investigator (Senmon Chousain) was deployed in Japan's de facto embassy in Taipei when former president, Lee, was given the opportunity to conduct a high-profile press conference during his visit to Tokyo in 2015 while Japan was debating the security bill. Here, a number of individuals in the 'green camps' in Taiwan argued that this constituted a recognition of the importance Japan attached to its relationship with Taiwan (Japan–Taiwan Exchange Association, 2015). In addition, as a political morale booster for Taiwan, Prime Minister Abe listed Taiwan as an independent actor along with South Korea and China during a speech he made on the 70th anniversary of WWII (Japan–Taiwan Exchange Association, 2015).

Japan is not obligated to defend Taiwan since, unlike the US, it is not bound by the TRA. While Japan is not taking concrete action to enact such a law, Kishi Nobuo, the state minister for foreign affairs, stated in an interview with the *Taipei Times* (2016) that the Diet has studied the issue. However, this point constitutes a unique development in Japan–Taiwan relations. As was discussed in Chapter 3, grassroots people-to-people ties are vital for enhancing bilateral cooperation, including in terms of the HADR, in Japan–Taiwan relations due to Taiwan's unrecognised status. This status requires the nation to rely on semi-institutionalised informal communication channels and civilian exchanges in the absence of official communication channels.

4.3.1 Japan's less committed yet incrementally significant security policy for Taiwan

Taiwan's ambiguous status in the Japanese concept of CSD is manifested in Japan's

lack of commitment to Taiwan's security. Japan has long avoided shouldering any explicit security responsibilities for Taiwan in view of pacifying the PRC and conducts the regional great power management in accordance with ESIR theory. However, according to Sasaki (2010), following the missile crisis in the Taiwan Strait, the stronger Japan–US alliance became the most significant factor for raising the threat perception regarding Japan-related security issues for the PRC. Further, the Taiwan factor within the Japan–US alliance became the most significant perceived threat for the PRC. The stronger Japan–Taiwan relations under Prime Minister Abe and President Tsai have been viewed with strong suspicion by the PRC.

4.3.2 Japan's non-engagement in arms sales to Taiwan

A Japan–Taiwan security cooperation that is potentially more robust is likely to create difficulties with the PRC. However, as discussed in Chapter 3, it is up to Japan to decide what it is willing to do and what it is unwilling to do, meaning any defence exchange with Taiwan that may excessively antagonise the PRC is not seriously considered by Japan. For example, at least two of the academics writing about Taiwan – Cole (2016) and Dreyer (2016) – expressed a desire for technological access to build indigenous submarines in Japan in the face of the US's apparent unwillingness to sell such high-profile military assets to Taiwan and the US's non-production of diesel submarines. However, Japan will not sell high-profile naval assets to Taiwan since this may have the obvious consequence of ruining the fragile Sino-Japanese relationship within the regional great power management.

Importantly, Monma (2017b) pointed out that it is considered illegal for Japan to sell or provide any form of military equipment or technology to Taiwan in the absence of a relevant defence agreement between the two nations. However, he also admitted the

possibility of Taiwan secretly approaching Japanese military technicians. In fact, Japan is legally allowed to jointly develop defence equipment, defence provisions, and defence transfers only with countries that have signed the ACSA with Japan, and is extremely unlikely to sign such a high-profile defence agreement with Taiwan (Monma 2017b). Furthermore, as Monma (2017b) highlighted, Japan and Taiwan have not even signed the GSOMIA, which would institutionalise intelligence sharing. Therefore, while informal exchanges of opinions between Japanese and Taiwanese officials may be possible, Taiwan's desire to acquire the technology used in Japanese submarines cannot be realised.

4.3.3 Japan's CSD in Asia Pacific and Taiwan

Shuhenjitaihou's geographical application did not specifically include Taiwan in its language; rather, it ensured the potential applicability vis-à-vis the Taiwan Strait was 'situational', clearly contributing to the deterrence for Taiwan vis-à-vis the PRC's potential aggression. The security bill enacted in 2015 expanded both the geographical and the situational limits of Japan's international security contribution, including in terms of Taiwan. The former de facto Japanese director in Japan-Taiwan Exchange Association in Taipei, Makoto Ogata, stated that Japan can now, in theory, defend the US military during any contingency in the Taiwan Strait, thereby offering Taiwan further protection vis-à-vis the PRC (Ogata, 2016). In addition, Japan's maritime forces can now defend the US navy, even during peacetime operations (Asahi Shimbun, 2016). This has further strengthened American military primacy in the Asia Pacific region, including in terms of deterrence measures against potential Chinese aggression towards Taiwan. From this perspective, it is fair to say that Japan's incremental shouldering of regional security leadership responsibilities in the US-led regional order in the Asia Pacific region has informally yet significantly contributed to creating a stable regional

security environment for Taiwan. This was necessary for the development of a vibrant democracy in Taiwan, especially given that the military threat from the PRC became more serious before and during the missile crisis in the Taiwan Strait between 1995 and 1996. Ogasawara (2012b) pointed out that without conducting a military takeover of Taiwan, it is unlikely that the PRC could enter into a realistic military conflict with Japan over the Senkaku Islands. In addition, as Ogata (2013) highlighted, Taiwan has been communicating the strategic message to Washington that it is also playing a role in the US's rebalancing by enhancing its defence and deterrence capabilities.

4.3.4 Political risks of high-profile defence cooperation

Taiwan has been long aware of the risks of having explicit security ties with Japan. For example, even when Taiwan was under the Chen administration, which called on Japan to take on a larger regional security leadership role and called for non-materialised Japan–Taiwan–US trilateral security ties, the US requested not to allow Japanese observers to attend the annual Han Kuang military drill in Taiwan in 2006 (Wikileaks, 2006). In 2007, Taiwan's then defence minister, Lee, expressed his desire for the US to facilitate explicit Japan–Taiwan–US military cooperation (Wikileaks, 2006). The then director of the American Institute in Taiwan, Stephen Young, expressed his appreciation for the ROCMND's decision to not invite nine Japanese observers – seven of whom were retired general officers or retired flag officers – to observe the military drill (Wikileaks, 2006). Such strategic compromises arose from Taiwan's communication with the PACOM regarding the Japanese participation, but Young expressed reservations for the subsequent year's Japanese observation participation (Wikileaks, 2006).

The prevention of cross-strait espionage by the PRC – as manifested in the cases of the

former ROC army major general, Hsu Nai-chuan, and the former air force colonel, Liu Chi-ju — must be strictly ensured if Taiwan is to attain Japan's political and strategic trust. Sharp (YouTube, 2016) discussed the view expressed by the former deputy defence minister, Lin Chong-pin, that the major motive behind Chinese espionage attempts that involved collaborators from Taiwan was to discredit Taiwan as an untrustworthy security partner to the US. Furthermore, upon retirement from his position as a de facto Japanese director in Japan-Taiwan Exchange Association in Taipei, Ogata (2016) stated that even though the numerous espionage cases in Taiwan have been ruining the political trust of Japan and the US toward the former, Taiwan has not undertaken effective measures to combat them.

Certainly, Japan has no political control over cross-strait espionage cases. Nonetheless, certain potential grassroots confidence-building measures may be considered. For example, if Japan delivers significant political capital, the defence education exchange between Japan and Taiwan may be an area for consideration (Sahashi, 2014). Here, Ogata (2013) pointed out that Japan–Taiwan relations constitute an unofficial substantive partnership as opposed to an alliance relationship, which excludes defence education exchanges. In 2017, 21 international students graduated from the National Defence Academy of Japan, and their countries of origin included Vietnam, the Philippines, Thailand, Indonesia, Mongolia, Cambodia, and East Timor but not Taiwan (Sankei News, 2017).

4.3.5 Japan remains ambiguous in its defence commitment to Taiwan while providing it with greater deterrence

For these reasons, a Japanese version of the TRA is yet to materialise, while it is highly possible that it will in the future, with the nation perhaps not openly committed to the military security aspects of its relations with Taiwan, as Dreyer (2016) pointed out.

However, Japan has been, albeit incrementally, providing space for Taiwan to expand its regional security contributions to its neighbour's security. The IIPS, a prominent Japanese think tank that, allegedly, has some influence on the Abe administration's strategic thinking, stated soon after President Trump's inauguration that 'a triangular framework consisting of Japan, the United States, and Taiwan should also be constructed' (IIPS, 2017)

While the question of how Taiwan fits into the Japanese concept of CSD remains ambiguous, Japan does provide deterrence for Taiwan vis-à-vis the PRC through its incremental accommodation of its neighbour into international society in defence terms. The Japan–China naval and para-military standoff has been providing the PLA and the Chinese Coast Guard with de facto battle experience in military and paramilitary aspects in the East China Sea (geographically just north of Taipei and Keelung). These battle experiences would no doubt prove useful in the case of a Taiwan contingency (Mazza, 2014). If the PRC succeeds in capturing the Senkaku Islands, Taiwan will risk facing the regular naval presence of both the PLA Navy and the Chinese Coast Guard there. As such, Taiwan will have to expand its own limited defence capabilities to defend its northern flank, complicating potential Japanese military intervention options. As will be discussed in the fisheries agreement chapter, the PRC claims that Taiwan is a part of China and that the Senkaku Islands are a part of Taiwan. This is the PRC's strategic rationale for claiming that the PRC is the 'true defender' of Taiwan's practical interests, such as its fishing rights, vis-à-vis Japan. This rational was discredited after Taiwan signed a fisheries agreement with Japan. It is in Taiwan's national security interests to distinguish itself from the PRC vis-à-vis the issue of the East China Sea lest it risks being isolated from both Japan and the US.

Meanwhile, Japan's military deployments indicate its interest in keeping Taiwan as a security asset in the regional security environment. Regarding the Ma administration's attempt to reconcile with the PRC and Japan's military deployment on Yonaguni, Lai L-Chung, the director of foreign policy studies at a Taiwan think tank, stated in 2009 that '[a]s a result of the government moving closer to China, distancing itself from the US and opposing Japan', Taiwan has gone from being 'a friend that cooperates with the US and Japan to a country that the two allies must defend themselves against. The stationing of troops on Yonaguni will be the first step in this change' (*Taipei Times*, 2009). Regarding this development, Japanese Vice Admiral Umio Otsuka explicitly admitted that the main purpose of deploying the SDF on Yonaguni was not simply to conduct a military defence of the Senkaku Islands but also providing deterrence for Taiwan (Sasakawa Peace Foundation, 2016). Taipei did not publicly protest the military activities at Yonaguni, such as the building of a radar station or the deployment of SDF troops, even though the island is only 67 miles from Taiwan and 93 miles from the Senkaku Islands (Leaf, 2014).

Mayor Yoshitaka Nakayama of the Ishigaki island announced that the island would accept the deployment of 500 to 600 GSDF soldiers equipped with surface-to-air-missiles and surface-to-ship-missiles in the face of the threat of Chinese intrusion in the East China Sea and North Korean ballistic missile launches (Asahi Shimbun, 2016). These forces are expected to engage in a naval and aerial blockade of the Nansei Islands during any contingency involving PLA invasion. In addition, Calvo (2013b) pointed out that Taiwan will monitor the development of Japan's amphibious capabilities (including its mobile anti-ship cruise missiles), which will require Japan to adopt a more integrated approach to military organisation. It has been suggested that surface-to-air missiles, which are more mobile and survivable and can challenge the PLA's superiority

in air defence more effectively, can offer a reasonable alternative to expensive fighter jets, which Taiwan has not been able to purchase for a long time (*Defense News*, 2016). Furthermore, in January 2017, Japan engaged in a military simulation drill involving Taiwan and the PRC amid rising cross-strait tensions. Now that Japan can, in theory, engage in CSD operations, it is attempting to determine what kind of logistical support it can provide during any Taiwan contingency with US observer participation (*Japan Times*, 2017). Again, in theory, the Japanese SDF can now rescue Japanese nationals and can provide military support for any US intervention. According to Reuters (2017), rescue missions for Japanese civilians living in Taiwan and the naval protection of the US Navy were included in the simulation drill conducted by the SDF in 2017, which greatly annoyed the PRC.

Japan's military consolidation of the Nansei Islands would lead to a potential naval blockade against the PLA Navy's attempt to cross the first island chains in order to advance into the wider Asia Pacific region. In addition, the Hard ROC's mine warfare and anti-ship cruise missiles will be crucial to keeping PLA aggression in check. The fact that Japan's national security strategy has become more aligned with the US's military intervention strategy for any contingency in Taiwan will indirectly yet significantly help Japan's own efforts to achieve military fortification of the Nansei Islands. The Hard ROC strategy is aimed at inflicting maximum costs on the PLA through a protracted war, meaning the US would likely come to the conclusion that Taiwan is worth defending, while Japan would provide safe maritime and air routes through its naval and air defence capabilities for US reinforcements. This would both help increase Taiwan's strategic survivability and enhance Japan's security. Under the Tsai administration, the military stance on enhancing Taiwan's asymmetric capability to contain Chinese aggression by increasing its military capabilities and inflicting

maximum kinetic and non-kinetic damage on the PLA is clear. Taiwan's 2017 QDR emphasises a 'multi-deterrence strategy' that uses asymmetrical methods to force invading PLA forces into 'multi-dilemmas'. Taiwan's military has admitted to possessing a certain strike capability in relation to the Chinese mainland, while Taiwan's defence minister, Feng Shih-kuan, revealed that the PRC has deployed a new type of medium-range ballistic missile that is capable of hitting Taiwan and other target locations in the first island chains, including the Philippines, Okinawa, and mainland Japan (*Focus Taiwan*, 2017).

4.3.6 Okinotorishima

Chiou I-jen, a high-profile figure in the National Security Council, became a representative of the East Asian Council, an avenue for Taiwan to diplomatically communicate with Japan under Tsai administration. Chiou was present during Taiwan's maritime talks with Japan, indicating the strategic emphasis the Tsai administration places on its efforts to boost the nation's ties with Japan (Nojima, 2016). Here, Ma Ying-Jeou attempted to dispute the legal status of Okinotorishima in terms of Taiwan's 'fishing rights' even though the island is strategically essential for both Japan and Taiwan. If cross-strait war breaks out and the US decides to send in reinforcements, the Japanese MSDF will have to conduct undersea warfare to secure safe passage for the US Navy (Kotani Asia Maritime Transparency Initiative, 2016). Geographically located between Guam and Taiwan, Okinotorishima is legally defined as an island in UNCLOS legislature and has strategic value for Japan since the vast seas that surround it consist of routes the US navy can use in the case of a Taiwan contingency. In such a scenario, the Japanese MSDF will have to destroy any adversarial submarines and sea mines (Kotani Asia Maritime Transparency Initiative, 2016). According to Kotani, if Taiwan disputes the legal status of Okinotorishima by engaging in the 'freedom of navigation',

this will benefit the PRC's anti-access and area-denial strategy for complicating US intervention efforts during a Taiwan contingency (Asia Maritime Transparency Initiative, 2016). Meanwhile, if Okinotorishima loses its status as an island, and Japan subsequently loses 200 nautical miles of the EEZ, then Chinese government ships will be able to engage in scientific surveys of the EEZ zones claimed by Japan in the seas around Okinotorishima without Japan's consent. These surveys will benefit China's naval warfare capability, including in terms of its anti-access and area-denial strategies, and will help streamline the PLA's naval ability to counter US intervention during a Taiwan contingency (Kotani Asia Maritime Transparency Initiative, 2016). Fortunately, the Okinotorishima issue was politically shelved after the Tsai administration decided to wait for arbitration by the UNCLOS (Lin Asia Maritime Transparency Initiative, 2016b).

Since Taiwan is not even a member of the UNCLOS, it is impossible for Taiwan to resolve the issue. This is a beneficial phenomenon for Japan wherein the absence of Taiwan's statehood, which is a requirement for membership in international maritime security efforts, functions to enhance Japan's superior position over subordinate Taiwan in international society.

4.4 Taiwan's Strategic Need to be an Asset for Japan

As discussed in Chapters 2 and 3, Taiwan must regularly obtain approval from Japan due to its unique status and its ontological security stance, which regards Japan as a 'positive other'. As such, Taiwan must regularly prove its worth as a defence asset to Japan in the hierarchical Japan—Taiwan relationship, and it is up to Japan to decide whether Taiwan is a valuable asset within international society. For certain, Taiwan remains a component of the Sino-Japanese regional balance of power. However, a non-

sovereign community such as Taiwan can be a component of the balance of power within the ESIR paradigm but not in that of the mechanistic school of realist power politics as argued by Waltz (1979). In terms of the former, a non-state actor such as Taiwan can be instrumentalised for Japan's security vis-à-vis the regional balance of power with the PRC, as discussed below.

4.4.1 Taiwan's defence contribution to Japan's security

The PLA's navy and air force have been crossing the Miyako Strait and the Bashi Channel on an increasingly frequent basis in order to advance into the Western Pacific areas, including the east coast of Taiwan. This is making both the Japanese and Taiwanese militaries nervous (Panda, 2017). On this point, Hideaki Kaneda (2016), a former Japanese vice admiral, stated that Taiwan is the most important country for Japan's security and, as such, Japan can no longer afford to exclude Taiwan from its political, diplomatic, and defence engagements, especially given the geographical connectivity between the East and South China Seas and the mutually reinforcing security threats, including the salami slicing tactics, posed by the PRC. Kaneda (2016) pointed out that even though the Nansei Islands can be heavily defended by the Japanese SDF, the Bashi Channel, which is located south of Taiwan, may be comparatively weak. This may enable the PLA's nuclear and diesel submarines, combat ships, and bombers to easily cross into the Western Pacific. Furthermore, currently, Taiwan only possess four submarines, while the PRC has 60, which heavily tilts the military balance of power in favour of the PRC. Therefore, Kaneda (2016) suggested that while the US was able to monitor the geopolitical developments in the Bashi Channel until just recently, it is becoming difficult for the country to sustain its military primacy, which means that Japan needs to enhance its security ties with Taiwan, the Philippines, and the US to enhance ISR activities for the defence of the Bashi Channel.

Considering the PLA's traditional weakness in the ASW (Goldstein, 2011), Taiwan's acquisition of P3Cs from the US has helped consolidate its defence of the Bashi Channel. The former president, Ma, mentioned the importance of conducting maritime patrols around Taiwan's vast maritime areas by the P3Cs, including the overlapping maritime boundaries with Japan and the Philippines (*Focus Taiwan*, 2013). Koda Yoji (2017), a retired vice admiral, argued that Japan can block the Bashi Channel using its naval capabilities because Japan and the US have the upper hand in securing geographical choke points. Therefore, Taiwan should use its aforementioned capabilities to supplement Japan's deterrence capabilities vis-à-vis the PLA's navy in the Bashi Channel. In addition, the Japan–US led 'Fish Hook Undersea Defence Line' — which connects various undersea intelligence network systems across key chokepoints from Japan to Southeast Asia, including Okinawa, Guam and Taiwan — is crucial to detecting any PLA's submarine movements in the first island chains (Singh 2016). Allegedly, this undersea intelligence system also covers the Bashi Channel.

In February 2017, Taiwan deployed PAC3 surface-to-air missiles to expand its air defence in its south-eastern airspace after the PLA's air force conducted several military drills (*Taipei Times*, 2017). Taiwan's ADIZ is missing a corner in the southeast due to the 1959 agreement with the US that allowed American fighter jets to fly from Okinawa to the Philippines. Taiwan's deployment of PAC3 surface-to-air missiles could offset this preparedness deficit (*Taipei Times*, 2017). Meanwhile, Chen Wei-han stated the following:

Japan, South Korea and Vietnam are also armed with supersonic anti-ship missiles with ranges of more than 150km, and Taipei should explore the possibility of cooperating with neighbouring nations to create a defence network

that extends to major bodies of water within the first island chain (*Taipei Times*, 2017).

It would appear that Taiwan is enthusiastic about demonstrating its intelligence capabilities to Japan and becoming a potential security asset for its powerful neighbour in view of the balance of power with the PRC.

In December 2016, Liaoning, the PRC's aircraft carrier, sailed through the Miyako Strait and then into the Bashi Channel to advance into the South China Sea (*The Independent*, 2017). Following this geopolitical development, Japan's MSDF submarines also crossed the Bashi Channel to advance into the South China Sea to enhance the nation's strategic ties with Vietnam and the Philippines and to send a strategic message to the PRC regarding the illegally occupied and militarised Chinese artificial islands (*Sankei News*, 2016). Following this, Japan's Izumo class helicopter carrier sailed in the South China Sea from Yokosuka in May 2017 to engage in a naval drill with India and the US in the Indian Ocean in July 2017. This required its helicopter carrier to cross the Bashi Channel, which is geographically located to the south of Taiwan and to the north of Luzon in the Philippines (Reuters, 2017).

4.4.2 Taiwan as an intelligence provider for Japan' military security

The extent to which Taiwan's intelligence-sharing cooperation with Japan's military has been institutionalised is unclear, despite the unambiguous evidence that Taiwan provided the military intelligence for a PLA nuclear submarine's intrusion into Japanese territorial waters in 2003 and for a North Korean missile launch in 2012. In 1999, retired Japanese SDF officers and diplomats frequently visited Taiwan to provide political and military advice and tours in Japanese military bases (Dreyer, 2016). The geopolitical development necessitating Japan's enhancement of its intelligence exchanges with

Taiwan involved how Japan lacked independent military threat assessment capabilities in relation to Taiwan during the missile crisis in the Taiwan Strait during the period of 1995 to 1996, as will be discussed in the chapter on the ICAO. In 2003, Japan dispatched Nagano Youichi, a retired Japanese military officer, as its first de facto defence attaché in Japan's de facto embassy in Taipei (Yang, 2009). According to the *Washington Post* (2006), Nagao engaged in talks with the Taiwanese government and military officers and frequently communicated military information and reports back to Japan. Yang (2009) argued that the dispatch was an indication that Japan had enhanced its semi-official ties with Taiwan and a symbolic recognition of the emphasis the Japanese government placed on cross-strait issues and Japan–Taiwan security relations. Japan's acknowledgement of Taiwan as an intelligence-sharing partner for Japan's SDF was also symbolic.

Uchida (2006), the former de facto Japanese ambassador to Taipei, admitted that Japan—Taiwan relations involve the intelligence and information sharing that is significant for Japan's national security. Meanwhile, Uchida (2006) also admitted to the existence of intelligence sharing related to situational analysis and individual information sharing in Japan—Taiwan relations. According to Uchida (2006), Taiwan immediately notified Japan of the intrusion of a PLA submarine into Japan's territorial waters in November 2003. While Uchida (2006) was not certain whether this was the first piece of intelligence Japan received regarding the intrusion, he believed that such intelligence sharing can be critical if both Japan's and the US's intelligence fail.

Chinese elites in the PRC have expressed some concern over Taiwan's provision of military intelligence to Japan. For example, Liu (2006), the deputy director at the Department of International Relations at the prestigious Tsinghua University, pointed

out, in reference to Taiwan's Zhongguo-Shibao, that Taiwan's ST anti-submarine aircraft immediately detected a Chinese nuclear submarine and sent this vital intelligence to Japan and the US in November 2004. Meanwhile, Sahashi (2014) pointed out that while Japan and Taiwan would both benefit from regular intelligence exchange arrangements, this would require Japan to make difficult political decisions. In addition, Uchida (2006) discussed the importance of military information sharing to avoid accidents both at the naval and aerial levels. Taiwan and Japan especially need to develop a good mechanism for prior notifications in designated seas. Through this mechanism, Japan could be informed of any Taiwanese naval drills beforehand, especially when the drills are being conducted in international waters (Uchida, 2006). Without this knowledge, Japanese civilian ships and fishing vessels may be put at risk (Uchida, 2006). Uchida (2006) also discussed the importance of intelligence sharing for air forces on a bilateral level, which will be discussed in the chapter on the ICAO.

With heightened tensions in the East and South China Seas, it is possible that the US and Japan will include Taiwan in their regional military intelligence cooperation. For example, Chen cited the RAND report in stating the following:

China could trigger deeper military partnerships between its neighbours and a stronger US presence in the region, with Taiwan reportedly establishing a cooperation with the US and Japan that allows the three nations' military aircraft to be identified as friendly by radar (Taipei Times, 2017).

Taiwan has placed an early warning radar surveillance system at the top of Le Shan Mountain, which can monitor, detect, and track incoming flying objects such as ballistic and cruise missiles, military aircrafts, and stealth fighters at a range of more than 3,000 miles, giving Taiwan time to make an immediate response through competent military

intelligence gathering and analysis capabilities. While the system is expected to be upgraded by 2018 (Military Aerospace Electronics, 2016), at present it can conduct ISR activities from the Korean Peninsula to the South China Sea, especially in terms of submarine-launched ballistic missiles (Military Aerospace Electronics, 2016). Furthermore, during the symposium hosted in Taipei on 14 December 2016, Taiwan's National Security Council's deputy secretary-general, Tseng Ho-jen, disclosed that Taiwan was sharing military intelligence with the United States and argued that the PLA's military activities constituted a threat to Taiwan, Japan, and the US. The council also confirmed that Taiwan is equipped with the capabilities and confidence to deal with such a threat (*Focus Taiwan*, 2016). Clearly, Taiwan wants to use ISR capabilities to prove that it can be a defence asset to Japan and an important component in the Japan—US alliance.

4.4.3 Taiwan as a potential regional cybersecurity contributor for Japan

There exists the potential for cybersecurity cooperation between Japan and Taiwan. While Japan's National Information Security Centre has its own 'cybersecurity strategy', which emphasises the importance of engaging in policy coordination dialogue with the US and other like-minded regional states regarding bilateral cybersecurity cooperation, Taiwan is completely excluded from the strategy (Cunico, Nien-chung, Uchimura at CSIS 14 January 2015). Taiwan is a major springboard for the PLA to test its cyberattack capabilities given the shared language for crime purposes and military operations, which includes massive cyberattacks intended to cripple Taiwan's command, control, and logistics during a cross-strait contingency (Cunico, Nien-chung, Uchimura at CSIS 14 January 2015). Informal, ad hoc anti-cybercrime cooperation may exist between Japan, Taiwan, and the US, and Japanese and Taiwanese firms may engage in informal cyber-security discussions, but what has not been addressed is that Japan and

Taiwan have a low-key yet 'official' cybersecurity cooperation arrangement based on the APEC.

The APEC is one of the few multilateral mechanisms that allow Taiwan to fully engage as an economic entity and to engage in security relationships with other countries, including with Japan on low-key grassroots yet 'official' levels. For example, in 2010, Taiwan's National Communications Commission hosted 18 APEC economies, including Japan, in a discussion on ICT developments related to, for example, information and cybersecurity. Similar events were hosted in both 1996 and 2003 (National Communication Commission, 30 June 2010). During the 2010 event, Taiwan proposed the 'Deployment of Disaster Information Distribution System and Service Model by NGO Exemplar Platforms' (National Communication Commission, 2010) and obtained co-sponsorship from Japan. According to the event, an estimated US\$50,000 would be invested to study how best to combine cloud computing, open resources, online community websites, and various governmental and private resources for the effective response to disasters and the subsequent necessary reconstruction projects (National Communication Commission, 2010). In the absence of formal diplomatic channels, Taiwan's grassroots yet official security channels with Japan, such as the APEC, are a vital component of Japan–Taiwan security relations. In fact, this is an underappreciated component of the regional security environment and includes operations such as cybersecurity vis-à-vis HADR cooperation.

Whether this form of security cooperation can affect the regional power politics in Japan's geopolitical calculous remains to be seen. Nevertheless, it does have symbolic value in relation to Taiwan's contribution to regional security. In addition, Matsuda (YouTube, 2017) pointed out that while Taiwan's guest membership in the ICAO was

rejected due to the PRC's sanctions, the nation continued to be represented by the APEC. An APEC event in Taiwan also discussed the nation's potential for law enforcement cooperation for cybersecurity operations with other countries, including Japan, which will be discussed in the chapter on the ICAO.

4.4.4 Japan's renegotiation of Taiwan as a security dialogue partner

As discussed in the previous chapter, for Taiwan, track 2 diplomacy is significantly different in nature from the track 2 diplomacy pursued by de jure states. For example, according to the *Taipei Times*, Edwin Feulner, the founder of the Washington-based Heritage Foundation, played a key role in the first Tsai—Trump phone call (*Taipei Times*, 2016b). Japan has been accommodating 'semi-official' track 2 diplomacy with Taiwan in relation to Japan's security and has incrementally expanded the 'official' side of its unofficial security dialogue channels with Taiwan. Japan renegotiated Taiwan's membership as a security dialogue partner in international society without explicitly recognising its officiality. In addition, much like in the US, as the former de facto director Ogata in Japan-Taiwan Exchange Association in Taipei (2013) pointed out, those who are working at prominent think tanks and universities in terms of the track 2 diplomacy in Taiwan can be easily deployed in the central political decision-making body if there is a change in administration. However, the concept of track 2 and track 2 diplomacy is ambiguous in Japan—Taiwan security dialogues, necessitating the creation of unique semi-institutionalised communication channels.

In September 2011, Abe attended a track 2 security dialogue as a high-profile political figure from the opposition party with former prime ministerial experience and argued that Taiwan was an important partner for Japan, one that shared its universal values of democracy, freedom, and peace (Nihon Ritouki-No-Kai in Aichi Branch, 2015).

However, the fact that he attended the dialogue as the leader of the opposition party may have diminished the symbolic political importance of his argument to a great degree (Nihon Ritouki No Kai in Aichi Branch, 2015). At the same time, it can be argued that the fact that Abe attended the dialogue at all is significant within the context of the reassessment of the strategic value of Taiwan's track 2 diplomacy in its security communications with Japan. Here, Yang (2009) pointed out that the shared perception of the threat the PRC poses for both nations' national security played a significant role in facilitating security talks between Japan and Taiwan. The track 2 security dialogue was instrumentalised to put political pressure on Taiwan to diffuse any maritime security tensions with Japan amid other maritime tensions in the East China Sea resulting from the perceived cross-strait united front against Japan before the conclusion of the fisheries agreement. For example, at the Heritage Foundation's conference in Washington that was held in early 2013, the former US deputy assistant secretary of state for East Asian affairs, Randy Schriver – while clearly indicating that the US prioritised its alliance with Japan over the TRA – stated that 'Japan is arguably Taiwan's second-most important security partner' and added that '[i]f Taiwan undertakes activities that cause problems with Tokyo, that will cause problems with the United States and that should be avoided' (Taipei Times, 2013). Furthermore, Su Tsengchang, the former director of the Executive Yuan, who competed with Tsai for the presidential nomination, also visited the LDP's think tank in Tokyo to exchange views with Japanese legislators regarding the negotiations on the impending fisheries agreement and insisted that Taiwan enhance its diplomacy based on its shared values with Japan, the US, and South Korea for stability and prosperity in Northeast Asia (Japan-Taiwan Exchange Association, 2013). Here, Ishihara (Japan-Taiwan Exchange Association, 2013) pointed out that the DPP was conscious of Abe's value-based diplomacy and expressed a keen sense of rivalry with a PRC that does not share its political values. Regarding this visit, the *Taipei Times* (2013) reported the following events:

[t]he senator — the brother of Japanese PM Abe — said that Ishiba did not propose to 'include Taiwan in the US-Japan Security Treaty', but underlined that policymakers and countries should get to understand Taiwan better before assessing Asia-Pacific security situations, and Taiwan should not be ignored or excluded from any security plans.

While at this time the DPP was an opposition party, and it was the KMT that ultimately concluded the fisheries agreement with Japan, the political heavyweights in both the KMT and the DPP were clearly conveying a strategic message about enhancing Taiwan's ties with the Japan–US alliance through 'track 2 diplomacy' with the LDP's think tank in Tokyo. The conclusion of the fisheries agreement provided symbolic leverage for Taiwan to strategically align with the alliance.

Such a security dialogue was instrumentalised to encourage Taiwan to improve its self-defence capabilities and to become an asset within the Japan–US alliance. For example, at the conference organised by Project 2049 in December 2016, Schriver insisted that Taiwan must develop its self-defence capabilities to deter the PLA from invading the nation (Project 2049, 2016). Meanwhile, Matsumura (2010 page123) insisted that, in partnership with the US, the Japanese government communicates its strategic message through 'unofficial' security talks to convince Taiwan to acquire military assets and to maintain a cross-strait balance of power amid the rapid development of the PLA's capabilities. Elsewhere, Matsumura (2010 page123) argued that Japan should welcome the professionalisation of the Taiwanese military, including in terms of the abolition of

the conscription system, and should allocate more financial resources to the acquisition of US-made weapons. Matsumura (2010) also insisted that Japan and the US jointly encourage Taiwan to develop self-defence capabilities qualitatively through track 2 security dialogues. Regarding Taiwan's acquisition of the P3Cs, President Ma argued in 2013 that the P3Cs acquired by Taiwan were qualitatively equal to the US's P3Cs (*Focus Taiwan*, 2013). This qualitative upgrade of Taiwan's military hardware clearly boosted the country's military morale. Meanwhile, Ogata (2016) further pointed out that rather than acquire second-hand military hardware, Taiwan acquired the upgraded version of its F16A/Bs. In addition, the P3Cs and Patriot PAC3s boosted Taiwan's air defence capabilities, anti-submarine warfare capabilities, and missile defence capabilities. In fact, these are still used by Japan and the US in terms of frontline defence, which has, in turn, increased Taiwan's confidence in its self-defence capabilities.

Foreign Minister Joseph Wu called on Japan to open a security dialogue mechanism between Japan and Taiwan in his interview with *Sankei Shimbun* in 2018. If Japan encourages more communication with Taiwanese think tanks, the Tokyo MOD's think tank, NIDS, may or may not engage with its Taiwanese counterpart in the future. An informal exchange of ideas through the publication of informal reports can shape policy-making decisions related to Japan–Taiwan relations. For example, NIDS published its China Security Report on Taiwan–China relations, stating the following:

[i]t is possible that if Taiwan devotes less of its efforts to self-defense, China's actions will become even more expansionistic. In that sense, Taiwan's attention to improving its self-defense capabilities may influence the stability of the region as a whole. Further, maritime affairs also become important since Taiwan lies at the nexus of international channels of navigation tying together the East

China Sea, the South China Sea, and the Pacific Ocean. As geographical confrontation in East Asia takes on new dimensions, the strategic importance of Taiwan will likely increase as well (NIDS China Security Report, 2017, p. 80).

As discussed, Taiwan's self-defence capabilities cannot be underestimated in their ability to supplement the military balance of power for Japan in terms of the issues of the East and South China Seas, the Bashi Channel, and the wider Pacific regions and in terms of preventing the PLA Navy's advancements. Russel Hsiao, the executive director of the Global Taiwan Institute – the only Washington-based think tank that solely focuses on issues related to Taiwan (Brooking Institution, 2017) – indicated that the informal reports published by NIDS on Taiwan–China relations are important for Taiwan's policy-making elites in terms of understanding Japan's strategic thinking and making better decisions regarding issues pertaining to the Taiwan Strait.

Furthermore, based on track 2 diplomacy, where high-profile figures are represented, Taiwan's attempt at regional strategic adjustment in accordance with the shift in the regional security environment is significant. For example, the former Indian defence minister, George Fernandes, attended a conference entitled 'India–Japan–Taiwan Trialogue: Prospects for Democratic Cooperation' that was hosted by a Taiwan think tank in Taipei (*Taipei Times*, 2004). Reportedly, this informal conference aimed to promote the awareness of potential areas of tripartite cooperation involving India, Japan, and Taiwan in the areas of economic growth and trade, high-tech industry, and democratic development, as well as other inter-regional issues that are of vital interest to these nations (*Taipei Times*, 2004). As will be discussed in the chapter on the ICAO, the issue of how to reduce Taiwan's excessive economic dependence on the PRC is one component of the regional security environment. Under the current Tsai administration,

as one of its foreign policy priorities, Taiwan is shifting its diplomatic focus toward Southeast Asia and India to address the issue of economic security (*Taiwan News*, 2017). Importantly, under the Modi administration, India is not interested in being bound by the One-China policy in the absence of the PRC's One-India policy (Taiwan News, 2017). Ogata (2016) expressed the view that under the New Southbound Policy initiated by the Tsai administration, Taiwan will seek multi-dimensional relations with both the ASEAN states and India. This will require Taiwan to reject unification with China while placing pivotal emphasis on its ties with Japan and the US in view of deepening its relationships with regional states other than the PRC. In addition, the brochure distributed in September 2016 by the New Delhi-based India Foundation at the Shangri-La Hotel in Singapore – where it functioned as a co-host with Nanyang University's RSIS – pointed out that, along with participant member states that discussed the ties between India and the ASEAN states, a Taiwanese representative, Hung Mao-tien president and board chairman of the institute for National Policy Research in Taiwan – somewhat surprisingly gave a talk on maritime security at the pre-conference comity workshop (Cordner, 2016). According to Cordner (2016), high-profile pro-Taiwan Japanese figures such as Kishi Nobuo, minister of state for foreign affairs, were present. Clearly, Taiwan is not excluded from the strategic cooperation between Japan and India in the Indo-Pacific Region.

4.5 Japan-Taiwan Non-Traditional Security Cooperation: Japan-Taiwan Coast Guard Relations

It is difficult for the PRC to sanction Japan-Taiwan non-traditional security cooperation at constabulary levels since coastguard exchanges are considered less-coercive and more normative. Bateman (2003) argued that the coastguard is suitable for facilitating regional maritime security cooperation given its confidence-building effects, the less escalatory nature of maritime policing, and the importance of protecting the marine environment. Furthermore, if Japan–Taiwan security interactions are conducted at the constabulary level, it is harder for the PRC to oppose to it than military cooperation, while the constabulary cooperation between Japan and Taiwan helps the latter to align itself more with the former within the context of Japan–China–Taiwan relations. As such, the coastguard cooperation between Taiwan and Japan has positive effects in terms of regional peace and stability, while the One-China policy remains unbreached within the context of Sino-Japanese regional great power management. Thus, Japan is instrumentalising Taiwan as a maritime constabulary asset for non-traditional security efforts within the context of Japan's One-China policy. However, this cooperation will be discussed in more detail in the chapters on CSD and the fisheries agreement since any Chinese maritime intrusions of the East China Sea are largely conducted by white hulls, and Japan is more concerned about grey zone contingency.

Taiwan cannot be represented in the ReCAAP or the Information Sharing Centre due to the sanctions imposed on it by the PRC, which have prevented the nation from gaining access to intelligence on regional maritime piracy and crimes and from communicating its own maritime piracy issues in its neighbouring seas (Glaser, 2015). However, it is precisely because Taiwan cannot be included in multilateral maritime security cooperation in the region that its bilateral coastguard cooperation with Japan and other countries, such as the Philippines, is even more critical for maritime peace and stability in the East and South China Seas and the Bashi Strait. In short, Japan can enhance its non-traditional maritime security cooperation with Taiwan bilaterally through these coastguard exchanges.

According to Glaser (2015), Taiwan engages in coastguard cooperation with other states, which, in addition to Japan and the Philippines, includes Malaysia, Thailand, Indonesia, and Vietnam, in relation to maritime anti-human trafficking and maritime anti-drug smuggling. The author also pointed out that Japan engages with Taiwan for annual mutual visits and working-level meetings for limited coastguard cooperation (Glaser, 2015). Furthermore, according to an interview with Foreign Minister Joseph Wu (*Taipei* Times, 2019), Taiwan has signed a coastguard agreement with the Marshall Islands, and President Tsai plans to sign coastguard agreements with Palau and Nauru during official visits to these countries – which still have official ties with Taiwan – to provide them with patrol vessels in the future. Reportedly, the new sea patrol cooperation pact concluded between Taiwan and Palau enables Taiwan to conduct training, exchange patrol personnel, fight maritime cross-border crimes, protect Palau's fishing vessels, and engage in a joint maritime rescue drill in the seas around Palau using Taiwan's 1,800-tonne patrol frigate (Focus Taiwan, 2019). In fact, Wu (Taipei Times, 2019) pointed out that despite the vast oceans surrounding these island states, their coastguard capabilities are limited, and Japan, Australia, and the US have thus already provided them with patrol vessels. Clearly, Taiwan can supplement Japan's maritime law enforcement capacity-building efforts for Palau in the South Pacific, which are a part of Japan's Indo-Pacific strategy.

4.5.1 Japanese constabulary asset sales to Taiwan

It is unknown whether the Japanese government or a non-state actor in Japan was responsible for maritime constabulary asset sales such as the export of a coastguard patrol ship to Taiwan despite the One-China policy. Even after the Japanese derecognition of the ROC, Japan's maritime security activities, including constabulary asset sales to Taiwan, continued. To understand Japan's maritime capacity-building

efforts for Taiwan's maritime law enforcement, the Taiwan Coast Guard's main missions must be discussed. Here, Morris (2017) explained that the three main missions of the coastguard are: (i) the regulation of coastal zones under national jurisdiction, (ii) the inspection of cargo entering and exiting ports, and iii) the investigation and seizure of smuggling operations and illegal vessels or people entering or exiting the country. Other important missions include search and rescue, intelligence gathering, and support for maritime research and development (Morris, 2017).

The Japan-made coastguard patrol ship that is officially named *Dexinguang* was produced in Japan's shipyard in 1977 and participated in Taiwan's maritime law enforcement activities for 37 years, with its security missions including anti-drug smuggling and search-and-rescue missions in Taiwan's waters (CNA, 2015). Reportedly, the *Dexinguang* succeeded in conducting an anti-drug smuggling mission in the ocean in the south of Taiwan, while it also engaged in escort missions to ensure the safety of fishing vessels during the Taiwan–Philippines maritime tensions that stemmed from the tragic killing of a Taiwanese fisherman by the Philippines Coast Guard in 2013 (CNA, 2015). Clearly then, the *Dexinguang* was utilised for operations involving both Taiwan and the Philippines, both of which are considered to be Japan's security partners.

4.5.2 Potential for Japan-Taiwan coastguard relations

Sahashi (2014) noted that coastguard exchanges have the potential for growth if Japan allocates political capital to enhancing Japan–Taiwan relations. Regarding this point, in an interview with *Yomiuri Shimbun* (2016), President Tsai discussed Taiwan's intentions to enhance its maritime search-and-rescue cooperation, to resolve any fisheries issues, and to enhance its marine scientific cooperation with Japan through

maritime cooperation dialogue. In fact, President Tsai expressed her strong political interest in improving the bilateral ties with Japan for regional stability and stated that 'from Taiwan's perspective, PM Abe is someone we [have been] quite familiar with over a long period of time. We also understand that he has goodwill toward Taiwan' (Reuters, 2016). On examining this non-traditional maritime security cooperation between Japan and Taiwan, Kotani argued that both nations can engage in a dialogue on regional maritime security issues such as maritime piracy and marine environmental protection (Global Taiwan Institute, 2016). Coastguard exchanges should be also enhanced via maritime cooperation dialogue. In terms of the Taiwanese perspective, the vice president of the Taiwan Brain Trust, Lin Ting-hui, suggested that the two coastguard operations can establish a 'notification system within the maritime law enforcement agencies and a hotline between the competent regulatory authorities on marine affairs' and added that '[t]he two sides could even conduct joint maritime rescue drills, loan each other equipment to fight marine pollution' and 'conduct bilateral training and exchanges for law enforcement and marine affairs personnel' (Asia Maritime Transparency Initiative, 2016a).

4.5.3 Concrete Japan–Taiwan coastguard relations

As discussed earlier, maritime-related tensions surfaced between Japan and Taiwan during the Okinotorishima issue, even though the EEZ around the island is a vital route for the defence of Taiwan within the context of Japanese CSD. However, in 2017, the two nations signed an MOU agreement over the emergency and rescue operations carried out by the coastguards and essentially shelved the legal dispute over Okinotorishima (*Focus Taiwan*, 2017). Monma (2018) admitted that Japan had dispatched a coastguard attaché to the Japan–Taiwan Interchange Association and hinted at the possibility of close daily communications with the coastguard in Taiwan

while suggesting that this will likely lead to bilateral joint search-and-rescue drills between the two coastguards in the near future. In 2018, a Taiwanese fishing vessel was rescued by an MSDF helicopter 190 kilometres from Iwojima after the Japanese coastguard received a rescue signal from the vessel and subsequently launched an aircraft search (Asahi Shimbun, 2018). Thus, the Japanese military and coastguard were instrumentalised to rescue a Taiwanese civilian entity in Japan's maritime domain, and interestingly, it appears this incident has not brought about a diplomatic crisis in Sino-Japanese relations.

4.5.4 Taiwan's misinterpretation of Japan's maritime security ties to the Philippines

As discussed in Chapters 2 and 3, the de facto state of Taiwan's political input has no effect on Japan's security policy given the strong hierarchical relationship, with the more powerful Japan able to impose rules and norms on its subordinate Taiwan, which is clearly manifested in the Japan–Taiwan–Philippines coastguard ties. According to Kotani, Taiwan has expressed concerns over Japan's maritime law enforcement capacity building in relation to the Philippines Coast Guard. This concern is related to the fact that Taiwan holds the largest land feature, Itu Aba, in the South China Sea and is worried about the Philippines' potential use of Japan-provided coastguard ships, which could weaken Taiwan's maritime security in the area (Asia Maritime Transparency Initiative, 2016). However, while maritime-related tensions between Taiwan and Japan or the Philippines do occasionally surface then, the major challenge to the Taiwan Coast Guard comes from the PRC (Morris, 2017). In fact, Taiwan's coastguard has engaged in tabletop exercises and other drills with its Philippines counterparts (Glaser, 2015). Nonetheless, Japan, Taiwan and the Philippines clearly require enhanced communication mechanisms to ensure the message regarding the

importance of the Philippines for Japan's regional security is clearly grasped by Taiwan. In short, the Japan-provided coastguard assets will safeguard the Philippines' interests in the South China Sea and will help it to combat maritime crime, piracy, and terrorism to defend the liberal maritime order in international society, which will, of course, also benefit Taiwan.

As discussed in Chapter 2, in the ESIR paradigm, great powers are accorded special privileges that smaller states are unable to enjoy. Japan is no exception to this rule within the regional context. As a regional great power, Japan can lay down rules and norms to compel smaller nations such as Taiwan to behave in accordance with Japan's preferences. Meanwhile, while the Philippines may not be a particularly strong country, it does enjoy the legal privileges afforded to a sovereign state. This is not the case for Taiwan. As such, if the two nations face maritime tensions, Japan will not side with Taiwan because the sovereign state of the Philippines has a higher legal status than the de facto state of Taiwan within international society.

4.6 Japan-Taiwan Non-Traditional Security Cooperation in the HADR

It is difficult for the PRC to sanction the development of Japan–Taiwan security relations if this involves humanitarian components such as the HADR, despite the Sino-Japanese great power management of Taiwan. This is due to the fact that such a sanction would damage the PRC's reputation within international society. Japan's CSD operation can enhance its regional soft security leadership, including in terms of the HADR, while Taiwan can be a supplementary asset for Japan in international society. Miller (2014)

pointed out that Japan's defence policy that is aligned with the concept of proactive contribution to peace will mean the nation remains largely defensive in the East China Sea but will be more proactive in soft security affairs such as the HADR. This became clear when Typhoon Haiyan struck the Philippines in 2013. In East Asia, where natural disasters have resulted in huge numbers of causalities, the HADR has become an essential component of regional security affairs. Japan and Taiwan are both natural disaster-prone areas in the region, which means both have expertise in this nontraditional security field. If the Japan-Taiwan HADR-based cooperation is enhanced, this will improve the crisis management capabilities of the two actors, which could then subsequently be applied to other contingencies. In fact, non-traditional security cooperation has special symbolic importance for Taiwan given that non-controversial humanitarian international soft security cooperation involving, for example, the HADR and non-military medical cooperation have significant political implications, as indicated by the improvement of Japan's perception of Taiwan after the latter donated generously and provided emotional support to the former during the triple disasters of 2011. Importantly, Japan can accommodate Taiwan into international society by cooperating in HADR activities with Taiwan on humanitarian grounds, while Taiwan can enhance its HADR cooperation with Japan to help increase its importance in the Japan-US alliance and to counter the PRC's attempts to exclude Taiwan from it. This will also help to further tip the Sino-Japanese regional balance of power in Japan's favour.

4.6.1 Taiwan's HADR capabilities and the PRC's sanctions

With the endorsement of the US, the Taiwanese Air Force transported relief supplies using a C-130 transport plane and used US military bases for refuelling and repairs to help Haiti during the massive earthquake of 2010 (Guilloux, 2016). Furthermore,

Taiwan's air force and navy provided both official and private relief supplies using the C-130 transport plane and a landing ship-tank for HADR operations during Typhoon Haiyan in 2013 (Guilloux, 2016). These activities demonstrate Taiwan's contributions to supplementing Japan's HADR activities both regionally and globally. However, given the PRC's sanctions, Taiwan has been severely restricted in carrying out HADR missions on either an international or regional scale. For example, in 2015, the PRC pressured the Nepalese government into refusing to accept an emergency response search-and-rescue team from Taiwan following the huge earthquake in Nepal, while the Nepalese government did accept medical help from Taiwan (New York Times, 2015). In fact, Taiwan has significant HADR capabilities, as manifested in its replenishment ships, which include the AOE 532 Panshih that forms part of the ROC navy. This provides Taiwan with naval and aerial re-supply capabilities with state-of-the-art medical facilities, even when it is physically away from its port (Strategic Review, 2015). In addition, on the grassroots level, Taiwan's massive donations and emotional support for Japan during the 3/11 triple disasters in the Tohoku region increased the awareness among the Japanese of Taiwan's potential to carry out HADR cooperation, as will be examined in the chapter on the fisheries agreement. In fact, the Ma administration cited humanitarian work as one of the major components of Taiwan's second line of defence, one that clearly enhances Taiwan's international development contribution (Glaser, 2015). These geopolitical developments have ensured that Japan-Taiwan HADR cooperation are critical given that the sheer scale of the damage resulting from natural disasters such as the 2004 tsunami in Southeast Asia, the 2011 triple disaster in Japan, and 2013's Typhoon Haiyan calls for major transnational efforts.

4.6.2 Japan's instrumentalisation of the HADR to enhance its security ties with Taiwan

The MOFA in Tokyo immediately announced the dispatch of a specialised search-andrescue team following the massive earthquake in eastern Taiwan in 2018 (MOFA, 2018).

Abe offered his condolences, displaying a picture of his calligraphy on Facebook
(*Taipei Times*, 2018). Reportedly, the specialised search-and-rescue team members
consisted of the MOFA, the JICA, the Tokyo Fire Department, the Tokyo Police
Department, and the Japan Coast Guard's Tokyo branch, while specialised HADR
equipment was also dispatched (*Taipei Times*, 2018). Clearly then, Japan had chosen to
openly dispatch the necessary relevant personnel and soft security equipment.
Furthermore, in 2018, Abe used Twitter to express his gratitude to Taiwan in Mandarin
Chinese in response to President Tsai's Twitter message expressing (in Japanese) her
condolences and readiness to provide support for the natural disaster that occurred in
western Japan (Yahoo Qima Xinwen, 2018).

In addition, according to the then mayor of Tainan City, Lai – who was known for having a strong background in foreign affairs and who later became a premier – Japan's investigation team arrived in Tainan's earthquake-stricken region to determine what support was needed by the local citizens following the huge earthquake of 2016 (Japan National Press Club, 2017). This development was appreciated by both Lai and the locals, especially given that other countries had been sending goods to Tainan without any consideration for the local needs (Japan National Press Club, 2017). This proves that the Japanese understand the importance of information gathering and intelligence analysis for effective HADR missions and demonstrates the significance of grassroots exchanges in times of natural disasters for Japan–Taiwan security relations. In fact, following the 1999 earthquake in Taizhong, the Japanese rescue team arrived in Taiwan

before those of any other country. In short, since both Japan and Taiwan face the risk of natural disasters such as earthquakes and typhoons, enhanced technical and personnel cooperation will undoubtedly increase the regional security.

The NGO Peace Winds visited Japan and Taiwan for disaster preparedness coordination in April 2017, bringing Japan, Taiwan, the US, and the Philippines together to improve the regional capacity and collaboration in disaster relief activity (Peace Winds, 2017). Based on previous Japan–Philippines–US and Japan–Korea–US trilateral preparedness initiatives, Taiwan is a natural partner for such initiatives due to its vulnerability to diverse natural disasters, its geography, and its strong interest in regional cooperation (Peace Winds, 2017). The dialogue for the Peace Winds initiative included participants from the Ministry of Foreign Affairs, the Ministry of National Defence, various other government stakeholders, a number of NGOs, and various private sectors from both Japan and Taiwan (Peace Winds, 2017). This initiative was aimed at making Taiwan a natural partner and improving its HADR cooperation with Japan, the US, the Philippines, and various other countries.

Japan also instrumentalises the APEC as an avenue to cooperate with Taiwan for regional security matters such as the HADR. Glaser (2015) pointed out that a large majority of Taiwan's HADR cooperation relates to participation in the APEC's Emergency Preparation Working Group, which is authorised to enhance emergency preparedness for natural disasters in the region. The Taiwanese Li Wei Sen is co-chair of the group alongside Kyoshi Natori (Glaser, 2015).

4.6.3 Taiwan's HADR contribution to Japan's regional soft security leadership and soft power projection in international society

Taiwan's air force and navy engaged in huge HADR operations during Typhoon Haiyan,

which severely affected the Philippines in 2013 (Glaser, 2015). If Japan's large deployment of the SDF in the post-war period during Typhoon Haiyan was a major component in enhancing the nation's soft power in the Philippines, then Taiwan's HADR missions certainly supplemented Japan's HADR operations, albeit indirectly, which is in sharp contrast to the PRC's delayed dispatch of one hospital ship and its provision of severely limited financial support, despite domestic and international criticism (Reuters, 2013).

Thus, the HADR is deemed to be a non-politicised political soft power contributor for Japan in the regional security environment. Taiwan can potentially play a supplementary role in Japan's regional soft security leadership and can enhance its regional soft power in partnership. According to the Philippines's National Security Policy implemented between 2011 and 2016, natural disasters were regarded as one of the main national security priorities for the Philippines (Philippines Government National Security Policy 2011–2016). Clearly, Japan–Taiwan HADR cooperation has relevant regional security implications for the Philippines and a number of other Southeast Asian member states. In fact, Japan participated in an anti-disaster drill with Malaysia and Australia in Taipei in March 2019 (*Taiwan Today*, 2019).

Japan's JICA undertook the training of Taiwanese HADR experts during the 2000s. Here, Japanese disaster relief team members trained with their Taiwanese, Singaporean, and Filipino counterparts to improve their knowledge and rescue skills for overseas disasters (Japan JICA, 2005). Following the huge landslide caused by the typhoon that hit Kaoshiung in 2010, Japan and Taiwan signed a mutual help agreement to improve sand control, with the two nations engaging in mutual exchanges involving anti-disaster experts and academics to improve sand control technology and minimise the risk of

landslides (Taipei Economic and Cultural Representative Tokyo, 2016). Currently, Taiwan is proposing that it cooperates with Japan in providing anti-disaster and post-disaster models to Southeast Asian countries to help them improve their capacity building and better handle the effects of climate change (Taipei Economic and Cultural Representative Tokyo, 2016). In fact, the tiny Pacific Islands state of Palau – which has official ties with Taiwan – called for anti-drought support from Japan and Taiwan after it imposed a state of emergency due to a lack of drinkable and useable water (AFP, 2016), with climate change cited as the reason for the drought. Here, Japan may have agreed with Taiwan's views on the security of Palau as an anomalous case, while Palau diplomatically recognises Taiwan as a de jure state. As discussed in Chapter 2, international society is biased in favour of sovereign states, which can be somewhat detrimental to non-sovereign political communities. Therefore, Taiwan can enjoy some equality in its relations with Palau in international society as a sovereign state, which increases Taiwan's political importance in the South Pacific and in relation to Japan's Indo-Pacific strategy.

4.6.4 HADR cooperation with Taiwan as a strategic shield for Japan's normative power in the Sino-US rivalry

The PRC may instrumentalise the HADR as leverage to lure Taiwan away from the Japan–US alliance and towards a cross-strait united front strategy. In other words, Japan's enhancement of its HADR cooperation with Taiwan can be viewed as a strategic asset used to firmly lock Taiwan into the Japan–US alliance. A similar view was expressed in Shang-Su Wu's 2016 research on the HADR in Taiwan. Here, Wu (2016) discussed the possibility of the PLA utilising the natural disasters that may strike Taiwan as a justification – on humanitarian grounds – for the deployment of the PLA on Taiwan's soil to 'safely' and 'legitimately' invade Taiwan. According to unclassified

documents from the US's de facto embassy obtained from Wikileaks, during Typhoon Morakot, Ma's incompetence in effectively handling the damage inflicted by the typhoon was heavily criticised. The PRC could have attempted to exploit this fact to dispatch its military and land on Taiwan's soil (Wikileaks, 2011).

In addition, the typhoon was a turning point in terms of reducing the number of black hawk helicopters in favour of heavy lift helicopters for HADR operations in Taiwan. This was welcome news to the PLA (Wikileaks, 2011) since it could easily disrupt the regional military balance in the Taiwan Strait (hitherto sustained by the Japan–US alliance; Wikileaks 2011). However, both US C-130 aircraft and naval vessels arrived in Taiwan to lead massive HADR operations. Meanwhile, for unspecified reasons, Japan's MSDF vessel's port call was declined (Wikileaks, 2011). Had things turned out differently, Japan–China–US relations over Taiwan may have been negatively affected due to the Taiwanese leadership's lack of judgment and mismanagement (Wu, 2016). Clearly, the geopolitical developments surrounding Typhoon Morakot in Taiwan have increased the strategic need for Japan to enhance its HADR cooperation with Taiwan as well as its normative power in the Sino-US rivalry.

4.7 Summary

Japan is navigating Taiwan's ambiguous status in the Japanese concept of CSD while instrumentalising Taiwan for its own security. Meanwhile, Taiwan is benefiting from Japan's greater deterrence capabilities vis-à-vis the PRC. Taiwan's self-defence capabilities are also an indirect military asset for Japan in reducing Chinese military pressure on Japan's southern flank. Japan–Taiwan security relations may develop further in the field of non-kinetic military cooperation – including in terms of intelligence sharing – and in the field of non-traditional security, including track 2

security dialogue, coastguard exchanges, and the HADR. If Japan enhances non-traditional security ties with Taiwan in ways that are regarded as normative, humanitarian and non-coercive, such as the coast guard exchanges, and the HADR, Japan can still appease its One China Policy. Furthermore, Japan can tilt Sino-Japanese power-political relations in its favour by instrumentalising Taiwan in terms of non-kinetic defence security and a non-traditional security asset. This will, in turn, enhance Japan's standing vis-à-vis the PRC within the context of the One-China policy. Meanwhile, Japan will continue to navigate the ambiguities of the regional great power management with the PRC. Finally, regardless of whether Japan adheres to the Yoshida Doctrine or shifts toward the Abe Doctrine with regard to its external security policy, the nation's security ties with Taiwan will continue to make incremental progress. The next chapter examines how Japan can socialise Taiwan through compliance with norms and rules generated in international security agencies, the membership to which requires statehood, such that Japan can enable Taiwan to contribute to Japan's security through aviation participation.

Chapter 5. International Civil Aviation Organization

5.1 Introduction

In this chapter, we examine how Japan has gradually accommodated Taiwan as a normative aviation security asset into the regional aviation security environment. This has allowed Taiwan to contribute to Japan's security by modifying the balance of power in relation to the PRC, while Japan continues to navigate the ambiguities of the regional great power management with the PRC within the context of the One-China policy. Taiwan can never become a full-fledged member of the International Civil Aviation Organization (ICAO) since the organisation requires statehood for membership, but Japan can enhance Taiwan's 'as if' membership, thereby allowing Taiwan to fill the void in Japan's international aviation security efforts. While Taiwan is attempting to prove itself as a valuable aviation security asset for Japan, according to the ESIR, this is, in fact, consolidating Japan's superior position in the hierarchical Japan-Taiwan relationship. This chapter also elucidates on how Japan can socialise Taiwan through compliance with international security norms and rules without affording it explicit membership; rather, Taiwan will be given an alternative opportunity to enable the nation to behave as if it is a member of international security efforts. Here, we adopt the area of aviation security as a case study.

First, we discuss what the ICAO is and how Taiwan has been a marginalised entity within the organisation in the absence of its membership, which requires statehood, even though Taiwan has ample expertise and resources to contribute to international aviation security, including in terms of Japan. Second, we discuss how Taiwan's marginalisation with regard to the ICAO affects Japan's aviation security through conceptualising what Japan can do to instrumentalise Taiwan in terms of international

aviation security efforts. Third, we examine the implications of Japan's enhancement of its aviation security ties with Taiwan within the context of the ICAO. Fourth, using an Open Skies agreement as a case study, we examine the geopolitical implications of this agreement and how Japan instrumentalises Taiwan as an economic security asset in the power politics related to Sino-Japanese relations. Fifth, we examine the PRC's non-normative Air Defence Identification Zone and how it negatively affects the regional legal order in aviation terms, while we also discuss how the role of Taiwan as an autonomous stakeholder in the ADIZ issue fits into the regional aviation security. Lastly, we examine Japan—Taiwan anti-international terrorism and anti-transnational organised crime cooperation to illuminate how Japan can enable Taiwan to behave as if it is a member of the ICAO, despite the lack of official membership in the absence of statehood.

5.2 What is the ICAO and what does it mean for Taiwan?

5.2.1 The main agenda of the ICAO

The ICAO is a UN-specialised agency that was established in 1944 by member states to manage the administration and governance of international civil aviation based on the Chicago Convention, where the member states and industry groups arrive at a consensus on the rules and norms of international civil aviation, and the recommended practices, policies for safety, efficiency, economic sustainability, and environmental responsibility in civil aviation. Furthermore, ICAO member states attempt to ensure that their national civil aviation practices and regulations conform to international norms. This allows global civil flights to operate safely. The ICAO also attempts to support the coordination of aviation capacity building for states to pursue multilateral progress in air safety and navigation, maintain air transport standards, and audit civil aviation capabilities for the air safety and security of the member states (ICAO, n.d.).

Clearly, since aviation is closely linked to airspace/airport management, the air force, law enforcement, air safety, aerial economy, and air transport, the ICAO is essential to the security of both Japan and Taiwan.

5.2.2 Taiwan as a marginalised entity in the ICAO

The ICAO restricts its membership to states that are members of the UN according to Article 93 of the Chicago Convention (ICAO, 2006 p. 42). Taiwan was formally excluded from the ICAO after it was expelled from the UN and the PRC replaced the ROC within the organisation. Taiwan's legally unrecognised statehood complicates its participation in international organisations. President Barack Obama's explicit statement regarding Taiwan's non-representation at the ICAO in 2013 illustrates how non-statehood becomes a hindrance to accessing international organisations that require statehood for membership:

[t]oday I have signed into law H.R. 1151, an Act concerning participation of Taiwan in the International Civil Aviation Organization (ICAO). The United States fully supports Taiwan's membership in international organisations where statehood is not a requirement for membership and encourages Taiwan's meaningful participation, as appropriate, in organisations where its membership is not possible. My Administration has publicly supported Taiwan's participation at the ICAO and will continue to do so. Consistent with my constitutional authority to conduct foreign affairs, my Administration shall construe the Act to be consistent with the 'One-China' policy of the United States, which remains unchanged, and shall determine the measures best suited to advance the overall goal of Taiwan's participation in the ICAO (The White House, 12 July 2013)

Clearly, the US has to consider the One-China policy before supporting Taiwan's participation in the ICAO. For Taiwan to be represented in the ICAO, both sides of the Taiwan Strait would have to make certain compromises. In fact, a compromise on Taiwan's statehood and the PRC's One-China principle was negotiated in the ICAO. The US supported Taiwan's bid for observer-ship in its negotiations with the PRC, but the PRC opposed the US and instead sought a way to subordinate Taiwan within the context of the policy (*Taipei Times*, 2013). In the US's domestic politics, the issue of how Taiwan can be represented in the ICAO, which honours the TRA and would not overtly breach the US's stance on the One-China policy, became a decidedly political one (Kan and Morrison, 2014).

Meanwhile, Taiwan recommenced its bid to regain observer-ship in the ICAO in 2009 under the Ma administration, despite its lack of member status in the UN, which, as noted, is a requirement for membership in the ICAO (Glaser, 2013). While Taiwan consistently attempted to regain its 'observer' status despite the lack of international recognition regarding its sovereignty, it was ultimately up to Sino-US negotiations to determine how Taiwan's creative representation in the ICAO would be achieved. Regarding Taiwan's limited success in achieving restricted representation in the ICAO, Glaser (2013, pp. 18–19) pointed out that Jean Shen – the director general of Taiwan's Civil Aeronautics Administration – led a delegation at the ICAO triennial meeting as a guest and stated that Taiwan would continue its efforts to obtain observer-ship for 'professional, pragmatic, dignified and meaningful' participation in the ICAO.

The above approach has created significant difficulties for Taiwan. Here, Glaser (2013) highlighted that while Taiwan's guest participation was meant to upgrade Taiwan's Flight Information Region to the ICAO standard, the guest-status was only applicable

as a one-time access pass to the Triennial ICAO assembly. In short, Taiwan's attendance at the technical meeting was not secured, and a new invitation was required for Taiwan to have its guest-status renewed such that it could attend the next assembly in 2016. These conditions are similar to those related to Taiwan's limited participation in the WHO. Meanwhile, the PRC can prevent Taiwan from participating in the ICAO if the DPP comes to power. To achieve some form of limited participation, Taiwan will have to downgrade itself to a local entity in the context of the One-China policy (*Taipei Times*, 2013). The PRC's Taiwan Affairs Office's spokesman stated the following:

[t]he way that the Chinese Taipei CAA was invited by the ICAO council president as his guest reflects our goodwill to our Taiwanese compatriots and our sincerity to maintain the momentum of the peaceful development of cross-strait relations (*Taipei Times*, 2013).

Taiwan is concerned that China's growing influence in global governance vis-à-vis international organisations may affect its international space in negative terms. Since the incumbent secretary general of the ICAO is a PRC national with a background in air safety issues and a prior position in the General Administration of Civil Aviation of China, collective anxiety surfaced in Taiwan regarding the sanctions imposed by the PRC against Taiwan's limited participation in the ICAO (*Focus Taiwan*, 2016b). During the ICAO triennial, after the exclusion of Taiwan as a guest became public, Taiwanese journalists were barred from attending the triennial assembly (*The China Post*, 2016). This development led the Reporters Without Borders organisation to issue the following statement denouncing the ICAO: '[j]ournalists should not be prohibited from covering events because of their nationality or the nationality of the media they are affiliated with' (*The China Post*, 2016). Reportedly, a Paraguayan officer in the ICAO revealed that the

secretary general of the ICAO, Fang Liu, who was from the PRC, warned that if representatives from other countries attempted to speak up for Taiwan, their microphones would be muted (Zhongyang Tongji She, 2016).

5.2.3 Japan's accommodation of Taiwan into international aviation security

Taiwan was prohibited from attending the triennial assembly in the ICAO in 2016, and the nation reportedly expressed its political will to ask Japan and the US for help (China Post 2016). Frank Hsieh, the current de facto ambassador for Taiwan in Tokyo, requested support for Taiwan's participation in the triennial assembly in 2016 from the Japanese general public and from the Government of Japan in an article published in *Asahi Shimbun* (2016). These developments suggest that Taiwan considers Japan to be a critical ally for international aviation security in the ICAO vis-à-vis the PRC's sanctions. Nevertheless, Japan's attempt to enhance Taiwan's international security profile will inevitably conflict with the PRC's aim to downplay Taiwan as a domestic issue in the context of the One-China principle. Japan thus needs to navigate its Sino-Japanese great power management regarding Taiwan and must instrumentalise Taiwan in terms of aviation security in accordance with the primary institutions in international society.

Taiwan does not possess the latest database provided by the ICAO that contains the most up-to-date international aviation security and safety intelligence and information since it is not represented in the organisation. This has put Taiwan's civil aviation in jeopardy due to the discrepancies in aviation safety and security standards and rules between Taiwan and the international community. Crucially, the exclusion of Taiwan conflicts with the current No Country Left Behind campaign proposed by the ICAO. This campaign assists member states of the ICAO, especially those from developing countries, to improve their capacity building and to harmonise international civil

5.2.4 Taiwan as a missing link in international aviation security: its implications for Japan's security

Since Taiwan is a highly developed democratic polity, its lack of access to the latest aviation security and safety database in the ICAO is unquestionably an anomaly. In fact, Taiwan is indispensable to international aviation activity. For example, the nation manages 18,000 nautical miles in its flight information region (FIR) and borders four other FIRs (Kao, 2016). Shen, the former de facto Taiwanese ambassador in Washington, pointed out that Taiwan's FIR is seven times larger than its own territory (YouTube, 2015). In fact, Taiwan's CAA handled a total of more than 1.5 million air traffic control services and 58 million air travellers in 2015 alone (Kao, 2016). From Japan's aviation and economic security perspectives, Taiwan's missing link in international aviation security is detrimental since Japan and Taiwan operate more than 700 direct flights on a weekly basis, which is numerically equal to the number of crossstrait flights (Hsieh, 2016). Crucially, even the Palestinian Liberation Organization was granted observer status in the ICAO in 1977 despite the fact it does not have a single aviation facility (Glaser, 2013).

Clearly, the Japanese government understands that the geopolitical implications of the PRC's sanctions on Taiwan in the ICAO negatively affect Japan's own aviation security. This is due to the risks posed to the air lines of communication between Japan and Taiwan and the sheer scale of the aviation-related economic activities between the two actors. The PRC sanctioned Taiwan's participation in the ICAO when it participated as a guest under the Tsai administration in September 2016 (Nikkei Shimbun, 2016). Here, the PRC shifted the blame to the Tsai administration in the DPP by stating that Taiwan

is merely a province of China and does not have the right to attend the ICAO (Nikkei Shimbun, 2016). Regarding this development, the chief cabinet secretary in the Abe administration, Suga Yoshihide, stated that 'it is desirable for Taiwan to participate in it [the ICAO] in one way or another', adding that Japan and Taiwan operate numerous regular direct flights and that Japan should ensure the safe and stable development of international civil aviation (Nikkei Shimbun, 2016). These comments clearly suggest that the Japanese government considers the absence of Taiwan in the ICAO to be detrimental both to international aviation security and Japan's security. This makes Japan's aviation security engagement with Taiwan even more critical.

5.2.5 Taiwan as an aviation security asset for Japan's security and beyond

It is essential that Taiwan maximises its aviation security ties with Japan and other countries to enhance its own aviation security and safety capabilities and to open up alternative revenues for Taiwan to obtain the most up-to-date aviation intelligence and information. As the Malaysian government's behaviour vis-à-vis the search-and-rescue operation for Malaysian Airlines Flight MH370 (most of the passengers were from the PRC) indicates (Reuters, 2015), updated airspace management and aviation security intelligence capabilities for each ICAO member, including Taiwan and China, is essential for international security. Indeed, this will have positive implications for Japan's aviation security. Japan has legitimate security interests vis-a-vis Taiwan's quasi-participation in the ICAO. Even if Taiwan's lack of statehood makes it difficult for it to engage in a high-profile security relationship with Japan, this does not mean that Taiwan cannot contribute to Japan's security or to international security, including in the East China Sea. In fact, one of the main contributions Taiwan can make to regional security relates to the field of intelligence. Non-traditional, transnational, and multi-dimensional aviation security do not depend on the concept of sovereignty, but

do depend on international cooperation to mitigate any issues. It is here that the exclusion of Taiwan from international aviation security cooperation and networking simply does not make sense. The Aviation Intelligence Report of 2015 stated the following:

[i]n short, tackling global issues will require global cooperation and the ability to move faster than the current regulatory process. If we are to be effective in combating these new kinds of threats, sovereignty will need to be left at the door and a new level of collaboration between countries, agencies and all industry stakeholders will be essential (Brooks, 2015, p. 4)

This quote applies to Taiwan, which is marginalised from international security efforts despite the strong capability, affluent resources, know-how, and technical knowledge it possesses. We can look at what ensued with Taiwan's HADR cooperation vis-a-vis Malaysian Airlines Flight 370 to understand the consequences of excluding Taiwan from international aviation security efforts. According to Glaser (2015), while Taiwan's air force and navy were immediately dispatched to Malaysia following the tragic disappearance of the civilian aircraft in 2014, the PRC pressured the Malaysian authorities to send them back. Thus, Taiwan's aerial and naval HADR missions were eventually withdrawn (Glaser, 2015, p.31).

As a potential contribution by Taiwan's intelligence to international aviation security in the ICAO, Kuang-shih Yeh, the minister of transportation and communication, pointed out that Taiwan was the first polity that brought a communications, navigation, and surveillance air traffic management system into service in 2011. In fact, the ICAO had proposed such a globally coordinated system of air navigation in the 1980s (Yeh ROC MOFA, 2013).

5.3 Japan's Instrumentalisation of Taiwan as an Aviation Security Asset in relation to the Sino-Japanese Balance of Power

Japan's instrumentalisation of Taiwan as an aviation security asset generates an alternative narrative regarding Taiwan in face of the conventional narrative that Taiwan is a geopolitical source of tension or pawn in great power politics. The latter narrative was manifested in 'the abandonment of Taiwan theory' in Sino-US relations. Here, Fan (VOA Chinese, 2016) argued that if Taiwan has no value for the US, it is possible that it would abandon Taiwan, as was the case in the 1970s. Therefore, Taiwan must emphasise its shared democratic values and strategic importance in handling the US's relations with the PRC (Fan VOA Chinese, 2016). The Japan–Taiwan aviation security cooperation will help to improve the geostrategic value of Taiwan in the eyes of the US in aviation security terms.

If bilateral Japan–Taiwan relations help improve international aviation security, Taiwan will become an aviation security asset for international security as opposed to merely being the subject of the 'Taiwan issue' in Sino-US and Sino-Japanese tensions, despite its status as a de facto state. Furthermore, Taiwan can diversify its strategic narratives for Japan's security beyond the excessively emphasised narrative of Taiwan's geographical importance for its neighbour's defence of its sea lines of communication. Furthermore, greater aviation security and safety cooperation between Japan and Taiwan would help to mitigate Japan's concerns regarding Taiwan's strategic alignment with the PRC on regional aviation security issues, especially in terms of the East China

Sea. Japan can instrumentalise Taiwan as an aviation security asset vis-à-vis the PRC to affect the regional balance of power – even though Taiwan is a non-state actor – in accordance with the primary institutions in international society related to the ESIR.

5.3.1 Risks posed by Taiwan's inability to receive aviation intelligence and information: Implications for Japan

Regarding the exclusion of Taiwan from the ICAO's triennial assembly, the de facto incumbent Taiwanese ambassador in Washington, Stanley Kao, stated the following:

[a]s aviation safety knows no boundaries and requires close cooperation of countries around the world, keeping Taiwan out for political reasons does not serve ICAO's mission, and has left the global aviation security system incomplete' (*Washington Times*, 2016b).

Taiwan has been obtaining ICAO data through informal diplomacy by engaging with ICAO member states, but this has proven to be costly in terms of both budget and time. According to Julie Yu Wen Chen (2013), Taiwan relied on receiving ICAO's air security intelligence and information from the US's FAA. However, due to certain US aviation security policy changes, this intelligence and information may have been Americanised and thus different from what the ICAO was expecting the member states to provide (Julie, 2013). In some cases, such a delay and disadvantage in obtaining aviation data jeopardised the aviation security in Taiwan's airspace and airport-related facilities despite the criticality of Taiwan's airspace. Taiwan was often not even aware of the technical and regional meetings conducted by the ICAO due to its lack of ICAO membership (ROC MOFA, 2011). For example, according to the ROC's MOFA, in 2006, the new security rules imposed by the ICAO regarding liquids, aerosols, and gels in cabin luggage was not relayed to Taiwan in a timely manner. In fact, the rules were implemented in 2007 without Taiwanese authorities being informed. Taiwan was not

even aware of this policy change until its aviation authority was asked for its comments on the matter, which led to massive confusion in Taiwan's aviation security and safety (ROC MOFA, 2011). The issue of incompatibility was also discussed (ROC MOFA, 2011). Given that Taiwan depended on the US's FAA for aviation safety information (Chen, 2013), and given that this information was often Americanised, it was generally not aligned with the ICAO's security policy. This, in turn, created the issue of incompatibility in the audit system (which was reported by the US's FAA). In short, it was difficult for Taiwan's FIR to adjust to the ICAO's standards (ROC MOFA, 2011). Furthermore, most of the ICAO member states underwent a universal safety program to improve their audit systems but, given Taiwan's exclusion, the credibility of its audit system was frequently questioned (ROC MOFA, 2011). In addition, Taiwan's information and intelligence sharing with ICAO member states was severely limited due to its non-representation (ROC MOFA, 2011).

The issue of the compliance of Taiwan's aviation rules and regulations with international standards and norms has direct implications for international civil aviation security and safety. For example, the fatigue issue among pilots is significant for Taiwan's aviation safety, as was manifested in Trans Asia's ground collision with urbanised areas in 2015 (*Aerotime News*, 2016). In a more recent case, China Airline, the largest airline in Taiwan, experienced its first major strike by its pilots, which was partly due to the fatigue issue (*Japan Times*, 2016). The strike affected air transport routes such as those between Taipei and Fukuoka. Clearly then, the fatigue issue in Taiwan's civil and commercial airlines that fly over the airspace between Taiwan and Japan must be addressed in compliance with the international standards and rules of the ICAO. This may have direct implications for aviation security and safety, as well as for the economic security of Japan and Taiwan. From this perspective, Taiwan must be

accommodated into international civil aviation security and safety networks by Japan.

5.3.2 Japan's socialisation of Taiwan through international aviation security norms for Taiwan's as if membership

Nevertheless, as stated earlier, for the PRC and the ICAO, Taiwan is considered neither a state nor an international organisation, and the PRC is extremely sensitive to any political gestures that may imply Taiwan's sovereignty. In fact, the PRC opposes any foreign intervention in this matter, and it was incensed by the US's overt ties with Taiwan. While Japan's signing of an Open Skies agreement with Taiwan was considered provocative from the standpoint of Taiwan's legal status and led to some discomfort in the PRC, Japan's low-key, less visible yet significant approach to upgrading Taiwan's air safety and security is certainly a creative one. Furthermore, even if the role Japan plays in cross-strait relations is low-key in comparison with the role played by the US due to the absence of a Japanese version of the TRA, Japan still provides Taiwan with a quasi-platform to enable the latter to behave as if it is a semi-legitimate aviation security actor, thus contributing to international aviation security. As such, Taiwan probably better meets the required criteria for potential future membership in the ICAO. Of course, while whether Taiwan can arrange with the PRC to obtain representation in the ICAO is not a matter for Japan to decide, Japan can still provide alternative representation options to Taiwan.

5.3.3 Japan's utilisation of the ICAO's regulations to improve Taiwan's aviation security contributions

The ICAO lacks a formal international aviation dispute resolution mechanism to deal with high-profile contentious aviation security issues, such as the Air Defence Identification Zone (ADIZ). Meanwhile, the ICAO is still the representative authority

on normative developments in international civil aviation security. The Chicago Convention regulates international civil aviation rules and standards, and the ICAO provides the most authoritative data available on civil aviation issues, including security and safety issues, to signatory member states. Taiwan's reincorporation into the ICAO would be a symbolic recognition of the nation's international aviation security contributions. In fact, Japan has used ICAO recognition to evaluate the strategic reliability of other state aviation actors. For example, only after the ICAO acknowledged that the Philippines had cleared certain aviation security and safety standard issues did Japan lift restrictions on international flights to the Philippines and sign an Open Skies agreement with it in 2013 (Rappler, 2013).

The ICAO also does not have a mechanism to carry out legal punishments against non-normative aviation actors. For example, even after the coercive ADIZ was unilaterally established by the PRC to regulate the airspace in the East China Sea in 2013, the legal clarification of the controversial ADIZ remains ambiguous, as will be examined in the ADIZ section. Nonetheless, the ICAO does have the normative authority to create international norms for international aviation security and safety, which defines certain aviation behaviours as unacceptable in accordance with international norms on aviation security. For example, Parameswaran (The Diplomat, 2015) pointed out that Thailand was downgraded to a category 2 nation by the ICAO in 2015. This means that Thailand was red flagged for deficiencies in several areas following an audit. As will be examined in the ADIZ section, if certain aviation behaviours are regarded as more normatively progressive and can be emulated within international society, then more peaceful and amicable Japan—Taiwan aviation security relations will be regarded as more normative in the ICAO in relation to the PRC's coercive aviation behaviours.

Japan may not have the authority to influence Taiwan's representation in the ICAO, but the sheer scale of Japan–Taiwan aviation relations requires Taiwan to be an aviation security asset. This helps contribute to peace and stability in terms of regional aviation security. It is within this context that Japan's engagement with Taiwan plays an incremental and low-key yet significant, substantive, and symbolic role in ensuring Taiwan attains 'as if' membership in international society, including in terms of aviation security and safety in the ICAO. Much like Japan used the UN's mandate as an avenue to overcome domestic constraints and incrementally enhance its international security in the post-Cold War era (Singh, 2013d), the nation has been utilising various bilateral political measures and concluding aviation security relevant agreements with Taiwan to afford its neighbour opportunities to behave like a semi-legitimate non-state actor in aviation security in accordance with the ICAO's standards.

5.4 The implications of Japan's enhancement of aviation security ties with Taiwan in the context of the ICAO

5.4.1 Japan's civil aviation security strategy and Taiwan

The development of civil aviation is essential for Japan's national security strategy. Regarding the centrality of the aviation development in Japan, which includes the Japanese commercial aviation that occupies the centrality of the third arrow in Abenomics, Cronin (2013) argued that Japanese civil aviation is a great indicator of whether the third arrow will be successful. Within the context of Sino-Japanese relations, Japan is currently competing with China to develop its own commercial jets, and will attempt to internationally promote them as an alternative civil aviation option for strategic use in view of the established Western commercial jets. This indicates the civil aviation's cruciality for Japan's global strategic and economic agendas (Forbes, 2015). Meanwhile, amid the rising tensions in the East and South China Seas, the

strategic importance of aviation is becoming ever more critical. These developments in Japan's aviation mean aviation security is becoming essential to Japan's overall security. Here, Japan cannot afford to have Taiwan as a missing link when shaping the emerging regional aviation security environment for its own security vis-à-vis the PRC.

The US is directly responsible for enhancing the international space of Taiwan based on the TRA (American Institute in Taiwan, 1979). The Government of Japan did not issue an explicit official statement regarding Taiwan similar to Obama's 2013 bill on Taiwan's ICAO participation. This indicates that Japan prefers a low-key approach in contrast to the US's TRA, given the lack of a solid domestic legal and institutional framework defining Japan—Taiwan relations. Therefore, Japan's usual trade-off with the US in terms of the opportunity to play a high-profile role in diplomatic leadership in return for obtaining maximum national security for Japan can be implicitly observed in the form of Japan's indirect support to Taiwan for expanding Taiwan's international security profile. This support was manifested in terms of Taiwan's de facto greater representation in the area of international aviation security.

Japan has utilised Taiwan's potential civil aviation security capabilities by concluding a number of high-profile agreements, if not state-centric treaties, with Taiwan, such as an Open Skies agreement. Japan has helped Taiwan become part of the regional Open Skies regime in East Asia, ensuring Taiwan is an alternative independent aviation security asset to the PRC. Finally, Japan has incorporated Taiwan within anti-global terror and anti-transnational organised crime networks.

5.4.2 Japan-Taiwan aviation-related law enforcement cooperation and its geopolitical implications

As noted, Taiwan has been excluded from international organisations that require

statehood for membership. Following the PRC's sanctions against its guest participation in the ICAO, Taiwan has expressed even greater strategic interest in participating in other international security agencies such as Interpol (*Taipei Times*, 2016). Interestingly, while Taiwan lost its official seat in the UN in 1971, it remained a member of Interpol until 1984 (Glaser, YouTube, 2015). This indicates the importance of Taiwan's participation in international policing efforts. In fact, it is not unfeasible that excluding Taiwan will make it a potential loophole for international terrorist and transnational crime organisations.

From this perspective, Japan's civil aviation security cooperation with Taiwanincluding in terms of international anti-terrorism and anti-transnational crime, airport security, and border and immigration security – benefits Japan's own security situation. Crucially, Japan and Taiwan concluded three bilateral law enforcement agreements that are essential for deterring anti-international terrorism and anti-transnational organised crime. The first was an agreement related to anti-terrorism financing through money laundering, while the second was an agreement over intelligence sharing for border and immigration security (Interchange Association Archive, 2014) and the third was a customs enforcement agreement for combating smuggling and enhancing the clearance system via information sharing and expertise (Taiwan Today, 2017), all of which serve to enhance Japan's international security. Regarding the importance of intelligence and information sharing for regional law enforcement efforts, including in financial terms, in recent years, Taiwan's neighbours with sizeable Islamic populations in Southeast Asia have been receiving unfettered access to financial support from Wahhabi foundations in Saudi Arabia to promote a less tolerant interpretation of Islam (Nikkei Asian Review, 2016). Furthermore, according to Lai L Chung, in 2006, North Korean ships frequented the port in Kaohsiung in Southern Taiwan largely because Taiwan is

an essential sea line of communication for North Korean shipping. Here, Chung recommended that Taiwan import stronger military warning radar intelligence capabilities from the US to initiate air defence intelligence sharing with Japan (*Taipei Times*, 2006). In fact, using the US's secret intelligence, Taiwan detained a North Korean ship and confiscated 150 barrels of chemical rocket fuel in 2003 (Glaser, YouTube, 2015). This suggests that Taiwan's security cooperation with other countries, including Japan, is not only essential for bilateral Japan–Taiwan law enforcement cooperation but that it also has huge geopolitical implications for international and regional strategic-security issues such as the non-proliferation of WMDs in the Korean Peninsula.

5.4.3 Taiwan's exclusion: its negative implications for aviation security, and Taiwan's potential role in aviation security for Japan

The former UN secretary general, Ban Ki-Moon (2016), stated in his ICAO address that no member state can be left behind as a specialised agency in the UN for the sake of global aviation safety and security. Nonetheless, as discussed earlier, Taiwan had been excluded from the ICAO until it gained inclusion in late 2013. However, the nation was again excluded from the ICAO's triennial in 2016 despite the fact that its airspace presented a highly critical flight route for civil aviation and its lack of representation in the ICAO may have put the global and regional civil aviation field at risk due to the lack of up-to-date aviation safety and security intelligence and information provided by the ICAO. Regarding the re-exclusion of Taiwan from the ICAO and its negative security implications for international security, Nigel Evans – a member of the UK's House of Commons and co-chair of the British–Taiwanese All-Party Parliamentary Group, who was a part of the visiting British delegation in Taiwan – asked what the big difference was between 2013 and 2016 (Focus Taiwan, 2016). The answer was that 'far

more aircraft and passengers' are flying in Taiwan's airspace, meaning 'it is more important today than it was three years ago that Taiwan participates in this aviation conference' (*Focus Taiwan*, 2016).

In addition, as was discussed in the literature review, Taiwan is essential for a stable balance of power and the enhancement of the legal order regarding regional maritime security issues in the East China Sea and beyond. The same applies to the defence of airspace. For example, Taiwan's Ministry of Defence revealed that a powerful early warning radar that was newly deployed by the military in Hsinchu county in Taiwan in 2012 detected and monitored missiles launched by North Korea (*Focus Taiwan*, 2012). Here, a ministry official stated how '[o]ur long-range early warning radar system detected the North Korean rocket flying over waters some 200 kilometers east of Taiwan' and added that 'the first and second stages of the rocket crashed into the waters off South Korea and the Philippines, respectively'. Furthermore, the *Taipei Times* described Taiwan's powerful air defence radar as 'the most powerful EWR installation on the face of the planet' one that can 'simultaneously track as many as 1,000 airborne targets the size of a golf ball within a range of 3,000km' (*Taipei Times*, 2012).

At least one former de facto Japanese ambassador and one former director in Japan-Taiwan Exchange Association have acknowledged the criticality of instrumentalising Taiwan's intelligence capability for Japan's security. For his part, the former de facto ambassador of Japan in Taiwan, Uchida (2006), stated that Japan and Taiwan need to have more consolidated security dialogue mechanisms and pointed out that inadequate information and intelligence sharing between the two actors regarding military aviation may increase the risk of air incidents. Here, Uchida (2006) revealed that once Taiwan falls into the hands of the PRC, missiles initially directed at Taiwan from mainland

China may be easily re-directed at Japan. Meanwhile, in 2013, a former director at the Japan-Taiwan Exchange Association, Ogata, who has a high-profile military background as the former director of the J1 Joint Staff Office MODA, suggested that, in addition to cooperating for non-military maritime security and the HADR, Japan and Taiwan should cooperate to implement guidelines for an aviation contingency in the geographically connected airspaces between the two nations (Ogata, 2013).

5.4.4 Japan's position on Taiwan's restricted international space

As discussed in the CSD chapter, if Japan's security ties with Taiwan are considered humanitarian in nature, as is the case with the HADR, it will be normatively difficult for the PRC to sanction them. According to Matsuda (2012), Japan would not accept the One-China policy in the PRC's favour if a global security issue stemming from Taiwan's exclusion was considered a humanitarian issue or had the potential to negatively affect Japan's public health issues. Japan will navigate the ambiguities of the regional great power management with the PRC if the security issue in question is considered humanitarian by international society within the context of Japan's One-China policy.

Taiwan was excluded from WHO until the Ma administration came to power. Nonetheless, Japan, under the Koizumi administration, explicitly supported Taiwan's representation in WHO in ways that would satisfy all the stakeholders (Matsuda, 2012). In fact, even the PRC was losing the moral high ground in resisting Taiwan's participation in WHO for humanitarian reasons. Domestically, the lack of access to essential medical intelligence and information in Taiwan, which were necessary to effectively minimise the outbreak of SARS in the nation, led to outrage among the Taiwanese population (Shen, YouTube, 2015). Furthermore, according to Shen, the former de facto Taiwanese ambassador in Washington, since Taiwan was the second

largest state affected by the outbreak of SARS, despite the fact that the virus was first transmitted from the PRC to Taiwan, international society was convinced to accommodate Taiwan (Shen, YouTube 2015). Along with the US and the EU, Japan supported Taiwan diplomatically in a high-profile manner. Meanwhile, while the PRC attempted to subordinate Taiwan into the One-China policy framework, much like in the case of the ICAO, the PRC's attempt to encourage WHO to reject Taiwan was a failure (Matsuda, 2012). Consequently, Taiwan was granted observer status in the World Health Agency, a major organ of WHO.

5.4.5 Japan's low-key stance vis-à-vis Taiwan in 2013 and high-profile stance vis-à-vis Taiwan in 2016 in the ICAO

Unlike the US and the EU, Japan did not take a strong stance regarding Taiwan's guest representation in the ICAO in 2013. More than 80 MEPs from the EU called upon Raymond Benjamin, the former secretary general of the ICAO. Among them was Charles Tannock – a MEP from the UK and chairman of the EP–Taiwan Friendship Group – who stated that '[t]he Taipei FIR plays a critical role in this area', adding that '[i]t is clear that it would be of significant benefit, both to Taiwan and to the ICAO, to allow Taiwan to participate in the organisation as an observer' (European Business Review, 2013). Meanwhile, as was discussed earlier, Obama signed a bill explicitly supporting Taiwan's representation in the ICAO.

Interestingly, unlike in 2013, Japan decided to take a stronger stance against Taiwan's ostracisation by the PRC regarding the ICAO in 2016. Suga Yoshihide, a part of the Abe administration, expressed some displeasure about the ostracisation (Nikkei Shimbun, 2016). Furthermore, in an official statement, the ROC's MOFA (2011) referenced the Japanese government's public support for Taiwan's aspiration to be represented in the ICAO. While the reason behind the change of stance by the Japanese

government requires further research, one interesting development is that the grassroots support for Taiwan in Japan has become ever more consolidated in recent years.

For example, around 130 pro-Taiwan councillors from across Japan and from delegations in Taiwan cities (headed by Hsieh) gathered in Wakayama City and created the so-called Wakayama Statement. According to the statement, since the number of Japan–Taiwan annual personnel exchanges had already reached six million, the delegation supported Taiwan's participation in the ICAO and other international organisations (*Taipei Times*, 2016c). Furthermore, Japan's public reacted to Taiwan's exclusion from the ICAO, with *Yomiuri Shimbun*'s (2016) editorial arguing that instead of intimidating Taiwan, China should enhance its economic activities and conduct dialogues with it. Even the liberal-leaning *Asahi Shimbun*'s (2016) editorial argued that the pressure China had put on the ICAO to exclude Taiwan had put aviation security at risk.

5.5 The Bilateral Japan-Taiwan Open Skies Agreement

5.5.1 What is the Open Skies agreement?

The EU has been attempting to harmonise safety regulatory standards by implementing consistent safety and security rules for different aircraft and operations based on potential risk assessments in accordance with the ICAO. Here, the union has effectively created one sky over the airspace of Europe, which has become known as a 'cabotage' (Shimoi, 2010). The EU's 'open skies' model inspired ASEAN states in the 1990s to create an integrated aviation market for aviation security and safety (European Commission Press Release Database, 2014).

The bilateral Japan-Taiwan Open Skies agreement symbolically accommodated Taiwan as an independent aviation security asset. Crucially, the agreement

institutionalised the deregulation of the amount of air traffic between Japan and Taiwan and allowed an unlimited number of scheduled flights to be operated between the two nations (*China Post*, 2011). Tokyo was initially exempted from these effects but was soon also subjected to a greater number of flights, as is examined below. Furthermore, the agreement gave both sides so-called 'beyond rights', allowing both Japanese and Taiwanese airlines to transport passengers to a second country from Japan or Taiwan and from that country to a third country, and so on (*China Post*, 2011).

Meanwhile, Shimoi (2010) argued for an East Asian regional open skies regime given that the East Asian region is the fastest growing commercial aviation market, faster even than the EU or North American markets. The bilateral Japan–Taiwan Open Skies agreement followed this global trend. The agreement itself aimed for greater commercial aviation activities between Japan and Taiwan. The fact that Japan negotiated and concluded such an agreement with Taiwan itself is a de facto recognition of Taiwan as an important regional aviation entity in the East China Sea, one that is independent from the PRC in the regional aviation security environment.

5.5.2 The Open Skies agreement's geopolitical implication from a historical perspective

The geopolitical implications of the Open Skies agreement become clear through an examination of the history of Japan–US aviation relations. Following the surrender of Imperial Japan in 1945, the GHQ ordered a ban on Japanese civil aviation. However, after the signing of the San Francisco Peace Treaty, the ban was lifted (Cronin, 2013). Subsequently, the US and Japan signed an agreement in 1952 that granted them carrier rights to increase or decrease their weekly number of flights without prior notification. This was, in essence, the beginning of the historical process for the Japan–US Open Skies agreement concluded in 2010 (Cronin, 2013). The US considered a ban on Japan's

civil aviation to be strategically beneficial to its interests in that it wanted to ensure that Japan would not revive its air force after WWII and become a threat again. However, with the intensification of the Cold War and the breakout of war in the Korean Peninsula, a political need emerged to revive Japan's aviation sector and airspace for geo-strategic reasons. Cronin (2013) pointed out that the treaty on aviation signed with Japan in 1952 during the Korean War granted US carriers unlimited beyond rights to utilise Japan's airspace and airports as a hub to other destinations in the rest of Asia.

In the case of the UK, regarding the strategic, sovereign, and national security implications of the Open Skies agreement, Dodds and Adey (2014) stated the following:

[w]e might be reminded of their historical evolution in this context, as Paul K. Saint Amour's alerts us to a 1955 summit held in Geneva when President Eisenhower proposed an open skies agreement with the Soviet Union, France and Britain, that would permit aerial reconnaissance into each other's territory for 'photographic reassurance' of each country's weapons arsenal. The Soviet's refusal would lead to the deployment of high-altitude spy planes like the U2.

5.5.3 Japan-Taiwan Open Skies agreement: Geopolitical Implications for Taiwan's airspace

The fact that Taiwan signed the Open Skies agreement with Japan is geopolitically critical. Following Japan's termination of the diplomatic recognition of the ROC, the most salient issue the country faced in terms of maintaining unofficial ties with Taiwan related to bilateral civil aviation (Kiyomizu, 2009). Japan signed a bilateral Open Skies agreement with Taiwan that is commercial in nature yet requires Japan to exercise significant political decisions. This indicates that Japan recognised Taiwan's de facto independent airspace as separate from the PRC since it was Taipei that Japan negotiated

the agreement with and not Beijing. According to Matsuda (YouTube, 2014), Japan's conclusion of the agreement with Taiwan certainly increased the PRC's ire. Nevertheless, the Open Skies agreement is cited as evidence of peaceful bilateral Japan—Taiwan relations in Taiwan's official online archive of the bilateral Japan—Taiwan fisheries agreement (ROC MOFA, 2013, p. 2). The Open Skies agreement symbolises the mutual political trust that binds Japan and Taiwan while the former navigates the ambiguities of the One-China policy in relation to the regional great power management in Sino-Japanese relations.

The UK and Russia instrumentalised the Open Skies agreement for a 'confidence-building measure'. This allowed the two sides to conduct de facto unarmed intelligence, surveillance, and reconnaissance activities using civilian aircraft within mutual airspace and aviation facilities, including aerial military facilities (UK Defence Journal, 2016). Following the conclusion of their bilateral Open Skies agreement, Japan and Taiwan were able to enhance their aviation security and safety-related procedures in equal measure. Defining it as one component of their commercial naval aviation activities, Japan and Taiwan also signed an agreement for naval aviation search-and-rescue cooperation (*Focus Taiwan*, 2013). This had positive effects in the East China Sea in terms of non-traditional security and safety. According to the Taipei Economic and Cultural Representative Office (2013) in the US, the function of this agreement is as follows:

[u]nder the agreement on sea search-and-rescue, the Taipei Mission Control Center and Tokyo Rescue Control Center will closely cooperate and coordinate their actions in aviation accidents, and arrange visits, training sessions and other activities on a regular basis to enhance capabilities.

5.5.4 Economic balance of power and Japan's accommodation of Taiwan into the regional commercial aviation regime

Unlike with the PRC, which from time to time boycotts Japanese products, stops inbound Chinese tourists from entering Japan, or even conducts violent anti-Japanese demonstrations, Japan's tourist ties with Taiwan are far less influenced by geopolitical risks; rather, they serve as an economic security instrument vis-à-vis the PRC's economic coercion. Taiwan has long considered Japan to be an economic balancer as Taiwan's economic reliance on mainland China continued to expand (Ogasawara, 2014). If economic security is a component of the balance of power theory in the ESIR, Japan, as the second largest trading partner of Taiwan, is essential for maintaining the crossstrait balance of power in Taiwan's favour. Furthermore, the fact that Japan and Taiwan signed an Open Skies agreement to streamline and expand civilian flights without governmental intervention can not only be understood within the context of greater economic security and commercial benefits but can also be interpreted as a consolidation of Taiwan's de facto airspace independence. The PRC cannot interfere with civilian flight routes between Japan and Taiwan since neither Japanese nor Taiwanese airlines are affected by governmental intervention. Thus, thanks to Japan's political and economic leadership, Taiwan is being incorporated into a growing regional aviation regime.

For Taiwan, the diversification of economic partners is a major component in the issue of the Taiwan Strait vis-à-vis Taiwan's survival in the face of the PRC's economic coercion. The question of how Taiwan can enhance this in creative terms in relation to Japan is an important component of the emerging regional security environment in East Asia, including in terms of the cross-strait issue. A Taiwan that is economically less reliant on the PRC may find it easier to strategically contribute to Japan's security in

the Japan-China-Taiwan power politics relationship. Crucially, economic power is an important component of the balance of power in primary institutions in international society according to ESIR theory, which means the Japan-Taiwan Open Skies agreement can help to tilt the Sino-Japanese and Sino-Taiwanese balance of power in Japan's favour.

5.5.5 The Open Skies agreement: Increased Japan-Taiwan tourism

Greater Japan-Taiwan bilateral civil aviation cooperation has been a critical factor in increasing the number of Taiwanese tourists visiting Japan after the triple disasters in the Tohoku Region in 2011, which has been achieved by institutionalising a greater number of airlines to connect Japan to Taiwan at various regional airports via the Open Skies agreement (Ikegami, 2015). For example, Haneda and Songshan airports commenced regular international flights following the opening of a new terminal at the former (Ikegami, 2015). According to Nikkei Business, the Open Skies agreement encouraged a greater number of low-cost carrier operators to fly between Taiwan and Naha. From March 2013 to March 2014, a record 627,000 international inbound tourists visited Okinawa, a 64% increase on the previous year. Meanwhile, Taiwan witnessed slightly more than 40% in international inbound tourist traffic and accounted for the largest nationality group of foreign tourists in Okinawa (Nikkei Business, 2014). Ikegami (2015) noted that the Open Skies agreement between Japan and Taiwan made it possible for local airports in both nations to open new flight routes between each nation, which served to remove restrictions on airlines that operate regular flights, to abolish restrictions on local destination airports besides Haneda and Narita, to start an unlimited number of flights, to remove restrictions on charter flights, non-regular flights, and cargo flights, and allowed free openings of new flights between Japan and Taiwan

in the local airports of each nation.

5.5.6 The PRC's instrumentalisation of economic coercion against Japan, Taiwan, and others

According to Guo (Taipei Times, 2016), China has been instrumentalising economic measures to coerce other states in recent years, most notably Japan, the Philippines, and Taiwan. When rare earth metal exports to Japan were suspended after a Chinese fishing vessel that had rammed into a Japanese coastguard vessel was detained in 2010, Japan suffered due to its over-reliance on the PRC for the supply of such metals, which are essential to the nation's high-tech industry (Hiranuma, 2016). According to Guo (*Taipei Times*, 2016), this coercive use of economic measures often backfires, in this case potentially encouraging the PRC to stop using them. Japan cooperated with the US and the EU to bring a lawsuit in the WTO against the PRC, and the PRC defence was lost (*Taipei Times*, 2016).

Furthermore, the PRC instrumentalised a reduction in the number of Chinese inbound tourists visiting Taiwan. For example, Guo pointed out, to force the Tsai administration to accept the 92 consensus or to admit that both sides of the Taiwan Strait belong to 'One China', the PRC restricted the number of inbound Chinese tourists visiting the nation (*Taipei Times*, 2016). A number of travel agencies in the PRC began to promote Chinese inbound tourism in terms of eight cities and counties, where voters chose the KMT camps as the best option for weeklong holidays for the Chinese (*Taipei Times*, 2016a). Meanwhile, the PRC authorities expressed their support for establishing a liaison centre in China to promote tourism in these eight cities and counties in Taiwan as well as various areas in China (*Taipei Times*, 2016a). In comparison with August 2015, the number of inbound Chinese tourists visiting Taiwan decreased by 30% (*Yomiuri Shimbun*, 2016).

Regarding this development, Taiwan's Mainland Affairs Council issued the following statement through its vice chairman, Chiu Chui-cheng: '[t]he government's policy of welcoming Chinese tourists remains unchanged, but we do not welcome (Beijing's) political arrangement of Chinese tourist schedules in Taiwan' (*Focus Taiwan*, 2016). President Tsai expressed her concerns over the over-reliance on China for Taiwan's economic security and expressed her willingness to explore the economic opportunities offered by the ASEAN and South Asian regions in partnership with Japan in her interviews with the *Yomiuri Shimbun* (2016). For example, China Airlines, Taiwan's largest airline, announced it would add another direct flight to Palau to boost Taiwanese tourism there. In fact, since 2017, Palau has been going through a crisis due to China's travel ban, which was implemented after Palau expressed its diplomatic recognition of Taiwan (Travel and Tour World, 2019). This development in Taiwan–Palau aviation ties can potentially contribute to Japan's Indo–Pacific strategy in the South Pacific if Taiwan can provide Palau with a sustainable supply of tourists.

5.5.7 The Japan-Taiwan Open Skies agreement: sustainable Japan-Taiwan tourism and commercial security vis-à-vis the PRC's economic coercion

Stable commercial ties with diverse economic partners will make Taiwan less reliant on the PRC. For example, the DPP administration led by Tsai Ing-Wen saw the decreasing number of Chinese inbound tourists visiting Taiwan as a form of economic coercion, one that was instrumentalised by the PRC. However, this was countered by increasing the number of visitors from Japan and South Korea (*Taipei Times*, 2016f; Dong Wang, 2016). As the aforementioned open skies model of the EU indicates, the Open Skies agreement between Japan and Taiwan aims to harmonise the bilateral aviation system, rules, and norms between the two actors, but it also requires Taiwan to implement more reforms in its aviation sector and airspace management in view of greater profit-making

in accordance with the international aviation norms of the ICAO. For example, the TransAsia airline in Taiwan opened new flight routes between Taiwan and Narita Airport following the bilateral Japan–Taiwan Open Skies agreement (Naritayakuwarigenjo, 2013). From January to June 2016, the number of Taiwanese tourists to Japan amounted to more than 2.15 million, the third largest inbound tourist group after Chinese and Korean tourists, with the rise of low-cost carriers between Japan and Taiwan the main factor behind the increase (*Focus Taiwan*, 2016).

The ROC's Ministry of Transportation and Communications also recognised the strategic importance of instrumentalising Japanese inbound tourism as a countermeasure to the PRC's use of tourism as a method of economic coercion. After the Tsai administration came to power in 2016, the ministry announced its intention to increase the number of Japanese inbound tourists to two million, stating that it would give out free tickets to one Japanese passenger from a Japanese tourist group that consisted of four people flying in from big cities in Japan (*Focus Taiwan*, 2016a). Japan and Taiwan can largely contribute to each other's sustainable economic development through reciprocal tourism and the bridging of regional aviation networks via the Open Skies agreement to mitigate the effects of PRC's coercive instrumentalisation of Chinese tourists.

5.6 The PRC's Non-Normative use of the ADIZ

This section first examines what the air defence identification zone (ADIZ) is and how the PRC's ADIZ issue affects Japan's aviation security in the regional security environment. In addition, it also examines how the PRC's ADIZ compares in relation to the normative use of the ADIZ in accordance with international law and norms. It then examines the PRC's recent hostile military aviation behaviours before it scrutinises

the role of Taiwan as an aviation security asset for Japan vis-à-vis the PRC's ADIZ issue in the regional aviation security environment. It finally elucidates how Japan instrumentalises Taiwan as an aviation security asset vis-à-vis the PRC's ADIZ to enhance Japan's security.

5.6.1 What is the ADIZ for?

Regarding the function of the ADIZ and its historical development, Welch (2013) stated the following:

[a]n ADIZ is a publicly defined area extending beyond national territory in which unidentified aircraft are liable to be interrogated and, if necessary, intercepted for identification before they cross into sovereign airspace. The concept is a product of the Cold War: in the 1950s, the United States declared the world's first ADIZs to reduce the risk of a surprise attack from the Soviet Union.

It would appear that the PRC is not an exception to utilising the ADIZ to achieve its own security in terms of the defence of its airspace. Regarding the normative functions of the ADIZ, Welch (2013) pointed out that if utilised properly, an ADIZ functions to reduce mid-air collisions, combat illicit narcotic trafficking, assist in search-and-rescue missions, and reduce the need for sorties in fighter jets for visual inspections. The last function is the most critical in increasing the transparency for stable aerial interceptions and improving the certainty in air defence.

As such, the ADIZ's normative function is to enhance the security of coastal states and protect their airspace against suspicious aircraft. If used properly, this would not present a geopolitical risk that would increase regional aviation security tensions. The normative practice of the ADIZ is conducted with the necessary aviation

communication tools, which include radio frequencies and transponders, and by engaging in communication with incoming aircraft to ensure aviation and airspace safety (Rinehart and Elias, 2015). Such communication applies to any aircraft that intends to fly over or through the national airspace of one state (but not to civilian aircrafts), as opposed to the state aircrafts that only intend to fly over or through the international airspace over one state's exclusive economic zone. These practices are shared by the US, Japan, South Korea, the UK, Australia, and various other nations (Welch, 2013). Furthermore, the Chicago Convention does not illegalise ADIZs, which allowed the PRC to claim that its ADIZ is not a violation of international laws and norms.

In the following subsections, the question of how the PRC's ADIZ has increased tensions over the airspace in the East China Sea is addressed. While the Chicago Convention does not have an explicit clause prohibiting the use of an ADIZ (Rinehart and Elias, 2015), the PRC's unilateral ADIZ has detrimental effects on international norms and regional aviation security.

5.6.2 Japan's defence minister's view of the PRC's non-normative ADIZ

Regarding the PRC's use of the ADIZ, the former defence minister, Tomomi Inada, stated at the CSIS in Washington that 'in November 2013, China announced the East China Sea Air Defence Identification Zone or ADIZ', adding that 'this ADIZ was declared over the large swath of international airspace over the East China Sea'. The Chinese announcement was also made in a way that suggested the airspace over the Senkaku Islands was a part of China's territorial airspace. According to China's announcement, any aircraft flying in China's ADIZ must abide by the rules imposed by China's defence ministry. If an aircraft did not follow China's instruction, Chinese armed forces would respond by taking emergency defensive measures. This presents a

serious infringement of the freedom of flight in international airspace. The Japanese government strictly protested against China and demanded that China revoke these measures. In fact, Japan remains steadfast in both word and deed (Inada, 2016).

5.6.3 The tension between the PRC's ADIZ and normative ADIZ

Only the Chicago Convention currently contains comprehensive rules and norms based on the framework of international law for defining how an airspace should be legally governed (Dutton, 2009). Here, Dutton (2009, p.692) pointed out that the Chicago Convention regulates civil aviation rules and practices adopted by the ICAO; however, crucially, the convention distinguishes between civil aircraft and state aircraft such as military, customs, and constabulary aircraft, while the convention lays down detailed rules pertaining to civil aircraft only.

The PRC deviates from international norms and practices in its ADIZ interpretation and practice. In fact, the nation has cited domestic law such as the PRC National Defence Law, the PRC Civil Aviation Law, and the PRC Flight Regulations as legal grounds to justify its unilateral ADIZ (Rinehart and Elias, 2015). This suggests that the PRC applies national law to aviation security and safety in international airspaces.

Dutton (2009) also noted the normative practice of state aircraft paying due regard to the safety of civil aircraft in accordance with the Chicago Convention. Since 1998, somewhat controversially perhaps, state aircraft cannot resort to the use of force against civil aircraft during flight or risk the lives and safety of people on board during an interception. Despite this, the PRC's ADIZ continues to threaten international aviation security. In short, the PRC does not distinguish between the state aircraft and civil aircraft that fly over or through the PRC's claimed exclusive economic zone in the East China Sea. The disputed exclusive economic zones in these seas deliberately overlap

with other claimants such as Japan, South Korea, and Taiwan (Pilger US-China Economic and Security Review Commission, 2016). Furthermore, the PRC has threatened to use defensive emergency measures against any aircraft that does not comply with the PRC's unilateral ADIZ (Pilger US-China Economic and Security Review Commission, 2016). In addition, the PRC's ADIZ requires both types of aircraft, civil and state, to maintain close communication with aviation authorities in the PRC, even if they have no intention of flying over or through PRC's national airspace (Pilger US-China Economic and Security Review Commission, 2016). A few days after the PRC's announcement of the ADIZ, as a sign of defiance, Japan, South Korea, and the US flew their military aircraft over the PRC's airspace to signal their non-recognition of the ADIZ (Pilger US-China Economic and Security Review Commission, 2016).

There is no provision in either the ICAO or UNCLOS that stipulates that the use of ADIZ gives one sovereign state authority to exercise a stronger sovereign claim over contested territories or seas. However, the PRC's ADIZ closely resembles the PRC's claimed exclusive economic zone that encompasses the Senkaku Islands (Kotani, 2014), thus creating the perception that the PRC intends to assert its sovereign claim over the islands. In addition, the PRC did not notify Japan, the US, South Korea, or Taiwan about its ADIZ even though the ADIZ overlaps with all of these states' own ADIZs. This created tensions with regard to regional security, as suggested by Inada's statement. As discussed earlier, the ADIZ does not give the PRC stronger sovereign claim over the Senkaku Islands due to the absence of clauses allowing this in the Chicago Convention-based ICAO and UNCLOS. However, it does pose a threat to regional aviation security in the East China Sea given the restrictive effects on the freedom of flight in the airspace over this area. Furthermore, Japan has avoided having its ADIZ cover the airspaces in Takeshima and the northern territories out of diplomatic consideration for the other

claimants to these territories (Hirakawa, 2013). From a geopolitical perspective, as Inada (2016) asserted, the PRC's unilateral announcement of the ADIZ coverage of the airspaces over the Senkaku Islands in the East China Sea is unquestionably provocative.

The US disappointed Japan when it announced that American commercial aircraft would notify its flight plans to Chinese aviation authorities in compliance with the PRC's ADIZ, even when the aircraft did not intend to fly over or through the PRC's national airspace. This created some discord in the aviation security policy coordination in the Japan–US alliance (Rinehart and Elias, 2015). Martin Dempsey, a former general of the US army, answered a question on this topic from Gideon Rose, the editor of Foreign Affairs, in June 2016 (Dempsey and Rose, 2016) by stating that the US needed to admonish the PRC through both diplomatic and military channels, but the Chinese insisted that the current international rules and norms were created at a time when it did not play a major role in global politics and now that it was stronger, it wanted to renegotiate these rules and norms. The US was somewhat successful in reinforcing the international rules of maritime behaviours with the PRC, but it has not made much progress in terms of aerial behaviour norms (Dempsey and Rose, 2016). Clearly, the US military recognises the importance of defending the rules-based aviation security order vis-à-vis the PRC, but needs to make more effort to ensure international aviation security norms prevail.

5.6.4 The PLA's coercive military aviation behaviours

Following the unilateral announcement of the ADIZ in the East China Sea by the PRC, a PLA military ship harassed the US Navy's aviation activity in 2013 in the South China Sea (Kawamura, 2014). Subsequently, from March 2014 to May 2014, PLA military aircraft approached US aircraft and maintained excessively close proximity, which is, of course, regarded as dangerous (Kawamura, 2014). Furthermore, in July 2014, the

PLA Navy unilaterally dispatched military spy ships to the RIMPAC in Hawaii (within the US's EEZ) for military intelligence gathering during a high-profile US-led multilateral naval exercise. This led the US to believe that similar intelligence-gathering activities conducted by their own agencies were acceptable to the PRC (Kawamura, 2014). However, the US Navy's P-8A aircraft was continuously harassed by the PLA's fighters in the South China Sea in August 2014. Finally, the PRC commenced the construction of an air force base on the artificial islands of the Paracel Islands in the South China Sea (Kawamura, 2014).

In recent years, the PLA's aviation behaviours have been causing Sino-Japanese tensions. For example, in 2014 and 2015, Japan's ASDF scrambled its fighter planes 943 and 873 times, respectively, which is numerically equivalent to the final stage of the Cold War era. A large bulk of the scrambling was conducted against PLA aircraft (Hokazono, 2016). Meanwhile, PLA pilots engaged in dangerous aviation manoeuvres during joint military drills conducted by the PRC and Russia in the East China Sea and a Japanese aircraft engaged in legitimate intelligence gathering was threatened by PLA aircraft manoeuvring in very close proximity. This behaviour deviated from the Chicago Convention's aerial rules defined in the ICAO (Hokazono, 2016).

In addition, as the recent high-profile case of the PLA's coercive aviation behaviours in the East China Sea suggests, the PLA's bombers and jet fighters flew over the Miyako Strait to advance into the Pacific Ocean in a formation involving 40 military aircraft, with the ostensible aim of conducting 'routine missions' in September 2016 (*Asahi Shimbun*, 2016). According to the *Asahi Shimbun*, the PLA's formation included H-6K strategic bombers and Sukhoi Su-30 fighter jets to improve reconnaissance, combat, aerial refuelling, and other capabilities, making this, according to the MOD, the first

time that Chinese fighter jets were detected in the Miyako Strait. This clearly sent a strategic signal to Tokyo (*Asahi Shimbun*, 2016). Furthermore, the PRC conducted its first military aviation exercise in the Western Pacific, flying over the airspace in the strategically vital Bashi Channel (*Asahi Shimbun*, 2016). Interestingly, the Japan-based US air force undertook its own military aviation activity, which was interpreted as a countermeasure to the PLA's military drill in the Bashi Channel. Clearly, the PRC regards the airspaces in the East and South China Seas as geographically essential choke points.

5.6.5 The role of Taiwan for Japan's security vis-à-vis the PRC's ADIZ issue and the Japan-Taiwan ADIZ issue

Here, we examine how Taiwan is attempting to become an aviation security asset for Japan vis-à-vis the PRC's ADIZ. The PRC has not declared an ADIZ over the airspace of the main island of Taiwan. According to Mark Stokes (2015), a retired Chinese general from the PLA suggested the Taiwan issue is a domestic one, meaning the PRC is not required to declare an ADIZ over Taiwan. However, despite the low visibility in the regional security environment, Taiwan is also a stakeholder. Chief Cabinet Secretary Suga Yoshihide deplored Taiwan's exclusion from the ICAO in 2016 and stated that one of the reasons Taiwan should be included in the ICAO is that Japan and Taiwan operate numerous direct flights. According to Michael Cole (2013), following the announcement of the ADIZ, Taiwan was forced to use 100 transponders per day to communicate its civilian flight charts to the PRC's military authorities. Furthermore, in 2010, aviation security tensions between Japan and Taiwan surfaced regarding the former's expansion of the ADIZ over the airspace west of Yonaguni Island in the East China Sea (*Taipei Times*, 2010). Clearly, despite its unrecognised status, Taiwan is a

quasi-independent regional stakeholder in the ADIZ issue in the East China Sea, and Japan clearly views Taiwan as an aviation security asset.

Regarding the above expansion, the retired JSDFA general, Toshimichi Nagaiwa (2014), complained that Taiwan's attitude toward Japan was more aggressive than Taiwan's more discreet attitude under the Ma administration toward the PRC's ADIZ. Meanwhile, Singh (2013c, p87) pointed out that the Japanese were concerned about potential PLA aggression towards the Yaeyama Islands during a Taiwan contingency. Here, Singh (2013c, p87) noted Japan's lack of independent threat assessment capability during the Taiwan Missile Crisis of 1995 to 1996. According to Sakashi (2014), Japan's expansion of the ADIZ over Yonaguni in 2010 was intended to enhance intelligence, surveillance, and reconnaissance based on the concept of dynamic defence and joint military training as well as the joint use of military facilities as a part of the Japan-US alliance. Furthermore, Sakashi (2014) argued that the lack of strong communication channels between the foreign and defence ministries of Japan and Taiwan was highlighted by the former's expansion of the ADIZ over the airspace west of Yonaguni Island. Japan is aware of the strategic need to communicate with Taiwan on matters of air defence and air security for the defence of the Nansei Islands. The military defence of air control in Taiwan and the Nansei Islands is strategically interlinked from the perspective of Japan's security in terms of both civil and military aviation.

Despite the initial restrained attitude of the former Ma administration vis-à-vis the PRC's ADIZ over the airspaces in the East China Sea, Taiwan conducted a large military drill involving the air force in the new ADIZ areas in 2014 (*Focus Taiwan*, 2014). While it cannot be confirmed whether Taiwan had a clear resolve to align itself strategically with Japan and the US during this military drill, Taiwan did intend to demonstrate its

independent diplomatic stance in relation to the PRC's ADIZ. Regarding this military drill, *Focus Taiwan* (2014) reported the following:

[s]ailors who joined the mission said it showed that Taiwan was not scared away by China's announcement in November 2013 of an ADIZ spanning a large part of the East China Sea that was seen as an attempt to assert territorial claims in the region.

In addition, following the unilateral announcement of the PRC's ADIZ, the Taiwan Air Force increased its air patrols within Taiwan's ADIZ (Schreer, 2014). If Japan is afraid of a potential cross-strait united front against it in the event of a contingency in the East China Sea, Taiwan's independent stance vis-à-vis the PRC's ADIZ may have tilted the balance of power in Japan's favour.

Crucially, the airspace over the East China Sea will be a strategically important air defence line against the PLA Air Force during a Taiwan contingency or a contingency in the East China Sea around the Senkaku Islands. Furthermore, the Senkaku Islands are geographically located just north of Taipei, which means it makes strategic sense that the Taiwanese military pays close attention to the aerial and maritime behaviours of the PLA in the East China Sea. Stokes (2014) argued that depending on the geographical boundaries of a potential future South China Sea ADIZ declared by the PRC, it will be possible for the PLA to geographically and militarily isolate Taiwan from other strategically vital first-island chains. As discussed in the CSD chapter, Taiwan appreciates the more solidified Japanese military presence in and around Taiwan since this indicates that both Japan and the US are willing to provide greater, if unstated, deterrence for Taiwan vis-à-vis the PRC. Tellingly, during the military drill in the Bashi Strait close to the northwest Philippines that was conducted by the PRC's

military, US military aircraft, including the F/A 18C of the marine corps, landed in southern Taiwan's Tainan airport (Stars and Stripes, 2015). The aircraft landed in Taiwan after departing from the Japan-based US military base in Misawa on 31 March 2015 to repair 'some mechanical issues' on its way to Singapore, where they were to participate in a military exercise (Stars and Stripes, 2015).

Uchida (2006) warned that Taiwan's Air Force and Japan's ASDF need to have greater awareness of the geographical boundaries on each side of the ADIZ between Japan and Taiwan, especially if Taiwanese military aircraft accidentally trespass into the Japanese side of the ADIZ. In such a scenario, Japanese military aircraft may scramble or use a laser beam against the intruder, resulting in panic among the Taiwanese aircrew and subsequently leading to a dangerous situation. The lack of bilateral strategic communication channels between Japan and Taiwan regarding regional aviation security issues was one of the factors that led to harsh protests by Taiwan against Japan in 2010. At the time, Japan had slightly expanded its own ADIZ over Yonaguni Island in Okinawa Prefecture (Taipei Times, 2010; Sakashi, 2014). As discussed earlier, one of the disadvantages of Taiwan's non-representation in the ICAO is its limited number of aviation security exchanges with contracting member states of the organisation. If Taiwan was better represented in the ICAO, it could create more solid communication channels with Japan to resolve regional aviation security issues, including the sensitive issue of airspace security management. Furthermore, following the unilateral announcement of the PRC's ADIZ over the East China Sea, the US and Japan officially submitted a letter to the ICAO asking whether a state has a right to dictate to or restrict commercial and civil aircraft over skies that are geographically out of the state's jurisdictional regions via the civilian air traffic controllers (Rinehart and Elias, 2015). Elsewhere, Yasuda (2016) argued that Taiwan's participation in the ICAO would be meaningful given the regional aviation security developments related to the issue of the ADIZ. In fact, it would be even more critical to examine the PRC's expansion and reorganisation of its sphere of influence over aerial domains to maintain peace and stability in the airspace in the Taiwan Strait and its periphery.

As a concrete counter-measure against the PRC's ADIZ, Japan convened a Japan–ASEAN summit on the freedom of overflight and civil aviation safety in Tokyo in 2014 without explicitly admonishing the PRC in reputational terms in relation to international law originating from the ICAO and UNCLOS (Rinehart and Elias, 2015). In addition, after Taiwan was allowed representation in the ICAO as a 'guest', the nation took the opportunity to protest against the PRC's coercive aviation measures. This included, for example, the PRC unilaterally announcing the use of civilian routes for commercial airlines in 2015, routes that are very close to Taiwan's FIR in the middle of the Taiwan Strait, which Taiwan saw as a coercive measure against it (*Focus Taiwan*, 2015).

5.7 Japan-Taiwan Anti-International Terrorism and Anti-Transnational Crime Cooperation

Following the 9/11 terrorist attacks, the Chicago Convention was reviewed to prevent such attacks from occurring again (IATA, 2001). At the same time, many states strengthened their anti-terror measures and established inter-state cooperation to combat terrorism. Here, the ADIZ issue discussed in the previous section remerged in the ICAO's national security discourse (Dutton, 2009). However, a lesser known fact is that both Japan and Taiwan conducted anti-terror measures after 9/11 (Uchida, 2006) to meet the new challenges that arose with regard to aviation security and safety according to the standards of the ICAO. According to Yang (2009), Japan's departure from the One-China policy of 1972 is clear from the fact that Japan and Taiwan conducted semi-

official security policy cooperation and coordination bilaterally and established a bilateral anti-terror security policy. This bilateral or multilateral law enforcement cooperation with Taiwan was not limited to international policing issues; rather, it was deemed as a significant political activity that required Japan to exercise political capital for its own security.

Japan has been instrumentalising Taiwan as a normative security asset to combat international terrorism and engaging in law enforcement cooperation for combating anti-international terrorism and anti-transnational crime activities. As discussed in the literature review, a de facto state often becomes a loophole for international terrorism and transnational crime activities. Here, Kawamoto (2016) argued that Japan must have the capability to deal with the threats of international terrorism and transnational crime, even during peacetime. In fact, Japan must also contribute to preventing states from becoming fragile and vulnerable to incubating violent threats. It requires further study to elucidate why the PRC appears to grudgingly acquiesce Japan-Taiwan constabulary cooperation. As in the case of the CSD, non-military policing cooperation appears not to cross the redline for the PRC in the context of Japan's One China policy. Japan-Taiwan anti-international terrorism and anti-transnational organised crime cooperation presents a suitable case study for elucidating how Japan has been enhancing Taiwan's as if membership in anti-international terrorism and anti-transnational crime efforts in accordance with the international security norms of the ICAO.

5.7.1 Geopolitical importance of airport security

Airports play a geopolitical role as a gateway for international travellers entering and leaving a country. They facilitate the free movement of people and must account for the consequences of unintentionally offering better mobility to international terrorists and transnational crime organisations. Airport security has become one of the most critical components of the war on terror, or the anti-terror efforts. Regarding the importance of airport security, Dodds and Adey (2014) stated the following:

[f]or one thing, as others have noted, the airport has been a key site for investigating how and with what consequences the war on terror has manifested itself in terms of security and surveillance with ever greater attention given to monitoring and evaluating the body and behaviour.

The ICAO regards airport security as an integral part of the global fight against international terrorism, and in 2018, ICAO Secretary General, Fang Liu, and Assistant Secretary-General and Executive Director of the Counter-Terrorism Executive Directorate, Michèle Coninsx, signed an agreement to enhance cooperation between the two agencies in terms of intelligence sharing on border control, aviation security, and counter-terrorism. Here, Liu stated that '[i]t [the agreement] also recognises our shared emphasis on prioritising more sustainable travel facilitation solutions', adding that this included 'travel document security, citizen identity management, and the international standardisation of passenger data exchange' (The ICAO, 2018).

5.7.2 Japan-Taiwan airport security cooperation

Clearly, Taiwan is being ignored by the ICAO and UNCTED's cooperative initiative due to its ostracisation by the PRC. However, on closer examination, it would appear that Japan has been enhancing Taiwan's 'as if membership' in international counterterrorism and anti-transnational crime efforts. In 2004, the ROC's Executive Yuan ratified a bilateral immigration control agreement based on the mutual decisions made by Japan's semi-official interchange association, Japan's de facto embassy in Taiwan, and Taiwan's East Asia Relations Commission (Yang, 2009). Both Yang

(2009) and the *Mainichi Shimbun* (Shijienian Jian, 2005) pointed out that this semi-official bilateral agreement between Japan and Taiwan enabled the staff in charge of immigration control and restriction issues from both countries to visit each other's airports when the security need arose and made it possible to screen visas and passports beforehand to detect and prevent suspicious figures such as potential terrorists and criminals from entering either nation. Elsewhere, *Mainichi Shimbun* (2004) noted that the immigration office in Japan considers such anti-terror efforts at airports to be efficient measures to minimise the security risks stemming from overseas terrorists entering Japan. For example, from 2002 to 2003, Al-Qaeda's high-ranking members clandestinely entered Japan via Singapore using forged passports.

Japan and Taiwan signed an agreement over intelligence and information sharing and law enforcement cooperation in terms of border and immigration security on 20 November 2014 (Interchange Association 2014). Due to this agreement, the two actors have been able to comprehensively cooperate on border and immigration security issues through the Ministry of Justice in Japan and the Ministry of Interior in Taiwan. For example, immigration officers from each side can now collaborate, conduct annual consultations, conduct functional exchanges on an ad hoc basis in times of need, can engage in data sharing on immigration issues and information sharing for technological cooperation to strengthen the border security on both sides, and can carry out emergency communication and coordination to prevent the use of forged passports and forged documents and the illegal entry of aliens within the legal boundaries of the relevant divisions of the Ministry of Justice and the Ministry of Interior. If a change occurs within any division, immediate notification is now required (Interchange Association 2014). Furthermore, an anti-terror financing and anti-money laundering agreement was signed between Japan and Taiwan in 2012. This is essential for

intelligence and information sharing and the effective implementation of bilateral law enforcement and security cooperation as well as for conducting bilateral economic activities for Japan–Taiwan relations (Interchange Association, 2012). In 2017, Japan and Taiwan concluded an agreement on customs enforcement that enabled the two sides to combat smuggling and to streamline clearance procedures via information and expertise sharing (*Taiwan Today*, 2017). Taiwanese nationals travelling to Japan have been subject to possible Japanese immigration and customs screening processes through the pre-clearance system at Taiwan Taoyuan International Airport since the middle of 2017 (*Taipei Times*, 2016).

5.7.3 The issues with airport security in Taiwan: Implications for Japan's security Cost-effective aviation has facilitated the movement of criminals and illegal goods on a global scale, including in Taiwan. Certain airport security issues can be major and may potentially affect Japan's security with regard to the proliferation of both light and heavy arms. For example, in 2012, a Taiwanese-American woman and her children who were travelling with a handgun and bullets were caught by the aviation police in Taiwan's Taoyuan Airport (Taipei Times, 2012). The woman and her children were flying from John F Kennedy Airport to Osaka, Japan, and then to Taipei, Taiwan. The security checks at these three airports could not detect the weapons, which called into question the robustness of airport security. Furthermore, in 2014, during a security check, the aviation police held up a China Airlines aircraft at Taiwan's Taoyuan Airport after it was found to be transporting the components of 36 AIM missile warheads (Taipei Times, 2014). Here, the Ministry of National Defence had to intervene to clarify that these weapons were intended to be secretly transported to the US for free repairs (Taipei Times, 2014). The poor secrecy management and lack of communication between the aviation police and the Ministry of National Defence was highlighted in

this case, raising widespread alarm regarding hypothetical situations in which such high-profile weapons could fall into the hands of terrorists (*Taipei Times*, 2014). On a positive note, it was proven that the aviation police in Taiwan are able to detect high-profile military components for missiles (*Taipei Times*, 2014).

The issue of corruption must be tackled if the reliability of airport security is to be improved. For example, the 'Honey Trap' case of 2016 revealed the extent of corruption among aviation police. Here, a female Chinese spy seduced a high-ranking Taiwanese aviation officer into procuring Chinese-manufactured X-ray security equipment by falsely claiming that the equipment was produced in Japan (*Taipei Times*, 2016).

5.7.4 Japan—Taiwan airport and port security intelligence sharing and multilateral efforts

Japan and Taiwan engage in intelligence and information-sharing cooperation in relation to anti-international terrorism and anti-transnational crime as well as contingencies that involve airport security and port security. Ito (2016) described how eels caught by Taiwanese fishermen – which are prohibited from being exported to Japan by the Japanese government – are shipped to Hong Kong and legally re-exported to the Japanese market using aerial and maritime transport. Tackling such cases of smuggling will require aviation security cooperation and transnational intelligence sharing among Japan, Taiwan, and various other countries.

During an interview done by Konno, the former head of the fisheries development foundation in the Taiwan region, Ling, expressed some dissatisfaction with the Japanese government's policy on legally importing eels from Hong Kong and suggested that Japan amend its laws that ban the import of such goods from Taiwan. In fact, Japan, Taiwan, South Korea, and the PRC have been conferring over the measures needed to

resolve this smuggling-related problem. Huang Hong-yen, who serves in the fisheries division of the agricultural committee of the Executive Yuan in Taiwan, stated in an interview that since 2012, Taiwan, Japan, South Korea, and China have been conducting informal consultations to achieve multilateral cooperation for data sharing on the smuggling of eels. Based on the multilaterally shared data, they voluntarily restricted the fishing of eels (Konno, 2016). However, Huang revealed his long-term desire to make this cooperation legally binding (Konno, 2016). Crucially, in certain cases, such as the eel smuggling case, Taiwan and the PRC have agreed to conduct transnational policing cooperation despite their adversarial relationship.

According to the Joint Statement of the Bureau of Fisheries of People's Republic of China, the Fisheries Agency of Japan, the Ministry of Oceans and Fisheries of the Republic of Korea and the Fisheries Agency of Chinese Taipei on International Cooperation for Conservation and Management of Japanese Eel Stock and Other Relevant Eel Species (2014 17 September), the Fisheries Agency in Chinse Taipei is represented as a "participant" due to its position as an "economic entity" in the APEC together with the PRC, Japan and South Korea. Chinese Taipei engages in intelligence and information sharing with the counterparts from those nations on this security issue.

Regarding the sensitive multilateral intelligence cooperation between political and military adversaries, Naim (2012) noted the low-trust issues among national police agencies (including Interpol), especially those in adversarial or unreliable states. While national law enforcement agencies go to great lengths to ensure the confidentiality of their intelligence, they generally remain wary of excessive sharing. Given this context, the Japan–China–Taiwan–Korea informal multilateral intelligence-sharing cooperation vis-à-vis the eel smuggling case is significant.

5.7.5 Travel documents, border security, and identity management

While Japan and Taiwan have no extradition agreement, de facto bilateral or even multilateral law enforcement cooperation between them in terms of aviation security is still possible. In 2011, the ICAO's regional seminar that was held in Singapore discussed the critical nature of biometric and border security as follows:

[a] key finding from the 9/11 Commission Report was that 'for terrorists, travel documents are as important as weapons.' The issuance and integrity of travel documents, border security and identity management remain an integral part of global counter-terrorism capacity-building efforts in furthering the objectives of UN Security Council Resolution 1373 (2001) (The ICAO, 2011).

Japan, Taiwan, and other countries in the region are cooperating to share intelligence related to travel documents, the identities of suspicious figures, and border security in accordance with the principles of the ICAO.

Japan's National Police Agency's white paper of 2012 described a transnational crime investigation case that involved examining travel documents, the identity of the criminal, and the border security for transnational policing efforts operating in accordance with the ICAO's practices involving biometric security technology. The paper stated the following:

[t]here was a South Korean national that committed the robbery of a luxury ring by using a false identity as a Japanese female in Tokyo, then proceeded to commit a similar robbery of a ring in Taipei, Taiwan. Nevertheless, he was identified through the security cameras and the manner in which the crime was committed was found to be similar to the case in Japan. By the time he entered

his next destination, which was South Korea, his passport was recorded as that of a robber because Japan–Taiwan–South Korea's relevant law enforcement agencies declared a transnational emergency and conducted speedy intelligence and information sharing regarding his transnational movements. The security camera image was shared among the three parties, and the South Korean Police investigated the transnational crime activities of the criminal and arrested him for the robberies he committed in Japan and Taiwan (Japan National Policy Agency, 2012).

Furthermore, the white paper confirmed that Japan and Taiwan cooperated in terms of sharing intelligence on the facial image, fingerprints, and transnational movements of the criminal. The stolen passport used when the criminal entered South Korea was on record with the police in Taiwan, and the confession made by the criminal's contacts led to the identification of the criminal. Clearly, Japan and Taiwan are cooperating in the form of making bilateral or multilateral law enforcement efforts in intelligence sharing.

Japan and the United Nations Office on Drugs and Crime (UNDOC) agreed to cooperate to facilitate the capacity building of airport security, including in terms of the introduction of the latest facial recognition systems and explosive detection techniques in Southeast Asian states in view of combatting international terrorism (*Mainichi Shimbun*, 2016a). Taiwan's Southeast Asian neighbours may be vulnerable to radicalisation and terrorism due to the regional diffusion of ISIS. As such, the UNODC conduct the training of airport crews and introduced the latest security devices developed by Japan (*Mainichi Shimbun*, 2016a). For certain, ISIS is utilising the porous borders of ungoverned de facto states in Southeast Asia as safe havens for conducting

international terrorism (Kurlantzick, 2016).

Reportedly, 69 Taiwanese nationals are listed as ISIS targets (Taipei Times, 2016e). Foreign Minister David Lee stated that his ministry received intelligence from the Taipei Representative Office in the UK in June 2016 and from London's Metropolitan Police Service Counter Terrorism Command (Taipei Times, 2016e). During the era of White Terror, the corrupt KMT regime of the ROC under Chiang Kai Shek had a criminalised economy (*The Economist*, 2014). Even after Taiwan's democratisation, the PRC supported pro-PRC political forces in Taiwan. For example, it supported the Bamboo Union, whose 'spiritual leader', Chang An-lo, was suspected of being a triad leader in Taiwan and spent 17 years in the PRC as a fugitive. However, the PRC denied this (The Economist, 2014). A crucial incident for Japan's security came in the form of Chiang An-lo's meeting with the local Okinawan mafia groups that were allegedly conducting narcotic trafficking in Okinawa (Ryukyu Shimpo, 2016). Considering Taiwan's importance to Japan's security, and given the upcoming Olympics event in Tokyo in 2020,6 ensuring Japan-Taiwan intelligence sharing on travel documents, border security, and identity management while facilitating Taiwan's 'as if membership' in the ICAO's anti-terrorism and anti-transnational crime efforts will be crucial.

5.7.6 The airport security cooperation between Japan and Taiwan: Implications for the contingencies in Taiwan and the East China Sea

Law-enforcement cooperation for anti-international terrorism and anti-transnational crimes at airports or ports between Japan and Taiwan presents a crucial countermeasure to any potential PRC invasion of Taiwan and the Nansei Islands in Japan. One of the characteristics of the grey zone scenarios is the greater use of non-traditional armed

 $^{^{\}rm 6}$ The games were subsequently postponed until 2021 due to the Covid-19 pandemic.

groups such as maritime militia, spies, and insurgents, which do not fit into the traditional definition of hard power. According to Sato, the former director general of the Japan Coast Guard, in the process of becoming a 'maritime power', the PRC has been utilising non-military tactics (Yomiuri Shimbun, 2016). For example, Mochida (2014), the former chief of staff of the GSDF Western Army 7th Brigade Field Operation Bureau, argued that, as the case of Crimea shows, the hybrid nature of warfare can involve a greater use of unofficial militia. Here, the Japanese SDF or the Taiwanese military would confront non-traditional adversary forces, including armed non-military militia, and this hostile militia may attempt to occupy airports or naval ports to disrupt any logistical advantages during a contingency. Mochida (2014) went on to argue that Japan-Taiwan-US militaries should conduct contingency analyses and consolidate the information and intelligence sharing regarding the results of the analyses among the three actors. After studies were conducted on the needs of the various police forces by the NSC and the relevant security organs in Taiwan, since May 2019, Taiwanese airport police have been equipped with submachineguns to better deal with terrorist threats (Taiwan News, 2019). Scott W. Harold (2016), the associate director of the RAND Center for Asia Pacific Policy, noted the importance of fortifying airfields and aviation-related facilities to improve the survivability of critical air assets during a contingency.

5.7.7 Taiwan's as if membership in international security and the hierarchical Japan–Taiwan relations

Interestingly, the PRC has not openly condemned this Japan–Taiwan anti-terror airport security cooperation. It is not implausible to assume that the PRC would react negatively to any Japan–Taiwan semi-official security cooperation as a violation of the One-China policy, especially if the cooperation was commenced under the Koizumi

and Chen administrations, both of which were perceived as hostile to the PRC. However, the aforementioned case of the smuggling of marine products such as eels suggests that China is willing to collaborate with Taiwan – albeit informally – when the need arises. Given this scenario, if any Japan-Taiwan security cooperation is conducted at a constabulary level, the PRC may begrudgingly acquiesce to it within the context of Sino-Japanese regional power management. Therefore, the issue of whether Japan can enhance Taiwan's as if membership in anti-international terrorism networks in accordance with the ICAO's security norms depends on how Japan navigates its One-China policy while securing a space for Taiwan to enhance its contribution to Japan's security. It is up to Japan to determine what form of security contributions Taiwan is permitted to make, provided Japan continues to instrumentalise Taiwan as an aviation security asset while pacifying the PRC by appropriately navigating the One-China policy. This hierarchical relationship grants Japan an influential moral authority to determine the normative security behaviours that Taiwan must adhere to. As discussed in Chapters 2 and 3, to be accommodated into the Eurocentric club of elite powers, 'non-civilised' peoples must meet Western civilisation standards and must be recognised as equals by the 'civilised' Europeans. However, according to the ESIR, this merely serves to consolidate the Eurocentric hierarchical global order within international society. Similarly, to meet the expectations set by the regional great power Japan, Taiwan needs to make an effort to meet its neighbour's security preferences. Taiwan's compliance with Japan's security expectations and its attempts to obtain recognition from Japan are consolidating the hierarchical Japan-Taiwan relationship. Japan rewards Taiwan through new bilateral agreements related to anti-terror efforts provided the latter meets the former's security requirements or if the international security environment requires Japan to conduct new anti-terror efforts. Japan cannot

leave Taiwan to fend for itself within the context of international security despite the latter's lack of international recognition. In short, Japan expects Taiwan to complement its own security priorities by regulating and structuring its security behaviours in accordance with the international security norms of the ICAO. Thus, Japan may give Taiwan an alternative avenue to improve Taiwan's international standing by enhancing Taiwan's *as if membership* in international security efforts. However, this may also force Taiwan into becoming, if not a disposable geopolitical pawn, a permanent international security policy instrument for maintaining Japan's balance of power in relation to the PRC.

Kawashima (2016) argued that since Japan–Taiwan relations improved under Ma Ying Jeou, who sought closer ties with the PRC, the PRC tolerated the stronger Japan-Taiwan relations until the DPP came to power in 2016 under Tsai. However, during the administration of Chen, who experienced more tensions with the PRC than the Ma administration, Japan and Taiwan engaged in more frequent high-profile contact (Nojima, 2016). Lee Teng Hui admitted that high-profile secret strategic security dialogues were conducted among Japan, Taiwan, and the US on a frequent basis, in which Tsai participated twice (Asahi Shimbun, 2016). In the written communication involving the author of this dissertation and Hideaki Kaneda – a former vice admiral of Japan's Defence Force – Kaneda expressed that, as a frequent participant in Japan– Taiwan security talks, he had a difficult time under Ma Ying Jeou. The PRC may attempt to hinder the development under Abe and Tsai of stronger Japan-Taiwan ties that enhance the security of both nations. In fact, it has already sanctioned Taiwan's international space, including in terms of its guest membership in the ICAO. That is precisely why Japan-Taiwan aviation security ties are essential and why it is worth investigating the role of Japan in accommodating Taiwan in its international aviation

security efforts.

5.8 Summary

Japan utilised the ICAO's security and safety regulations to instrumentalise Taiwan for its own aviation security. Here, it encouraged Taiwan to behave in accordance with the ICAO's principles regarding the Open Skies agreements, the ADIZs, and law enforcement. All of these aspects have significant security implications for Japan's security and the balance of power vis-à-vis the PRC in the East China Sea and beyond. Despite Taiwan's non-access to the ICAO due to the PRC's sanctions and the absence of membership due to its lack of statehood, Japan can socialise Taiwan in terms of complying with the ICAO's international aviation security norms and rules, such that the former can enable the latter to contribute to Japan's aviation security.

Meanwhile, Japan's non-traditional aviation security cooperation such as the airport security and the international anti-terrorism efforts appear to make the PRC acquiesce incrementally growing Japan-Taiwan security cooperation despite the former's One China policy. Clearly, if Japan adheres to security cooperation based on international security norms, it is difficult for the PRC to sanction security ties between Japan and Taiwan.

The next chapter examines Japan's direct intervention that acknowledged Taiwan as a de facto stakeholder in the maritime security issues in the East China Sea through the fisheries agreement. This agreement propelled Taiwan to an almost equal level to Japan and China as a semi-legitimate actor in the issue of the East China Sea, which ensured Taiwanese perspectives more appreciated by the Japanese. However, the agreement also antagonised the Chinese since it was regarded as a violation of Sino-Japanese great power management. It also drove a wedge into the potential cross-strait united front

against Japan, while Taiwan's subordinate position vis-a-vis its powerful neighbour remains unchanged within the hierarchical Japan-Taiwan relations in international society.

Chapter 6. The Fisheries Agreement

6.1 Introduction

Japan's conclusion of the bilateral fisheries agreement with Taiwan has significant geostrategic implications for Japan's policy on Taiwan-China relations and for enhancing Japan's balance of power in the East China Sea area in relation to the PRC. The agreement helped Taiwan assert itself as a quasi-legitimate security actor in the East China Sea issue. This case study chapter is different in nature from the other two case study chapters since, within this context, Japan has politically acknowledged Taiwan as a quasi-legitimate actor. This has created the need for Taiwanese perspectives to be further examined, as opposed to CSD and ICAO perspectives, which effectively render Taiwanese political inputs nonrelevant in the hierarchical Japan-Taiwan relationship. This is because unlike the CSD and ICAO perspectives wherein we discussed Japan's dealings of Taiwan as Japan's foreign policy, Japan's signing of the fisheries agreement with Taiwan acknowledged Taiwan as a semi-independent stakeholder in the marine issue in the East China Sea. Thus, this chapter discusses not merely Japan's Taiwan policy as part of its foreign policy, but also bilateral Japan-Taiwan relations. Moreover, the agreement was regarded as Japan's attempt to drive a wedge into the cross-strait united front and to undermine the PRC's concept of One China (which includes Taiwan). From the PRC's perspective, the agreement was a violation of the One-China policy and a detriment to Sino-Japanese great power management. The agreement also enhanced Taiwan's as if participation in international maritime security efforts in terms of law enforcement and economic security. Furthermore, Japan concluded the fisheries agreement with Taiwan despite the maritime tensions in the East China Sea. We can use this point to elucidate how Japan and Taiwan were able to peacefully solve their maritime issues and achieve amicable

Japan–Taiwan relations through the perspective of the ESIR.

First, it discusses how the fisheries policy has been traditionally conceptualised as part of comprehensive security issues without consideration of the power politics in Japan, with the exception of the Japan–Taiwan fisheries agreement. Second, it examines the geopolitical implications of Japan's instrumentalisation of the fisheries agreement with Taiwan in relation to Japan–China–Taiwan relations. Third, due to the nature of the fisheries issue, it examines how the agreement affected law enforcement and economic security for Japan, Taiwan, and Okinawa. Fourth, with reference to the ESIR perspective, it examines the normative dynamics behind how Japan and Taiwan aimed to create amicable relations and addresses how these dynamics allowed the peaceful signing of the fisheries agreement despite the maritime tensions in the East China Sea and the fact that the hierarchical Japan–Taiwan relationship remained unchanged.

6.2 What is the Traditional Fisheries Policy in Japan? Japan's Comprehensive Security and Food Security Policy

Smith (2015) argues that Japan's fisheries policy is deeply rooted in the concept of comprehensive security, which encompasses the food security of the entire Japanese populace. As with all international security issues, Japan must alleviate any uncertainty and risks stemming from other state or non-state actors. Regarding how this fisheries policy became an important foreign policy for Japan, Smith (2015, p. 5) noted that as distant-water fisheries became more critical, the contact among foreign fishermen increased. However, this also increased the international friction, which meant the international management of fisheries resources became a more urgent issue for the Japanese government. In short, this was no longer simply a domestic issue, but an issue firmly related to foreign policy.

The international fisheries issue has not involved traditional security concerns for Japan. In fact, the nation's fisheries policy prior to the bilateral Japan–Taiwan fisheries agreement was not regarded as a major strategic instrument in the regional balance of power. Regarding the conceptual differences between comprehensive security and national security, Smith (2015, p. 5) pointed out that unlike traditional power politics, comprehensive security regards the pursuit of reliable trading partners and routes, economic security, and stable access to natural resources to be as critical as the conventional militarised conception of security.

As noted, Japan's fisheries policy has traditionally been conceptualised within the context of comprehensive security, especially in terms of the nation's food security policy, as opposed to a strategic instrument for achieving regional maritime peace and stability, which would involve enhancing the regional maritime power balance and the legal maritime order. However, due to the unique status of Taiwan and the PRC's courting of Taiwan in the issue of the East China Sea to create a cross-strait united front vis-à-vis Japan, Japan's conclusion of the fisheries agreement with Taiwan was utilised as a foreign policy instrument to defend Japan's southern flank in the East China Sea surrounding the Senkaku Islands while simultaneously recognising Taiwan as a quasi-legitimate stakeholder that was (almost) equal to Japan and the PRC in the issue of the East China Sea.

6.3 Japan's Instrumentalisation of the Fisheries Agreement as a Security Asset

6.3.1 Japan's geopolitical instrumentalisation of the fisheries agreement with Taiwan vis-à-vis the PRC

Japan instrumentalised international law to sign the fisheries agreement with Taiwan as

a hard security asset vis-à-vis the PRC as opposed to in terms of traditional food security in comprehensive security domains. The agreement may be viewed as a political refutation of the PRC's claim over Taiwan and the Senkaku Islands (Kawashima, 2013). Japan's treatment of Taiwan as a quasi-legitimate actor, one that is quasi-equal to Japan and the PRC in the issue of the East China Sea, helped Taiwan assert its 'statehood'. In short, Japan allowed Taiwan's quasi-official participation in the issue of the East China Sea through the fisheries agreement. Unlike the ambiguous status of Taiwan in the CSD and the ICAO (both require statehood for membership), Japan's treatment of Taiwan propelled the latter to a quasi-official level in this maritime issue even though Taiwan was not involved in the United Nations' Convention on the Law of the Sea (UNCLOS). This ensured the Taiwanese political input regarding Japan's security policy was considerably more appreciated, albeit that the hierarchical Japan—Taiwan relationship, wherein Japan unilaterally affords Taiwan quasi-recognition as a reward for its normative behaviour in relation to Japan's security, remained unchanged.

The UNCLOS regulates and handles international maritime disputes, and claimant states can instrumentalise it for arbitration, with the convention instrumentalised as a set of modern international maritime laws in bilateral or multilateral maritime disputes when the disputes are considered unresolvable by the claimant states alone. Here, the states need to make a request to the International Court of Justice to obtain a legal ruling. In the bilateral Japan—Taiwan fisheries agreement, both nations agreed to set aside their sovereign disputes over the territories in question and attempted to resolve the issue of marine resources and fishing rights in the East China Sea. The Japan—Taiwan fisheries agreement is viewed as a geostrategic success for Japan since it bilaterally created a dispute and conflict resolution mechanism at both the legal and institutional levels, including in terms of the need for legal readjustments on the side of Taiwan. After de

facto Japanese ambassador Masaki Saito retired in 2009, he noted the importance of creating a legal and institutional mechanism for the bilateral fisheries issues, stating that it was an important unresolved business of his semi-abruptly and semi-forcibly terminated ambassadorship (Sahashi, 2014). Now that such a legal and institutional mechanism exists, the two sides can initiate an annual fisheries committee to 'negotiate fishing affairs, jointly conduct maritime resource protection, and discuss possible cooperation' (Po and Pei, 2013, p. 13). In fact, the two nations can now take joint legal action against the PRC if the PRC's vessels encroach on Japanese territorial seas around the Senkaku Islands.

The PRC has a different interpretation of the UNCLOS to that within mainstream international society, as was manifested in the PRC's 'nine dash line' demarcation in the South China Sea. For example, regarding the Philippines' initiative to bring regional maritime disputes in the South China Sea against the PRC to the International Court of Justice, the PRC argued that it was impossible to legally settle such disputes if a legal solution to the territorial disputes in the South China Sea had not yet been found (Press Release at Permanent Court of Arbitration, 2016). As such, the Philippines focused its arbitration case on maritime rights rather than territorial disputes. The PRC did not participate in the hearing, in which the ICJ attributed merits to seven of the 15 submissions (Press Release at Permanent Court of Arbitration, 2016). The Philippines arbitration case may not have led to material changes vis-à-vis the tensions in the South China Sea, but the Philippines' legal attempt to resolve the dispute certainly affected the PRC's reputation. In short, it was seen as a bully, one that had resorted to coercive measures vis-à-vis smaller states such as the Philippines. Similarly, Japan and Taiwan attempted to find a peaceful solution to regional maritime disputes in the East China Sea via the UNCLOS. This also resulted in the PRC appearing to be a hostile party,

despite the fact that the nation is a signatory member of the UNCLOS. In contrast, while, as we know, Taiwan cannot be a member of any international organisation that requires statehood for membership, it attained a reputation as a regional peacemaker.

To elucidate the difference between the Japan–Taiwan fisheries agreement and the Japan–PRC fisheries agreements in the East China Sea, a brief examination of the Japan–ROK fisheries agreement and the Japan–PRC fisheries agreement must be conducted. Both cases highlight the non-existence of power balance calculations vis-à-vis the third-party state, which is not the case with the Japan–Taiwan fisheries agreement. The changes in domestic law and the conclusion of the UNCLOS in 1982 required Japan to renegotiate its fisheries agreements with both the PRC and South Korea. New legal zones such as the EEZ expanded Chinese and Korean fisheries activities in Japan's EEZ. Regarding Japan's need to ratify the UNCLOS, Smith (2015, p. 114) stated the following:

[o]n the basis of earlier bilateral agreements and Japan's 1977 fisheries law, Japanese waters beyond the 12 mile territorial sea remained the high seas for Chinese and Korean trawlers, prompting these fishermen to overexploit fish stocks in grounds contiguous to the Japanese coast.

Thus, the increasing number of disputes with fishing vessels operated by Japan's neighbours required a renegotiation of the fisheries agreements among the nations in question. Japan ratified the UNCLOS to secure its marine resources and better protect its coastal fisheries within its own EEZ. Based on the UNCLOS, Japan signed a fisheries agreement with the PRC, and accordingly, both sides provisionally extended the EEZ waters. Provisional waters were established, and the EEZ's boundaries were blurred to minimise any conflicting claims. Furthermore, a joint fisheries committee

was established to resolve any fisheries-related issues. Regarding the Japan–South Korea fisheries agreement, while negotiations commenced in 1996, it took three years to conclude the agreement due to the Takeshima dispute. Both sides had to compromise by denoting the waters around Takeshima as 'middle waters' and by agreeing to conduct joint management of the contested waters (Smith, 2015).

Meanwhile, the Japan–Taiwan fisheries agreement was a success for Japan since it rendered the twelve nautical miles of territorial waters around the Senkaku Islands as exclusively Japanese territory (Japan–Taiwan Fisheries Agreement, 2013). Here, Japan and Taiwan agreed to disagree about the legal status of the Senkaku Islands (Japan–Taiwan Fisheries Agreement, 2013). While both the Japan–PRC and the Japan–South Korea fisheries agreements considered different dispute management methods to alleviate tensions, neither cases took regional power politics dynamics into consideration vis-à-vis the third-party state. Notably, despite the existence of a controversy over the issue of fishing rights, it took Japan 17 years to finally conclude its fisheries agreement with Taiwan (Japan–Taiwan Fisheries Agreement, 2013). In fact, this became possible only after maritime tensions seriously escalated in the East China Sea following Japan's nationalisation of the Senkaku Islands. From the language used in the official document on the Taiwan–Japan fisheries agreement from Taiwan's MOFA, we can clearly observe the geopolitical dimension of the agreement. Here, MOFA's archive includes the following:

Article 1 of the Taiwan–Japan Fisheries Agreement stipulates that 'This Agreement aims to ensure peace and stability in the East China Sea, promote friendly and reciprocal cooperation...' This stipulation does not appear in fisheries agreements of other countries such as the Japan–Korea and the

Japan-mainland China fisheries agreements. It is included in the Taiwan–Japan agreement due to Japan's positive response to President Ma's East China Sea Peace initiative (The Taiwan–Japan Fisheries Agreement Archive, 2013, p. 2).

6.3.2 Taiwan and Sino-Japanese regional great power management

If Taiwan remains under the shadow of the PRC, any international agreements it signs with Japan that may infringe on the strict interpretation of the One-China policy will be viewed negatively by the PRC. For example, soon after Japan and the PRC signed their fisheries agreement, the 'Taiwan factor' became an inevitable issue due to China—Taiwan tensions (Tseng and Ou, 2010), which needed to be resolved for future Japan—Taiwan and Sino-Taiwanese relations. As a claimant of the Senkaku Islands, depending on how it behaves, Taiwan can either contribute to increasing or alleviating regional maritime tensions in the East China Sea (Jakobson, 2013). Therefore, a non-traditional security instrument such as the fisheries agreement has geopolitical implications for the sovereignty factor in Taiwan—China relations. In short, it is likely that the PRC regarded Japan's conclusion of the fisheries agreement with Taiwan as a unilateral change in the status quo.

Any security arrangements that accommodate Taiwan need to help the nation consolidate its position as an autonomous player. Japan's engagement with Taiwan will increase the chances of Taiwan's strategic survival and its international legitimacy by helping improve its record on human rights, democracy, economic liberalism, and the environment. All the normative concepts that Taiwan embraces are shared by Japan, and the latter's human rights record and socio-economic development have presented a model for Taiwan to emulate since the post-war period. The CCP considers 'universal values' to be irrelevant to the national conditions of China and has enacted laws that

regard such values as a threat to national security (The PRC's Guojia Anquan Fa, 2015). Crucially, these laws claim to judicially encompass Taiwan and regard the 'Taiwan Independence Force' as China's gravest national security threat (The PRC's Guojia Anquan Fa, 2015). Taiwan has its own constitution and domestic laws, which are independent of the PRC. A number of academics in Taiwan have described the PRC's the enactment of the aforementioned law as a legal annexation (Brooking's Institution, 2015). Here, China's final goal vis-à-vis Taiwan – as manifested in its non-interference policy and anti-secession law, which legalises the use of force against Taiwan if Taiwan moves for de jure independence (Fan Guojia Fenlie Fa, 2005) – clashes with Japan's acknowledgement of Taiwan as a quasi-legitimate security actor in the issue of the East China Sea. Therefore, from the PRC's perspective, Japan's signing of the fisheries agreement was regarded as a violation of both the Sino-Japanese regional great power management and the One-China policy. The PRC subsequently retaliated by informally disputing Okinawa's status (*New York Times*, 2013).

Taiwan appears to be less rigid regarding the sovereignty issue since it could potentially serve Taiwan's consolidation of its de facto independence. Provided Japan has the political will, this consolidation will benefit its instrumentalisation of international law vis-à-vis the PRC. For example, following Japan's derecognition of the ROC in 1972, Taiwan could only use the private-based Sino-Ryukyuan Cultural and Economic Association to handle any bilateral ties. The ROC government was outraged when the US returned Okinawa to Japan without consulting it (*Taipei Times*, 2006). Reportedly, in 2006, Taiwan admitted that Okinawa was a part of Japan in return for upgrading the private association to the quasi-diplomatic office in Okinawa in 2006 (*Taipei Times*, 2006). However, in a personal written communication with the current author, a former state minister, Furuya Keiji (2019), pointed out that the upgrade of the consular-like

entity was carried out based on Taiwan's voluntary behaviour. Meanwhile, the PRC is yet to establish a diplomatic consular there (Zhonghua Renmin Gonhuguo Riben Guo Dashiguan, 2015).

Japan accommodated Taiwan into its regional oceanic governance in the East China Sea by instrumentalising international law, namely the UNCLOS, albeit in a private manner. This is significant given that Taiwan has not signed any high-profile fisheries agreements other than that signed with Japan in relation to the East China Sea. The fisheries agreement is viewed as Japan's foreign policy instrument for elevating Taiwan to a quasi-equal stakeholder alongside Japan and China in the issue of the East China Sea, despite the nation's vulnerable position and its limited material ability to contribute to Japan-China-Taiwan relations. The fisheries agreement made the PRC look like a bully alongside the more genial Taiwan. Through the fisheries agreement, Japan ensured its security in terms of the Senkaku Islands and Taiwan in relation to the PRC by enhancing Taiwan's as if membership in international maritime security efforts. Meanwhile, it would appear from the discussions on the Senkaku issue conducted in the PRC that the mere existence of the East China Sea Peace Initiative proposed by Ma Ying Jeou frustrated the PRC. Ma's initiative indicated that Taiwan was not interested in cooperating with the PRC and was pursuing its own diplomatic tactics to ensure it remained an independent security actor (Hickey, 2014).

6.3.3 Taiwan and the East China Sea

Given the semi-enclosed nature of the East China Sea, conflicts and disputes over fisheries rights among stakeholders such as Japan, Taiwan, South Korea, and China are inevitable (Yeh et al., 2014). Many claims over EEZ demarcations in this area geographically overlap. There have thus been numerous cases where the Japan Coast Guard detained or even sank Taiwanese fishing vessels amid the absence of a bilateral

fisheries agreement. This led to a diplomatic standoff between the nations' respective governments in 2008, according to Ikeda, a former de facto Japanese ambassador in Taipei (Ikeda, 2010). Here, Executive Chairman Liu Zhaoxuan was questioned by a lawmaker in the Legislative Yuan regarding the possibility of deploying force against Japan, and he answered 'Buxiyizhan', which literally means 'do not hesitate to use force'. This shocked Japan (Ikeda, 2010), and the tension was only alleviated when the Japanese government apologised to Taiwan. The Ma administration then officially designated Japan as a special partner to Taiwan and highlighted the strategic importance the nation attaches to its relationship with its neighbour (Ikeda, 2010; Ogasawara, 2014). This bilateral maritime crisis between Japan and Taiwan that ensued throughout 2008 revealed the importance both sides attached to maritime security cooperation. The incident could have strained the amicable relationship between the two nations, and, as such, it was widely considered as desirable to replace the ad hoc dispute management that was in place with a more official bilateral conflict management mechanism to manage any maritime crises involving the two nations (Sun, 2013). In fact, at the beginning of the Ma administration, it was unclear whether Ma had a good grasp of the available diplomatic channels to communicate with Japan and whether he could display the leadership skills required to alleviate the attendant maritime crisis (Shimizu, 2009). For certain, Ma was not adequately prepared to de-escalate maritime tensions with Japan (Shimizu, 2009). The official bilateral conflict management system mentioned above was thus formally created after the conclusion of the bilateral fisheries agreement. The above crisis also made Japan suspicious of the possibility that the Ma administration would render Japan-Taiwan relations somewhat unstable or uncertain (Ikeda 2010). At any rate, Ma's image as an anti-Japanese figure was solidified in Japan.

While it was unlikely that Taiwan would have sought a cross-strait united front vis-à-

vis Japan over the issue of the Senkaku Islands and the PRC – especially given Taiwan's national security policy of maintaining its de facto independence from the PRC – Japan's conclusion of the fisheries agreement with Taiwan effectively erased the Japan's strategic concerns regarding such a front. As examined in the literature review, Taiwan's political input in Japan's security policy has no influence, and it is up to Japan to determine the extent to which Japan–Taiwan security relations evolve as a part of the hierarchical Japan–Taiwan relationship. Japan's concerns over a cross-strait united front was a significant determining factor in its signing of the fisheries agreement.

In fact, the agreement put an end to the PRC's strategic desire for a cross-strait united front against Japan vis-à-vis the East China Sea issue. Here, Japan secured Taiwan as a political friend in East Asia amid the rising tensions between Japan and its neighbours, including the PRC, North Korea, and South Korea. This Japan-Taiwan political and legal cooperation also helped Taiwan maintain its ties with the Japan-US alliance (Ishihara, 2013; Ko, 2013). According to Ogasawara (2014), the international image of Ma as an anti-Japanese figure was created after the publication of his famous PhD dissertation on the issue of the Senkaku Islands, while this image was also based on his proactive involvement in anti-Japanese activities along with other Chinese nationalists when he was studying international law at Harvard University. Nevertheless, despite his tendencies as a ROC Chinese nationalist, as a pragmatic legal scholar in the field of international law, he concluded that Taiwan unquestionably required the political and strategic backing of both the US and Japan for its survival. Meanwhile, a cross-strait united front with the PRC was out of the question (Ogasawara, 2014). Ogasawara (2014) argued that the Ma administration's East China Sea Peace Initiative was an embodiment of Ma's academic knowledge as a legal scholar and a way for Taiwan to maintain strong relations vis-à-vis the Japan-US alliance and, by extension,

international society (Ogasawara, 2014). Thus, Ma decided to seek marine resources that were practically accessible for Taiwan through its fisheries agreement with Japan while rhetorically continuing to insist on its sovereign claim to the Senkaku Islands. For his part, Cole (2015) argued that Taiwan has neither strong territorial ambitions regarding the Senkaku Islands within its domestic policy nor the military capability, or the intention, to invade the area. On this point, the current author agrees with these two authors. Taiwan is heavily reliant on the Japan–US alliance for its own survival and the anti-Japanese sentiment in Taiwan is both weak (include a public opinion survey to show this here or earlier in the thesis) and marginal, except among diehard fanatics. In short, Taiwan is clearly at the mercy of Japan and the US for its survival given the political sanctions placed on it and the spectre of a military invasion by the PRC.

6.3.4 Implications for the PRC's claims over Taiwan and the Senkaku Islands

As noted, Japan's signing of the fisheries agreement with Taiwan seriously antagonised the PRC, which continued to make diplomatic overtures to Taiwan to cooperate with it against Japan. For example, one Chinese news website described Ma as a 'sinner against the Chinese Race' due to his rejection of cross-strait cooperation with the PRC (*Duowei Xinwen*, 2013) and other CCP newspapers also lambasted the Japan–Taiwan fisheries agreement (Global Times 2013). Crucially, the Chinese government was displeased, stating (Reuters, 2013) that '[w]e are extremely concerned about Japan and Taiwan discussing and signing a fishing agreement'. Elsewhere, the Chinese Ministry of Foreign Affairs' spokesperson, Hong Lei, stated at a daily news briefing that '[w]e hope that Japan earnestly abides by its promises on the Taiwan issue and acts cautiously and appropriately' (Reuters, 2013). The PRC also informally questioned Japan's ownership of Okinawa (*New York Times*, 2013), claimed that the Senkaku Islands belong to Taiwan Province, which is a part of China, and that Japan and Taiwan's

cooperation on issues involving sovereignty undermined the PRC's claim to the Senkaku Islands. The PRC views 'Taiwan Province' as an indivisible part of China, so it does not allow Taiwan to conduct inter-state relations with other countries, including Japan, as an independent entity (Jakobson, 2013).

Clearly, the Japan-Taiwan fisheries agreement touched on both the PRC's territorial claims over the Senkaku Islands and Taiwan and Japan-China-Taiwan relations. Here, historical and legal analysis deserves some merit. The Japanese government continues to hold the view on the issue of the Senkaku Islands that since the islands were not incorporated into Japan under the Treaty of Shimonoseki, Japan's renouncement of its foreign territories under the San Francisco Peace Treaty did not apply to the Senkaku Islands (Hickey, 2014). For the PRC, the ROC ceased to exist in 1949 when the Nationalists were defeated in the Chinese Civil War and the KMT, which fled to Taiwan, is simply a local authority that has no right to exercise statehood (Hickey, 2014). As such, the PRC does not recognise the legal validity of the San Francisco Peace Treaty even though the ROC resorts to the treaty as legal evidence to bolster its claims on the Senkaku Islands (Hickey, 2014). Since the PRC does not recognise the existence of the ROC, it cannot recognise the legal treaties signed between it and Japan, such as the Sino-Japanese Peace Treaties. However, the bilateral Japan-Taiwan fisheries agreement is reasonably estimated to constitute a political and legal refutation of the PRC's claims on both the Senkaku Islands and Taiwan to a degree – even if the PRC would never admit this. At any rate, following the conclusion of the fisheries agreement with Japan, Taiwan is considered to be a quasi-independent political actor in relation to the issue of the East China Sea vis-à-vis the PRC.

The PRC's argument that Taiwan exploited the overtures from the PRC for its own

political gain is worthy of some attention. Japanese and Western scholars that focus on Taiwan seem to be divided on this issue. Certainly, Taiwan does appear to have behaved in a way that confuses Western observers in terms of whether the nation exploited the PRC for its own benefit (Hickey, 2014). Crucially, the PRC regarded Taiwan's signing of the fisheries agreement with Japan as a betrayal. The Global Times (2013) stated that '[t]he strong stance from the mainland side in safeguarding the sovereignty of the islands has undoubtedly strengthened Taiwan's status in its negotiations with Japan'. Even if Taiwan was interested in resolving the issue peacefully from the beginning, as Ogasawara (2013a, 2014) pointed out, a number of Japanese who were paying close attention to the Japan–Taiwan tensions were somewhat confused about Ma's acceptance of the fisheries agreement. Here, Hickey (2014) pointed out that while Ma supported the 40 or so fishing vessels with Taiwanese activists on board that were escorted by eight Taiwanese coastguard ships when they clashed with Japanese coastguard ships around the Senkaku Islands on 25 September 2012 - with both sides using water cannons against each other – the PRC deployed five patrol boats a mere 12 nautical miles away. In effect, this amounted to a de facto policy coordination between Taiwan and the PRC (Hickey, 2014). Meanwhile, Ogasawara (2013c) noted that on 24 January 2013, when Taiwanese coastguard patrol boats escorted a number of civilian vessels with Taiwanese activists on board, the PRC's three coastguard cutters approached. The Taiwanese fleet drove them away crying 'diàoyútái shì zhōnghuá mínguó lǐngtǔ, zhèlǐ shì zhōnghuá mínguó diàoyútáihǎiyù, qǐngmǎshànglíkāi (Diaoyutai is the territory of the ROC. These are the waters of the Diaoyutai ROC; please leave immediately)'. Ogasawara (2014) argued that this was a part of the Taiwanese government's noncooperation policy that it had worked out with the PRC beforehand. While determining which Taiwan expert's observation is correct is beyond the scope of this discussion, if Ogasawara's view is indeed correct, it will be linked to the paramilitary balance of power by the two respective coastguards regarding Japan–Taiwan maritime security cooperation according to the fisheries agreement. Ogasawara (2013c) pointed out that immediately before Taiwan dispatched its fleet in the 24 January 2013 case, the Ma administration had insisted it had no authorisation to prevent civilian ships from approaching the Senkaku Islands. Taiwan had informed the Japanese side about this beforehand, enabling Japanese coastguards to identify the location of the fleet and deploy Japanese patrol boats against it effectively.

However, the loudest vocal minority of Taiwanese activists was right to be concerned about the fisheries rights to secure their immediate jobs and employment opportunities. Ogasawara (2013b) expressed some concern about the possibility of the Japanese coastguard having to deal with Taiwanese protest ships and fishing vessels simultaneously with the PRC's government ships if Japan-Taiwan fisheries negotiations failed. This could have led to a cross-strait united front vis-à-vis Japan. In addition, while Taiwan is not a member of the UNCLOS, it has internalised the convention's norms and has sought peaceful resolutions to conflicts. This was recognised by Japan when it signed the fisheries agreement. Japan's engagement with Taiwan encouraged the latter to behave peacefully as if it was officially participating in international maritime security efforts. The former de facto Taiwanese ambassador to Japan, Fu-Chen Lo, argued that territorial disputes must be resolved based on international law and consultations. However, China resorted to the use of force and intimidation. Therefore, if Taiwan cooperates with China, this will not only negatively affect Taiwan's international image but will also prove disadvantageous to Japan (Ishihara, 2013). Taiwan's appreciation of international legitimacy and norms is clearly manifested in the statements made by the former ambassador. From this perspective,

the PRC generates uncertainty and risk in the international legal order.

Japan expressed reservations about the Ma administration's East China Sea Peace Initiative on the grounds that some of the 'Implementation Guidelines' were not acceptable (Taipei Times, 2012). In the current author's personal written communication with Furuya Keiji, a former state minister, Keiji stated that while he was aware of Taiwan's unique ideas regarding the East China Peace Initiative proposed by Ma, it is clear that the Senkaku Islands belong to Japan from a historical and international legal perspective. Japan owns these islands in actuality, which means the issue of territorial rights is not open for discussion. Meanwhile, while Japan has not explicitly explained what was unacceptable about the 'Implementation Guidelines', Cabestan (2014) argued that it was concerned about the proposal for the creation of a code of conduct and about trilateral negotiations between Japan, China, and Taiwan regarding the issue of the East China Sea. Meanwhile, Chief Cabinet Secretary Yoshihide Suga rejected the Ma administration's calls for trilateral talks involving the three nations (Japan Times, 2013). Cabestan (2014, p. 7) outlined three factors behind why the East and South China Sea Peace Initiative would not be effective, namely, there was '[a] piecemeal endorsement of international law and UNCLOS by some of the claimants, particularly China', there was 'a lack of common political values among the region's countries, some being authoritarian, others democratic' and there was 'a rise of nationalism in most of them'.

As will be discussed in the economic security section below, the fisheries agreement expanded the zones of access for Taiwanese fishermen to engage in fishing operations. This was considered a victory for Taiwan's national interests since it ensured the economic security of the nation's fishermen. As discussed above, unlike in the case of

the Japan–South Korea fisheries agreement – which denoted the territorial waters around Takeshima as 'middle waters' for joint management by South Korea and Japan – Japan effectively retained exclusive jurisdiction over fishing rights around the Senkaku Islands for 12 nautical miles through its fisheries agreement with Taiwan. This did not affect the territorial claims of either Japan or Taiwan (Japan–Taiwan Fisheries Agreement, 2013) and was viewed as a diplomatic victory for Japan, essentially maintaining Japan's sovereignty over the Senkaku Islands while allowing Taiwan to save face regarding its territorial claims.

Jimbo (2015) argued that Japan and the PRC used ambiguous language regarding the issue of the East China Sea. Their mutual face-saving measures made it possible to conduct an APEC meeting between Abe and Xi in 2014 where it was stated that Japan and China tacitly agreed to shelve the issue of the Senkaku Islands for the future in 1972. Taiwan and China abided in terms of the 1992 consensus, according to which both the PRC and the ROC may claim both sides of the Taiwan Straits and may represent 'China' to begin the cross-strait negotiations. This was considered to be an exemplary case of diplomatic wisdom in that it ensured both sides were dissatisfied with the diplomatic outcome. According to Article 4 of the bilateral fisheries agreement, the agreement does not affect Japan's and Taiwan's territorial positions on the issue of the East China Sea. Regarding Taiwan's claim to the Senkaku Islands, the mainstream views on Taiwan's domestic debates are divided among two main groups, one of which believes that Taiwan tacitly endorses Japan's sovereignty over the Senkaku Islands in return for fisheries rights, while the other argues that although Taiwan has certain fisheries rights, this does not affect its claim on the Senkaku Islands (Tsai, 2013). These opposing views were manifested in the remarks of two former presidents of Taiwan regarding the status of the Senkaku Islands. Here, the former president, Lee, argued that the Senkaku Islands belong to Japan (*Tokyo Foreign Press*, 2015), while President Ma reacted by stating that the Diaoyutai belongs to the ROC (*Zhongguo Shibao*, 2015). Japan's intentions behind giving Taiwanese fishermen greater oceanic space for their fishing activities in the East China Sea despite opposition from the Japan Fisheries Agency (which is supported by Okinawan fishermen) deserve some scrutiny. From the Japanese strategic perspective, the Japan–Taiwan fisheries agreement liquidated the little remaining leverage the PRC could have instrumentalised to play the role of the benevolent protector of the fishermen in Taiwan Province in the face of Japanese activity. Regarding this Japanese strategy, the former state minister, Furuya Keiji (2019), pointed out in his e-mail communication with the current author that immediately before the signing of the fisheries agreement between Japan and Taiwan in February 2013, the Ma administration revealed its strategy of non-cooperation with the PRC visà-vis the Senkaku Islands. Clearly, this strategy worked out well for Japan's security.

6.3.5 Paramilitary balance of power

Yang, the deputy secretary general of the National Security Council, stated, '[i]t is not acceptable that mainland China's fishermen operate in the applicable area. The ROC will negotiate with mainland China in this regard if necessary' (Po and Pei, 2013, p. 14). This is an essential national security development for Japan since it helps enhance the alignment of similar strategic views in bilateral terms, if not the strictly defined operational interoperability, between the Japanese and Taiwanese coastguards. Meanwhile, in terms of Japan–China–Taiwan paramilitary relations, Alex Calvo (Shigeru News Agency, 2013a) stated the following:

[w]hile Taipei is not renouncing its sovereignty claims, once Japan has agreed to let the island's fishermen operate in these disputed waters, there should be no pressing reason for the two countries' coast guards to clash again. Furthermore, the agreement provides for the setting up of a joint fisheries committee and therefore may lead to increased regular bilateral exchanges between Japan and Taiwan.

This leads to another important point. Once this source of tension between Tokyo and Taipei is eliminated, it should become much more difficult for Chinese and Taiwanese coastguard units to act, or to be seen as acting, in concert with one another. As discussed in the CSD chapter, the coastguard cooperation between Japan and Taiwan has improved, leading to a search-and-rescue agreement, and allowing actual operations around the Bonin Islands.

The PRC has been building up its paramilitary forces and strengthening its maritime law enforcement, including in terms of obtaining armed fishing vessels with ISR capabilities. It has created a nominally integrated coastguard. The PRC has been encouraging the gradual escalation of maritime security tensions by sending many of these paramilitary or governmental ships into the East China Sea. Furthermore, according to *The Strategist* (2019), on May 2018, the PLAN, the Chinese Coast Guard, and the local authorities staged joint patrols for the first time in the Paracel Islands, which are under the control of the PRC but are also claimed by both Taiwan and Vietnam. Less than two weeks later, China's coastguard blocked a Taiwanese supply ship near Kinmen in the Taiwan Strait and unilaterally cancelled its annually held maritime rescue drill with Taiwan. Crucially, the PRC's unilateral ADIZ, which covers the designated areas mentioned in the Japan–Taiwan fisheries agreement and covers Taiwan's airspace, indicates the PRC's hostile attitude vis-à-vis the Japan–Taiwan bilateral fisheries agreement. The fisheries agreement between Japan and Taiwan contributed to the strategic need for a ROC military and for the Taiwan Coast Guard to

cooperate and coordinate the strategy more closely to handle the PRC's new threat. Regarding this point, Po and Pei (2013, p. 15) stated the following:

[i]t may become a necessity for the Ministry of National Defense and Coast Guard Administration to strengthen communication, coordination and cooperation in order to develop policies to respond to diverse incidents, and especially mainland Chinese provocations, that may occur in the applicable area.

Po and Pei (2013) also noted that the PRC dispatched the PLA South China Fleet, eight surveillance ships, and more than 40 fighter jets to the Senkaku Islands on 23 April 2013. Soon after the conclusion of the Japan–Taiwan fisheries agreement, the PRC made a massive aerial military and naval paramilitary threat aimed at north Taipei and the areas in the East China Sea designated by the Japan–Taiwan bilateral fisheries agreement. Here, the authors argued that the fisheries agreement created a new national security threat to Taiwan from the PRC that stemmed from a combination of military and paramilitary forces. In other words, the threat did not just involve direct military attacks on Taiwan in view of a Chinese unification effort. However, Taiwan's political activism vis-à-vis the Senkaku Islands was weakened or, rather, disincentivised following the conclusion of its fisheries agreement with Japan.

The Japan–Taiwan bilateral fisheries agreement contributed to further divisions between the PRC and Taiwan. The PRC introduced more competent cutters into its law enforcement fleet, some of which were transferred from the PLA Navy and were immediately deployable (Goldstein, 2013). According to Goldstein (2013), the Chinese felt dismayed at the state of their law enforcement ships compared to those of the more competent Japanese coastguards around the Senkaku Islands in the East China Sea. A

Japan—Taiwan paramilitary alignment would help to alleviate maritime security threats posed by the PRC in relation to the islands. For Japan, this constitutes a grey zone scenario, and for Taiwan, it creates a new national security challenge in the northeast zone of Taipei stemming from the Sino-Japanese grey zone scenario's strategic implications for Taiwan's own defence needs. The PLA's navy, air force and coastguard operate near the Senkaku Islands, which are geographically close to northeast Taiwan, which includes Taipei and major military ports such as Keelung. This would clearly help the Chinese military and paramilitary forces to gain the strategic and operational experience that would be essential in the event of a Taiwan contingency.

China and Taiwan have only a limited and low-key mechanism for dispute resolution. However, the two sides enjoy more robust paramilitary cooperation, including in terms of coastguard drills (Glaser and Vittelo, 2015). During semi-official negotiations over cross-strait issues that were held in 2015, Taiwan and the PRC discussed the maritime communication mechanism, the law enforcement mutual judiciary mechanism, and the extradition of criminals. These issues remained within the range of the talks and appeared to be confined to low-key development (Monma, 2015). Considering the 'Sunflower' movement and the KMT debacle in the municipal election of 2014 (Huang, 2014), it is unlikely that security cooperation on both sides of the Taiwan Strait will improve. In 2019, a Taiwan Coast Guard patrol ship fired a stun grenade at Chinese fishing vessels on Taiwan's side of the strait (Taiwan News, 2019). From this perspective, regarding the maritime issues and the balance of soft power in the Taiwan Strait, it appears that Japan is not necessarily in a disadvantageous position. The Taiwan Coast Guard also launched two new 3000-class coastguard cutters in 2015 (Monma, 2015), and the nation officially proposed to enhance its maritime security cooperation with both Japan and the Philippines (Reuters, 2015).

It is unclear whether the supposedly integrated Chinese Coast Guard can conduct maritime law enforcement activities in an integrated and holistic manner (*The Diplomat*, 2015). Here, Gady quoted the following statement from Ryan D Martinson: '[w]ith new "China Coast Guard" ships entering service at regular intervals, it is easy to forget that the China Coast Guard as an organisation does not yet exist in any complete sense' (Gady, *The Diplomat*, 2015a). The Chinese have been endeavouring to enhance their maritime law enforcement capabilities by building more and larger maritime law enforcement cutters, with both the Chinese government and the military believing that their law enforcement capability at sea lags behind that of other Pacific maritime states (Goldstein, 2013). Furthermore, accidents at sea such as collisions and typhoon-related sinking are frequent among Chinese fishing vessels, which raises numerous maritime safety issues (Goldstein, 2013). Japan and Taiwan formed a naval aviation search-andrescue pact to minimise bilateral safety issues after the conclusion of the fisheries agreement (Taiwan Today, 2013). Meanwhile, the Chinese government made efforts to enhance its ISR capabilities for both fishing vessels and the satellite communication between the fishing boats and the maritime enforcement ships (Goldstein, 2013).

6.4 The Fisheries Agreement with Taiwan as Japan's Comprehensive Security Policy for Law Enforcement and Economic Security

Japan's fisheries agreement with Taiwan is in line with Japan's fisheries policy. The policy approaches maritime issues in the East China Sea from multifaceted security perspectives, ranging from national security and law enforcement to economic security. Taiwan clamours for international recognition, wishing to be seen as a quasi-legitimate actor for its own survival and for securing its ontological security. For Taiwan, attaining as much international recognition as possible is one of its most important national

security agendas, along with improving its relationships with the US, Japan, and the PRC. Japan has supplemented its security with regard to Taiwan through better law enforcement and a stable economy. These factors will help contribute to Taiwan's moving away from China and, in turn, will tip the regional balance of power in Japan's favour. Furthermore, stringent law enforcement and sustainable economic security are essential factors for measuring how consolidated the international legal order is in the East China Sea based on the Japan—Taiwan bilateral fisheries agreement.

Maritime security is not synonymous with or equal to national security. Indeed, generally speaking, national security, law enforcement, and economic security are interconnected within the realm of maritime security. That is why the existing literature related to the Japanese fisheries policy prior to the bilateral Japan—Taiwan fisheries agreement discusses the matter within the context of comprehensive security without considering the issue of geopolitical power politics in relation to international security. In view of this, this case study chapter on the fisheries agreement examines the law enforcement and economic security implications stemming from the bilateral fisheries agreement between Japan and Taiwan.

In fact, Taiwan, in the absence of statehood, is excluded from participating in international and regional institutions related to maritime security affairs, including fisheries, marine labour, maritime piracy issues, and so on (Hsu and Southerland, 2015). The Japan–Taiwan fisheries agreement gives Taiwan a unique opportunity to improve Taiwan's 'as if membership' in international maritime security efforts in terms of law enforcement and ensures Japan's economic security. Black (2014) observed that Japan has been taking on a role as an international norm entrepreneur for international and regional maritime security in international society thanks to its innovative

instrumentalisation of the Japanese coastguard. Similarly, the nation is instrumentalising the non-state actor, Taiwan, to enhance maritime security.

6.4.1 Law enforcement

Chinese fishermen who are observed to be conducting fishing activities without proper legal procedures and permissions in Taiwan's seas are considered to be operating illegally. Taiwan has a tacit de facto middle line with the PRC in the Taiwan Strait that geographically divides the ocean in two. Chinese fishermen who illegally engage in marine activities such as fishing on the Taiwanese side of the strait are considered violators of Taiwanese marine interests. Due to the tacit nature of the line, the Japan–Taiwan fisheries agreement is considered a de facto acknowledgement of Taiwan as a quasi-independent political actor.

Taiwan's ability to defend its coastal waters has been jeopardised by the intrusion of illegal Chinese boats amid the absence of a fisheries agreement with the PRC. The ROC's Act Governing Relations between Peoples of the Taiwan Area and the Mainland Area determines the management of the aforementioned line (Tseng and Ou, 2010). If PRC vessels attempt to illegally intrude into the Taiwan side of the water, the Taiwanese coastguard will deal with them. The problem is that due to the geographical proximity of China and Taiwan in the Taiwan Strait – with the narrowest point of the strait being only 80 miles long – and given the wide availability of economically important species of fish within the Taiwanese waters, many illegal Chinese maritime activities are conducted on the Taiwanese side (Lin et al., 2013).

According to the Taiwanese Coast Guard Administration (2017), the nation's coastguard encountered 1,376 Chinese fishing vessels that had illegally intruded into Taiwan's waters in 2016 alone. Clearly then, a significant number of illegal Chinese

fishing vessels trespass into the Taiwanese ocean. However, due to the sheer number, the Taiwanese coastguard simply cannot deal with all of them (Lin et al., 2013). Furthermore, smugglers exploit the short geographical proximity of the two nations to transport illegal products that are in short supply in Taiwan, and often bring weapons with them, which negatively affects the security on both sides of the Taiwan Strait (Ou and Tseng, 2010).

While Japan and Taiwan have established conflict and dispute management mechanisms such as the bilateral fisheries committee, the PRC and Taiwan have not. Japan and Taiwan also have an MOU agreement related to anti-human trafficking (*Focus Taiwan*, 2015), while, as discussed in the CSD chapter, the two nations also signed a maritime search-and-rescue agreement. Furthermore, intelligence cooperation on coastguard exchanges do exist. For example, as Fukuda (2017) noted, the Taiwanese coastguard deployed coastguard officers to Taiwan's representative office in Okinawa to cooperate with the Japanese coastguard in view of resolving any fishing conflicts, conducting search-and-rescue operations, and engaging in other types of maritime security cooperation. Fukuda (2017) also pointed out that the Japanese coastguard deployed officers to the Japan–Taiwan Exchange Association in Taipei.

Regarding the protection of Taiwanese fishermen's rights, the Japan–Taiwan fisheries agreement states the following:

ROC fishing vessels are now able to operate within the designated zone without interference by Japanese government vessels. They will be overseen by the ROC Fisheries Agency in accordance with relevant ROC fisheries laws and regulations (The Taiwan–Japan Fisheries Agreement, ROC, 2013).

Before the bilateral fisheries agreement between Japan and Taiwan was concluded, Japanese inspection vessels and patrol boats were infamous in Taiwan for their harsh treatment of Taiwanese fishermen on the Japanese side of the EEZ (Yeh et al., 2014). On 6 June 2005, around 50 Taiwanese fishing vessels staged a marine demonstration in the East China Sea against this harsh treatment from the Japanese coastguard. One ruling party lawmaker from the DPP even described this an 'act of piracy' (Taiwan Info, 2005), while the famously pro-Japanese figure, Lee Teng Hui, commented on the Japanese treatment of the Taiwanese fishermen while discussing the change in Japan's uncompromising attitude towards Taiwan, the China factor, and the huge donations Taiwan made to Japan in 2011 immediately after the conclusion of the bilateral fisheries agreement (Ishihara, 2013). Reportedly, on 13 September 2013, shortly after the conclusion of the fisheries agreement, Japanese fishing boats were being rammed into by Taiwanese fishing vessels, at a time when neither side had to worry about intervention from the respective coastguards of Japan and Taiwan (Yaeyama Mainichi Shimbun, 2013). Clearly, more rules-based fishing activities in the East China Sea are critical for the sustainable maintenance of the Japan-Taiwan fisheries agreement.

The Sino-Taiwanese fisheries relations lack regulation and management due to the lack of a high-profile cross-strait fisheries agreement. In fact, only minor punishments can be meted out for illegal Chinese fishing activities conducted on the Taiwanese side of the strait (Lin et al., 2013). Illegal Chinese fishermen conduct cost-benefit analyses and conclude that these punishments (which include fines and confiscation of the fish) are minor in comparison to the greater economic benefits of conducting illegal fishing activities in Taiwanese waters (Lin et al., 2013). Often, these illegal fishermen exploit the notoriously bad weather in the Taiwan Strait to reduce the risk of being caught and even resort to violence against Taiwanese patrol boats (Lin et al., 2013). For example,

The Diplomat (2017) reported how, in January 2017, the captain of an illegal Chinese fishing vessel was wounded while resisting inspection by the Taiwanese coastguard, and how, in February of the same year, a Chinese fishing vessel intentionally rammed a Taiwanese coastguard ship.

The more stringent the punishments meted out by the Taiwanese coastguard, the more the illegal fishermen resist. Some illegal fishermen from the PRC even resort to criminally dangerous physical measures such as the use of weapons, while others attempt to take patrol crewmembers as hostages (Lin et al., 2013). In August 2015, a high-profile maritime case emerged involving illegal Chinese fishermen, who captured a number of Taiwanese coastguards after they came onboard for inspection. With the hostages onboard, the Chinese fishing vessels attempted to escape to the PRC (*Taipei Times*, 2015). Three Taiwanese patrol boats with specially trained and heavily armed coastguard crew members mounted a commando-style rescue operation to save the captured Taiwanese crew and arrest the fishermen (*Taipei Times*, 2015).

Notwithstanding the significant amount of work it will involve to conduct maritime law enforcement with limited personnel, the Taiwanese coastguard must make better use of its resources to conduct effective deterrence operations (Lin et al., 2013). Now that the Japanese and Taiwanese can engage in joint legal actions and the two nations signed the agreement on search and rescue by the respective coastguards, it remains to be seen whether this will have a positive impact on enhancing law enforcement in the East China Sea. For certain, Japan–Taiwan maritime security cooperation will contribute to maintaining the legal order in the area, especially given the fact that Taiwan is interested in further coastguard cooperation with Japan and the Philippines (Reuters, 2015).

While Taiwanese fishermen were concerned about the possibility of more Japanese

crackdowns on non-Japanese fishing activities in the East China Sea (Jakobson, 2013), the conclusion of the Japan–Taiwan fisheries agreement effectively eliminated this concern. However, from Okinawa's perspective, the agreement was biased towards Taiwan. For example, since Okinawan fishing boats are smaller both in terms of size and the number of crew members on board, this disadvantages Okinawan fishermen and forces them to retreat from encounters with Taiwanese fishermen in the East China Sea (Jakobson, 2013). According to Takuya Takahashi (*Yaeyama Mainichi Shimbun*, 2013), a member of the fishermen's union from Ishigaki Island in Okinawa, Taiwanese fishermen openly engage in longline fishing to the north of Ishigaki Island. This negatively affects Okinawan fishermen from Yaeyama Islands. As such, Takahashi expected unruly Taiwanese fishermen to trespass into the waters to the south of Ishigaki, which are not covered by the Japan–Taiwan fisheries agreement.

6.4.2 Economic security for Taiwan and economic insecurity for Okinawa: Implications for power politics in Japan–Taiwan–China triangle

Japan's bilateral fisheries agreement with Taiwan suggests that it recognises its neighbour's marine economic rights in the East China Sea. This ensures that Taiwan is an important economic and fishing actor in the area – albeit at the expense of the economic and fishing rights of Okinawan fishermen. As examined in the literature review and the ICAO chapters, economic security is an important component of the balance of power in the ESIR paradigm.

From the economic security perspective, the fishing industry has made limited contributions to the overall economy of Taiwan, especially in comparison with the hydrophone or oil industry in the East China Sea. However, it is essential to secure jobs for the fishermen and to stabilise the coastal economy in a sustainable manner. This also has political implications since it keeps Diaoyutai political activism to a minimum. The

most vocal of these activists are largely confined to the fishing community in Yilan County in Northern Taiwan, which is geographically closer to the Senkaku Islands. Local politicians with sympathy for the fishing industry will likely get maximum votes in this region (Ogasawara, 2012a). Furthermore, some Diaoyutai activists in Yilan County who were onboard the fishing vessels that attempted to get close to the Senkaku Islands displayed the following protest slogan: 'Defend the Fishing Rights to Survive!' (*Focus Taiwan*, 2012). The Japan–Taiwan bilateral fisheries agreement enhanced the economic security of the Taiwanese fishermen and alleviated the risk of a Sino-Taiwanese united front against Japan regarding the issue of the Senkaku Islands.

Taiwanese coastguard ships are equipped with reinforced equipment, and they must patrol the greater area in the face of the significant presence of illegal Chinese fishing vessels. The destructive fishing activities of illegal Chinese fishermen have been posing serious threats to Taiwanese marine living resources and making the life of Taiwanese fishermen financially and physically more difficult and more dangerous (Lin et al., 2013). One of the major reasons why many Chinese fishermen trespass into the Taiwanese side of the Taiwan Strait is that China's littoral seas have been experiencing a dangerously severe depletion of their overall fish stocks, and Chinese fisheries experts are highly pessimistic about the future prospects regarding this issue (Goldstein, 2013). If the two zones on the Japanese side of the new demarcation line in the East China Sea — which the Taiwanese have long argued is their traditional fishing area — are more abundant in fish stocks (Jakobson, 2013), it is quite reasonable for Taiwan to expect major commercial benefits from its fisheries deal with Japan. This may have motivated the Taiwanese government to negotiate a fisheries agreement with Japan in the first place.

However, the fisheries agreement has markedly antagonised the Okinawans. For example, *Yaeyama Mainichi Shimbun*, a local newspaper in the Yaeyama Islands in Okinawa (2013), expressed the view that the central government in mainland Japan prioritised more than 800 Taiwanese fishing fleets over the dozens of Okinawan fishing fleets while ignoring the views of the local Okinawans. This destroyed the traditional ties between Taiwan and Okinawa, even though Okinawa was ready to make certain concessions to Taiwan considering the geostrategic importance of accommodating Taiwan into Japan's strategy vis-à-vis the PRC over the issue of the East China Sea. *Ryukyu Shimpo* (2013) also expressed the view that while the fisheries agreement negatively affected China—Taiwan relations, Okinawa was forced to pay the price.

One of the Taiwanese negotiators for the fisheries agreement with Japan, Chú xiàn chāng – who is First Secretary at the Taipei Economic and Cultural Presentative Office in Japan (2013) – argued that among the numerous reasons that it took 17 years to conclude a fisheries agreement with Japan, the most important one was the fact that the Fisheries Agency had been unable to convince the fishermen in Okinawa of the value of the agreement. The Okinawan fishermen had long opposed the signing of a fisheries agreement with Taiwan in relation to seas they considered to be their own (Jakobson, 2013). In fact, the bilateral Japan–Taiwan fisheries agreement gave access to the Taiwanese fishermen to the Japanese EEZ, where the Okinawan fishermen had had a monopoly on fishing activities and this consequently affected their economic interests (Smith, 2013). The local Okinawan authorities and the Okinawan fishermen considered the swiftly concluded fisheries agreement between Japan and Taiwan to be an unjustifiable act that negatively affected their economic and daily activities in the name of diplomatic victory (Smith, 2013).

The former state minister Furuya Keiji (2019) pointed out that Prime Minister Abe responded to questions raised in the budgetary committee in the Upper House on 23 May 2013 as follows:

[t]he effective signing of the fisheries agreement, which has been a traditional irritant in the friendly relations between Japan and Taiwan, will achieve significant progress in the regional security environment in Asia. Although we must make further efforts to gain understanding from the people in Okinawa, Prime Minister Abe recognises it (the fisheries agreement) was a historic agreement. Furthermore, prior to the signing of the fisheries agreement in February 2013, Taiwan expressed its position of non-cooperation with China vis-à-vis the Senkaku Islands. Based on these developments, the signing of the fisheries agreement was concluded.

Keiji (2019) also stated that the Japanese government must make efforts to obtain the understanding of Okinawan fishermen, thus, it does not believe that it is more important to ensure that China and Taiwan do not jeopardise Japan's security than to defend the interests of Okinawa.

Nonetheless, the fifth Japan–Taiwan meeting on the fisheries issue in 2016 received significant attention from Okinawan fishermen, who wanted the fisheries agreement amended. Many of them were dissatisfied with the economic damage inflicted by their reduced access to fishing activities in the seas to the north of Yaeyama Islands (*Taipei Times*, 2016d).

6.5 Normative Factors that Facilitated the Japan-Taiwan Fisheries Agreement despite the Tensions in the East China Sea

With reference to the ESIR, this case study chapter now examines four conceptual factors, the historical, cultural, democratic, and political dynamics factors, in Japan—Taiwan relations in international society to explain why the high-profile fisheries agreement was concluded despite the increasing tensions. Examining the normative factors that contributed to the conclusion of the agreement will help to elucidate the fact that Japan—Taiwan bilateral ties are increasingly being governed by the unique dynamics of each nation and that both are creating a unique space that will help balance the emerging regional security environment in relation to the PRC. The normative developments surrounding Japan—Taiwan relations on both the grassroots and governmental levels are encouraging both nations to create stronger ties between them in view of their respective complicated relationships with the PRC. This is beneficial for Japan's power politics vis-à-vis the PRC in the East China Sea.

6.5.1 Different historical interpretations of Japanese imperialism

The different historical interpretations of Japanese imperialism are a contributory factor in the significant differences between China and Taiwan's contemporary ties with Japan as well as in Sino-Taiwanese tensions, as was discussed in Chapter 3. However, according to the ESIR, the more the post-colonial populations attempt to gain recognition from the post-imperialists, the more this consolidates the hierarchical relations between the two groups. This also applies to the Japan–Taiwan relationship in regional settings.

In Taiwan, the perspectives regarding the Japanese colonisation of Taiwan are diverse. Some groups argue that while Taiwan has suffered from Japan's colonisation, the nation has also become more modernised itself. This is why the Taiwanese did not engage in high-profile anti-Japanese activities or boycotts in the post-colonial period (Yeh et al., 2014), which is in sharp contrast to the PRC's historical antagonism of Japan. As discussed in the literature review, Taiwan's unique post-colonial interpretation of Japanese colonial rule will help to improve Japan—Taiwan relations and will help create some momentum for the social and political forces in Taiwan that favour amicable relations with Japan. While the political links between Japan's historical legacy and the conclusion of the bilateral fisheries agreement are difficult to ascertain in definitive terms, Taiwan's colonial links with Japan is one of the facilitators of the former's attempts to enhance its ties with the latter rather than the PRC. This post-colonial positive sentiment vis-à-vis Japan has created a domestic mechanism in Taiwan that has consolidated Taiwan's ideationally subordinate position vis-à-vis 'the civilised Japan' as opposed to alongside 'uncivilised others' such as China in the hierarchical regional order.

After the end of WWII, having witnessed the activities of the violent and corrupt KMT oppressors, the Taiwanese were disillusioned with the party's rule (Dreyer, 2008) and did not feel an emotional connection with China. They felt abandoned by the Qing Empire (Dreyer, 2008) and a sense of nostalgia for Japanese colonial rule (Dreyer 2008; Sun, 2013). It is common for Japanese people to encounter elderly Taiwanese people who speak Japanese and have positive memories of the Japanese colonial era in Taiwan. In fact, Sun (2013) pointed out that in Tsai Ching-tang's study, 94.47% of the 205 elderly Taiwanese participants who had received a Japanese education in the Japanese colonial era described their Japanese teachers as either 'good' or 'quite good'.

Indeed, a politicised Taiwanese imagination of the Japanese as the 'positive other' exists

in Taiwanese domestic politics. For example, Wakabayashi (2015) argued that the rise of China as the 'negative other' functioned, for Taiwanese nationalists, as a political restraining factor that minimised the otherness of the Japanese as past occupiers, adding that this phenomenon corresponds with the image of Japan as a model advanced country that Taiwan should aspire to emulate. It can be stated here that many Taiwanese nationalists continue to instrumentalise their internalised Japanese historical and cultural influences to reduce Chinese historical and cultural ownership of Taiwan for their own political benefit. Taiwanese historical imagination regarding Japanese imperialism renders the mainstream Taiwanese indifferent or unsympathetic to the greater Chinese nationalism that the PRC instrumentalises to appeal to its Taiwan compatriots. In fact, this appeal is, in reality, a political call for a cross-strait united front against Japan in the East China Sea.

6.5.2 Cultural dynamics

According to the ESIR, an international society that is marked by shared normativity and values, including in terms of culture, will likely experience successful consolidation. This also applies to Japan–Taiwan relations. The following sections conduct a close examination of the cultural dimension of the Japan–Taiwan relations, which helped to facilitate the Japan–Taiwan fisheries agreement. Regarding the inter-relationship between cultural exchanges and their effect on diplomacy, Sun (2013, p. 132) stated that '[c]ultural exchanges can either be applauded for their vibrancy or lamented for their irrelevance'.

As a unique aspect of Japan–Taiwan relations, cultural factors – including both high-cultural and sub-cultural exchanges – can be critical to enhancing bilateral ties. For certain, as noted, the cultural dimension of Japan–Taiwan relations contributed to the conclusion of the bilateral Japan–Taiwan fisheries agreement. The mutually positive

perceptions of the two nations (MOFA, 2013) that stem from their socio-cultural exchanges are essential to enhancing their bilateral ties. For one, they have contributed to the strong foundations of their respective societies and effectively prevented Japan—Taiwan relations from being negatively affected by the tensions in the East China Sea following the nationalisation of the Senkaku Islands by the Japanese government. For example, the former Taiwanese de facto ambassador, Shen Ssu-tsun (Taipei Economic and Cultural Representative Office, 2012), emphasised Japan—Taiwan high-cultural ties such as the performance of the Takarazuka in Taiwan and the artefact exhibition at the National Palace Museum in Japan. He also called for the commencement of fisheries-related negotiations with Japan and for a declaration of Taiwan's intention of non-cooperation with the PRC in their conflict with Japan vis-à-vis the issue of the East China Sea.

Both Japan and Taiwan have undertaken numerous measures to enhance the sociocultural exchanges between the two sides. This has improved Japan–Taiwan relations
at the civilian and grassroots levels and ensured the two countries support each other
during natural disasters (Sakashi, 2014). For example, when the Great East Japan
Earthquake hit Japan in 2011, Taiwan's emotional support and huge financial donation
(the largest in the world on a per capita basis) significantly boosted the positive image
of Taiwan among the Japanese (Japan MOFA, 2013). However, the Taiwanese rescue
team was politically pressured into waiting for the arrival of the PRC's rescue team
during the initial stages of the disaster, which outraged the Taiwanese population
(Sakashi, 2014). On the first anniversary of the earthquake, the Japanese government
did not invite a Taiwanese representative to the commemorative event out of
consideration for the PRC. The Japanese mass media and public expressed its
resentment and protested, which subsequently put political pressure on the Noda

administration to change its Taiwan policy (Ogasawara, 2014). The Japanese government decided it no longer saw the need to consider the demands of the PRC regarding Taiwan given Taiwan's significant humanitarian contributions to the nation. In September 2012, Prime Minister Noda issued an open apology to Taiwan for not inviting it to the aforementioned event, while Japan's de facto embassy in Taipei published a half-page thank you note in every major newspaper in Taiwan (*Taipei Times*, 2013). A month later, the Emperor of Japan invited Taiwanese representatives to Tokyo for the first time (Ogasawara, 2014).

On the second anniversary of the Great East Japan Earthquake, a Taiwanese representative was invited. The representatives gave flowers and expressed their condolences to the victims (Ishihara, 2013). According to Ishihara (2013), *Ziyou Shibao* argued that these representatives attended the ceremony as an official Taiwanese delegation that was on an equal footing with the delegations from other countries. This antagonised the PRC's government, which chose to boycott the event (Ishihara, 2013). Subsequently, due to the rising tensions in the East China Sea and the presence of a Taiwanese representative at the third anniversary of the earthquake, the PRC ambassador boycotted the event along with South Korea. Regarding this development, the Chinese spokesman at MOFA stated the following:

[w]e believe that Japan's behaviour has violated the principles and spirit of the China-Japan Joint Statement and the commitments of the Japanese side...We urge Japan to match its words and deeds regarding the issue of Taiwan, to honour its commitment and not play games (*Taipei Times*, 2013).

It is no great leap to presume that the Japanese people detected a sharp contrast between Taiwan and the PRC. At the third anniversary event, Prime Minister Abe expressed his gratitude to Taiwan and asked the Taiwanese representative to present commemorative flowers at the event on his personal Facebook page. This Facebook post received the largest number of 'likes' on record, indicating the Japanese people's positive perception of Taiwan (Ogasawara, 2014).

According to the Taiwan–Japan fisheries agreement document archived in the ROC MOFA website, the emotional and financial support the Taiwanese offered Japan contributed to improving Japan–Taiwan relations and facilitated the eventual conclusion of the fisheries agreement. According to the document,

[t]he ROC government sees relations with Japan as a special partnership. The two countries enjoy close exchanges at all levels and have built strong trust and friendship. Following the massive earthquake that hit eastern Japan on March 11, 2011, donations came from all sectors of Taiwan, for which Japan remains grateful today. In his congratulatory message on the 40th anniversary of the interchange association, Japan, Japanese Minister for Foreign Affairs Fumio Kishida said that Taiwan and Japan are important partners, and a valuable and solid friendship exists between the two countries (ROC MOFA: The Japan–Taiwan Fisheries Agreement 2013, p. 2).

Since Japan does not enjoy friendly relations with its immediate neighbours, the PRC, South Korea, and North Korea, it is essential that it maintains its amicable relationship with Taiwan. Many people in Japan and Taiwan, including both elites and ordinary citizens, believe that important Japan—Taiwan sociocultural ties should not be ruined by the maritime tensions in the East China Sea. This attitude helped pave the way for the bilateral fisheries agreement (Ogasawara, 2014).

However, the grassroots contact between the PRC and Taiwan is not achieving the political goals the PRC envisioned in terms of its cross-strait civilian exchanges. In fact, the greater the number of cross-strait grassroots and civilian exchanges, the more the Taiwanese perceive the socio-cultural and political differences between themselves and China. According to Rowen (2014c), Taiwanese citizens were likely to feel more alienated by their interactions with PRC citizens after the legal restrictions on crossstrait tourism were relaxed thanks to Ma's greater engagement with the PRC. After the Ma administration took office in 2008, the expected economic boom due to Chinese tourism was meant to change the negative perceptions of Chinese mainlanders and create political forces that would favour closer ties between Taiwan and the PRC. However, this merely resulted in Taiwanese people realising just how much they differed from the Chinese in cultural terms. The huge presence of the Chinese also led to a social and cultural sense of alienation among the Taiwanese (Rowen, 2014c). Meanwhile, in the PRC, the political views of Taiwanese celebrities are highly contested, which often poses a political risk to the celebrities and also further alienates the Taiwanese people (Sun, 2013). In fact, Taiwanese celebrities harbouring political inclinations that the PRC deems as unacceptable are often ostracised, which leads to further resentment among the Taiwanese (Sun, 2013). For example, the 16-year-old Taiwanese K-pop singer, Chou Tzu-yu, had to make an online video apologising to the Chinese public for waving the ROC flag and the South Korean national flag. This led to outrage in Taiwan and caused President-elect Tsai to intervene (New York Times, 2016). Moreover, soon after Taiwan's landmark legalisation of the first gay marriage in Asia in 2019, Taiwan's MOFA was infuriated by the *People's Daily*, a PRC publication for taking credit for Taiwan's achievement, claiming that the local government of Taiwan, China, was the first to legalise same-sex marriage in Asia, which prompted the

ROC MOFA to tweet '<u>@PDChina</u> is a commie brainwasher & it sucks' (Taiwan MOFA Twitter, 2019).

6.5.2.1 Taiwan's evolving views on Japan and its positive implications for Japan–Taiwan relations

Amako (2013) discussed the changing nature of a pro-Japanese Taiwan in relation to the issue of the East China Sea before and after the conclusion of the bilateral Japan-Taiwan fisheries agreement. Elsewhere, Sun (2013) pointed out that while Taiwan is the friendliest of Japan's immediate neighbours, part of this friendliness could perhaps be attributed to Taiwan's partisan politics. Here, we must consider the emerging pro-Japanese sentiment among young people in Taiwan, who have witnessed unique cultural interactions between the two nations (e.g. the mutual support during natural disasters) and who have grown up with Japanese popular culture. This pro-Japanese sentiment has a historical continuity with the Japanese colonial influence. One critical social event in Taiwan that played out in 2014 was a manifestation of its youth's positive sentiments towards Japan and ambivalent feelings towards the PRC. The so-called 'Sunflower Movement' ensued throughout much of 2014, and numerous Taiwanese students took to occupying the Legislative Yuan. The Japanese mass media covered the event more excessively than any other country apart from Taiwan itself. The students were discontented with the perceived overly rapid development of cross-strait economic relations and the perceived undemocratic decision-making by the KMT regarding the cross-strait service trade agreement with the PRC. These activities had been conducted without engaging in appropriate consultations with those who could potentially be negatively affected. The students watched a pro-Japanese film entitled Kano (2014) on a large screen at the Legislative Yuan to combat the fatigue of their long occupation (Taipei Times, 2014). Kano is about Taiwanese and aboriginal teenagers who form part of a youth baseball team playing under the supervision of a

Japanese coach and which makes it to the final of the Koshien tournament during the Japanese rule of Taiwan. The film is based on historical events and has often been described as beautifying Japanese colonial rule – or at least depicting it in positive terms.

Some of the old KMT mainlanders who fled to Taiwan following their total defeat in the civil war against the CCP may harbour anti-Japanese sentiments, as manifested in the recent case of the anti-Japanese military parade in Taiwan (Nojima, 2015). However, they are a fairly marginal group given their lack of numbers and the fact that they are rapidly aging. Furthermore, the KMT has always struggled to obtain international and domestic credit for its contribution to the effort against the Japanese invasion of mainland China. In fact, the authority has been largely discredited by the PRC, whose CCP has monopolised the historical narrative and demonised or ignored the KMT's contributions. Under Japanese rule, Taiwan was mobilised as a part of the Japanese war machine. Here, the Taiwanese fought for the Japanese colonisers (Nojima, 2015). Nojima (2015) also pointed out that for most Taiwanese people, their conception of an anti-Japanese struggle lies in the anti-Japanese colonialist movement in Taiwan, as manifested in the internationally acclaimed 2011 Taiwanese movie, Warriors of the Rainbow: Seedig Bale. Therefore, the anti-Japanese sentiment in Taiwan is conceptually different from that of the 'war of resistance against Japanese aggression' sentiment that prevails in the PRC.

Matsuda (2009) argued that the democratisation of Taiwan resulted in diverse perceptions of Japan among the Taiwanese people, which includes both positive and negative views of Japan's historical legacy. In fact, the democratisation complicated Japan–Taiwan relations from the 1990s onwards. Nevertheless, Japan is the most favoured country among the people in Taiwan (MOFA, 2013) and their shared

democratic values helped enhance their ties and resolve their tensions in peaceful terms. To generalise the Taiwanese mainlanders (*Waishengren*) simply as anti-Japanese may risk ignoring the complexity of the issue. Shimizu (2009) stated that even among the mainlander groups in Taiwan, the perceptions of Japan are fairly varied due to the generational factors. Elsewhere, Yeh (2015) pointed out that even among the mainlanders that arrived in Taiwan after 1949, the new generations tended to associate themselves with daily life in Taiwan as opposed to the Second Sino-Japanese War. For certain, Taiwanese domestic politics has no influence in Japan's accommodation of Taiwan into international society, but the hierarchical relationships between the post-colonial people of Taiwan and the post-imperialist people of Japan should be closely observed in the long term. Taiwan's domestic cultural development favours external policies that facilitate stable ties with Japan, rather than enhance the PRC's cross-strait campaign against Japan.

6.5.3 Democratic identity and democratic peace theory

Democracy is now recognised as a vital component of the normative criteria for civilisation in international society (Stivachtis, 2015). Taiwan shares this view with Japan. As discussed in the previous chapters, obtaining international respect and dignity is essential to Taiwan's national identity since, by extension, this will help the nation gain international legitimacy and national security. While the PRC cannot grant Taiwan a sense of dignity and respect due to the latter's democratic structure, Japan certainly can. For example, the former Taiwanese de facto ambassador, Shen Ssu-tsun (Taipei Economic and Cultural Representative Office, 2012), emphasised the value of shared democracy, freedom, and human rights with Japan and discussed the Ma administration's intention to recommence negotiations on fisheries rights with Japan to improve Japan—Taiwan relations, while he announced Taipei's intention not to

cooperate with the PRC vis-à-vis the issue of the East China Sea.

The PRC has limited political imagination, and the best and only political deal it can make with Taiwan regarding the final resolution of Taiwan's status is that of 'One country, two systems'. This deal has been completely rejected by all the political spectrums of Taiwanese domestic politics (Sun, 2013). Taiwan's Sunflower Movement and Hong Kong's Umbrella Movement were mutually inspiring due to the shared social, political, and cultural concerns and fears vis-à-vis Chinese influence over the two societies (Rowen, 2014a, 2014b). Thus, it is highly unlikely that the people of Taiwan will change their negative perception of the decidedly undemocratic 'One country, two systems' deal. The aforementioned new national security law of the PRC, which obliges Taiwan and Hong Kong to defend the national integrity and national sovereignty of the PRC and regards the Taiwan Independence Force as a grave national security threat visà-vis the PRC, will not help alleviate Taiwanese resentment towards the PRC.

During the 1980s, the authoritarian regime that ruled Taiwan under the KMT and which had a brutal human rights record, was transformed into a democratic regime. This helped to change the Japanese perception of Taiwan in a positive way, unlike the Tiananmen Massacre in the PRC, which unquestionably shocked the whole of Japan (Takagi, 2006). Taiwan also used its democratic identity to forge ties with Japan (Takagi, 2006; Matsuda, 2009; Sun, 2013). As indicated by pro-Japanese figures such as Lee Teng Hui, Taiwan's democratisation made it possible to revive long-suppressed, yet widespread pro-Japanese sentiments under the authoritarian dictatorship of the KMT among the Taiwanese populace (Sun, 2013). Following the diplomatic de-recognition of the ROC, the Japanese felt a sense of guilt vis-à-vis Taiwan, but the Tiananmen Massacre gave the Japanese greater confidence and the political space to enhance ties

with Taiwan (Matsuda, 2009). Furthermore, in sharp contrast to the PRC, Taiwan's democratisation expanded the political support base for Taiwan across the political spectrums of Japanese domestic politics; this support was no longer just confined to traditional conservative camps (Sun, 2013).

Regarding the strategic importance of shared democracy for Japan–Taiwan relations vis-à-vis the PRC's brutal human rights record, Sun (2013, p. 139) stated the following:

[w]hile the security discourse strikes Japanese nerves about the consequences of a Chinese invasion, the democratic discourses bring Japan and Taiwan closer by highlighting a value-based common identity – that is, Japan is no longer the lone democracy in the region but has been joined by a democratic, gracious, and friendly Taiwan.

Furthermore, despite the fact that the Sino-Japanese trade volume has long surpassed that of Japan and Taiwan, the PRC failed to coerce Japan into capitulating to the PRC's demands on the Taiwan issue. Japan's strong resistance indicated its increasingly rigid attitude vis-à-vis the PRC's use of its rising material capability and power for coercion (Sun, 2013). In the post-Cold War era, with Japan's normalisation underway, Japan began to make a political commitment to promote Western democracy and the Western concept of human rights in a more pronounced manner in terms of its own foreign policy agenda (Takagi, 2006). The current Abe administration places a strong emphasis on 'upholding universal values such as freedom, democracy, respect of fundamental human rights, and the rule of law' (Japan MOFA Diplomatic Blue Book, 2014). Thus, Taiwan's and Japan's shared political and democratic values have a more binding effect on bilateral Japan—Taiwan relations. The conclusion of the bilateral Japan—Taiwan fisheries agreement fits into this geopolitical development within Japan—China—Taiwan

relations. Here, Japan and Taiwan can instrumentalise civilised democratic measures to communicate with each other and resolve maritime tensions peacefully, while, the undemocratic PRC can only resort to violence, including in terms of violent anti-Japanese demonstrations.

Regarding this point, Sun (2013, p. 136) stated the following:

[t]he Japan-Taiwan case shows that for lesser powers, knowing their place does not mean being totally powerless. A shrewd use of soft power factors such as identity and culture values may enhance weaker countries' presences in international politics.

While it is debatable whether Japan is a weaker power given its military and economic might, the above statement captures the essence of how Japan–Taiwan relations are governed and flourishing under their own dynamics, including the shared democratic identities and cultures.

6.5.4 Political dimension

Black (2014) pointed out that what is generally less discussed in ESIR is that many nongreat power states must undergo significant domestic reforms in accordance with international norms in order to be accommodated into international society. Meanwhile, Singh (2013b) argued that in the post-Cold War era, Japan's domestic political and legal decision-making bodies became equipped to effectively tackle regional and international security affairs as Japan's security identity transitioned from that of a peace state to an internationalist state. In studying the case of the bilateral fisheries agreement, this chapter demonstrated how Japan's national security decision-making body on the Japanese domestic political level has evolved vis-à-vis the issue of the East China Sea and how this elevated Japan's status within international society. Singh (2013b) pointed out that in the post-Cold War era, the domestic political decision-making power in terms of foreign and security policy shifted to Prime Minister Kantei and the cabinet following Japan's normalisation from traditional bureaucratic approaches. Japan's contribution to regional and international security affairs faced fewer barriers in relation to domestic leadership decision-making. The fisheries agreement between Japan and Taiwan reflected these changes in the nature of the domestic political leadership structure in Japan.

Japan needs strong political and decision-making will in its leadership to enhance its political, strategic, and security ties with Taiwan while navigating any tensions with the PRC. Japan's conclusion of the bilateral fisheries agreement with Taiwan was deemed as signalling a change in the status quo in Japan–China–Taiwan relations by the PRC. This change challenged the complicated structure of this triangulated relationship. Domestically, despite strong domestic opposition from the Okinawan fishermen and the local Okinawan authorities, Abe and his political allies had to make the difficult decision to finalise the agreement with Taiwan. In short, Japan's geo-strategic compulsions encouraged it to expedite the conclusion of the bilateral fisheries agreement with Taiwan after 17 years of negotiation. Meanwhile, Japan's major economic relations with the PRC did not force the nation into making greater political concessions regarding the Taiwan issue (Sun, 2013). This strong leadership is a concrete example of the normalisation effect on the Japanese domestic political decision-making process. MOFA continues to interact with Taiwan at the Kachou level and has not created an independent Taiwan division, with any issues related to Taiwan resolved under the supervision of the China division (Sun, 2013). Thus, it was unlikely that MOFA would play a leadership role in the conclusion of the bilateral fisheries agreement. Both Kawashima (2013) and Sakashi (2014) suggested that there was a high likelihood that Prime Minister Abe, a pro-Taiwan individual, would take on a leadership role within the Kantei diplomacy and would make a final decision regarding the Japan–Taiwan bilateral fisheries agreement.

Even under the DPJ-led administration, Japan's political leadership's unwillingness to yield to the PRC over Taiwan was evident. Following the nationalisation of the Senkaku Islands by the Japanese government, the Japan–China–Taiwan triangulated relationship experienced certain maritime security tensions. Koichiro Gemba, a former foreign minister under the DPJ administration, published a letter to the people of Taiwan on the homepage of the Interchange Association in Taipei, which expressed Japan's desire to reopen negotiations on the bilateral fisheries agreement with Taiwan (Interchange Association, 2012). From the conclusion of the bilateral fisheries agreement, it is clear that regardless of which political party is in power in Japan, Japan will likely favour strong ties with Taiwan (Sakashi, 2014). For example, despite the fact that the DPJ had fewer Nikka-Kon members in the DPJ-led cabinet compared with 11 ministers from the Nikka-Kon in the LDP-led Aso administration, the two incoming chief cabinet secretaries in the DPJ administration, Sengoku Yoshito and Edano Yukio - both of whom were powerful players in the DPJ – sought to enhance ties with Taiwan (Sakashi, 2014). In the current author's personal written communication with former state minister Keiji Furuya (2019), Keiji stated that since Japan does not have a formal diplomatic relationship with Taiwan as a parliamentary diplomacy, the Nikka-Kon played an important role in the signing of the fisheries agreement. Another interesting domestic political development in Japan under the DPJ administration was the role of high-profile LDP politicians in the opposition party in enhancing Japan-Taiwan relations. For example, during the DPJ era, a record number of former Japanese prime ministers visited Taiwan (Sakashi, 2014), including Junichiro Koizumi, Aso Taro, and Shinzo Abe. These politicians ensured that Japan's interactions with Taiwan continued to be robust despite the nation's change of administration.

After the fall of the DPJ administration, the Abe administration, with its pro-Taiwan political inclinations, took on a political leadership role. Taro Aso became deputy prime minister and financial minister, Furuya Keiji, became state minister, while Hakubun Shimomura became the minister of education (Sakashi, 2014). Abe's sympathy for Taiwan is famous (his grandfather had ties with the ROC in the post-war period) and he rejected Wen Jia Bao's demand to explicitly oppose Taiwan's independence in an official written document during Wen's official visit to Japan in 2007 (Shimizu, 2009). Evidently, Abe made the final decision on the bilateral fisheries agreement. Abe even expressed his gratitude to Taiwan for its support during the triple disasters in Tohoku and mentioned the Japan—Taiwan bilateral fisheries agreement on his official Facebook page (Ogasawara, 2014). Recognising Abe's positive sentiments towards Taiwan, the ROC MOFA stated the following:

[a]fter the agreement was signed, Japanese Prime Minister Shinzo Abe received Mitsuo Ohashi, chairman of the Interchange Association, Japan, and head of the Japanese negotiating team, at the Prime Minister's Office. Prime Minister Abe said that Taiwan is an important partner of Japan and that the signing of the fisheries agreement was very positive for both sides. On the day the agreement was signed, Chief Cabinet Secretary Yoshihide Suga said that he welcomes such an historic agreement and believes that it will enhance regional stability (The Japan–Taiwan Fisheries Agreement, 2013, p. 2).

6.6 Summary

Japan instrumentalised Taiwan as a geostrategic asset through the conclusion of the

fisheries agreement to enhance the balance of power in the East China Sea. It was motivated by the PRC's attempt to conduct a cross-strait campaign against it to acknowledge Taiwan as a quasi-independent actor in the East China Sea issue. Depending on how Japan approaches the One-China policy, Japan can ensure Taiwan remains a quasi-independent actor whose perspectives are more appreciated in regional security affairs. However, the hierarchical nature of Japan—Taiwan relations remains foundationally the same: Japan has an influential authority to grant quasi-recognition to an unrecognised Taiwan, and Taiwan remains on the receiving end. However, the PRC views this as a violation of the One-China policy within the context of the Sino-Japanese great power management. Japan and Taiwan enhanced their law enforcement and economic security cooperation through the fisheries agreement to the detriment of Okinawan interests. Furthermore, the multifaceted normative developments in Japan—Taiwan relations helped achieve amicable relations between the two actors and helped pave the way for the fisheries agreement in accordance with the ESIR paradigm. The next chapter now presents the conclusions of this study.

Chapter 7. Conclusion

7.1 Overview

This dissertation elucidated how Japan instrumentalises Taiwan as a foreign policy instrument within the context of Japan's One-China policy. This policy is conducted on the basis of how Japan socialises Taiwan to behave 'as if' it is a 'member' of international security efforts, which officially requires statehood. This sheds some light on Japan–Taiwan relations within the context of the ESIR paradigm. As demonstrated, Japan incrementally accommodates unrecognised Taiwan into the international society of states without recognition of Taiwan's actual statehood in view of Japan's security vis-à-vis the PRC. Due to this international non-recognition of Taiwan, and the PRC's sanctions, members of the international society (including Japan) tend not to treat Taiwan as an equal member. As such, Japan's innovative security policy means the nation presents a unique case study due to the way it has instrumentalised unrecognised Taiwan as an international security asset for its own security vis-à-vis the rising threat of the PRC. Notably, Japan is navigating its One China policy to find out what it is willing and unwilling to do in terms of its non-traditional security ties with Taiwan. In the hierarchical Japan-Taiwan relationship, it is clear that Japan is both the master and the main protagonist and that Taiwan is merely a foreign policy instrument used by its more powerful neighbour to improve its security vis-à-vis the PRC.

The study thus partially focused on how Japan has been a central protagonist in promoting international security norms and rules in the emerging regional security environment through its incremental accommodation of Taiwan into international society in relation to its regional power politics vis-à-vis the PRC. This contrasts with

notions of Japan becoming a socialised member of the Eurocentric global order by accepting the prevailing Eurocentric norms and rules.

7.2 Outline of the Argument

Chapter 1 first outlined the research question of this dissertation, which was aimed at addressing how the de facto state of Taiwan fits into Japan's regional security strategy. Following this, it discussed the debates surrounding the implications of the status of Taiwan from a Japanese perspective and outlined how Taiwan is a non-sovereign political community due to the absence of Japan's recognition within the context of its One-China policy. This significantly reduces Taiwan's foreign policy leverage vis-àvis its interactions with Japan and renders the nation a mere foreign policy instrument for Japan in its great power struggle with the PRC. Taiwan's marginal place within international society affords a regional great power such as Japan the opportunities to exploit Taiwan's vulnerability for its own national interests. Japan achieves this through the accommodation of Taiwan into the regional security environment, which allows the former to gain a strategic advantage over emerging major powers such as the PRC. As was highlighted, the existing social constructivist approach to Japan's foreign policy lacks the conceptual framework for examining "as if membership in international security efforts that require statehood, which is a key concept in understanding Japan's state behaviour vis-à-vis the de facto state of Taiwan.

Chapter 2 built upon this discussion by tracing the overall reviews of the ESIR and by investigating how international society has engaged with de facto states in order to demonstrate how Japan has been portrayed as simply a 'good pupil' of the Eurocentric global order and not a core creator of international society, which was a major focus of this study. In addition, this chapter demonstrated how de facto states have been

discussed as troublemakers or the generators of international security issues within the existing literature. Here, we showed that despite the volume of studies that deal with how international society engages with de facto states, none have as yet discussed the role of Japan as a paternalistic accommodator of unrecognised Taiwan within the international efforts to enhance the peace and security in international society.

Chapter 3 then outlined the literature related to how Japan has utilised foreign policy instruments to enhance its own security via multiple leverages including in terms of its defence, non-traditional security means, security networks and pacifist identity. It also looked at the literature related to Japan-Taiwan relations in general. These reviews were conducted to highlight the general lack of research regarding Japan's instrumentalisation of Taiwan as an international security asset, if not a geopolitical pawn, one that could potentially allow it to gain the upper hand in its power politics with the PRC within the regional security environment. Especially if Japan's security interactions with Taiwan are deemed normative and less coercive, it is difficult for the PRC to sanction Japan-Taiwan non-traditional security ties in view of Japan's One China policy. It is especially the case in non-military policing cooperation. Due to Taiwan's ambiguous status in the Japan–US alliance, the nation's historical ties with Japan, its reliance on Japan for its own security, and its strategic need to gain semirecognition from Japan for its normative security behaviours, the hierarchical nature of the Japan-Taiwan relationship has been strengthened. These social structures within international society afford Japan an influential authority to dictate Taiwan's contributions to Japan's security. The attendant discussion demonstrated that regardless of whether Japan makes radical changes in its security policy or continues to follow an incrementalistic approach to international security, its accommodation of Taiwan unquestionably follows such an incremental approach due to its regional great power management with the PRC in the context of Japan's One-China policy. In the literature review related to Japan—Taiwan relations, which was presented in Chapter 3, it was observed that a number of studies exist on how Taiwan's view of Japan affects its policy for its interactions with its neighbour. However, the political insignificance of Taiwan's input into Japan's security policy regarding Taiwan was clearly demonstrated.

Thus, following the arguments and theories discussed above, Chapters 4, 5 and 6 empirically examined how Japan's instrumentalisation of Taiwan manifested itself in each case study, which related to the areas of CSD, international civil aviation, and the fisheries agreements.

This conclusion highlights the main findings of this study. In doing so, it outlines how it is ultimately up to Japan to define the boundaries regarding the progress of Japan—Taiwan security relations, concluding that Japan will never make Taiwan an equal partner within the hierarchical Japan—Taiwan relationship due to the latter's unrecognised status in international society. These social structures are constructed in relation to the Sino-Japanese balance of power, the attendant regional great power management, and Japan's enabling of Taiwan to serve as a normative security asset for Japan's security. As was shown in Chapter 4, Japan has granted Taiwan ambiguous status in its approach to the CSD due to Taiwan's unrecognised position. However, Taiwan can still benefit from this unequal arrangement while its defence capabilities continue to contribute to Japan's security as a military buffer against the PRC. This unequal relationship provides Japan with various opportunities to enhance its non-traditional security cooperation with Taiwan for its own security and that of the region and beyond. In Chapter 5, it was shown that Taiwan is only afforded semi-participation in the ICAO due to its absence of international recognition and the PRC's sanctions.

Nonetheless, Japan has instrumentalised Taiwan to enhance international aviation security through enabling the latter to behave as if it is a member in ICAO's security principles. Chapter 6 was somewhat different in nature to the previous chapters since, within the maritime context, Japan has acknowledged Taiwan as a semi-legitimate participant in the attendant security issues, specifically in the East China Sea. Here, Japan conducted matters within a limited space, which created some need for an examination of Taiwan's perspectives, albeit that this was also undertaken in a paternalistic manner. For example, the Taiwanese view of Japan as a 'positive other' and China as a 'negative other' in relation to Taiwan's ontological security created the political momentum for Taiwan's foreign policy preference to enhance its ties with Japan rather than the PRC, as was examined in Section 6.3 of the fisheries agreement chapter.

7.3 Contribution to Japanese Studies

It was demonstrated that Japan has used Taiwan as a foreign policy instrument to balance its relationship with the PRC, incrementally enhancing Taiwan's security contribution to Japan's security while appeasing the PRC within the context of its One-China policy. Despite Japan's less visible approach to the issue of the Taiwan Strait – and contrary to the prevalent claim (Soeya 2001) that Japan is simply rather meekly following the One-China policy by keeping its involvement in the cross-strait relations to a minimum – the nation has, in reality, strategically instrumentalised its One-China policy and the Taiwan issue to meet its own national interests. In short, Taiwan is more than merely one issue in Sino-Japanese relations, and Japan can transform Taiwan into a foreign policy instrument in the defence, aviation and maritime security dimensions for its own security depending on how this is instrumentalised vis-à-vis the PRC while navigating the complexity of the One-China policy. These findings potentially provide

a new perspective on how Japan's foreign policy has been characterised as largely reactive (Funabashi, 1991/1992; Calder 1988, 2003).

This dissertation elucidated how Japan has been utilising the non-state actor Taiwan to enhance its own security and to navigate its power relations with the PRC. If Japan is deemed an anomaly in the realist paradigm since the nation is seen as not having translated its economic heft into military power and not engaging in power politics, its interactions with Taiwan in the context of its One-China policy offers a different picture within this scenario. Furthermore, Japan has been using state-centric international law to disseminate international security norms to Taiwan for its own security interests. This encourages Taiwan to behave as a foreign policy instrument that is being socialised through international security norms by Japan. Taiwan has no access to international security forum that require statehood for membership, but Japan is making great efforts to encourage Taiwan's as if membership. This helps to eradicate the loophole that stems from Taiwan's absence in international security efforts and ensures Taiwan is a more important component of the regional power politics related to Sino-Japanese rivalry in view of tilting the balance in Japan's favour. This demonstrates how a regional great power such as Japan can accommodate a non-sovereign political community such as Taiwan into international society within a regional setting.

The way Japan incrementally engages with Taiwan is different from the way the US uses Taiwan as a bargaining chip or a geopolitical pawn vis-à-vis the PRC. Furthermore, while the China factor remains a major constraint in the enhancement of Japan–Taiwan security relations, it was found that Japan skilfully manoeuvres the ambiguity of its One-China policy to enhance Taiwan's contributions to Japan's security vis-à-vis the PRC. Moreover, as the section on political dimensions (Section 6.5.4) in Chapter 6 demonstrated, the role of Nikkakon and Kantei diplomacy proved the existence of a

strong political leadership that led to the signing of the Japan–Taiwan fisheries agreement, which does not correspond to the ineffective domestic decision-making system in Japan described by Calder (2003).⁷

Despite the increasing amount of literature related to the radical shift in Japan's security policy under Abe, which allegedly renders Japan susceptible to resorting to the use of force in order to assist the US's war effort while weakening the domestic hurdles to restrain Japan's foreign policy behaviours – characterised as the Abe Doctrine by Hughes (2017)⁸ – Japan–Taiwan security relations present the unique aspect of Japan's security policy. As examined, Japan's security interactions with Taiwan remain, at best, incrementalistic. Japan's security commitment to Taiwan in the face of potential invasion by the PRC also remains ambiguous, regardless of whether Japan radically alters its pursuance of international security policy or remains incrementalistic in its approach. This is due to Taiwan's ongoing ambiguous status in the Japanese conception of CSD due to Taiwan's lack of statehood and the delicate nature of Japan's One-China policy vis-à-vis the Sino-Japanese regional great power management. As such, Taiwan presents an under-researched case of how Japan's security policy is not determined by

.

⁷ Funabashi (1991) argued that the shock of the global under-appreciation of Japan's financial contribution to the Gulf War exposed Japan's structural absence of both the external and internal mechanisms to create effective foreign policy initiatives. It also made Japan unwilling to be a rule-maker in international society due to its lack of regional strategy, post-war reliance on the asymmetric alliance with the US, mercantilist policy, excessive pacifism, and neighbours' historical distrust of Japan. Therefore, in Funabashi's (1991) view, Japan relied on Gaiatsu to pursue its own diplomacy without engaging in healthy policy debates on its foreign policy.

⁸ Hughes (2017) argued that despite Abe's claims regarding the enhanced democratic hurdles to collective self-defence and the enhanced deterrence capabilities against potential adversaries, the Abe Doctrine is more likely to heighten the possibility for Japan to resort to the use of force to assist the US's war efforts. This is due to factors such as considerations of the Japan–US alliance, the growing indivisibility of the US's national security interests with that of Japan's, and the weakened domestic constraints, including fewer legal, bureaucratic and executive hurdles in decision-making processes in Japan. Hughes (2017) also pointed out that the still substantial pacifist public opinions in Japan were unable to stop the legal passage of collective self-defence. Furthermore, Hughes (2017) argued that the ambiguous nature of the three conditions imposed on collective self-defence would be unlikely to be consolidated hurdles.

foreign pressure alone. Precisely because Taiwan remains a de facto state, Japan is able to enhance unique security interactions between the two nations, most notably in the field of the non-kinetic intelligence cooperation and non-traditional security ties, in ways that do not excessively breach Japan's One-China policy.

Meanwhile, while Japan refuses to explicitly acknowledge any security-related commitment to Taiwan, this does not mean that Japan demonstrates no independent security behaviour in relation to Taiwan—China relations. Therefore, as was highlighted, in terms of an internationally isolated actor such as Taiwan, Japan accommodates it into international society and transforms it into an international security asset for its own security enhancement. At the same time, it prevents such an internationally isolated entity from being brought further into the geopolitical orbit of the PRC, minimising the risk of Taiwan being an incendiary issue in Sino-Japanese relations. Finally, as examined in Section 3.2.5, in the literature related to the hierarchical relationship between Japan and Taiwan, it is often debated whether Taiwan is regarded as an asset or an irritant in Sino-Japanese relations. This is precisely the reason why security studies on Japan's instrumentalisation of Taiwan as an international security asset are titled toward Japanese studies. This provides the opportunity to engage in an in-depth study of Japan's conception of security via the instrumentalisation of a sensitive issue

٠

⁹ The widely held claim of Japan always being reactive to Gaiatsu, or foreign pressure, as opposed to Japan's role of enhancing active norm entrepreneurship needs to be reconsidered, at least in the context of Japan–China–Taiwan relations. Calder (2003) argued that Japan is a reactive state, which has no independent foreign policy. This is due to the constraints imposed on it by its asymmetric alliance with the US for security, historical antagonism from China and Korea, and ineffective domestic decision-making systems, despite Japan's enormous financial and investment capabilities. Calder (2003) admitted that Japan has some room for independent foreign policy initiatives in the field of official development assistance, proactive quasi-diplomatic endeavours by local governments and NGOs, and occasional major independent foreign policy initiatives when the international environment allows Japan to do so – including Koizumi's visit to North Korea. Finally, Calder (2003) argued that without the international shock, especially from the US vis-à-vis its unreliability as an ally, becoming a reality, Japan would not produce Japanese foreign policy activism.

such as Taiwan in relation to Sino-Japanese relations, which assists in the enhancement of Japan's security within the emerging regional order.

7.4 Contribution to the ESIR Paradigm

The ESIR is known to be Eurocentric, and many of its contributors (e.g. Bull, 1977; Watson, 2008; Suzuki, 2014) openly admit such a tendency. For example, Bull and Watson (1984, p. 2) stated that 'it is not our perspective, but the historical record itself that can be called Eurocentric'. As examined in the literature related to the English School, a number of academics suggest that Japan has responded to the global expansion of international society in terms of being a 'good pupil' of Eurocentric international society. Meanwhile, others such as Suzuki (2008) suggest that both Japan and China have remained frustrated great powers within international society amid their aspirations to be recognised as equals vis-à-vis the privileged Eurocentric great powers. Black (2013) looked at Japan's innovative foreign policy approaches, including its *Kakehashi* diplomacy, and its innovative maritime security policy approaches, such as its coastguard diplomacy (Black, 2014).

However, as yet, none of the studies working in this area has positioned Japan as a central creator of international society, either in the regional setting or beyond. In fact, the current study on Japan's accommodation of Taiwan into international society is the first to construct the narrative of Japanese-centric regional order, which places Japan at the central core of international society in the regional setting while clearly positioning unrecognised Taiwan as a pupil of Japan's international norms and rules of international security vis-à-vis the emerging China. As a regional and global counterweight to the PRC, the strategic role of Japan is becoming ever more critical, which means the ESIR paradigm, as one of the major disciplines of security studies, requires some adjustment in terms of its interpretations of Japan. For example, Suzuki (2014) observed that the

expansion of international society into 'Asia' produced contradictory reactions. Here, we witnessed Japan's aspiration to be a legitimate member of international society, one on a par with the European states, which was manifested in its attempt to socialise itself with the Eurocentric norms and rules. However, as Suzuki (2014) pointed out, Japan resented the nominal equality that harboured racist views against the Japanese by the 'civilised' European people, even after Japan emerged as a great power. Japan's pan-Asian advocacy also contradicted Japan's sense of superiority over other Asian peoples, meaning that the nation failed to gain the trust of the Asian populations. Nonetheless, in the process of the current research, one point became clear: Japan is a leader in international society in the regional setting, with Taiwan willing to accept Japan-centric security norms regardless of Eurocentric narratives. In other words, the ESIR must move beyond the conceptual representation of Japan as a learner of the Eurocentric global order and must accordingly place Japan's regional strategy vis-à-vis Taiwan-China relations at the central narrative of international society regarding the region. It must also leave behind the historical essentialism of European supremacy in its conceptual analysis of contemporary Japan's creation of international society.

This study of Japan's instrumentalisation of Taiwan as an international security asset elucidates the conceptual and empirical reifications of Japan's security policy in the regional security environment vis-à-vis the emerging challenge of the PRC. Through this study, the author demonstrated that it is entirely possible to challenge the narrative that describes Japan as simply being a good pupil of Eurocentric international society in an attempt to be recognised as an equal member of international society on the global stage. In fact, it is entirely possible to conceptualise Japan as a core member of international society, one that has the influential authority to incrementally grant semi-legitimacy via as if membership to state-like entities such as Taiwan as a reward for

their contribution to Japan's security in the Japan-centric regional setting. Japan may have been a latecomer to the global stage, but it was the pioneering actor in the regional modernisation process. As such, the author considered why no study exists that positions Japan as a centrepiece of international society in the regional setting, which set the wheels in motion for this PhD dissertation related to the ESIR.

7.5 Contribution to Taiwan Studies

Through the author's PhD studies on Japan's utilisation of Taiwan as one of its foreign policy instruments for its own security enhancement, it was demonstrated that it is up to Japan to decide how far Japan—Taiwan security relations will be developed. As such, Taiwanese political input in Japan's security policy on Taiwan has little influence. Equally, Taiwan's domestic policy on Japan is not considered in Japan's security interactions with its neighbour due to the absence of Taiwan's foreign policy leverage vis-à-vis Japan, as was examined in Chapter 3. Thus, through this research, Taiwan's international marginality has been elucidated in the context of the hierarchical Japan—Taiwan relations in international society in a regional setting. Taiwanese perspectives on international security efforts are ignored in the multilateral security efforts, if these efforts require members to have a 'recognised statehood'. Thus, Taiwanese perspectives matter only if great powers including Japan choose to accommodate the former's perspectives in creative ways.

Furthermore, if Taiwan remains a de facto state, there is no Taiwan without China, and Japan–Taiwan relations do not exist without considering the China factor. However, if Taiwan falls into the hands of the PRC, this would clearly be detrimental to Japan's security. Therefore, Japan needs to consider what it must do to prevent the risk of the US's abandonment of Taiwan. In this context, it can be argued that Taiwan's enhanced strategic and normative value for Japan's security is critically important to ensuring the

US continue to engage with Taiwan's security. On this point, the empirical knowledge this study produced could potentially help Japan to avoid the risk of the US's abandonment of Taiwan, that is, Japan could incrementally increase Taiwan's security contribution to its own security. In fact, the empirical knowledge produced in this study is essential for scholars working within the field of Taiwan studies in terms of reflecting on how Taiwan could be a better national security asset for Japan in view of securing its own survival. For certain, a Taiwanese perspective has no place in Japan's elite debates on Japan's security policy; however, as was demonstrated in the three case study chapters, Taiwan can voluntarily enhance its internal reforms and internal balancing since it enjoys some internal autonomy as a de facto state, and its efforts may ultimately be noted by Japan. The hope is that Taiwan's internal efforts will be accommodated by Japan in its facilitation of Taiwan's "as if membership in the multilateral security efforts as an international security asset for Japan's security, as was examined in the ICAO chapter. As for Taiwan's international space, if, with Japan's help, the nation behaves as if it is a member of the international organisation that require statehood for membership, it will have alternative methods for improving its international contributions and some hope for its future entry into the state-centric international organisations.

7.6 Future Research on Japan's Accommodation of Internationally Isolated Actors for International Security

From this study, one major finding has been clarified: Japan's innovative accommodation of internationally isolated entities, including Taiwan, into the international community (and its attempt to discipline them enough to be able to maintain their linkage with international society) in ways that contribute to Japan's security is different from how Eurocentric Western international society ostracises non-

sovereign political communities in the international society of states. This is significant in Japan's regional security strategy because when the West ostracises a specific state or non-sovereign political actor in the Asia Pacific region, it is the PRC that attempts to lure such an actor into its own geopolitical sphere of influence, which may subsequently reduce Japan's regional power. Thus, Japan accommodates the membership of isolated actors in the region in view of managing any threat from other rising powers such as the PRC to enhance its security and to elevate its regional standing. As such, Japan's policy regarding the accommodation of such entities into international society in the regional setting requires more scholastic attention.

Black (2013) argued that Japan's Kakehashi diplomacy helped Burma maintain ties with the West through Japan. In addition, despite the controversial war on drugs that Duterte has carried out as his flagship national security policy (at the expense of human rights and the erosion of the rule of law in the Philippines that shocked the Eurocentric Western great powers), it would appear that Japan is fine to accommodate Duterte's Philippines. For Duterte, Japan presents a great ally in terms of maintaining ties with the West, especially the US. Therefore, in the case of an ostracised political entity that has significant difficulty in maintaining ties with international society, Japan plays a linking role in the process of maintaining any ties. It is essential to state here that both Burma and Duterte's Philippines have been geopolitically pursued by the PRC in ways that may undermine the Western influences in the region. In recent years, Japan attempted to accommodate Iran vis-à-vis the rising tensions with the US, in which case Japan's accommodation of such entities may have incrementally expanded into the global setting. Nonetheless, more research needs to be conducted to determine whether or not this is indeed the case. If it is, it could be argued that Japan is transitioning from a great regional great power to a global great power. Certainly, Japan has a great deal of regional expertise and experience in accommodating the internationally isolated entities in its region; however, in terms of the global arena, it remains to be seen whether Japan can take on a similar, but yet-to-be-explored, global security role to help supplement the maintenance of the international liberal order.

Bibliography

Abb, P. and Yang, A.H., (2018). Democratization, political culture, and diplomatic isolation on 000000000think-tank development in Taiwan. *Pacific Affairs* [online]. **91**(1), 73-94. [Viewed 28 October 2018]. Available from: doi: https://doiorg.sheffield.idm.oclc.org/10.5509/201891173.

Aerotime News., (2016). EASA rules on pilot fatigue – Time for a strike? *Aerotime News* [online]. 21 July. [Viewed 24 July 2016]. Available from: http://www.aerotime.aero/en/civil/12818-pilot-fatigue-br-time-for-a-strike.

AFP., (2016). *Taiheiyouno Shimaguni Parao*, *Kanzendansuinokiki Nihontotaiwan Nishienyouse* (Island state of Palau faces the crisis of a complete water shutdown, requesting Japan and Taiwan for help. *AFP*. [Online]. 4 April. [Viewed 25 March 2017]. Available from: http://www.afpbb.com/articles/-/3082890.

Aibara, R., (2016). SDF can protect U.S. naval ships under guidelines approved by NSC. *The Asahi Shimbun* [online]. 22 December. Updated 22 December 2016, 17:10 JST. [Viewed 22 December 2016]. Available from: http://www.asahi.com/ajw/articles/AJ201612220061.html.

Alderman, L., (2016). Terrorism scares away the tourists Europe was counting on. *New York Times* [online]. 29 July. [Viewed 29 July 2016]. Available from: http://www.nytimes.com/2016/07/30/business/international/europe-economy-gdp-terrorism.html? r=0

Amako S., (2013). Senkakuno Souten (The controversy on Senkakus). In: *Nicchuu Tairiritsu Shuukinheino Chuugokuwoyomu* (Sino-Japanese rivalry, understanding Xi-Jingping's China). Tokyo: Chikuma Shinsho. pp. 113–164.

Anno, T., (2018). National identity and democracy: Lessons from the case of Japan. *The Asan Forum* [online]. [Viewed 23 January 2019]. Available from: http://www.theasanforum.org/national-identity-and-democracy-lessons-from-the-case-of-japan/

Antunes, S., and Camisão, I., (2018). Introducing Realism in International Relations Theory. E-International Relations [online]. 27 February. [Viewed 3 October 2020]. Available from: https://www.e-ir.info/2018/02/27/introducing-realism-in-international-relations-theory/

Arrington, C., and Yeo, A., (2019). Japan and South Korea can't get along – Why America needs to help its allies mend fences. *Foreign Affairs* [online]. [Viewed 12 March 2020]. Available from: https://www.foreignaffairs.com/articles/japan/2019-07-31/japan-and-south-korea-cant-get-along.

Asahi Shimbun., (2016). Beijing's shifting strategy behind air drills in Western

Pacific. *The Asahi Shimbun* [online]. 26 September. [Viewed 26 September 2016]. Available from: http://www.asahi.com/ajw/articles/AJ201609270076.html

Asahi Shimbun Editorial., (2016). Chutai Kankei Atsuryoku Wataiwawo Touzakeru (China-Taiwan relations: The pressure would push the dialogue away). Asahi Shimbun. 2nd October.

Asahi Shimbun., (2018). Iwojimaokide Taiwangyosen Sounan, Kaijiheri, Norikumiin 8ninzeninkyuujo (A Taiwanese fishing vessel went Missing around the sea in Iwojima. MSDF Helicopter rescued all the 8 crews). *Asahi Shimbun* [online]. [Viewed 28 October 2018]. Available from:

https://www.asahi.com/articles/ASLBX71YTLBXULOB00H.html.

Asahi Shimbun Interview., (2016). [intabiuu] [Kouhuku] na Taiwane Taiwanmotosoutou Ritoukisan [Interview] (Make happy Taiwan the former Taiwanese President Lee-teng Hui San). *Asahi Shimbun* [online]. 6 June. [online]. [Viewed 27 July 2016]. Available from:

http://digital.asahi.com/articles/DA3S12408955.html?_requesturl=articles%2FDA3S12408955.html&rm=150.

Asplund, A., (2018). Normative power Japan: Settling for 'Chinese democracy'. *Contemporary Japan* [online]. **30**(1), 117-134. [Viewed 30 December 2018]. Available from: doi: 10.1080/18692729.2018.1422913.

Auslin, M., (2016). Japan's new realism. *Foreign Affairs* [online]. **95**(2),125-134. [Viewed 29 August 2018]. Available from: https://search-proquest-com.sheffield.idm.oclc.org/docview/1778090093?accountid=13828.

Babones, S., (2016). Taipei's name game – It's time to let Taiwan Be Taiwan. *Foreign Affairs* [online]. **2016** (December). [Viewed 27 November 2018]. Available from: https://www.foreignaffairs.com/articles/taiwan/2016-12-11/taipeis-name-game.

Bacon, P. and Burton, J., (2018). NATO-Japan relations: Projecting strategic narratives of 'natural partnership' and cooperative security. *Asian Security* [online]. **14**(1), 38-50. [Viewed 21 September 2018]. Available from: doi: 10.1080/14799855.2017.1361730

Bartmann, B., (2008). Between de jure and de facto statehood: Revisiting the status issue for Taiwan. *Island Studies Journal* [online]. **3**(1), 113–128. [Viewed 12 January 2019]. Available from: https://search-proquest-com.sheffield.idm.oclc.org/docview/1953353254?accountid=13828

Bateman, S., (2003). Coast guards: New forces for regional order and security. *Asia Pacific Issues* [online]. 63(January), 1-8. [Viewed 21 January 2019] Available from: https://www.eastwestcenter.org/system/tdf/private/api065.pdf?file=1&type=node&id=31902

Berg, E. and Toomla, R., (2009). Forms of normalisation in the quest for de facto statehood. *The International Spectator* [online]. **44**(4), 27–45. [Viewed 15 January 2019]. Available from: doi: 10.1080/03932720903351104

Berg, E. and Kuusk, E., (2010). What makes sovereignty a relative concept? Empirical approaches to international society. *Political Geography* [online]. **29**(1), 40–49.

[Viewed 20 January 2019]. Available from: https://doi.org/10.1016/j.polgeo.2010.01.005.

Berger, T., (2011). Norms, identity, and national security. In: C. Hughes, and L. Meng, eds. *Security Studies A Reader*. Oxon: Routledge. pp. 187–190.

Black, L. and Hwang, Y. J., (2012). China and Japan's quest for great power status: Norm entrepreneurship in anti-piracy responses. *International Relations* [online]. **26**(4), 431–451. [Viewed 29 December 2018]. Available from: doi: 10.1177/0047117812452441.

Black, L., (2012). Debating Japan's intervention to tackle piracy in the Gulf of Aden: Beyond mainstream paradigms. *International Relations of the Asia-Pacific* [online]. 12(2), 259–285. [Viewed 29 December 2018]. Available from: https://doiorg.sheffield.idm.oclc.org/10.1093/irap/lcs004

Black, L., (2013). Bridging between Myanmar and international society – Japan's self-identity and *kakehashi* policy. *The Pacific Review*. [online]. **26**(4), 337-359. [Viewed 27 June 2019]. Available from: DOI: 10.1080/09512748.2013.788064

Black, L., (2014). *Japan's maritime security strategy: The Japan Coast Guard and maritime outlaws. Critical studies of the Asia Pacific*. London: Palgrave Macmillan UK.

Brady, H. E. and Collier, D., eds. (2010). *Rethinking social inquiry: Diverse tools, shared standards*. New York: Rowman and Littlefield.

Brookings Institution., (2015). Relations across the Taiwan Strait: Retrospective and prospects for future development. Brookings Institution [online]. *13 July*. [Viewed 15 August 2015]. Available from: http://www.brookings.edu/events/2015/07/13-taiwan-cross-strait-relations

Brookings Institution., (2017). Cross-strait relations at a juncture: Japanese and American perspectives [online]. Speaker: Yasuhiro Matsuda. Washington D.C.: Brookings Institution. YouTube. [Viewed 15 March 2017]. Available from: https://www.youtube.com/watch?v=t1AX0O1snIU

Brooks, N., (2015). AIR aviation intelligence report. InterVISTA [online]. May. [Viewed 10 August 2016]. Available from: https://issuu.com/intervistasconsulting/docs/intervistas_aviation_intelligence_r_80cb1 497ccc3ab?e=6192829/13558446

Brown, C., (2009). Understanding international relations. Basingstoke: Palgrave.

Bukh, A., (2015). Shimane Prefecture, Tokyo and the territorial dispute over Dokdo/Takeshima: Regional and national identities in Japan, *The Pacific Review*, 28:1, 47-70, DOI: 10.1080/09512748.2014.970040

Bull, H. and Watson, A., (1984). *The expansion of international society*. [online]. Oxford: Oxford University Press. p. 2 [Viewed 12 August 2019]. Available from: scholar.google.com.

Bull, H., (1977). The *anarchical society: A study of order in world politics*. New York: Columbia University Press.

Bull, H., (1979). *The anarchical society: A study of order in world politics*. London: Macmillan.

Bush, R., (2019). An open letter to Mr. Kuo Pei-hung and his colleagues [online]. *Brookings*. [Viewed 22 February 2019]. Available from: https://www.brookings.edu/blog/order-from-chaos/2019/02/11/an-open-letter-to-mr-kuo-pei-hung-and-his-

colleagues/?fbclid=IwAR2Gd0PSVkefdInxfXs0pOh_4OxbnNhvpR1HdV8fA1wvo15 0phZxYpIY-sA

Buzan, B., (2014a). *An introduction to the English School of International Relations: The societal approach*. Cambridge: Polity Press.

Buzan, B., (2014b). The 'standard of civilisation' as an English School concept. *Millennium: Journal of International Studies* [online]. **42**(3), 576–594. [Viewed 31 January 2019]. Available from: doi: 10.1177/0305829814528265

Buzan, B., Wæver, O. and de Wilde, J. (1998). Security: A new framework for analysis. Boulder, CO: Lynne Rienner

Cabestan, J., (2014). Implementation of East China Sea Peace Initiative: Perspectives from non-claimant states. *Paper prepared for the International Conference on Implementation of East China Sea Peace Initiative – Search for New Approach, 6 November 2014, Taiwan* [online]. Taipei: Chinese Council of Advanced Policy Studies. pp. 1–10. [Viewed 23 July 2015]. Available from: http://gis.hkbu.edu.hk/files/Implementation%20of%20East%20China%20Sea%20Peace%20Initiative with%20map.pdf

Calder, K., (1988). Japanese foreign economic policy formation: Explaining the reactive state. *World Politics*. **40**(4), 517-541.

Calder, K., (2003). Japan as a post-reactive state? *Orbi* [online]. 47(4), 605-616. [Viewed 30 June 2019]. Available from: https://doi.org/10.1016/S0030-4387(03)00081-4

Calvo, A., (2012). Manila, Beijing, and UNCLOS: A test case? Academia, education. *The Asia Pacific Journal: Japan Focus* [online]. **10**(34). [Viewed 30 December 2015]. Available from:

https://www.academia.edu/5419798/Manila_Beijing_and_UNCLOS_A_Test_Case

Calvo, A., (2013a). Adult wisdom: The Japan–Taiwan fisheries deal in Shigeru News Agency. *Academia* [online]. [Viewed 11 May 2016]. Available from: https://www.academia.edu/10107442/Adult_Wisdom_The_Japan–Taiwan_Fisheries_Deal

Calvo, A., (2013b). Marines, missiles, and the Iron Lady: The military leg in Japan's ocean strategy. *The Asia Pacific Journal Japan Focus* [online]. **11**(49) | Number 1. [Viewed 7 August 2015]. Available from: http://apjjf.org/2013/11/49/Alex-Calvo/4045/article.html

- Canada. IATA., (2001). *The impact of September 11, 2001 on aviation* [online]. Montreal: IATA. [Viewed 22 January 2015]. Available from: http://www.iata.org/pressroom/documents/impact-9-11-aviation.pdf
- Canada. ICAO., (n.d.) About ICAO [online]. Montreal: ICAO. [Viewed 23 June 2018]. Available from: http://www.icao.int/about-icao/Pages/default.aspx
- Canada. ICAO., (2006). *Article 93 Convention on International Civil Aviation. 9th edition* [online]. Montreal: ICAO. [Viewed 22 January 2019]. Available from: https://www.icao.int/publications/Documents/7300 cons.pdf
- Canada. ICAO., (2014). No country left behind [online]. Montreal: ICAO. [Viewed 10 August 2016]. Available from: http://www.icao.int/abouticao/NCLB/Pages/default.aspx
- Canada. ICAO., (2018). ICAO and UNCTED to strengthen cooperation on border control, aviation security, and counter-terrorism issues, 17 May [online]. Montreal: ICAO. [Viewed 17 May 2018]. Available from: https://www.icao.int/Newsroom/Pages/ICAO-and-UN-CTED-to-strengthen-cooperation-on-border-control,-aviation-security,-and-counter-terrorism-issues.aspx
- CAPA Center For Aviation., (2016). Taiwan's LCCs Tigerair Taiwan and V Air focus expansion on Japan; other markets incubating. *CAPA Center For Aviation* [online]. 17 May. [Viewed 17 May 2016]. Available from: http://centreforaviation.com/analysis/taiwans-lccs--tigerair-taiwan-and-v-air--focus-expansion-on-japan-other-markets-incubating-278461
- Caspersen, N., (2018). Recognition, status quo or reintegration: Engagement with de facto states. *Ethnopolitics* [online]. **17**(4), 373–389. [Viewed 13 January 2019]. Available from: doi: 10.1080/17449057.2018.1495360
- Center for Strategic & International Studies., (2015). Taiwan's marginalized role in international security [online]. *Directed by Bonnier Glaser. Washington D.C.: CSIS. YouTube.* [Viewed 1 March 2015]. Available from: https://www.youtube.com/watch?v=k3mPb0n0jTc
- Chan, R., (2012). Ma welcomes Beijing's ICAO support. *Taiwan Today* [online]. [Viewed 22 July 2016]. Available from: http://taiwantoday.tw/ct.asp?xItem=195986&ctNode=2175
- Chang, Y., (2006). Taiwan to set up its first Okinawa TECRO office. *Taipei Times* [online]. 31 May. [Viewed 3 March 2016]. Available from: http://www.taipeitimes.com/News/front/archives/2006/05/31/2003310890
- Chen, B. and Hwang, S., (2015). Senkaku/Diaoyu Islands dispute and Taiwan's netizens' sentiments toward Japan and China. *East Asia: An International Quarterly* [online]. **32**(4), 385-399. [Viewed 1 Oct 2015]. Available from: doi: http://dx.doi.org.sheffield.idm.oclc.org/10.1007/s12140-015-9245-3
- Chen, C. I. (2017). How Beijing could squeeze Taiwan A test for cross-strait relations. *Foreign Affairs* [online]. **2017** (April). [Viewed 22 July 2020]. Available from: https://www.foreignaffairs.com/articles/taiwan/2017-04-13/how-beijing-could-

squeeze-taiwan.

Chen, D.P., (2014). Constructing peaceful development: The changing interpretations of 'One China' and Beijing's Taiwan Strait Policy. *Asian Security* [online]. **10**(1), 22-46. [Viewed 21 June 2020]. Available from: doi: https://doi.org/10.1080/14799855.2013.874337

Chen, H. and Low, Y. F., (2015). Taiwan, Nauru ink MOU to combat human trafficking. *Focus Taiwan* [online]. 8 June. [Viewed 10 July 2015]. Available from: http://focustaiwan.tw/news/aipl/201506080008.aspx

Chen, H., (2004). Bashi Strait: A lesson in geography. *Taipei Times* [online]. 23 September. [Viewed 12 August 2015]. Available from: http://www.taipeitimes.com/News/editorials/archives/2004/09/23/2003204030

Chen, J., (2013). Taiwan continues to avoid international irrelevance. *China Policy Institute: Analysis* [online]. 14 October. [Viewed 19 September 2016]. Available from: https://cpianalysis.org/2013/10/14/taiwan-continues-to-avoid-international-irrelevance/

Chen, M., (2004). Former Indian minister to attend meet. *Taipei Times* [online]. 11 November. [Viewed 18 March 2017]. Available from: http://www.taipeitimes.com/News/taiwan/archives/2004/11/11/2003210581

Chen, W-h., (2017). Taiwan's options as PLA ramps up pressure. *Taipei Times* [online]. 5 February. [Viewed 5 February 2017]. Available from: http://www.taipeitimes.com/News/taiwan/print/2017/02/05/2003664381

The China Post., (2011). Taiwan, Japan sign 'breakthrough' open skies accord. The China Post [online]. 11 November. [Viewed 15 October 2016]. Available from: http://www.chinapost.com.tw/taiwan/foreign-affairs/2011/11/11/322617/Taiwan-Japan.htm

The China Post., (2016). Press freedom group urges ICAO to allow in Taiwanese media. The China Post [online]. 2 October. [Viewed 7 October 2016]. Available from: http://www.chinapost.com.tw/taiwan/intl-community/2016/10/02/479946/Press-freedom.htm

The China Post., (2016). Delegations to go to ICAO venue without invite. China Post [online]. 17 September. [Viewed 17 September 2016]. Available from: http://www.chinapost.com.tw/print/478639.htm

Chu, M., (2016). No need to beg China? Taiwan's membership of the Asia-Pacific Economic Cooperation as a contested state. *The China Quarterly*, [online] Volume 225, 169-189. [Viewed 23 June 2018]. Available from: doi: 10.1017/S030574101500171X.

Chung, L. and Hetherington, W., (2016). Government preparing to take part in Interpol meet. *Taipei Times* [online]. 26 September. [Viewed 26 September 2016]. Available from: http://www.taipeitimes.com/News/front/archives/2016/09/26/2003655951

Cohen, E., (2019). America's long goodbye - The real crisis of the Trump era. *Foreign Affairs* [online]. (January/February 2019). [Viewed 12 March 2019]. Available from:

- https://www.foreignaffairs.com/articles/united-states/long-term-disaster-trump-foreign-policy
- Cohen, J.A., (1971). Recognizing China. *Foreign Affairs* [online]. July/August (2020). [Viewed 21 July 2020]. Available from: https://www.foreignaffairs.com/articles/asia/1971-10-01/recognizing-china
- Cole, J. M., (2015). Taiwan and the Diaoyutai spat: Is all that noise really necessary? *China Policy Institute* [online]. 29 July [Viewed 27 August 2015]. Available from: http://blogs.nottingham.ac.uk/chinapolicyinstitute/2015/07/29/taiwan-and-the-diaoyutai-spat-is-all-that-noise-really-necessary/
- Cole, J. M., (2016). The plan Taiwan needs to defend against China. *The National Interest* [online]. 28 May. [Viewed 20 May 2016]. Available from: http://nationalinterest.org/feature/the-plan-taiwan-needs-defend-against-china-16381?page=show
- Cole, M., (2013). China's ADIZ: Taiwan's dilemma. *The Diplomat*. [Viewed 4 October 2016]. Available from: http://thediplomat.com/2013/11/chinas-adiz-taiwans-dilemma/
- Collier, D., (2011). Understanding process tracing. *PS: Political Science & Politics*. Cambridge University Press, 44(4), pp. 823–830. doi: 10.1017/S1049096511001429.
- Collier, P. and Hoeffler, A., (2002). *Greed and grievance in civil war*. Centre for the Study of African Economies Working Paper Series, paper 160 [online]. Oxford: Centre for the Study of African Economies. [Viewed 2 January 2017]. Available from: https://www.csae.ox.ac.uk/materials/papers/2002-01text.pdf
- Cordner, L., (2016). CO16231 | Indian Ocean Conference 2016 in Singapore: India's soft power on display. *RSIS Commentary* [online]. [Viewed 18 March 2017]. Available from: https://www.rsis.edu.sg/rsis-publication/rsis/co16231-indian-ocean-conference-2016-in-singapore-indias-soft-power-on-display/#.WMoe9 1942w
- Crawford, J., (1976). The criteria for statehood in international law. *British Yearbook of International Law.* **48**(1976–77), 93–182.
- Cronin, P., (2013). Taking off civil aviation, forward progress and Japan's third arrow reforms. *Center for a New American Security* [online]. October. [Viewed 1 August 2016]. Available from: http://www.cnas.org/publications/taking-civil-aviation-forward-progress-and-japans-third-arrow-reforms#.V7vf9piLQ2w
- Curtis, G., (2013). Japan's cautious hawks: Why Tokyo is unlikely to pursue an aggressive foreign policy. *Foreign Affairs* [online]. [Viewed 12 February 2019]. Available from: https://www.foreignaffairs.com/articles/china/2013-03-01/japans-cautious-hawks
- Dalay, G., (2017). After the Kurdish independence referendum. *Foreign Affairs* [online]. 2 October. [Viewed 8 January 2019]. Available from: https://www.foreignaffairs.com/articles/middle-east/2017-10-02/after-kurdish-independence-referendum
- Deans, P., (2000). Contending nationalisms and the Diaoyutai/Senkaku dispute. *Security Dialogue* [online]. **31**(1), 119-131. [Viewed 20 October 2018]. Available from:

Deans, P., (2001). Taiwan in Japan's foreign relations: Informal politics and virtual diplomacy. *The Journal of Strategic Studies*. (**24**)4, 151-176. [Viewed 22 January 2018]. Available from: doi: 10.1080/01402390108437859

deLisle, J., (2011). Taiwan: Sovereignty and participation in international organizations. *Foreign Policy Research Institute* [online] [Viewed 9 November 2015]. Available from: https://www.fpri.org/article/2011/07/taiwan-sovereignty-and-participation-in-international-organizations/

deLisle, J., (2016). Taiwan's quest for international space: Ma's legacy, Tsai's options, China's choices, and U.S. policy. *Orbis* [online]. **60**(4), 550-574. [Viewed 3 October 2018]. Available from: doi: https://doi.org/10.1016/j.orbis.2016.08.007

Department of Japanese Studies, National University of Singapore., (2015). NUS Public Symposium '70 years on: How to manage changing power relations in Asia' August 5, 2015 [online]. *Speaker: Jimbo. Singapore: Department of Japanese Studies, National University of Singapore.* [Viewed 20 August 2015]. Available from: https://www.youtube.com/watch?v=86tGGxro4f8

Dobson, H.J., (1998). *Japan and United Nations peacekeeping: Foreign policy formulation in the post-cold war world*. Ph.D. thesis, University of Sheffield. [Viewed 22 October 2019]. Available from: http://etheses.whiterose.ac.uk/14769/

Dodds, K and Adey, P., (2014). Come fly with me: Airports and geographies of rendition. *E-International Relations* [online]. 18 December. [Viewed 9 Jan 2015]. Available from: http://www.e-ir.info/2014/12/18/come-fly-with-me-airports-and-geographies-of-rendition/

Dong Wang., (2016). Guānguāng jú shùzì:5 Yuè lù kè fù tái jiǎnshǎo zŏng rénshù fǎn zēng (Tourist bureau number: The number of mainland tourists decreased in May, but the overall number of tourists increased) [online]. *Dong wang*. 16 June. [Viewed 16 July 2016]. Available from: http://tw.on.cc/tw/bkn/cnt/news/20160616/bkntw-20160616010632984-0616_04011_001_cn.html

Dreyer, J., (2008). Japan and the security of the Taiwan Strait. In: Chow, P. C. Y., ed. *The 'One China' dilemma*. New York: Palgrave Macmillan US. pp. 291–304.

Dreyer, J. T., (2016). Taiwan and Japan in the Tsai era. *Orbis* [online]. **60**(4), 592–608. [Viewed 27 February 2017]. Available from: doi: https://doi.org/10.1016/j.orbis.2016.08.003

Dreyer, J. T., (2019). The Japan—Taiwan Relationship: An Unstable Stability. *NBR* [online]. **14**(1), 161-166. [Viewed 28 February 2019]. Available from: https://www.nbr.org/wp-content/uploads/pdfs/publications/asiapolicy14-1_japan_roundtable_jan2019.pdf

Drifte, R., (1998). Japan's quest for a permanent seat on the Security Council. *Asia - Pacific Review* [online]. **5**(2), 87-109. [Viewed 5 October 2018]. Available from: doi: 10.1080/13439009808719981

Duplain, J., (1995). Chisinau and Tiraspol's faltering quest for accord. *Transition*. **1**(19), 10–13.

Dutton, P. A., (2009). Caelum liberum: Air defense identification zones outside sovereign airspace. *American Journal of International Law* [online]. Cambridge University Press. **103**(4), 691–709. [Viewed 22 April 2017]. Available from: doi: 10.1017/S0002930000159834.

Easley, L-E., (2017) How proactive? How pacifist? Charting Japan's evolving defence posture. *Australian Journal of International Affairs*, **71**(1), 63–87, DOI: 10.1080/10357718.2016.1181148

The Economist., (2014). Taiwanese gangsters – The white wolf. The Economist [online]. 3 September. [Viewed 16 August 2016]. Available from: http://www.economist.com/blogs/banyan/2014/09/taiwanese-gangsters

Eisenhardt, K. M. (1989). Building theories from case study research. *The Academy of Management Review*, 14(4), 532-550. doi: 10/06/2011 00:03.

Englehart, N., (2010). Representing civilization: Solidarism, ornamentalism, and Siam's entry into international society. *European Journal of International Relations* [online]. **16**(3), 417–439. [Viewed 25 December 2018]. Available from: doi: 10.1177/1354066109350050

EU. European Commission., (2014). *Aviation: EU-ASEAN aviation summit proposes* 'open skies' agreement [online]. Belgium: European Commission. [Viewed 16 October 2016]. Available from: http://europa.eu/rapid/press-release_IP-14-133 en.htm

European Business Review., (2013). Taiwan's ICAO observer-bid garners European and global support. *EBR* [online]. 20 August. [Viewed 4 October 2016]. Available from: http://www.europeanbusinessreview.eu/page.asp?pid=1083

Everington, K., (2019). Taiwanese Coast Guard ship fires stun grenade at rock-throwing Chinese fishermen. *Taiwan News* [online]. 19 April. [Viewed 19 April 2019]. https://www.taiwannews.com.tw/en/news/3683575?fbclid=IwAR2aQI5AtJFS2rBdwx ARCgq0uDzZekERni-Cl3lKsHZVeTYYTOWw09MKqpY

Fabry, M., (2010). New states in the post-Cold War period. *Recognizing States: International Society and the Establishment of New States Since 1776* [online]. Oxford: Oxford University Press. [Viewed 22 June 2019]. Available from: doi: DOI: 10.1093/acprof:oso/9780199564446.001.0001

Faiola. A. (2006). Japan–Taiwan ties blossom as the regional rivalry grows. *Washington Post*. 24 March. p. 117.

Fan, P., (2016). Hăixiá lùn tán: "Tóu chuān pǔ jiù táiwān?" Chuān pǔ duì zhōngguó shì dí shì yǒu? (Straits talks, will Trump save Taiwan? Is Trump a friend or foe of China?). *Měiguó zhī yīn zhōngwén wăng VOA Chinese* [online]. 24 July. [Viewed 26 July 2016]. Available from: https://www.voachinese.com/a/strait-talk-trump-taiwan-china-20160724/3432575.html

Fatton, L.P., (2018). 'Japan is back': Autonomy and balancing amidst an unstable China–U.S.–Japan triangle. *Asia and The Pacific Policy Studies* [online]. **8**(5), 264–278. [Viewed 23 February 2019]. Available from: doi: https://doi.org/10.1002/app5.240

Fazal, T., (2018). Go your own way – Why rising separatism might lead to more conflict. *Foreign Affairs* [online]. 14 June. [Viewed 17 February 2014]. Available from: https://www.foreignaffairs.com/articles/2018-06-14/go-your-own-way

Feng, Y., Wang, W. and Pan, J., (2014). Trade pact siege: Producer screens 'Kano' for students in legislature. *Taipei Times* [online]. 2 April. [Viewed 5 July 2015]. Available from: http://www.taipeitimes.com/News/taiwan/archives/2014/04/02/2003587101

Focus Taiwan., (2012). Taiwan's new radar system helps detect North Korea rocket launch. Focus Taiwan [online]. 12 December. [Viewed 12 January 2015]. Available from: http://focustaiwan.tw/news/aipl/201212120021.aspx

Focus Taiwan., (2013). Taiwan, Japan sign 5 accords to expand cooperation. Focus Taiwan. [Viewed 28 November 2016]. Available from: http://focustaiwan.tw/news/aipl/201311050038.aspx

Focus Taiwan., (2014). Taiwan ships, aircraft drill inside China's ADIZ in East China Sea. Focus Taiwan [online]. 26 February. [Viewed 7 August 2016]. Available from: http://focustaiwan.tw/news/aipl/201402260023.aspx

Focus Taiwan., (2015). Taiwan submits objection to new Chinese flight routes to ICAO. Focus Taiwan [online]. 15 January. [Viewed 7 August 2016]. Available from: http://focustaiwan.tw/news/aipl/201501150006.aspx

Focus Taiwan., (2016). UK parliamentarians voice support for Taiwan's ICAO bid. Focus Taiwan [online]. 1 October. [Viewed 15 September 2020]. Available from: https://focustaiwan.tw/politics/201610010018

Focus Taiwan., (2016a). Kankoukyoku, Nihonjinno Taiwanryokouwo Sekkyokuyuuchihe Koukuukenno Muryoukanadowo Keikaku (Tourism authority proactively promotes the Japanese tourists including the plan for the free tickets). Focus Taiwan [online]. 11 October. [Viewed 11 October 2016]. Available from: http://japan.cna.com.tw/news/atra/201610110010.aspx

Focus Taiwan., (2016b). Taiwan Ikaosoukaisankanimuke Chuugokkutairikuno Bougaikeikai Beikokuwa Shiji Taiwan (To aim for participation in the ICAO, being cautious against Chinese obstruction, the US supports it). Focus Taiwan [online]. 5 August. [Viewed 5 August 2016]. Available from: http://japan.cna.com.tw/news/apol/201608050011.aspx#.V6UgqEQkDb8.facebook

Foot, R., (2017). Power transitions and great power management: three decades of China–Japan–US relations. *The Pacific Review* [online]. **30**(6), 829-842. [Viewed 11 September 2018]. Available from: doi: 10.1080/09512748.2017.1303535

Foreign Affairs., (2016). Notes from the chairman. *Foreign Affairs* [online]. [Viewed 19 August 2016]. Available from: https://www.foreignaffairs.com/interviews/2016-08-01/notes-chairman

Fukuda, M., (2017). Daijyuisshou Taiwanno Kaiyouanzenhoshou Kantorii • Purofairu (Chapter Eleventh: Taiwan's maritime security/country profile). *The Japan Institute of International Affairs*. [Accessed 5 May 2019]. Available from: http://www2.jiia.or.jp/pdf/research/H28_Indo-Pacific_country_profile/11-fukuda.pdf

Fukuda, M., (2018). Japan's policy toward China and Taiwan. *Center for Strategic & International Studies* [online]. [Viewed 29 June 2019]. Available from: https://csis-prod.s3.amazonaws.com/s3fs-

public/180402_Strategic_Japan_Madoka%20Fukuda_paper%20%28final%29.pdf?mUJOCZqOcRDXRZvov5do5ZJI4vudPoZu

Funabashi, Y., (1991/1992). Japan and the new world order. *Foreign Affairs* [online]. 1991/92 (Winter). [Viewed 28 June 2019]. Available from: https://www.foreignaffairs.com/articles/asia/1991-12-01/japan-and-new-world-order

Gady, F., (2015a). How China is expanding its Coast Guard – Beijing is converting old frigates into coastguard patrol vessels. *The Diplomat* [online]. 30 July. [Viewed 30 July 2015]. Available from: https://thediplomat.com/2015/07/how-china-is-expanding-its-coast-guard/

Gady, F. S., (2015b). Japan at peace – The improbable military resurgence. *Foreign Affairs* [online]. **2015** (September). [Viewed 27 March 2020]. Available from: https://www.foreignaffairs.com/articles/2015-09-16/japan-peace

Gibson, I., (2016). Actors and action for peace in Japan. *Peace Review* [online]. **28**(1), 99-107. [Viewed 23 December 2018]. Available from: doi: 10.1080/10402659.2016.1130402

Gilley, B., (2010). Not so dire straits. *Foreign Affairs* [online]. **2010** (January/February). [Viewed 17 August 2018]. Available from: https://www.foreignaffairs.com/articles/china/2010-01-01/not-so-dire-straits

Glaser, B., (2013). Taiwan's quest for greater participation in the international community. CSIS [online]. [Viewed 7 August 2014]. Available from: http://csis.org/publication/taiwans-quest-greater-participation-international-community

Glaser, B. S. and Vitello, J. A., (2015). Taiwan's marginalized role in international security – Paying a price. *CSIS* [online]. January. [Viewed 2 August 2015]. Available from:

http://csis.org/files/publication/150105 Glaser TaiwanMarginalizedRole WEB.pdf

Gladstone, R., (2019). Taiwan president risks infuriating China with U.S. visit. *The New York Times* [online]. 11 July. [Viewed 13 July 2019]. Available from: https://www.nytimes.com/2019/07/11/world/asia/taiwan-president-united-states-china.html?rref=collection/timestopic/China&fbclid=IwAR3yRFSMQn6wNjwsUH7 HBlH_nqYzEqqgg0A5eOKhuBsSjA7g8_rq2M-w9lA

Global Times., (2013). Taiwan should consider mainland's feelings on Diaoyu. *Global Times* [online]. [Viewed 24 January 2017]. Available from: http://www.globaltimes.cn/content/774416.shtml

Goldstein, L., (2011). Beijing confronts long-standing weakness in anti-submarine warfare. *China Brief, James Town* [Online]. 29 July. [Viewed 20 March 2017]. Available from: https://jamestown.org/program/beijing-confronts-long-standing-weakness-in-anti-submarine-warfare/

Goldstein, L. J., (2013). Chinese fisheries enforcement: Environmental and strategic implications. *Marine Policy* [online]. **40**, 187–193. [Viewed 2 August 2015]. Available from: doi: https://doi.org/10.1016/j.marpol.2012.12.007

Green, M. and Hornung, J., (2014). Ten myths about Japan's collective self-defense change. *The Diplomat* [online]. 10 July. [Viewed 20 December 2016]. Available from: http://thediplomat.com/2014/07/ten-myths-about-japans-collective-self-defense-change/

Greenstein, C., and Tensley, B., (2016). Japan, the United States, and public memory: The blurred lines between victims and perpetrators. *Foreign Affairs* [online]. 2016 (November). [Viewed 29 March 2020]. Available from: https://www.foreignaffairs.com/articles/united-states/2016-11-24/japan-united-states-and-public-memory

Greer, T., (2019). Taiwan's defense strategy doesn't make military sense - But it does make political sense. *Foreign Affairs* [online]. **2019** (September). [Viewed 30 September 2019]. Available from:

https://www.foreignaffairs.com/articles/taiwan/2019-09-17/taiwans-defense-strategy-doesnt-make-military-sense

Gries, P. and Wang, T., (2019). Will China Seize Taiwan? Wishful Thinking in Beijing, Taipei, and Washington Could Spell War in 2019. *Foreign Affairs* [online]. 15 February. [Viewed 15 February 2019]. Available from: https://www.foreignaffairs.com/articles/china/2019-02-15/will-china-seize-taiwan

Grønning, B. E. M., (2018). Japan's security cooperation with the Philippines and Vietnam. *The Pacific Review* [online]. **31**(4), 533-552. [Viewed 14 November 2018]. Available from: doi: 10.1080/09512748.2017.1397730

Gschwend T. and Schimmelfennig F., (2007) Introduction: Designing research in political science — A dialogue between theory and data. In: Gschwend T., Schimmelfennig F., eds. *Research Design in Political Science*. Palgrave Macmillan, London.

Guilloux, A., (2016). Taiwan's humanitarian aid/disaster relief: Wither or prosper? *Brookings* [online]. August. [Viewed 4 January 2019]. Available from: https://www.brookings.edu/opinions/taiwans-humanitarian-aiddisaster-relief-wither-or-prosper/

Guo, Y., (2016). Taipei must act amid tourist ploy by Beijing. *Taipei Times* [online]. 28 September. [Viewed 28 September 2016]. Available from: http://www.taipeitimes.com/News/editorials/print/2016/09/28/2003656080

Gustafsson, K., (2015) Identity and recognition: Remembering and forgetting the postwar in Sino-Japanese relations. *The Pacific Review*, **28**(1), 117-138, DOI:

Gustten, S., (2014). Students flock to universities in Northern Cyprus. *The New York Times* [online].16 February. [Viewed 17 February 2014]. Available from: http://www.nytimes.com/2014/02/17/world/europe/students-flock-to-universities-in-northern-cyprus.html?ref=europe& r=0

Grix, J., (2019). Print Media. *The foundations of research*. 3rd edn. London: Macmillan. pp. 131–132.

Haas, R., and Sacks, D., (2020). American Support for Taiwan Must Be Unambiguous - To Keep the Peace, Make Clear to China That Force Won't Stand. *Foreign Affairs* [online]. [Viewed 5 October 2020]. Available from:

https://www.foreignaffairs.com/articles/united-states/american-support-taiwan-must-be-unambiguous

Hamanaka, S., (2010). *Asian regionalism and Japan: The politics of membership of regional diplomatic, financial and trade groups*[online]. Abingdon: Routledge. [Viewed 7 March 2020]. Available from: https://read.amazon.co.jp/?asin=B00332KM0K

Hagström, L., and Hanssen, U., (2015) The North Korean abduction issue: Emotions, securitisation and the reconstruction of Japanese identity from 'aggressor' to 'victim' and from 'pacifist' to 'normal'. *The Pacific Review*, **28**(1), 71-93, DOI: 10.1080/09512748.2014.970043

Hagström. L. and Gustafsson, K., (2015) Japan and identity change: Why it matters in international relations. *The Pacific Review*, **28**(1), 1-22, DOI: 10.1080/09512748.2014.969298

Hara, K., (2015). Okinawa, Taiwan, and the Senkaku/Diaoyu Islands in United States—Japan—China relations. *The Asian-Pacific Journal — Japan Focus* [online]. **13**(28), Number 2 Article ID 4341. [Viewed 22 July 2020]. Available from: https://apjjf.org/2015/13/28/Kimie-Hara/4341.html

Harner, S., (2015). Japan and China unveil aircraft to compete with Boeing, Airbus, Bombardier, Embraer. *Forbes* [online]. 18 November. [Viewed 28 July 2016]. Available from: http://www.forbes.com/sites/stephenharner/2015/11/18/japan-and-china-unveil-aircraft-to-compete-with-boeing-airbus-bombardier-embraer/#7e2631262db1

Harold, S., (2016). Don't weaken the U.S.–Japan alliance, strengthen it. *The National Interest* [online]. 12 August. [Viewed 14 October 2016]. Available from: http://nationalinterest.org/blog/the-buzz/dont-weaken-the-us%E2%80%93japan-alliance-strengthen-it-17330?page=show

Harvey, J., (2010). Over-stating the unrecognised state? Reconsidering de facto independent entities in the international system. Ph.D. thesis, University of Exeter. [Viewed 30 December 2018]. Available from: https://ore.exeter.ac.uk/repository/bitstream/handle/10036/116412/HarveyJ.pdf?seque nce=4&isAllowed=y

Heginbotham, E. and Samuels, R., (2018). A new military strategy for Japan. *Foreign Affairs* [online]. **2018** (July 16). [Viewed 16 June 2019]. Available from:

https://www.foreignaffairs.com/articles/asia/2018-07-16/new-military-strategy-japan

Heng, Y.-K., (2015). Smart power and Japan's self-defense forces. *Journal of Strategic Studies* [online]. **38**(3), 282-308. [Viewed 13 September 2018]. Available from: doi: 10.1080/01402390.2014.1002911

Heydarian, J., (2017a). Crisis in Mindanao. *Foreign Affairs* [online]. 28 June. [Viewed 29 June 2017]. Available from:

https://www.foreignaffairs.com/articles/philippines/2017-06-28/crisis-mindanao

Heydarian, J., (2017b). After ISIS, what's next for Marawi? *Foreign Affairs* [online]. 9 November. [Viewed 1 December 2017]. Available from: https://www.foreignaffairs.com/articles/philippines/2017-11-09/after-isis-whats-next-marawi

He, Y., (2014). Identity Politics and Foreign Policy: Taiwan's Relations with China and Japan, 1895–2012. *Political Science Quarterly* [online]. **129**(3), 469–500. [Viewed 3 July 2017]. Available from: doi: 10.1002/polq.1221

Hickey, D., (2014). Taiwan and the rising tensions in the East China Sea: A mouse that roared. *Asian Survey*[online]. **54**(3), 492–514. [Viewed 3 August 2015]. Available from: doi:10.1525/as.2014.54.3.492

Hirakawa. S., (2013). [Boukuu Shikibtsuken] De Nozomareru Atarashii Nittaikankei - Chuugoku Kara Ichimoku Okareru Gaikouryokuwo (The New Japan—Taiwan relations that are desired because the [ADIZ] needs the diplomatic power that gets special attention from China). *Waseda Online Yomiuri Shimbun* [online]. 24 December. [Viewed 6 August 2016]. Available from: http://www.yomiuri.co.jp/adv/wol/opinion/gover-eco 131224.html

Hiranuma, H., (2016). Breaking free of dependence on rare earths China's resource nationalism spawns alternative technologies. *The Tokyo Foundation* [online]. 29 September. [Viewed 29 September 2016]. Available from: http://www.tokyofoundation.org/en/articles/2016/dependence-on-rare-earths

Hoch, T., (2011). EU Strategy towards post-Soviet de facto states. *Journal of Contemporary European Studies* [online]. **6**(2), 69–85. [Viewed 2 February 2014]. Available from:

https://www.researchgate.net/publication/235985285_EU_Strategy_towards_Post-Soviet De Facto States

Hokazono. K., (2016). Shinkokusawomasu Nihonnoryoudo Bouei Hujyuubunna Houtekikonkyoeno Keikoku (The defence of the Japanese territory increases the severity of the warning about the lack of legal settings). *Gaikou.* **39**, 30–35.

Hornung, J., (2016). Constitutional revision in Japan - Why change is hard to come by. *Foreign Affairs* [online]. 2016 (July). [Viewed 28 March 2020]. Available from: https://www.foreignaffairs.com/articles/japan/2016-07-26/constitutional-revision-japan

Hornung, J., (2018). Strong but constrained Japan—Taiwan ties. *Brookings* [online]. [Viewed 13 March 2018]. Available from: https://www.brookings.edu/opinions/strong-but-constrained-Japan—Taiwan-ties/.

- Hseuh, R., (2014). Taiwan's treaty trouble The backlash against Taipei's China deal. *Foreign Affairs* [online]. 2014 (June). [Viewed 22 June 2017]. Available from: https://www.foreignaffairs.com/articles/china/2014-06-03/taiwans-treaty-trouble
- Hsieh, C. and Kao, E., (2017). Taiwan confirms China's deployment of DF-16 missiles. *Focus Taiwan* [online]. 20 March. [Viewed 20 March 2017]. Available from: http://focustaiwan.tw/news/acs/201703200015.aspx
- Hsieh, F., (2016). Watashinoshiten Taiwanno koukuukaigisanka Ryokoushano anzennihukaketsu (My perspective, Taiwan's participation in the international conference is necessary for the safety of tourists). *Asahi Shimbun*. 17 Sep. p.15.
- Hsieh, F., (2018a). Zhuugokuga Taiwankaikyou no Kinchou wo Takamete Iru (China is Raising the Tension in the Taiwan Strait). Seiron. April 2018, 110-112. Tokyo: Sangyou Keizai Shuppansha
- Hsieh, F., (2018b). Taiwan wo Torimaku Kokusai Kankyou (International Environment Surrounding Taiwan). *Shukan Sekai To Nihon (Weekly the World and Japan)*. 18 June. p.2.
- Hsiu-chuan, S., (2010). Japan extends ADIZ into Taiwan space. *Taipei Times* [online]. 26 June. [Viewed 15 January 2015]. Available from: http://www.taipeitimes.com/News/front/archives/2010/06/26/2003476438
- Hsu, K., (2015). Taiwan's global fisheries modestly advance its 'international space'. *U.S.-China Economic and Security Review Commission* [online]. [Viewed 16 July 2015]. Available from: http://origin.www.uscc.gov/sites/default/files/Research/Taiwan%20Fisheries.pdf.
- Hsu, S., (2019). Interview: Taiwan unfazed by Chinese obstruction: Joseph Wu. *Taipei Times* [online]. 22 March. [Viewed 22 March 2019]. Available from: http://www.taipeitimes.com/News/taiwan/print/2019/03/22/2003711958
- Huang. E., (2019). China's coastguard fuelling tensions in the Taiwan Strait. *The Strategist* [online]. 5 February. [Viewed 5 February 2019]. https://www.aspistrategist.org.au/chinas-coastguard-fuelling-tensions-in-the-taiwan-strait/
- Huang, M. (2014). Taiwan's changing political landscape: The KMT's landslide defeat in the nine-in-one elections. *Brookings Institute* [online]. 8 December. [Viewed 19 March 2015]. Available from:
- http://www.brookings.edu/research/opinions/2014/12/08-taiwan-political-landscape-elections-huang
- Hughes, C., (2015). Japan's foreign and security policy under the 'Abe Doctrine' New dynamism or new dead end? London: Palgrave Macmillan
- Hughes, C. W., (2017). Japan's strategic trajectory and collective self-defense: Essential continuity or radical shift? *The Journal of Japanese Studies* [online]. 43(1), pp.93-126. [Viewed 20 June 2019]. Available from: doi:10.1353/jjs.2017.0005
- Hughes, C., (2018). Japan's emerging arms transfer strategy: Diversifying to re-centre

on the US–Japan alliance. *The Pacific Review* [online]. 31(4), 424-440. Available from: doi: 10.1080/09512748.2017.1371212

ICAO (The International Civil Aviation Organization)., (2016). The UN Secretary-General addresses ICAO staff [online]. *Made a Speech by Bang Ki-moon and Directed by Fang Liu. Montreal: ICAO.YouTube.* [Viewed 20 June 2016]. Available from: https://www.youtube.com/watch?v=Y7vCsAwliMQ

Ikeda, T., (2010). Dai Nana Shou Nittai Kankei No Doutai (Chapter Seven The movement of Japan—Taiwan relations). *Nihon. Taiwan. Chugoku Kizukeru ka Aratana Kouzu, (Japan Taiwan China can we establish the new structure?).* Tokyo: Sankei Shimbun Shuppan. pp. 185-209

Ikeda, T., (2010). Taiwan wa Nihonto Soenninaruka (Will Taiwan drift away from Japan?). *Nihon • Taiwan • Chuugoku-Kizukeruka Aratana Kouzu* (Japan–Taiwan-China: Can we establish the new structure?). Tokyo: Sankeishimbun Shuppansha. pp. 32–54.

Ikegami, H., (2015). Baeikyu seiken shita no taiwan Keizai (Taiwanese economy under the Ma Ying Jeou's administration). *Toua [East Asia]*. **571**(January), 34–43.

Inada, T. (Center for Strategic & International Studies)., (2016). The evolving Japan-U.S. Alliance – Keeping Asia and the Pacific peaceful and prosperous [online]. *Hosted by Michael Green. YouTube.* Washington D.C.: CSIS. [Viewed 10 October 2016]. Available from: https://www.youtube.com/watch?v=A2nLfurKyOk

Isezaki., K, (2016). Japan self-defense forces participation in UN peacekeeping: An idea whose time is past. *Nippon.com* [Online]. 5 December. [Viewed 23 December 2016]. Available from: http://www.nippon.com/en/currents/d00275/

Ishizuka, K., (2013). The crisis management capability of Japan's self defense forces for UN peacekeeping, counter-terrorism, and disaster relief. *Japanese Journal of Political Science* [online]. **14**(2), 201-222. [Viewed 22 October 2018]. Available from: doi: http://dx.doi.org.sheffield.idm.oclc.org/10.1017/S1468109913000030

Ito. S., (2016). Doyouno shinohi wa iranai (Don't need a Dodge day). *Wedge*. **28**[8].58-70.

Iwami, T., (2018). Shinzo Abe's security legislation reform and peacebuilding in Mindanao. *The Pacific Review* [online]. **31**(5), 617–634. [Viewed 16 November 2017]. Available from: doi: 10.1080/09512748.2017.1391868

Jackson, R., (1987). Quasi-states, dual regimes, and neoclassical theory: International jurisprudence and the third world. *International Organization*. **41**(Autumn 1987), 531.

Jackson, R., (1993). *Quasi-states: sovereignty, international relations and the third world.* Volume 12. Cambridge: Cambridge University Press.

Jakobson, L, (2013) Implications of Taiwan-Japan landmark fishing agreement. *In: Tensions in the East China Sea-Papers presented at an international workshop organised by the Lowy Institute for International policy, Sydney, June 2013.* [Viewed 11 August 2019]. Available from: http://www.lowyinstitute.org/publications/tensions-

James, A., (1986). Sovereign statehood: The basis of international society. London: Allen & Unwin.

Jamieson, T., (2018). Securitization theory – Toward a replicable framework for analysis. In: M. Bertucci, J. Hayes, and P. James, (eds). *Constructivism reconsidered* – *Past, present, and future*. Michigan: University of Michigan Press. pp. 155–179.

Japan. Fisheries Agency of Japan., (2014). Joint Statement of the Bureau of Fisheries of People's Republic of China, the Fisheries Agency of Japan, the Ministry of Oceans and Fisheries of the Republic of Korea and the Fisheries Agency of Chinese Taipei on International Cooperation for Conservation and Management of Japanese Eel Stock and Other Relevant Eel Species, 17 September 2014[online]. Tokyo: Fisheries Agency of Japan. [Viewed 7 September 2020]. Available from: https://www.jfa.maff.go.jp/j/saibai/pdf/140917jointstatement.pdf

Japan Institute for International Policy Studies., (2017). Policy paper, announcement of proposal: 'The Trump administration and Japan: Challenges and visions for Japan's foreign and security policy in the new era', *IIPS*, 19 January [Online]. Tokyo: *Institute for International Policy Studies*. [Viewed 23 January 2017]. Available from: http://www.iips.org/en/research/2017/01/19093412.html

Japan International Cooperation Agency., (2005). *Japan disaster relief team members undergo training with counterparts from Taiwan, Singapore, and the Philippines, July* [online]. Tokyo: *Japan International Cooperation Agency*. [Viewed 9 March 2017]. Available from:

https://www.jica.go.jp/english/news/announcements/jica archive/2005/0507.html

Japan Interchange Association in Taipei. Ishihara, T., (2013). Taiwannaisei, Nittaikankei Wo Meguru Doukou (Nisenjyuusannen Sangatsu Jyoujyun - Nisenjyuusannen Gogatsu Jyoujyun)-[Nittaiminkan Gyougyou Torikime]Noteiketsuto Daiyongenpatsu Kensetsuno Kahi Wo Meguru Tenkai] (Internal development in Taiwan: Development regarding Japan—Taiwan relations [Early March 2013 — Early May 2013] — The signing of the Japan—Taiwan private fisheries pact and the controversy regarding the fourth nuclear power plant]), 23 May [online]. Taipei: Taiwan Jouhoushi Kouryuu. 866, 16-29. [Viewed 27 July 2015]. Available from: https://www.koryu.or.jp/Portals/0/images/publications/magazine/2013/5/05-03.pdf

Japan Interchange Association., (2013). Koueki Zaidan Houjin Kouryuukyoukai To Toua Kankei Kyoukai Tonoaidano Gyogyouchitsujo Ni Kansuru Torikime (Agreement on the establishment of a new order in the fishing industry between the Interchange Association, Japan, and the Association of East Asian Relations) [online]. Taipei: Japan Interchange Association [Viewed 4 March 2015]. Available from: http://www.koryu.or.jp/ez3_contents.nsf/15aef977a6d6761f49256de4002084ae/122d5 ed43161e84049257b490024d3d4/\$FILE/torikime.pdf

Japan Interchange Association., (2012). Kouryuu Kyoukaiwo Tsuujita Taiwanno Minasamaeno Genba Gaimudaijin No Messeeji (Foreign Minister Gemba's message to the people in Taiwan through the Interchange Association, 5 October 2012) [online]. Taipei: Japan Interchange Association [Viewed 24 May 2015]. Available from:

http://www.koryu.or.jp/taipei/ez3_contents.nsf/8e90bf63626460b049256dfa0026e3b2/8595d637b3d1966c49257a8e000e39d9?OpenDocument

Japan. Liberal Democratic Party., (2017). Seisaku, Dandoumisairu Boueino Jinsokukatsukonpontekina Kyouka Nikansuru Teigen Policy (The proposal on the swift and resolute consolidation of the Ballistic Missile, 30 March 2017. [Online]. Tokyo: Liberal Democratic Party. [Viewed 31 March 2017]. Available from: https://www.jimin.jp/news/policy/134586.html

Japan. Ministry of Defense., (2020). *Defense of Japan 2020: 4 Taiwan's Military capabilities Section 2 China Chapter 2* [online]. Tokyo: Ministry of Defense. [Viewed 12 August 2020]. Available from:

https://www.mod.go.jp/e/publ/w paper/wp2020/pdf/R02010202.pdf

Japan. Ministry of Defense., (1957). Fundamental concepts of national defense, May 1957 [online]. Tokyo: Ministry of Defense – Defense Policy. [Viewed 6 April 2017]. Available from: http://www.mod.go.jp/e/d_act/d_policy/dp02.html

Japan. MOFA., (2013). *Diplomatic bluebook 2014 summary* [online]. Tokyo: MOFA. [Viewed 17 May 2015]. Available from:

http://www.mofa.go.jp/policy/other/bluebook/2014/html/chapter1/japansdiplomacy.ht ml

Japan. MOFA., (2018). Dispatch of the expert team in response to the damages caused by the earthquake in eastern coast of Taiwan, 8 February [online]. Tokyo: MOFA. [Viewed 30 March 2019]. Available from: https://www.mofa.go.jp/press/release/press4e 001899.html

Japan. MOFA Foreign Minister Press Conference Record., (2019). *Gaimudaijin Kaikenkiroku, Nittaikan No Anzenhoshou Kyouryoku, Taiwa (Foreign minister press conference record, Japan–Taiwan Security Cooperation, Dialog, 8 March 2019)*. [Online]. Tokyo: security cooperation). MOFA. [Viewed 20 March 2019]. Available from: http://www.mofa.go.jp/mofaj/m/press/kaiken/gaisho/2019/03/08-3.html

Japan. MOFA., (2013). Saikinno Nittaikankei To Taiwan Jousei (The recent Japan—Taiwan relations and the situation in Taiwan, July) [online]. Tokyo: MOFA. [Viewed 14 July 2015]. Available from: http://aranishi.hobbyweb.net/3web ara/taiwan kankei.pdf

Japan. Naikakukanbou Naikakuhu Gaimushou Boueishou Cabinet Secretariat Cabinet Office MOFA MOD., (2015). [Heiwaanzen Housei] No Gaiyou Wagakuni Oyobi Kokusaishakai No Heiwa Notameno Kiremeno Nai Taiseinoseibi. The Guideline on [Peace and Security Legislation] (The arrangement for the seamless regime for our nation and international society's peace and security) [online]. Naikakukanbou Naikakuhu Gaimushou Boueishou Cabinet Secretariat Cabinet Office MOFA MOD. [Viewed 13 March 2017]. Available from: http://www.cas.go.jp/jp/houan/150515 1/siryou1.pdf

Japan. National Diet Library and Rippou KousaKyoku., (2016). [Kankoku] Heiwaanzenhousei No Seiritsuni Taisuru Kankokugawano Hannou [South Korea] (The reaction from the Korean side regarding the enactment of peace and security

legislation) [online]. Tokyo: Kikuchi Yuji. [Viewed 23 October 2017]. Available from: http://dl.ndl.go.jp/view/download/digidepo_9593148_po_02660117.pdf?contentNo=1

Japan. National Institute for Defence Studies., (2017). NIDS China security report 2017 change in continuity: The dynamics of the China-Taiwan Relationship, February [online]. Japan: National Institute for Defence Studies. [Viewed 18 March 2017]. Available from:

http://www.nids.mod.go.jp/publication/chinareport/pdf/china_report_EN_web_2017_ A01.pdf

Japan. National Police Agency., (2012). *Keisatsu Shou Hakusho Dainisetsu Hanzaino Guroubarukani Taiousuru Tameno Senryaku 6 Guroubaruna Kokusai Kyouchou Taiseino Kouchiku* (National Police Agency White Paper Chapter Two: The strategy to cope with the globalization of crime: The build-up of the global cooperation structure) [online]. Tokyo: National Police Agency. [Viewed 10 August 2016]. Available from: https://www.npa.go.jp/hakusyo/h22/honbun/html/md260000.html.

Japan National Press Club., (2017). Lai Ching-Te Tainanshi shichou (Lai Ching-Te Mayor, Tainan city) [online]. Directed by Kenji Bando.Tokyo: Japan National Press Club. *YouTube*. [Viewed 22 February 2019]. Available from: https://www.youtube.com/watch?v=L0aq93k0PVs

Japan. Nihon Ritouki No Kai in Aichi Branch., (2015). Abeshushou, Sangiinno Iinkaishitsugide [Taiwanwa Jyuuyouna Paatonaa, Nihonnotaisetusna Yuujin] To Touben (PM Abe said Taiwan is an important partner and friend for Japan in his answer during the committee Q&A at the upper house) *Nihon Ritoukinokai Aichikenshibu* [online]. 1 August. [Viewed 1 April 2017]. Available from: http://ritoukiaichi.com/kyoei/%E5% AE% 89% E5% 80% 8D% E9% A6% 96% E7% 9B% B8% E3% 80% 81% E5% 8F% 82% E8% AD% B0% E9% 99% A2% E3% 81% AE% E5% A7% 94% E5% 93% A1% E4% BC% 9A% E8% B3% AA% E7% 96% 91% E3% 81% A7% E3% 80% 8C% E5% 8F% B0% E6% B9% BE% E3% 81% AF% E9% 87% 8D% E8% A6% 81% E3% 81% AA/20150801

Japan. NHK., (2016). *KenpouNi Kansuru Ishikichousa (The investigation regarding awareness of constitution)* [online]. NHK. [Viewed 8 January 2017]. Available from: http://www.nhk.or.jp/bunken/research/yoron/pdf/20160506 1.pdf

Japan. Prime Minister of Japan and His Cabinet., (2014). *Abe Naikaku Souridaijin Kishakaiken (Abe prime ministerial press conference). 15 May* [online]. Tokyo: Prime Minister of Japan and His Cabinet. [Viewed 8 January 2017]. Available from: http://www.kantei.go.jp/jp/96 abe/statement/2014/0515kaiken.html

Japan. Prime Minister of Japan and His Cabinet., (1946). *The Constitution of Japan, Chapter 2, Renunciation of war, Article 9. 3 November 1946*. [online]. Tokyo: Prime Minister of Japan and His Cabinet. [Viewed 3 March 2017]. Available from:http://japan.kantei.go.jp/constitution_and_Jgovernment_of_japan/constitution_e. html

Japan. Taipei Economic and Cultural Representative Office., (2016). Shinnankou Seisaku Kouryo wo Saitaku, Chuugoku Tairiku Tono Taiwamo-Soutouhu (Implementing

the New Southbound Policy, Dialogue with Mainland China is Also Possible-Presidential Palace, 18 August) [online]. Taipei: *Economic and Cultural Representative Office Tokyo*. [Viewed 23 August 2016]. Available from: http://web.roctaiwan.org/jp_ja/post/37437.html

Japan. Taipei Economic and Cultural Representative Office., (2016). *Taiwantonihon, Sabougijyutukouryuude Tounanajiano Bousaini Kouken (Taiwan and Japan want to enhance sand control cooperation to contribute to anti-disaster efforts in Southeast Asia, 20 September* [Online]. Tokyo:). *Taipei Economic and Cultural Representative Office*. [Viewed 4 February 2017]. Available from: http://www.taiwanembassy.org/jp_ja/post/39090.html

Japan. The House of Representatives., (2015). Dai 189 KaiKokkai Wagakuni Oyobi Kokusaishakai No Heiwahousei Nikansuru Tokubetsuiinkai Dai 10 Gou The 189th Session on the National Diet (The special committee for our nation and international society's peace and security legislation: The Tenth Session, 15th June [Online]. Tokyo: The House of Representatives. [Viewed 30 March 2017]. Available from: http://www.shugiin.go.jp/internet/itdb_kaigirokua.nsf/html/kaigirokua/029818920150 615010.htm

Japan. The Ministry of Defense., (2014). *Measures for self-defense permitted under Article 9 of the Constitution, the government's view on Article 9 of the Constitution, the Constitution and the basis of defense policy white paper section 2* [online]. Japan: The Ministry of Defense. [Viewed 3 March 2017]. Available from: http://www.mod.go.jp/e/publ/w paper/pdf/2014/DOJ2014 2-1-2 web 1031.pdfJapan.

Japan. The Ministry of Defense., (2013). The right of collective self-defense, the government's view on Article 9 of the Constitution, section 2 the Constitution and the right of self-defense white paper [online]. Tokyo: The Ministry of Defense. [Viewed 1 March 2017]. Available from:

http://www.mod.go.jp/e/publ/w_paper/pdf/2013/22_Part2_Chapter1_Sec2.pdf

The Japan Times., (2013). Taiwan leader urges three-way dialogue on Senkaku resources. *Japan Times* [online]. [Viewed 20 March 2017]. Available from: https://www.japantimes.co.jp/news/2013/06/07/national/politics-diplomacy/taiwan-leader-urges-three-way-dialogue-on-senkaku-resources/#.XMi3Y-hKg2w

The Japan Times., (2016). Taiwan's China airlines announces strike, disrupting flights to Japan. Japan Times [online]. 24 June. [Viewed 24 June 2016]. Available from: http://www.japantimes.co.jp/news/2016/06/24/business/corporate-business/taiwans-china-airlines-announces-strike-disrupting-flights-japan/#.V6dOWbiLQ2w

The Japan Times., (2017). SDF to carry out simulated exercise based on possible China-Taiwan clash. The Japan Times [online]. 19 January. [Viewed 20 January 2017]. Available from: https://www.japantimes.co.jp/news/2017/01/19/national/sdf-carry-tabletop-exercise-based-possible-china-taiwan-clash/#.XJ1ZrJj7Q2x

Jeffrey, J., (2016). Somaliland's search for place - The unofficial republic's struggle for acknowledgement. *Foreign Affairs* [online].26 May. [Viewed 8 January 2019]. Available from: https://www.foreignaffairs.com/articles/somalia/2016-05-

Jiang, Y., (2013). Shòu měi yālì qì liǎng'àn bǎo diào lù gāocéng jí bùmǎn mǎyīngjiǔ [Being pressured by the US to abandon cross-strait defence: Diaoyutai Gao Cengji is dissatisfied with Ma-Ying-Jeou. *DWnews* [online]. 5 March. [Viewed 14 July 2015]. Available from: http://taiwan.dwnews.com/news/2013-03-05/59153048.html

Jing, Q., (2015). Nihonseino Jyunshiseiga Intai Mitsuyutorishimari Nadonikatsuyaku (Taiwan Japan-made coast guard patrol ship has resigned played a role in missions like anti-drug smuggling/Taiwan) *Focus Taiwan*. [Online]. 1 February. [Viewed 29 December 2016]. Available from:

http://japan.cna.com.tw/news/asoc/201502010001.aspx#.VM3O75fMII0.facebook

Jing, S. (2013). Taiwan: Negotiating self-identity and security. *Japan and China as Charm Rivals – Soft Power in Regional Diplomacy*, 122–149.

Johnson, C., (1972). How China and Japan see each other. *Foreign Affairs* [online]. **2020**(July). [Viewed 22 July 2020]. Available from:

https://www.foreignaffairs.com/articles/asia/1972-07-01/how-china-and-japan-see-each-other

Kallender, P. and Hughes, C. W., (2019) Hiding in plain sight? Japan's militarization of space and challenges to the Yoshida Doctrine. *Asian Security*, **15**(2), 180–204, DOI: 10.1080/14799855.2018.1439017

Kaneda, H., and Morimoto, S., (2016). Zatsudan • Chuugokuno Kaiyoushinshutsu Ni Nihonwa Doutaiou Subekika Chitchat • (How should Japan deal with Chinese maritime advancement). In: S., Morimoto ed. "Kaiyou Kokka" Chuugokuni Nihonwa Doutachimukaiuka (How should Japan stand up to China as a maritime state?). Tokyo, Japan: Nihonjitsugyoushuppansha. pp. 249–321

Kaneda, H. (2019). Written correspondence by the author of this dissertation 25 May 2019.

Kan, F., Center For Security Studies at National Cheng Chi University., (2012). Taiwan's new foreign relations and national security. *MCSSTW* [online]. [Viewed 5 August 2016]. Available from:

http://www.mcsstw.org/web/content.php?PID=5&Nid=947

Kao, S., (2016a). Taiwan to ICAO: Let us in. *Aviation Week Network* [online]. 21 September. [Viewed 21 September 2016]. Available from: http://aviationweek.com/commercial-aviation/taiwan-icao-let-us

Kao, S., Washington Times., (2016b). Locking Taiwan out of ICAO unwise. *Washington Times* [online]. [Viewed 28 September 2016]. Available from: http://www.washingtontimes.com/news/2016/sep/28/letter-to-the-editor-locking-taiwan-out-of-icao-un/

Kawashima, S., (2013). Rekishiteki Nittai Gyougyoukyoutei Teikeitsu-Sono Igito Kadai (Historic signing of the Japan–Taiwan fisheries agreement – What it means and the issues). *Nippon.Com Homepage* [online]. [Viewed 2 February 2016]. Available

from: http://www.nippon.com/ja/currents/d00081/

Kawashima, S., (2016). The formation of the new government in Taiwan and Japan—Taiwan relations. *Asia-Pacific Review* [online]. 23(1), 42-64. [Viewed 7 July 2016]. Available from: DOI: 10.1080/13439006.2016.1200320

Kawashima, S., (2019). Taiwan policy at a moment of rising US-China tension, A Japanese perspective. *The Asian Forum* [online]. **7**(4). [Viewed 10 July 2019]. Available from: http://www.theasanforum.org/a-japanese-perspective-11/?fbclid=IwAR3hV4DO6dxF30rDieK0AMV4Nf4yigsx-t4PFH4_AnbLE9KFY30X47FrhlM

Keene, E., (2002). Beyond the anarchical society: Grotius, colonialism and order in world politics. New York, NY: Cambridge University Press.

Keiji, F., (2019). Personal written correspondence with Former State Minister Furuya Keiji (24 May 2019)

Keller, J., (2016). Raytheon to upgrade Taiwan missile-defense surveillance radar to mitigate obsolescence issues. *Military Aerospace Electronics* [online]. 30th November. [Viewed 15 March 2017]. Available from: http://www.militaryaerospace.com/articles/2016/11/radar-surveillance-upgrade.html?utm_source=feedburner&utm_medium=feed&utm_campaign=Feed%3 A+LandTechnology+(Land+Technology)

Kelly, T. and Kubo, N., (2017). Exclusive: Japan plans to send largest warship to South China Sea, sources say. *Reuters* [online]. 4 March. [Viewed 4 March 2017]. Available from: http://www.reuters.com/article/us-japan-navy-southchinasea-exclusive-idUSKBN16K0UP?utm_campaign=trueAnthem:+Trending+Content&utm_content=5 8c66ed204d301315ae75398&utm_medium=trueAnthem&utm_source=facebook

Ker-Lindsay, J. and Berg, E., (2018). Introduction: A conceptual framework for engagement with de facto states. *Ethnopolitics* [online]. **17**(4), 335–342. [Viewed 4 January 2019]. Available from: doi: 10.1080/17449057.2018.1495362

Ker-Lindsay, J., (2013). Engaging with contested states: how much interaction constitutes recognition? *E-International Relations* [online]. 10 September. [Viewed 3 March 2014]. Available from: http://www.e-ir.info/2013/09/10/engaging-with-contested-states-how-much-interaction-constitutes-recognition/

Ker-Lindsay, J., (2015). Engagement without recognition: The limits of diplomatic interaction with contested states. *International Affairs* [online]. **91**(2), 267–285. [Viewed 30 December 2018]. Available from: doi: 10.1111/1468-2346.12234

King, C., (2001). The benefits of ethnic war: Understanding Eurasia's unrecognized states. *World Politics* [online]. **53**(4), 524–552. [Viewed 24 May 2014]. Available from: http://www.jstor.org/stable/25054164

Kingston, J., (2018). Shadow boxing: Japan's para-diplomacy with Taiwan. In: J. Brown, and J. Kingston, eds. *Japan's Foreign Relations in Asia*. New York: Routledge. pp. 201–217.

Kiyomizu, M., (2010). Kan Seiken Ga Minogashita Chuugoku Tsuyoki No Ayausa (The danger of China's aggressiveness which Kan administration evaded). *ChuoKoron* [online]. Nov. [Viewed 2 April 2018]. Available from: http://www.chuko.co.jp/chuokoron/2010/12/post_43_2.html

Knudsen, T., (2016). Solidarism, pluralism and fundamental institutional change. In: P. Wilson, (ed.). *The English School in retrospect and prospect: Barry Buzan's An Introduction to the English School of International Relations: The Societal Approach*, September 2013, Warsaw, Poland. London: SAGE Journals, pp. 102–109. [Viewed 2 January 2019]. Available from: doi: 10.1177/0010836715610595

Koda, Y., (2017). China's blue water navy strategy and its implications – China's Bluewater navy series. *Center for a New American Security* [online]. [Viewed 20 March 2017]. Available from:

https://s3.amazonaws.com/files.cnas.org/documents/Koda BWN.pdf

Koldas, U., Tuzunkan, M., Ayaz, E., and Cirakli, M., (2018). Branding and internationalization of higher education in an unrecognized state through sports: The women basketball team of Near East University in Northern Cyprus. *Quality and Quantity* [online]. **52**(1), 725–735. [Viewed 31 December 2018]. Available from: doi: 10.1007/s11135-017-0649-6

Kolosov, V. and O'Loughlin, J., (1998). Pseudo-states as harbingers of a new geopolitics: The example of the Trans-Dniester Moldavian Republic (TMR). *Geopolitics* [online]. 3, 151–176. [Viewed 29 September 2013]. Available from: doi: 10.1080/14650049808407612.

Kolstø, P., (2006). The sustainability and future of unrecognized quasi-states. *Journal of Peace Research* [online]. **43**(6), 723–740. [Viewed 17 July 2014]. Available from: doi: 10.1177/0022343306068102

Konno. T., (2016). Mitsuyuno Jittai Wo Haakushiteiruga Torishimari Wa Konnan [Despite (ROC Government) having a grasp of the smuggling issue, cracking down on it is difficult]. *Wedge.* **28**(8),67.

Kono, T., (2018). Interview: Foreign Minister Kono Taro Japan's diplomatic landscape in 2018 pushing North Korea toward denuclearization by applying continued pressure. *Discuss Japan - Japan Foreign Policy Forum* [online]. **47**. [Viewed 27 February 2019]. Available from: https://wp-en.japanpolicyforum.jp/archives/diplomacy/pt20180529193613.html?fbclid=IwAR2y xpMitbFCemTORSzeLBy_Qk-RHqfvZC9kw5H0foAldtiiu4sJA3eMUeM

Kotani, T. and Auer, J., (2005). Reaffirming the 'Taiwan Clause': Japan's national interest in the Taiwan Strait and the U.S.-Japan alliance. *NBR Analysis Japan–Taiwan Interaction: Implications for the United States* [online]. 1st edn. **16**(1), 58–83. [Viewed 5 July 2017]. Available from:

http://www.nbr.org/publications/nbranalysis/pdf/vol16no1.pdf

Kotani, T., (2014). Reviewing the first year of China's ADIZ: A Japanese perspective. *Asia Maritime Transparency Initiative* [online]. 25 November. [Viewed 30 July 2016]. Available from: https://amti.csis.org/reviewing-the-first-year-of-chinas-adiz-a-

japanese-perspective/

Kotani, T., (2016a). A Japanese perspective on how Japan and Taiwan should deal with Okinotori-shima. *Global Taiwan Institute* [online]. **1**(8). 9 November [Viewed 10 November 2016]. Available from: http://globaltaiwan.org/2016/11/09-gtb-1-8/#TetsuoKotani110916

Kotani, T., (2016b). Taipei's risky agreement with Beijing on Okinotorishima. *Asia Maritime Transparency Initiative* [online]. 10 May. [Viewed 11 May 2016]. Available from: https://amti.csis.org/taipeis-risky-agreement-beijing-okinotorishima/

Koyama, K., (2016). Ishigaki accepts deployment of GSDF forces to shore up defense. *The Asahi Shimbun* [online]. 27 December. [Viewed 27 December 2016]. Available from: http://www.asahi.com/ajw/articles/AJ201612270055.html

Krejsa, H., Cho, A., (2017). Is Beijing adopting an ethnonationalist foreign policy? Warning signs emerge. *Foreign Affairs* [online]. **2017** (October). [Viewed 11 March 2020]. Available from: https://www.foreignaffairs.com/articles/asia/2017-10-23/beijing-adopting-ethnonationalist-foreign-policy

Kui-hsiang, W. and Hsu, E., (2019). Taiwan, Palau sign sea patrol cooperation pact. *Focus Taiwan* [online]. 22 March. [Viewed 23 March 2019]. Available from: http://focustaiwan.tw/news/aipl/201903220021.aspx?fbclid=IwAR2_iVv9AU9ixDJyr9-hmPmiJ48hsoalphETQGYinXayvNaKS5o4ZTEfQXc

Kurlantzick, J., Carnegie Council., (2016). Southeast Asia—The Islamic state's new front? *Carnegie Council* [online]. [Viewed 7 October 2016]. Available from: http://www.carnegiecouncil.org/publications/ethics_online/0122

Lai, I-C., (2009). Yonaguni plans raise questions of Taiwan. *Taipei Times* [online]. 6 July. [Viewed 14 April 2017]. Available from: http://www.taipeitimes.com/News/editorials/archives/2009/07/06/2003447948/1

Lai, J., (2014). Sovereignty and 'civilization': International law and East Asia in the nineteenth century. *Modern China* [online]. **40**(3), 282–314. [Viewed 27 December 2018]. Available from: www.jstor.org/stable/24575657

Lande, E., (2018). Between offensive and defensive realism — The Japanese Abe Government's security policy toward China. *Asian Security* [online]. **14**(2), 172-192. [Viewed 1 August 2019] Available from: doi: 10.1080/14799855.2017.1323882

Leaf, P., (2014). Promise and potential peril: Japan's military normalization. *The Diplomat* [online]. 4 September. [Viewed 30 December 2016]. Available from: http://thediplomat.com/2014/09/promise-and-potential-peril-japans-military-normalization/

Lebow, R., (2016). *National and International Relations*. Cambridge: Cambridge University Press.

Lee, T. H. (1999). Understanding Taiwan: Bridging the Perception Gap. *Foreign Affairs* [online]. **1991**(November/December). [Viewed 22 July 2020]. Available from:

- https://www.foreignaffairs.com/articles/asia/1999-11-01/understanding-taiwan-bridging-perception-gap
- Leng, T. and Liao, N., (2016) Hedging, strategic partnership, and Taiwan's relations with Japan under the Ma Ying jeou administration. *Pacific Focus* [online], **31**(3), 357–382. [Viewed 30 September 2018]. Available from: doi: 10.1111/pafo.12080.
- Lieberthal, K., (2005). Preventing a war over Taiwan. *Foreign Affairs* [online]. **2005**(March/April). [Viewed 22 July 2020]. Available from: https://www.foreignaffairs.com/articles/asia/2005-03-01/preventing-war-over-taiwan
- Lim, E. and Wang, J., (2013). 'No harm' in fishery talks with Taiwan: Philippine president. *Focus Taiwan* [online]. 9 June. [Viewed 3 July 2015]. Available from: http://cached.newslookup.com/cached.php?ref_id=270&siteid=2210&id=2208555&t =1370782121
- Lim, J-C. T., (2018). External and internal reconciliation: War memories and views of history regarding Japan in post-war Taiwan. *Japan Review* [online]. **2**(2), 61-65. [Viewed 2 February 2019]. Available from: https://www.jiia-jic.jp/en/japanreview/pdf/JapanReview_Vol2_No2_06_LIM.pdf?fbclid=IwAR10nrUJ esS2tbWZSh7UAKK9IltIsaWNBfQ3U-q3UfwiF6w70AwKFgWWJKM
- Lin. C., (2019). Rìběn qián zhù tái dàibiǎo zhǎotián: Cài hán chājù kŏng zài 5%zhī nèi (The former representative Numata in Taiwan: Tsai and Han's gap is within the five percent). *Ziyou Shibao* [online]. 23 December. [Viewed 23 December 2019]. Available from:
- https://news.ltn.com.tw/news/politics/breakingnews/3017232?utm_medium=P&fbclid =IwAR3WbHbRrldEC9Poh3NrRZAKg-tGOk5cmpAjvi5UN1bVzTbmvQUOiUbbOa8
- Lind, J., (2009). The perils of apology What Japan shouldn't learn from Germany. *Foreign Affairs* [online]. **2009** (May/June). [Viewed 1 March 2020]. Available from: https://www.foreignaffairs.com/articles/japan/2009-05-01/perils-apology
- Link, P., (2015). What it means to be Chinese Nationalism and identity in Xi's China. *Foreign Affairs* [online]. **2015** (May/June). [Viewed 11 August 2019]. Available from: https://www.foreignaffairs.com/articles/china/2015-04-20/what-it-means-be-chinese
- Lin, S. C., Shih, Y. C. and Chiau, W. Y., (2013). An impact analysis of destructive fishing and offshore oil barges on marine living resources in Taiwan Strait. *Ocean & Coastal Management.* **80**, 119–131.
- Lin, T., (2016a). Okinotori: A new beginning for Taiwan-Japan maritime dialogue. *Asia Maritime Transparency Initiative* [online]. 17th June. [Viewed 2 January 2017]. Available from: https://amti.csis.org/okinotori-new-beginning-taiwan-japan-maritime-dialogue/
- Lin, T., (2016b). The strategic significance of the Taiwan-Japan maritime affairs dialogue [online]. *Asia Maritime Transparency Initiative* [online]. 21 December. [Viewed 3 January 2017]. Available from: https://amti.csis.org/taiwan-japan-maritime-

- Lin, X., (2014). Taiwan Kaikyou No Anzen Hoshou Ni Okeru Nihon No Yakuwari (The role of Japan in security of the Taiwan Strait) (Symposium). *Higashi Ajia Kokusai Kankei no Dainamizumu [The Dynamism of the International Relations of the East Asia]*, 2014, Aoyamagakuin University and National Taiwan Normal University. Tokyo: Aoyama Journal of International Studies, pp. 25–39. [Viewed 22 January 2017]. Available from: http://jairo.nii.ac.jp/0132/00040504
- Little, R., (2015). Reassessing the expansion of international society. In: R. Murray, (ed.). *System, society and the world exploring the English School of International Relations* [online]. (2nd ed). Bristol: E-International Relations Publishing, pp. 24–33. [Viewed 5 January 2019]. Available from: https://www.e-ir.info/2015/12/05/edited-collection-system-society-and-the-world-exploring-the-english-school/
- Liu, J., (2006). Rì měi tóngméng zhuǎnxíng jí qí duì zhōngguó de yǐngxiǎng (The transformation of the Japan-US alliance and its impact on China). *Guójì guānchá International Observation* [online]. 1, 9-16 [Viewed 5 March 2019]. Available from: file:///C:/Users/michinori/Downloads/%E6%97%A5%E7%BE%8E%E5%90%8C%E 7%9B%9F%E8%BD%AC%E5%9E%8B%E5%8F%8A%E5%85%B6%E5%AF%B9%E4%B8%AD%E5%9B%BD%E7%9A%84%E5%BD%B1%E5%93%8D_%E5%88%98%E6%B1%9F%E6%B0%B8%20(2).pdf
- Liu, L., (2013). Taiwanga P-3C Shoukaikiwo Jyuryou Basoutoumo Manzokuge (Taiwan received P-3 maritime patrol aircraft; even President Ma seems satisfied). *Focus Taiwan* [Online]. 31 March. [Viewed 1 April 2017]. Available from: http://japan.cna.com.tw/news/achi/201310310005.aspx
- Li, Y., and Zhang, E., (2017). Changing Taiwanese identity and cross-strait relations: A post-2016 Taiwan presidential election analysis. *J of Chin Polit Sci* **22**, 17–35 (2017). https://doi.org/10.1007/s11366-016-9452-9
- Lostumbo, M., (2016). Taiwan forced to rethink its air defense strategy. *Defense News* [online]. 13 April. [Viewed 20 December 2016]. Available from: http://www.defensenews.com/story/defense/commentary/2016/04/13/taiwan-forced-rethink-its-air-defense-strategy/82897760/
- Lo, T-p., (2017). Patriot missiles deployed on east coast. *Taipei Times* [online]. 5 February. [Viewed 5 February 2017]. Available from: http://www.taipeitimes.com/News/front/archives/2017/02/05/2003664364
- Lowther, W., (2013). Schriver says 'constructive' role needed in Diaoyutais. *Taipei Times* [online]. 10 February. [Viewed 18 March 2017]. Available from: http://www.taipeitimes.com/News/taiwan/archives/2013/02/10/2003554664
- Lu, H. and Hou, E., (2017). Taiwan can strike enemy troops at their home bases: Defense official. *Focus Taiwan* [Online]. 16 March. [Viewed 16 March 2017]. Available from: http://focustaiwan.tw/news/aipl/201703160011.aspx
- Lynch, D., (2018). Playing the Taiwan card. *Foreign Affairs* [online]. 19 March. [Viewed 2 October 2018]. Available from: https://www.foreignaffairs.com/articles/china/2018-03-19/playing-taiwan-card

Mainichi Shimbum., (2004). Taiwan-nyukoku jizen shinsa wo kaishi (Taiwan commenced the pre-clearance system). Mainichi Shimbun. 8 Aug.

Mainichi Shimbum., (2016a). Seihuto Kokuren Tai-tero Ajiashiende Goui, Mizugiwa Taisakuwo Kyouka (The UN and governments agreed on capacity building for anti-terror in Asia: The consolidation of measures before entering the country). *Mainichi Shimbun* [online]. 9 Oct. [Viewed 9 October 2016]. Available from: https://mainichi.jp/articles/20161009/k00/00m/030/050000c

Mainichi Shimbun., (2016b). ASDF rescue flight for Japanese in South Sudan will also evacuate foreigners. Mainichi Shimbun [online]. 12 July. [Viewed 12 July 2016]. Available from: http://mainichi.jp/articles/20160712/ddm/002/030/116000c

Mainichi Shimbun Editorial., (2017). Rikuji PKO Tesshuu Hutoumeisa Nokoru Seihusetsumei (The withdrawal of the GSDF PKO: The lingering non-transparency remains in governmental explanation) Governmental Explanation. 13 March. Mainichi Shimbun [Online].13 march. [Viewed 13 March 2017]. Available from: http://mainichi.jp/articles/20170313/ddm/005/070/002000c

Maoz, Z., (2002). Case study methodology in international studies. In: F. Harvey, and M. Brecher, eds. *Evaluating methodology in international studies: Realism and institutionalism in international studies*. Michigan: Michigan Press. pp. 161–194.

Marysia, Z., and Enloe, C., (1995). Questions about identity in international relations', in Booth, K. and Smith, S., eds. *International relations theory today*. Cambridge: Polity Press. p. 299.

Matsuda, Y., (2009). Dairokushou Taiwanno Minshukato Aratana Nittaikankeino Mosaku-1988-94nen (Chapter Six: Taiwan's democratization and the search for new Japan – Taiwan Relations from 1988–94). In: Kawashima, S., Shimizu, R., Matsuda, Y. and Yang, Y., eds. *Nittaikankeishi 1945 – 2008 (History of Japan–Taiwan Relations 1945 – 2008*). Tokyo: The Tokyo University. pp. 231 – 241.

Matsuda, Y., (2012). Taiwan mondai no shintenkai (The new development of the Taiwan issue). In: R. Iechika, Y. Matsuta and R. Duan, eds. *Kiro Ni Tatsu Nicchuu Kankei-Kako Tono Taiwa-Mirai E No Sousaku (The Sino-Japanese relations in conundrum-The dialogue with the past and search for the future*). Kyoto: Koyoshoubou. pp. 222-251.

Matsuda, Y., (2013). Japan—Taiwan relations under DPJ and KMT administrations in international context. Japan and Taiwan in a new era possible effects and influences towards its relationship. 2 March 2013, Minato-ku, Tokyo, Japan [online]. Tokyo: Ocean Policy Research Foundation pp. 118–136. [Viewed 7 April 2020]. Available from: https://www.spf.org/_opri_media/publication/pdf/201303_16.pdf

Matsuda, Y., (Center for Strategic & International Studies., (2014). Japanese perspectives on China, Taiwan and Cross-Strait relations - 1 [online]. *Directed by Bonnie Glaser. YouTube*. Washington D.C.: CSIS. [Viewed 13 January 2015]. Available from: https://youtu.be/IT7iCx3pNiE

Matsuda, Y., (2018). The strategic impact of the Taiwan Issue on the U.S.-Japan alliance.

Japanese views on China and Taiwan. In: *Implications for U.S.-Japan alliance*.1 March 2018. Washington D.C.: CSIS, pp.1-13. [Viewed 1 March 2019]. Available from: https://csis-prod.s3.amazonaws.com/s3fs-public/180301_CSIS_US-Japan%28Matsuda%29.pdf?s8OjBJE9eIIvqXHYkyDAaNfeSx_CtHRH

Matsuda, Y., (2019). Indo-Pacific 'strategies' and their implication for Taiwan: A Japanese perspective. *Global Taiwan Brief* [online]. **4**(3). [Viewed 29 January 2019]. Available from: http://globaltaiwan.org/2019/02/vol-4-issue-3/?fbclid=IwAR0P2VPxej6izudYjPMrPjViknIkoJtWm3cRCvbuvOmcVCTZMv0p2I pgdT0#YasuhiroMatsuda02132019

Matsumura, M., (2010). Daisanshou Taiwankaikyouni Detantowa Kanouka-Nihonno Taiwanseisakuni Okerusentakushi (Chapter 7 Is a Détente in the Taiwan Strait possible? The choices regarding Japan's Taiwan policy). *Higashiajia Chitsujoto Nihonno Anzenhoshousenyaku* (The East Asian order and Japan's security strategy). Tokyo: Ashishoubou. pp.110-127

Ma, Y., (2015). Mă zŏngtŏng tóu shū: Diàoyútái dāngrán shì zhōnghuá mínguó lǐngtŭ lǐdēnghuī yīng dàoqiàn (President Ma's Writing: Diaoyutai is certainly part of the ROC – Lee-Teng-hui must apologize). *Chinatimes* [online]. 3 August. [Viewed 3 August 2015]. Available from: http://www.chinatimes.com/cn/newspapers/20150803000427-260109

Mazza, M., (2018). Is a storm brewing in the Taiwan Strait? *Foreign Affairs* [online]. [Viewed 2 October 2018]. Available from: https://www.foreignaffairs.com/articles/asia/2018-07-27/storm-brewing-taiwan-strait

McDevitt, M., (2005). Taiwan: The tail that wags dogs. *NBR* [online]. **16**(1), 14–37. [Viewed 15 February 2019]. Available from: https://www.nbr.org/publication/Japan—Taiwan-interaction-implications-for-the-united-states/

McDonald, M., (2013). Constructivisms. In: P. D. Williams, ed. *Security studies an introduction*. 2nd ed. Oxon: Routledge. pp. 63–76

McNamara, K., (2014). The EU after Ukraine. *Foreign Affairs* [online]. 4 March [Viewed 5 March 2014]. Available from: http://www.foreignaffairs.com/articles/140991/kathleen-r-mcnamara/the-eu-after-ukraineMearsheimer, J., (2001). *The Tragedy of Great Power Politics*. New York: W. W. Norton.

Mearsheimer, J., (2014). Say goodbye to Taiwan. *National Interest* [online]. (25 February). [Viewed 18 January 2019]. Available from: https://nationalinterest.org/article/say-goodbye-taiwan-9931

Miller, J. B., (2014). Battle-ready Japan? The real story behind Tokyo's first national security strategy. *Foreign Affairs* [online], 10 January. [Viewed 9 January 2017]. Available from: https://www.foreignaffairs.com/articles/china/2014-01-10/battle-ready-japan

Miller, J. B., (2017). Japan's North Korea options, will Tokyo equip itself for a preemptive strike? *Foreign Affairs* [online]. 15 March, [Viewed 9 April 2017]. Available from: https://www.foreignaffairs.com/articles/japan/2017-03-15/japans-

north-korea-options

Ministry of Foreign Affairs, ROC Taiwan [MOFA_Taiwan]., (2019). [WRONG! The bill was passed by our national parliament & will be signed by the president soon. Democratic #Taiwan is a country in itself & has nothing to do with authoritarian #China. @PDChina is a commie brainwasher & it sucks.] [Twitter]. 19 May. [Accessed 19 May 2019]. Available from: https://twitter.com/MOFA_Taiwan/status/1129901169906552833Japan.

Ministry of Foreign Affairs of Japan., (1972). *Joint Communique of the Government of Japan and the Government of the People's Republic of China, 29 September 1972* [online]. Tokyo: Ministry of Foreign Affairs of Japan. [Viewed 11 March 2019]. Available from: https://www.mofa.go.jp/region/asia-paci/china/joint72.html

Monma, R., (2015). Hong Xiuzhu Shi, Kokumintou Soutou Kouho He Zenshin, Taiwan No Doukou Gogatsu ~ Rokugats (Mrs Hong Xiuzhu advances as a presidential candidate in the KMT: The developments in Taiwan From May to June). Toua (East Asia). 577, 54 - 64.

Monma, R., (2017a). Shin Gunji Seisenryaku Wa - Bouei Koshu, Jyuusou Yokushi (New defence strategy is 'resolute defence, multi-domain deterrence'). Toua [East Asia]. 598 (April), pp. 56–65.

Monma. R., (2017b). Zen Chuugoku shakaiin taiwankenkyuujo shochouga chuutatouitsu no jikanhyouni genkyuu (The former director of the Chinese Academy of Social Sciences mentioned the timetable for China-Taiwan unification). Toua [East Asia]. 603 (September), pp. 60–70.

Monma. R., (2018). Dendousha hukyuu he ookiku kajiwo kitta Taiwan (Taiwan made a large steer toward the electric cars). *Toua [East Asia]*. **608** (February), pp. 58–68.

Mori, S., (2015). The new security legislation and Japanese public reaction. *The Tokyo Foundation for Policy Research* [online]. [Viewed 15 October 2017]. Available from: https://www.tkfd.or.jp/en/research/detail.php?id=542

Morris, A., (2015). Introduction: Living as left behind in postcolonial Taiwan. In: A. Morris, (ed.). *Japanese Taiwan colonial rule and its contested legacy*. London: Bloomsbury. pp. 3-23.

Morris, L., (2017). History and current developments regarding Taiwan's Coast Guard administration. *Global Taiwan Institute* [online]. **2**(7). [Viewed 16 February 2017]. Available from: http://globaltaiwan.org/2017/02/15-gtb-2-7/

Morris, L., (2017). History and current developments regarding Taiwan's Coast Guard administration. *Objective Analysis. Effective Solutions* [online]. 15 February. [Viewed 20 February 2017]. Available from: http://www.rand.org/blog/2017/02/history-and-current-developments-regarding-taiwans.html

Mo, Y., (2012). Taiwan-Japan relations solid, Ma says. *Taipei Times* [online]. 7 October. [Viewed 4 October 2015]. Available from: http://www.taipeitimes.com/News/taiwan/archives/2012/10/07/2003544589/1

Munro, R.H., (1994). Giving Taipei a place at the table. *Foreign Affairs* [online]. **1994** (November/December). [Viewed 22 July 2020]. Available from: https://www.foreignaffairs.com/articles/asia/1994-11-01/giving-taipei-place-table

Naím, M., (2012). Mafia states – Organized crime takes office. *Foreign Affairs*. **91**(3), pp. 100–111.

Naritayakuwarigenjo., (2013). Oupunsukai Ga Tekiyouto Natta Narita Kuukou, Naritakuukou~Sono Yakuwarito Genjyouu (Narita Airport that the open skies applies, Narita Airport – The role and the current state – the year 2013). *Narita Airport Homepage* [online]. [Viewed 30 July 2016]. Available from: http://www.naa.jp/jp/issue/yakuwarigenjyo/2013/pdf/sp1 01.pdf

Navari, C., (2009). Introduction: Methods and methodology in the English School. In: C. Navari. *Theorising international society: English School methods* [online]. Basingstoke, New York: Palgrave Macmillan. pp. 12. [Viewed 22 December 2019]. Available from:

http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.474.2510&rep=rep1&type=pdf

The New York Times., (2016). Singer's apology for waving Taiwan flag stirs backlash of its own. The New York Times [online]. [Viewed 6 May 2019]. Available from: https://www.nytimes.com/2016/01/17/world/asia/taiwan-china-singer-chou-tzu-yu.html

Nikkei Business., (2014). Oupunsukaide Honmonono Ajiarizoutowo Mezasu Okinawa (Okinawa that aims to create the authentic Asian resort via the open skies). *Nikkei Business* [online]. 7 May. [Viewed 29 September 2016]. Available from: http://business.nikkeibp.co.jp/article/report/20140424/263440/?rt=nocnt

Nikkei Shimbun., (2015). Shudanteki Jieikenkoushi, Zensankounin Ga [Iken] Shuiin Kenpoushin (Regarding the collective self-defense, all the attorneys and witnesses regarded it 'unconstitutional': The lower house legislature). *Nikkei Shimbun*[online]. [Viewed 8 January 2017]. Available from: http://www.nikkei.com/article/DGXLASFS04H3U_U5A600C1PP8000/

Nikkei Shimbun., (2016). Taiwan, Gaikou ni chuugoku no kabe ICAO ni sanka deki zu (Taiwan, the wall in front of its diplomacy, [Taiwan] could not attend the general assembly). *Nikkei Shimbun* [online]. 23 September. [Viewed 22 August 2018]. Available from:

https://www.nikkei.com/article/DGXLASGM23H6N T20C16A9FF1000/

Nojima, T., (2015). Taiwan ni Totte No [Kounichi Shouri Nanajyuu Nen] No Imi (What the seventieth anniversary of anti-Japanese war means for Taiwan). *Foresight* [online]. 8 July.[Viewed 8 July 2015]. Available from: http://www.fsight.jp/articles/-/40212

Nojima, T., (2017). Taiwan Saieibun To [Higunjidoumei Wo Musube] Daitokushuu Konokunitowa Koutsukiae (Taiwan create a [non-military alliance] with Tsai-ing wen special coverage: This is how you get along with this country). *Bungeishunjyu*. **95** (January), pp. 292–296.

Nojima. T., (2016). Taiwan Towa Nanika [What is Taiwan?]. Tokyo: Chikuma Shinsho.

Nye, J., (2004). Soft power: The means to success in world politics. New York, NY: Public Affairs.

Ogasawara, N., (2013a). Nittai Kankei Nittai Gyougyoukyoutei Ni Ugoita Baeikyuu No Senryaku [Japan–Taiwan relations: Ma-Ying-Jeou's strategy that instrumentalized the Japan–Taiwan Fisheries Agreement]. *Gaikou* (Diplomacy). 20 (July). pp. 100–105.

Ogasawara, N., (2013b). Senkaku Meguri Jyuuyousei Takamaru Taiwan Kaginigiru Gyogyou Koushou (The Taiwan that increases its importance regarding the Senkakus: The fisheries negotiations occupy the key). Wedge Infinity [online]. 26 March. [Viewed 23 March 2015]. Available from: http://wedge.ismedia.jp/articles/-/2657?page=1

Ogasawara, N., (2013c). Baeikyuu Seikein Wa Senkaku Shotoumondai De Chuugokuto Renkei Shinai (Ma-Ying-Jeou's administration will not collaborate with China on the Senkakus). Ogasawara Homepage [online]. 8 February. [Viewed 17 July 2015]. Available from:

http://www.tufs.ac.jp/ts/personal/ogasawara/analysis/mayingjeoupeaceinitiative3.html

Ogasawara, N., (2014). Baeikyuuno Hakaseronbun Kara Yomitoku Nittaigyogyou Koushou [Tokushuu Haneito Jiritsuno Direnma: Posuto Minshuka Taiwanno Kokusai Seiijikeizaigaku] (Comprehending the Japan—Taiwan fisheries negotiations from Ma-Ying-Jeou's PhD dissertation [Special: The dilemma between prosperity and autonomy: Studies on international political economy in the post-democratization Taiwan]). *Touyoubunka* [Oriental Culture]. **94**, pp. 61–87.

Ogasawara, Y., (2012a). Senkaku Mondai to Nittai Kankei (The Senkaku issue and Japan–Taiwan relations). Ogasawara Homepage [online]. 25 October. [Viewed 23 March 2015]. Available from:

http://www.tufs.ac.jp/ts/personal/ogasawara/analysis/mayingjeoupeaceinitiative2.html

Ogasawara, Y., (2012b). Baeikyuu Soutouno [Higashishinakai Inishiachibu] (President Ma's [East China Sea Peace Initiative]). *Ogasawara Yuko's Homepage* [Online]. 24 September. Updated 10 April 2017. [Viewed 10 April 2017]. Available from: http://www.tufs.ac.jp/ts/personal/ogasawara/analysis/mayingjeoupeaceinitiative.html

Ogasawara, Y., (2015). Ma Ying-jeou's doctoral thesis and its impact on the Japan—Taiwan Fisheries Negotiations. *Journal of Contemporary East Asia Studies*[online]. **4**(2), 67-92. [Viewed 27th June 2017]. Available from: doi: http://www-tandfonline-com.sheffield.idm.oclc.org/doi/pdf/10.1080/24761028.2015.11869085?needAccess=tr ue

Ogata, M., (2013). Taiwan Wo Meguru Anzenhoshou Gunjikouryuu Ni Tsuite (Regarding Taiwan's security and defense exchanges). *Toua*. **557** (November), pp. 18–27.

Ogata, M., (2016). Taiwan No Anzenhoshou Kokubouseisaku Baeikyuu Seikenno Soukatsuto Saieibun Seikenno Tenbou (Taiwan's security policy and national defense policy: The reflection on Ma administration and the outlook on Tsai administration). Toua. 592 (October), pp. 10–21

Oros, A., (2015) International and domestic challenges to Japan's postwar security identity: 'norm constructivism' and Japan's new 'proactive pacifism', *The Pacific Review*, 28:1, 139-160, DOI: 10.1080/09512748.2014.970057

Oros, A., (2017a). Japan's security renaissance: New policies and politics for the twenty-first century [online]. 16 February 2017. *Vimeo*. [Viewed 18 February 2017]. Available from: https://vimeo.com/204545270

Oros, A., (2017b). *Japan's Security Renaissance New Policies and Politics for the Twenty-first Century*. New York: Columbia University Press.

Otsuka, U., (2016). Keynote remarks [oral remarks]: A panel discussion co-organized by The Sasakawa Peace Foundation and Sasakawa USA 'East Asia's Maritime Challenges: Overview of the Current Situation 14 October'. *Sasakawa Peace Foundation* [online]. [Viewed 14 October 2016]. Available from: https://www.spf.org/en/seminar/list/21859.html

Panda, A., (2017). Chinese military aircraft transit Miyako Strait, leading Japan to scramble jets China returns to the Miyako Strait. *The Diplomat* [online]. 3 March. [Viewed 3 March 2017]. Available from: https://thediplomat.com/2017/03/chinese-military-aircraft-transit-miyako-strait-leading-japan-to-scramble-jets/

Pan, J., (2015). Coast guard rescues three from Chinese fishermen. *Taipei Times* [online].14 August. [Viewed 14 August 2015]. Available from: http://www.taipeitimes.com/News/front/archives/2015/08/14/2003625307

Pan, J., (2016). Top aviation police official indicted. *Taipei Times* [online]. 27 July. [Viewed 7 October 2016]. Available from: http://www.taipeitimes.com/News/taiwan/archives/2016/07/27/2003651893

Parameswaran, P., (2015a). Philippines buys 100 new patrol boats for war on poaching. *The Diplomat* [online]. 24 June. [Viewed 24 June 2015]. Available from: http://thediplomat.com/2015/06/philippines-buys-100-new-patrol-boats-for-war-on-poaching/

Parameswaran, P., (2015b). US downgrades Thailand on aviation safety: Civil aviation body goes to Category 2 for the first time in nearly two decades. *The Diplomat* [online]. 2 December. [Viewed 12 May 2017]. Available from: https://thediplomat.com/2015/12/us-downgrades-thailand-on-aviation-safety/

Parameswaran, P., (2017). Japan, Philippines boost defense ties in Duterte-Abe Summit. *The Diplomat* [online]. 16 November. [Viewed 16 November 2017]. Available from: https://thediplomat.com/2017/11/japan-philippines-boost-defense-ties-in-duterte-abe-summit/

Patalano, A., (2015). Beyond the gunboats. *The RUSI Journal* [online]. **160**(2), 32-39. [Viewed 7 November 2018]. Available from: doi: 10.1080/03071847.2015.1031523

Patalano, A., (2016). 'Natural partners' in challenging waters? *The RUSI Journal* [online]. **161**(3), 42-51. [Viewed 28 October 2018]. Available from: doi: 10.1080/03071847.2016.1193356

Patey, L., (2017). Japan's misadventure in South Sudan – How a botched peacekeeping mission is undermining Abe. *Foreign Affairs* [online]. [Viewed 5 October 2018]. Available from: https://www.foreignaffairs.com/articles/south-sudan/2017-08-23/japans-misadventure-south-sudan

Peace Winds America., (2017). Peace Winds continues dialogue with Japan, U.S., and Taiwan on disaster relief. *Peace Winds America*. [online].6 April. [Viewed 13 April 2017]. Available from: http://peacewindsamerica.org/japantaiwandialogue/

Pegg, S. and Berg, K., (2016). Lost and found: the WikiLeaks of de facto state—great power relations. *International Studies Perspectives* [online]. **17**(3), 267–286. [Viewed 11 January 2019]. Available from: https://doiorg.sheffield.idm.oclc.org/10.1111/insp.12078

Pegg, S. and Kolstø, P., (2015). Somaliland: dynamics of internal legitimacy and (lack of) external sovereignty. *Geoforum* [online]. **66** (November 2015), 193–202. [Viewed 29 December 2018]. Available from: doi: https://doi.org/10.1016/j.geoforum.2014.09.001.

Pegg, S., (1998a). *De facto states in the international system*. Institute of International Relations, The University of British Columbia, Working Paper No. 21. Place of publication: The University of British Columbia.

Pegg, S., (1998b). *International society and the de facto state*. Michigan, MI: The University of Michigan Press.

Pegg, S., (2000). The Taiwan of the Balkans? The de facto state option for Kosova. *Southeast European Politics* [online]. **1**(2), 90–100. [Viewed 22 December 2018]. Available from: https://www.researchgate.net/publication/267797599_The_%27Taiwan_of_the_Balkans%27 The De Facto State Option for Kosova

Pehlivantürk, B., (2016). From peace state to peacekeeping state: Japan's changing national role conception and foreign policy norms. *Perceptions* [online]. *21*(1), 63-82. [Viewed 12 October 2018]. Available from: https://search-proquest-com.sheffield.idm.oclc.org/docview/1853880534?accountid=13828

People's Republic of China. Embassy of the People's Republic of China in the Republic of South Africa., (2005). Făn fēnliè guójiā fâ" quánwén (Full text of the Anti-Secession Law). Beijing: Dì shí jiè quánguó rénmín dàibiǎo dàhuì dì sān cì huìyì (The Third Session of the Tenth National People's Congress). [Viewed 11 April 2020]. Available from: http://za.china-embassy.org/chn/zgxw/t187178.htm

Perlez, J., (2013). Calls grow in China to press claim for Okinawa. *The New York Times* [online]. 13 June. [Viewed 14 May 2015]. Available from: http://www.nytimes.com/2013/06/14/world/asia/sentiment-builds-in-china-to-press-claim-for-okinawa.html? r=0

The Philippines. Malacañang Palace., (2011–2016). *National security policy: Securing the gains of democracy*, 8 *August* [online]. Manila: *Gov.ph*. [Viewed 25 March 2017]. Available from: http://www.gov.ph/downloads/2011/08aug/NATIONAL-SECURITY-POLICY-2011-2016.pdf

Po, C. and Pei, Y. (2013). The challenges and implications of the Taiwan-Japan fisheries agreement. *Defense Security Brief.* **3**(3), 12–15.

Pool, D., (1993). Eritrean independence: The legacy of the Derg and the politics of reconstruction. *African Affairs* [online]. **92**(368), 389–402. [Viewed 2 December 2018]. Available from: http://www.jstor.org/stable/723289

Popescu, N., (2007a). Europe's unrecognised neighbours, the EU in Abkhazia and South Ossetia. CEPS Working Document No. 260 [online]. Brussels: Centre for European Policy Studies. [Viewed 13 January 2014]. Available from: http://aei.pitt.edu/7384/2/7384.pdf

The PRC. Embassy of People's Republic of China in the Kingdom of Norway., (2004). White Paper--The One-China Principle and the Taiwan Issue - The Taiwan Affairs Office and The Information Office of the State Council [online]. PRC: Embassy of People's Republic of China in the Kingdom of Norway. [Viewed 21 July 2020]. Available from: https://www.fmprc.gov.cn/ce/ceno/eng/ztxw/twwt/t110655.htm

The PRC. Ministry of National Defense., (2015). *Zhōnghuá rénmín gònghéguó guójiā ānquán fǎ* (The PRC National Security Law, 1 July) [online]. Beijing: The Ministry of National Défense. [Viewed 19 May 2019]. Available from: http://www.mod.gov.cn/auth/2015-07/02/content 4592636 2.htm

The PRC. PRC's Embassy in Japan. (n.d.), Zhù rìběn zŏng lǐngshìguǎn jiǎnjiè (Introduction to the consulate generals in Japan) [online]. PRC Embassy in Japan [Viewed 4 March 2015]. Available from: http://www.chinaembassy.or.jp/chn/sgjss/t62273.htm

Project 2049 Institute., (2016). Going ballistic: The Taiwan Strait crisis at 20 [online]. *YouTube*. [Viewed 18 March 2017]. Available from: https://www.youtube.com/watch?v=5fClSmSaIfA

Qimaxinwen., (2018). Zhōngxīn gǎnxiè zhēnxīn wèiwèn! Ānbèijìnsān tuītè yòng zhōngwén huíyīng càiyīngwén (Sincerely thank you for your sincere condolence! Abeshinzo tweets back to Tsai-Ingwen in Chinese). 10 July. *Qimaxinwen* [online]. [Viewed 10 July 2018]. Available from: https://tw.news.yahoo.com/%E8% A1%B7%E5%BF%83%E6%84%9F%E8% AC%9 D%E7%9C%9F%E5%BF%83%E6%85%B0%E5%95%8F%EF%BC%81-%E5% AE%89%E5%80%8D%E6%99%89%E4%B8%89%E6%8E%A8%E7%89%B9%E7%94%A8%E4%B8%AD%E6%96%87%E5%9B%9E%E6%87%89%E8%94%A1%E8%8B%B1-104741829.html?soc_src=social-sh&soc_trk=fb

Radio Taiwan International Japanese., (2017). Gaikoubu: Nihon Tono Gaikou Kankei Keizoku Wo Kibou (The MOFA hopes to continue the diplomatic relations with Japan)[online]. 6 June. [viewed 6 June 2017]. Available from: http://japanese.rti.org.tw/news/?recordId=69219

Ramzy, A., (2015). Nepal rejects Taiwanese offer of rescue help but accepts medical aid. *The New York Times* [online]. 27 April. [Viewed 1 March 2017]. Available from: https://www.nytimes.com/live/earthquake-katmandu-nepal-updates/nepal-rejects-taiwans-offer-to-help-rescue-efforts-but-accepts-medical-help/

Rappler., (2013). Finally, Japan allows PH to increase direct flights. *Rappler* [online]. 22 September. [Viewed 13 August 2016]. Available from: http://www.rappler.com/business/industries/171-aviation-tourism/38825-japan-allows-philippines-to-increase-direct-flights

Republic of China. Ministry of Foreign Affairs., (2013). *Tairi yutan (Taiwan-Japan fisheries negotiation), The Taiwan-Japan Fisheries Agreement—Embodying the ideals and spirit of the East China Sea Peace Initiative* [online]. ROC: MOFA Online Archive 2013. [Viewed 18 June 2015]. Available from: https://mail.google.com/mail/u/0/#search/glenn+fish/154dc3861b438c6a?projector=1

Republic of China Taiwan. Office of the President., (2020). Inaugural address of ROC 15th-term President Tsai Ing-wen [online]. London: Office of the President. [Viewed 22 July 2020]. Available from: https://english.president.gov.tw/News/6004

Reuters., (2013). China angered as Japan and Taiwan sign fishing agreement. *Reuters* [online]. [Viewed 11 January 2017]. Available from: https://www.reuters.com/article/us-china-Japan—Taiwan-idUSBRE93909520130410

Reuters., (2017). Japanese navy boosts overseas force projection capability with second big helicopter carrier. *Reuters* [online]. 22 March. [Viewed 22 March 2017]. Available from: https://www.reuters.com/article/us-japan-navy-carrier-idUSKBN16T0FJ

Rich, S., (2020). A Survey of public perceptions on diplomatic recognition of Taiwan. *Global Taiwan Institute*, 1-20. [Viewed 11 March 2020]. Available from: http://globaltaiwan.org/wp-content/uploads/2020/03/GTI-A-Survey-of-Public-Perceptions-on-Diplomatic-Recognition-of-Taiwan-Mar-2020-final.pdf

Rinehart, I. and Elias, B., (2015). Congressional Research Service, China's air defense identification done (ADIZ). *USC University of Southern California* [online]. 30 January. [Viewed 8 December 2015]. Available from: http://china.usc.edu/congressional-research-service-china%E2%80%99s-air-defense-identification-zone-adiz-january-30-2015

Ringmar, E., (2014). Recognition and the origins of international society. *Global Discourse* [online]. **4**(4), 446–458. [Viewed 26 December 2018]. Available from: doi: 10.1080/23269995.2014.917031

Rikki, K., (2016). Contextualising Australia–Japan security cooperation: The normative framing of Japanese security policy. *Australian Journal of International Affairs*, 70:1, 6-23, DOI: 10.1080/10357718.2015.1087965

Robson, C., (1993). Real world research: A resource for social scientists and practitioner-researchers. Blackwell, Oxford, UK; Cambridge, Mass., USA.

Robson, S., (2015). Misawa F-16s make emergency landing at Japanese airport. *Stars and Stripes* [online]. 15 April. [Viewed 4 October 2016]. Available from: http://www.stripes.com/news/pacific/misawa-f-16s-make-emergency-landing-at-japanese-airport-1.340416

The ROC. MOFA Taiwan – Department of International Organizations., (n.d.). *Important I.G.O> ICAO> Introduction* [online]. Taipei: MOFA Taiwan. [Viewed 22

March 2019]. Available from: https://csis-prod.s3.amazonaws.com/s3fs-public/legacy_files/files/publication/131121_Glaser_TaiwansQuest_WEB.pdf

The ROC. National Communications Commission., (2010). Taiwan hosts APEC TEL41, 18 member economies discuss ICT development [online]. Taipei: National Communications Commission. [Viewed 21 March 2019]. Available from: https://www.ncc.gov.tw/english/content_field_detail.aspx?site_content_sn=215&is_hi story=1&pages=0&sn_f=29

The ROC. Ocean Affairs Council. Coast Guard Administration., (2017). Table 8-1 The statistics of banning of illegal trespass fishing—by month. *Administration Ocean Affairs Council* [online]. [Accessed 10 May 2019]. Available from: https://www.cga.gov.tw/GipOpen/wSite/public/Attachment/2016StatisticalReport/EY A0801.pdf

The ROC. President Office., (2019). Zŏngtŏng jiēshòu `rìbĕn chǎnyè jīngjì xīnwén shè'zhuānfǎng [President Accepts Japanese Sankei Shinbun's Interview], 2 March 2019 [online]. Taipei: Zhonghua Mingguo Zongtonghe [ROC President Office]. [Viewed 2 March 2019]. Available from: https://www.president.gov.tw/NEWS/24118?fbclid=IwAR0VQHq_bRb_OXQ6OzmPt3on8qP1QR6m5t1PYuOhq6Vahh8t37HDWVUxr2g

The ROC. Schreer, B. at Defense Security Brief Office of Defense Studies Ministry of National Defense., (2014). *The strategic implications of China's air defense identification zone, May* [online]. Taipei: Defense Security Brief Office of Defense Studies Ministry of National Defense. [Viewed 15 January 2015]. Available from: http://www.mnd.gov.tw/UserFiles/File/%E5%9C%8B%E9%98%B2%E6%99%BA%E5%BA%ABPDF%E6%AA%94/Defense%20Security%20Brief%20VOL.4%20ISS UE2.PDF

The ROC. Taipei Economic and Cultural Representative Office, Shen, S., (2012). *Tainichi Kankei No Mirai Ni Mukete (For the future of Taiwan-Japan Relations)* [online]. Tokyo: Taipei Economic and Cultural Representative Office. [Viewed 19 May 2019]. Available from https://www.roc-taiwan.org/jp_ja/post/6663.html

The ROC. Taipei Economic and Cultural Representative Office., (2013). *Taiwan, Japan sign five cooperation pact, 9 November* [online]. Taipei: Taipei Economic and Cultural Representative Office. [Viewed 28 September 2016]. Available from: http://www.roctaiwan.org/us_en/post/3932.html

The ROC Taiwan. Ministry of National Defense., (2014). *The Diaoyutai contretemps and Taiwan's long-term security* [online]. Taipei: Office Defense Studies In the National Defense Think Tank - Mazza, M. [Viewed 5 August 2015]. Available from: https://www.mnd.gov.tw/UserFiles/File/%E5%9C%8B%E9%98%B2%E6%99%BA%E5%BA%ABPDF%E6%AA%94/Defense%20Security%20Brief-Vol3Issue3.PDF

The ROC Taiwan. MOFA., (2013). *The Taiwan-Japan Fisheries Agreement, August 2013* [online]. Taipei: Ministry of Foreign Affairs, Republic of China (Taiwan). [Viewed 18 June 2015]. Available from: https://www.mofa.gov.tw/Upload/WebArchive/979/The%20Taiwan-Japan%20Fisheries%20Agreement%20(illustrated%20pamphlet).PDF

The ROC Taiwan. MOFA., (2013). *The Taiwan-Japan fisheries agreement, August 2013* [online]. Taipei: Ministry of Foreign Affairs. [Viewed 18 June 2015]. Available from: https://cdn.fbsbx.com/hphotos-xpf1/v/t59.2708-

21/11916567_10152891695801853_382868232_n.PDF/The-Taiwan-Japan-Fisheries-Agreement-illustrated-

pamphlet.PDF?oh=9c9b52a2df21a6fa83e28a20df605b2d&oe=55E5EFDB&dl=1

The ROC. Yeh, K. at MOFA ROC Taiwan., (2013). Dr. Kuang-shih Yeh Minister of MOTC: Taiwan seeks meaningful participation in the International Civil Aviation Organization, 24 September [online]. Taipei: MOFA ROC Taiwan. [Viewed 13 August 2016]. Available from:

http://www.mofa.gov.tw/enigo/News_Content.aspx?n=F9C9A875D719DB70&sms=DE4C0FE1DE37E6E0&s=F508CA8B5B203ACF

Rose, G., (2016). Notes from the Chairman – A controversy with Martin Dempsey. *Foreign Affairs* [online]. **95**(5), 1–9. [Viewed 19 October 2016]. Available from: https://www.foreignaffairs.com/interviews/2016-08-01/notes-chairman

Rosemary, F., (2017). Power transitions and great power management: Three decades of China–Japan–US relations, *The Pacific Review* [online]. **30**(6), 829-842. [Viewed 23 January 2019]. Available from: DOI: 10.1080/09512748.2017.1303535

Ross. R.S., (2006). Taiwan's Fading Independence Movement. *Foreign Affairs* [online]. **2006**(March/April). [Viewed 22 July 2020]. Available from: https://www.foreignaffairs.com/articles/asia/2006-03-01/taiwans-fading-independence-movement

Rowen, I., (2014a). A tale of sunflowers and umbrellas. *Thinking Taiwan* [online]. 2 October. [Viewed 28 June 2015] Available from: http://thinking-taiwan.com/tale-of-sunflowers-and-umbrellas/

Rowen, I., (2014b). In search of sunflower seeds in Hong Kong. *Thinking Taiwan* [online]. 25 November. [Viewed 15 August 2015]. Available from: http://thinkingtaiwan.com/in-search-of-sunflower-seeds-in-hong-kong/

Rowen, I., (2014c). Tourism as a territorial strategy: The case of China and Taiwan. *Annals of Tourism Research* [online], **46** 62–74. [Viewed 4 September 2015]. Available from: https://doi.org/10.1016/j.annals.2014.02.006

Reus-Smit C. (2009) Constructivism and the English School. In: Navari C. (eds.) *Theorising international society*. Palgrave Studies in International Relations Series. Palgrave Macmillan, London.

Ryukyu Shimpo., (2013). Nittaigyogyou Kyoutei Sunaoni Yorokobenai Naiyouda (Japan—Taiwan fisheries agreement, this is frankly an unsatisfactory content. Ryukyu Shimpo [online]. [Viewed 10 May 2019]. Available from: https://ryukyushimpo.jp/editorial/prentry-205172.html

Ryukyu Shimpo., (2016). 600 kilo kakuseizai mitsuyu misui, Okinawa keiyu jyoutai kaka?(The attempted smuggling of the illegal drugs, Has the transit in Okinawa been routinised?) Ryukyu Shimpo. 29 June.

Sakashi, R., (2014). Japan–Taiwan relations since 2008: An evolving practical, non-strategic partnership. In: J.P. Cabestan and J. Delisle, eds. *Political changes in Taiwan under Ma Ying-jeou – Partisan conflict, policy choices, external constraints and security challenges.* Oxon: Routledge. pp. 232–246.

Sankei News., (2016). Kaijisensuikan Ga Firipin Kikoue, 15nenburi Minamishinakaide Chuugoku Kensei MSDF (Submarine makes a port call to the Philippines after 15 years, deterring China in the South China Sea). Sankei News [Online]. 16 March. [Viewed 20 March 2017]. Available from: http://www.sankei.com/politics/news/160315/plt1603150051-n1.html

Sankei News., (2017). Abeshinzo Shushouga Boudaisotsugyoushiki De Kunji [Boueiryokuwo Kyoukashi, Yakuwarino Kakudaiwo] (Prime Minister Abe Shinzo made a speech for graduation ceremony at National Defense Academy [strengthening defense capabilities and expanding the roles]). Sankei News [Online]. 19 March. Updated 10 April 2017, 22:44. [Viewed 10 April 2017]. Available from: http://www.sankei.com/politics/news/170319/plt1703190011-n2.html

Sankei Shimbun., (2020). Taiwan. Chin Fukusoutou Intabyuu Shouhou Nihon No Shien Ni Kansha (Taiwan. The Summary of Chen Deputy President Interview, Expressing Gratitude to Japan's Support). Sankei Shimbun [online]. 26 February. [Viewed 26 February 2020]. Available from: https://www.sankei.com/world/news/200226/wor2002260030-n1.html?fbclid=IwAR2vgiLaPuilYrt_VsS_pwM2-c4-O0E_ilNISmL2gX1wDlvNnhEaK9Gb-WQ

Sasaki, T., (2010). China eyes the Japanese military: China's threat perception of Japan since the 1980s. *The China Quarterly* [online]. **203**, 560–580. [Viewed 23 January 2019]. Available from: doi:10.1017/S0305741010000597

Schofield, C. H. and Townsend-Gault, I., (2011). Choppy waters ahead in 'a sea of peace cooperation and friendship'? Slow progress towards the application of maritime joint development to the East China Sea. *Marine Policy* [online]. *35*(1), 25–33. [Viewed 22 June 2015]. Available from: doi: https://doi.org/10.1016/j.marpol.2010.07.004

Schouenborg, L., (2012). Exploring Westphalia's blind spots: exceptionalism meets the English School. *Geopolitics* [online]. **17**(1), 130–152. [Viewed 25 December 2018]. Available from: doi: 10.1080/14650045.2010.523096

Scott, D., (2019). Taiwan's pivot to the Indo-Pacific. *Asia-Pacific Review* [Online]. **26**(1), 29-57. [Viewed 22 July 2020]. Available from doi: 10.1080/13439006.2019.1618602

Security and Strategy Research Institute of Japan., (2019). Nihon Senryaku Taiwan Kounaa (Japan Strategic Dialogue Corner) [online]. *Security and Strategy Research Institute of Japan*. [Viewed 11 March 2020]. Available from: http://www.ssrij.com/Japan—Taiwan%20strategic%20dialogue.html

Sejrup, J., (2012). Instrumentalized history and the motif of repetition in news coverage of Japan—Taiwan relations. *Pacific Affairs* [online]. *85*(4), 745-865. [Viewed 15 December 2018]. Available from: doi: http://dx.doi.org.sheffield.idm.oclc.org/10.5509/2012854745

- Shang-Su,W., (2016). Nuclear disaster in Taiwan: A multidimensional security challenge. *Global Change*, *Peace & Security* [online]. **28**(3), 289–305. [Viewed 25 February 2019]. Available from: doi: 10.1080/14781158.2016.1196660
- Shan S., (2016). Japan travel preclearance could start next summer. *Taipei Times* [online]. 26 August. [Viewed 9 October 2016]. Available from: http://www.taipeitimes.com/News/taiwan/archives/2016/08/16/2003653231
- Sharman, J., (2017). China sends aircraft carrier 'Liaoning' through Taiwan Strait as tensions remain high. *Independent* [online]. 11 January. [Viewed 11 January 2017]. Available from: http://www.independent.co.uk/news/world/asia/china-sends-aircraft-carrier-liaoning-taiwan-straits-tensions-south-china-sea-one-china-a7521366.html
- Sharp, W., (2016). Taiwan's floundering military needs to up its game. *The National Interest*[online]. 10 August. [Viewed 10 March 2017]. Available from: http://nationalinterest.org/blog/the-buzz/taiwans-floundering-military-needs-its-game-17303?page=show
- Shen, L. at Center for Strategic & International Studies., (2015). Taiwan's marginalized role in international security [online]. *Directed by Bonnie Glaser. Washington D.C.: CSIS. YouTube.* [Viewed 1 March 2015]. Available from: https://www.youtube.com/watch?v=k3mPb0n0jTc
- Shih, H., (2013). Taiwan only 'guest' due to China: ICAO. *Taipei Times* [online]. 26 September. [Viewed 13 October 2016]. Available from: http://www.taipeitimes.com/News/front/print/2013/09/26/2003573020
- Shimizu, R., (2009). Keizokuto Henyouno Nakano Nittaaikankei (Amid the continuity and change in Japan–Taiwan relations). In: Kawashima, S., Shimizu, R., Matsuta, Y. and Yang, Y., eds. *Nittaikankeishi 1945–2008* (History of Japan–Taiwan relations 1945–2008). Tokyo: The Tokyo University. pp. 231–241.
- Shimoi, N., (2010). Higashi ajia ni okeru o-punsukai (Open-skies in East Asia). NIRA kenkyu houkokusho. Chapter 5. pp. 76-88.
- Singapore. ICAO., (2011). Security and facilitation ICAO Regional Seminar on MRTDS: Biometrics and border security, 30 November 2 December [online]. Singapore: ICAO Security and Facilitation. [Viewed 30 July 2016]. Available from: http://www.icao.int/security/mrtd/singapore/Pages/default.aspx
- Singh, A., (2016). India's 'undersea wall' in the Eastern Indian oceans. *Asia Maritime Transparency* Initiative [online]. 15 June. [Viewed 1 April 2017]. Available from: https://amti.csis.org/indias-undersea-wall-eastern-indian-ocean/
- Singh, B., (2013a). *Japan's security identity from a peace state to an international state Sheffield Centre For Japanese Studies/Routledge Series*. New York: Routledge.
- Singh, B., (2013b). Security policymaking regime. Japan's security identity From a peace state to an international state. Oxon: Routledge. pp. 132–155.
- Singh, B., (2013c). Territorial conception of national security. In: G. Hook, ed. Japan's

- security identity, from a peace state to an international state. Sheffield Center For Japanese Studies: Routledge Series. pp. 77–96.
- Singh, B., (2013d). Regional and international (global) security. Japan's security identity From a peace state to an international state. Oxon: Routledge. pp. 97–131.
- Singh, B., (2016). Japan embraces internationalism: Explaining Japanese security policy expansion through an identity-regime approach. *Japanese Journal of Political Science* [online]. 17(4), 589-613. [Viewed 28 November 2018]. Available from: doi: http://dx.doi.org.sheffield.idm.oclc.org/10.1017/S1468109916000256
- Smith, M., (2013). Saizensen Toshiteno Okinawa-Nihonkokumin To Gaikou Seisaku (Okinawa as a frontline Japanese nationals and foreign policy). In: Inoue. T., Hatano, S., Sakai, T., Kokubun, R. and Oshiba, R., eds. *Nihon No Gaikou-Nihon Gaikou No Saikouchiku* (Japanese foreign policy The restructuring of Japanese foreign policy). Tokyo: Iwanami Shoten. pp. 235–254.
- Smith, R. D., (2015). *Japan's international fisheries policy: Law, diplomacy and politics governing resource security.* Oxon: Routledge.
- Soeya, Y., (2001). Taiwan in Japan's security considerations. *The China Quarterly* [online]. 165, pp.130-146. [Viewed 12 June 2019]. Available from: doi:10.1017/S0009443901000079
- Soeya, Y., (2005). Changing security and political contexts of Japan–Taiwan relations: A view from Japan. *NBR Analysis Japan–Taiwan Interaction: Implications for the United States* [online]. 1st ed. **16**(1), 38-56. [Viewed 3 July 2017]. Available from: http://www.nbr.org/publications/nbranalysis/pdf/vol16no1.pdf
- Soeya, Y., (2011). A 'normal' middle power: Interpreting changes in Japanese security policy in the 1990s and after. In: Y. Soeya, D. Welch and M. Tadokoro eds. *Japan as a 'normal Country'? A Nation in search of its place in the world*. 1st edn. London: University of Toronto Press. pp. 72–97
- Stivachtis, Y., (2015). Civilizations and global hierarchies: An English School approach. *E-International Relations* [online]. [Viewed 31 January 2019]. Available from: https://www.e-ir.info/2015/05/28/civilization-and-global-hierarchies-an-english-school-approach/
- Stokes, M., (2015). China's air defense identification system: The role of PLA air surveillance. *Arlington: The Project 2049 Institute* [online]. [Viewed 17 Jan 2015]. Available from: http://www.project2049.net/documents/Stokes_China_Air_Defense_Identification_System PLA Air Surveillance.pdf
- Stokes, M., (2017). The United States and future policy options in the Taiwan Strait. *Project 2049* [Online]. [Viewed 20 January 2017]. Available from: http://blog.project2049.net/2017/01/the-united-states-and-future-policy.html
- Strong, M., (2019). Taiwan to arm airport police with submachine guns. *Taiwan News* [online]. 12 March. [Viewed 12 March 2019]. Available from: https://www.taiwannews.com.tw/en/news/3656300?fbclid=IwAR0bxt7tJ-

7LCnYpXkjDawFejzTpcro2FmMXAqF-tYgcztXNR1NQDJg4Gxc

Sutton, D., (2016). International civil aviation organization shuts out Taiwan. *The Diplomat Magazine* [online]. 27 September. [Viewed 28 September 2016]. Available from: http://thediplomat.com/2016/09/international-civil-aviation-organization-shuts-out-

taiwan/?utm_content=buffer32e44&utm_medium=social&utm_source=facebook.com &utm_campaign=buffer

Suzuki, S., (2005). Japan's socialization into Janus-faced European international society. *European Journal of International Relations* [online]. **11**(1), 137–164. [Viewed 25 December 2018]. Available from: doi: 10.1177/1354066105050139

Suzuki, S., (2008). Seeking 'legitimate' great power status in post-Cold War international society: China's and Japan's participation in UNPKO. *International Relations* [online]. **22**(1), 45–63. [Viewed 29 December 2018]. Available from: doi: 10.1177/0047117807087242

Suzuki, S., (2014). Imagining 'Asia': Japan and 'Asian' international society in modern history. In: B. Barry and Y. Zhang, eds.. *Contesting international society in East Asia*. Cambridge: Cambridge University Press. pp. 51-72

Suzuki, S., (2015). The rise of the Chinese 'Other' in Japan's construction of identity: Is China a focal point of Japanese nationalism? *The Pacific Review*, **28**(1), 95-116, DOI: 10.1080/09512748.2014.970049

Suzuki, S., (2019). Japanese revisionists and the 'Korea threat': Insights from ontological security. *Cambridge Review of International Affairs*, **32**(3), 303–321, DOI: 10.1080/09557571.2019.1573806

Swaine, M.D., (2004). Trouble in Taiwan. *Foreign Affairs* [online].**2004**(March/April). [Viewed 22 July 2020]. Available from: https://www.foreignaffairs.com/articles/asia/2004-03-01/trouble-taiwan

Taipei Times., (2012). Gun found in US citizen's bag. *Taipei Times* [online]. 9 July. [Viewed 7 October 2016]. Available from: http://www.taipeitimes.com/News/taiwan/archives/2012/07/09/2003537318

Taipei Times., (2013). Japan 'disappointed' by China no show. *Taipei Times* [online]. 13 March. [Viewed 16 March 2015]. Available from: http://www.taipeitimes.com/News/front/archives/2013/03/13/2003556943

Taipei Times., (2014). Editorial: Ministry of Defense needs a review. *Taipei Times* [online]. 3 October. [Viewed 7 October 2016]. Available from: http://www.taipeitimes.com/News/editorials/print/2014/10/03/2003601130

Taipei Times., (2016a). Beijing's divisive tactics on tourism could backfire. *Taipei Times* [online]. 4 October. [Viewed 4 October 2016]. Available from: http://www.taipeitimes.com/News/taiwan/archives/2016/10/04/200365648

Taipei Times., (2016b). Groundbreaking Call: Tsai, Trump talk of defense, economics. *Taipei Times* [online]. 4 December. [Viewed 4 December 2016] Available from:

http://www.taipeitimes.com/News/front/archives/2016/12/04/2003660530

Taipei Times., (2016c). Japanese councillors support Taiwanese involvement in ICAO. *Taipei Times* [online]. 5 September. [Viewed 30 January 2017]. Available from: http://www.taipeitimes.com/News/biz/archives/2016/09/05/2003654501

Taipei Times., (2016d). Taiwan retains its fishing rights north of Yaeyama. Taipei Times [online]. 5 March. [Viewed 17 April 2016]. Available from http://www.taipeitimes.com/News/front/archives/2016/03/05/2003640841

Taipei Times., (2016e). Taiwanese allegedly targeted in IS list. Taipei Times [online]. 6 October. [Viewed 6 October 2016]. Available from: https://www.taipeitimes.com/News/taiwan/archives/2016/10/06/2003656618

Taipei Times., (2016f). The Liberty Times editorial: China tourism tactics not so costly. *Taipei Times* [online]. 3 July. [Viewed 3 July 2016]. Available from: http://www.taipeitimes.com/News/editorials/print/2016/07/03/2003650216

Taipei Times., (2017). Frank Hsieh says Japanese 'TRA' difficult to pass. *Taipei Times* [online]. 3 November. [Viewed 23 March 2018]. Available from: http://www.taipeitimes.com/News/taiwan/archives/2017/11/03/2003681569

Taiwan. American Institute Taiwan., (2006). Director's introductory call on Minister of Defense *LEE JYE*, *MARCH 31*, 2006 [electronic mail to Central Intelligence Agency, China Beijing, China Chengdu, China Guangzhou, China Hong Kong, China Shanghai, China Shenyang, Defense Intelligence Agency, Japan Tokyo, Joint Chiefs of Staff, Secretary of Defense, Secretary of State, South Korea Seoul, United States Pacific Command]. *Wikileaks* [online]. 6 April. [Viewed 28 March 2019] Available from: https://wikileaks.org/plusd/cables/06TAIPEI1194 a.html

Taiwan American Institute., (2009). Media Reaction: U.S.-China-Taiwan relations. *WikiLeaks* [online]. [Viewed 27 January 2019]. Available from: https://wikileaks.org/plusd/cables/09AITTAIPEI1033_a.html

Taiwan. American Institute in Taiwan., (1979). *Taiwan Relations Act* [online]. Taipei: American Institute in Taiwan. [Viewed 23 January 2015]. Available from: http://www.ait.org.tw/en/taiwan-relations-act.html

Taiwan Info., (2005). Taiwanese fishermen protest Japanese actions in disputed waters. *Taiwan Info* [online]. [Viewed 17 March 2016]. Available from: http://taiwaninfo.nat.gov.tw/ct.asp?xItem=21372&ctNode=103

Taiwan. Interchange Association., (2012). Maneerondaringu Oyobi Teroshikin Kyouyuni Kansuru Kinyuujyouhouno Koukanni Kansuru Kouekizaidan Houjin Kouryuu Kyoukai To Atou Kankei Kyoukai Tonoaidano Oboegaki (The pact on financial information sharing regarding money laundering and terrorism financing between the Public Foundation Interchange Association and Association of East Asian Relations, 25 April) [online]. Taipei: Interchange Association. [Viewed 24 July 2016]. Available

 $https://www.koryu.or.jp/ez3_contents.nsf/8e90bf63626460b049256dfa0026e3b2/e57276a5bd7b7783492579eb0003c093?OpenDocument$

Taiwan Interchange Association., (2014). Kouekizaidan Houjin Kouryuu Kyoukai To Atou Kankei Kyoukai Tonoaidano Shutsunyuukyou Kanribunya Ni Okeru Jyouhouno Koukanto Kyouryokuni Kansuru Oboegaki [Nittai Shutsunyuukyoku Kanrikyouryoku Oboegaki] (The pact on information sharing and cooperation regarding the exit and entry control between the Public Foundation Interchange Association and Association of East Asian Relations [Japan—Taiwan Exit and Entry Control Cooperation Pact], 27 November) [online]. Taipei: Interchange Association. [Viewed 25 July 2016]. Available from:

https://www.koryu.or.jp/ez3_contents.nsf/04/80395B92D77B38F249257D9B003B5C AB?OpenDocument

Taiwan. Japan—Taiwan Exchange Association., (2013). Taiwan Naisei, Nittaikankei Wo Meguru Doukou (2013nen 1gatsu Joujyun-3gatsu Joujyun) Kougika Naikakuno Seiritsuto Daiyonpatsu Kensetsuhantaidemono Jicchi (The development of internal politics in Taiwan, Japan—Taiwan relations [From early January to early March in 2013]: The birth of the Jiang Yi-huah administration and the initiation of opposition demonstration against the fourth nuclear power [online]. Taipei: Ishihara T. Taiwan Jouhoushi Kouryuu. [Viewed 29 March 2019]. Available from: https://www.koryu.or.jp/Portals/0/images/publications/magazine/2013/3/03-07.pdf

Taiwan. Japan—Taiwan Exchange Association., (2015). Taiwan Naisei, Nittaikankei Wo Meguru Doukou (2015nen 7gatsu Joujyun-9gatsu Joujyun) Soutousenkyo No Tenkai, Ritouki Motosoutou No Hounichi The Development on Internal Politics in Taiwan, Japan—Taiwan Relations (From Early July to Early September in 2015) The Development of The Presidential Election, Lee-Teng-Hui The Former President's Visit in Japan, September [online]. Taipei: Ishihara T. Taiwan Jouhoushi Kouryuu. [Viewed 28 March 2019]. Available from: https://www.koryu.or.jp/Portals/0/images/publications/magazine/2015/9/201509_04.p

Taiwan. MOFA Taiwan., (2011). *Taiwan's quest for meaningful participation in the International Civil Aviation Organization* [online]. Taipei: MOFA. [Viewed 15 January 2015]. Available from: http://www.mofa.gov.tw/Upload/RelFile/1061/1940/8d6243ff-e966-4d1a-ba01-8afcc115f7d2.pdf

Taiwan News., (2017). Engagement with India a cure for Taiwan's economic plight. *Taiwan News* [online]. 16 February. [Viewed 18 March 2017]. Available from: http://www.taiwannews.com.tw/en/news/3096681

Taiwan Today., (2013). Taiwan, Japan sign five cooperation pacts. Taiwan Today [online]. 11 June. [Viewed 14 June 2015]. Available from: http://www.taiwantoday.tw/ct.asp?xItem=211247&ctNode=445

Taiwan Today., (2017). Taiwan, Japan sign agreements on customs enforcement, cultural exchanges. *Taiwan Today* [online]. 23 November. [Viewed 29 November 2017]. Available from:

https://taiwantoday.tw/news.php?unit=2,6,10,15,18&post=125413

Taiwan Today., (2019). Indo-Pacific cooperation. Taiwan Today [online]. 8 March. [Viewed 20 March 2019]. Available from:

https://www.taiwantoday.tw/news.php?unit=36&post=151004&unitname=Snapshots&postname=Indo-Pacific-

cooperation&utm_source=Facebook_PicSee&fbclid=IwAR3byaoA770XObQ9P9_9e w05rStEN-npVF_N94kErzxOjEFtkrFzLxQMFA4

Takagi. S., (2006). The Taiwan factor in Japan-China relations. In: Er, L. P., ed. *Japan's relations with China – Facing a rising power*. pp 107–125.

Takeuchi, T., (2012). Minami Shina Kai To Senkaku Shotou Wo Meguru Baeikyuu Seiken No Ugoki(The behaviors vis-à-vis the South China Sea and the Senaku Islands by Ma Ying Jeou Administration). *IDE-JETRO* [online].Oct. [Viewed 1 April 2018]. Available from: https://www.ide.go.jp/Japanese/IDEsquare/Overseas/2012/ROR201216_001.html

Taku, T., (2015). The persistence of reified Asia as reality in Japanese foreign policy narratives. *The Pacific Review*, **28**(1), 23-45, DOI: 10.1080/09512748.2014.970036

Tanaka, Y., (2018). Taichuuanpo [Nittaidetaiwawo] Taiwanno Gaikoubuchouga Ireinoyobikake Tandokuintabyuu (Security vis-à-vis China [Japan—Taiwan dialogue] Taiwan's foreign minister made an unprecedented request). *Sankei Shimbun*[online].27 July [Viewed 27 July 2018]. Available from: https://www.sankei.com/world/news/180627/wor1806270007-n1.html

Tang, A., (2014). The Implications of Taiwan's Declining Defense. *Asia-Pacific Review*. **21**(1), 41-62. [Viewed 3 October 2020]. Available from: doi: https://doi.org/10.1080/13439006.2014.921135

Tang, P., (2016). Taiwan, Beikokuni Gunjijyouhouteikyou...Chuugokutairiku Gunyoukinohikou [Kyoui]de (Taiwan provided military intelligence to the US...due to the [threat] of Chinese mainland's military aircrafts. *Focus Taiwan* [Online]. 14th December. [Viewed 14 December 2016]. Available from: http://japan.cna.com.tw/news/apol/201612140011.aspx

Tatsumi, Y., (2017). Japan self-defense force withdraws from South Sudan: The decision highlights the domestic constraints on Japan's defense activities abroad. *The Diplomat* [Online]. 13 March. [Viewed 13 March 2016]. Available from: http://thediplomat.com/2017/03/japan-self-defense-force-withdraws-from-south-sudan/.

The Netherlands. Press Release at Permanent Court of Arbitration., (2016). *The South China Sea Arbitration (The Republic of the Philippines V. The People's Republic of China)* [online]. The Hague: Press Release at Permanent Court of Arbitration. [Viewed 17 September 2020]. Available from: https://docs.pca-cpa.org/2016/07/PH-CN-20160712-Press-Release-No-11-English.pdf

Thim, M., (2015). Taiwan's supply ship: Panshih, *Strategic Vision* [online]. **4**(21), pp. 18–22. [Viewed 2 March 2017]. Available from: https://issuu.com/strategic_vision/docs/sv21

Thomas, N. and Williams, B., (2017). Taiwan's sub-national government relations with Japan: Post-1979 Developments. *Journal of Contemporary Asia* [online]. **47**(1), 116-141. [Viewed 5 October 2018]. Available from: doi: 10.1080/00472336.2016.1238499

Toal, G. and O'Loughlin, J., (2014). How people in South Ossetia, Abkhazia and Transnistria feel about annexation by Russia. *Washington Post* [online]. 20 March. [Viewed 13 September 2014]. Available from: http://www.washingtonpost.com/blogs/monkey-cage/wp/2014/03/20/how-people-in-south-ossetia-abkhazia-and-transnistria-feel-about-annexation-by-russia/

Torode, G., (2013). China hospital ship to set sail for Philippine typhoon zone. *Reuters* [online]. 21 November. [Viewed 21 November 2013]. Available from: http://www.reuters.com/article/us-philippines-typhoon-china-idUSBRE9AJ04C20131121

Travel and Tour World., (2019). China Airlines to add another flight to Palau to promote tourism. *Travel and Tour World* [online]. 16 April. [Viewed 17 April 2019]. Available from: http://www.travelandtourworld.com/news/article/china-airlines-add-another-flight-palau-promote-

 $tourism/?fbclid=IwAR29vlisQNGJgL7ZJ_olt1jHdyDU6uhz3jRZKe1Mf2-7BH1-kMGL_F8f9VE$

Tsai, T. (2013). Cóng dōng huă hépíng chàngyì dào tái rì yúyè xiédìng: Yīgè [guójìfă-guójì guānxì] de guāndiǎn (From the East China Sea Peace Initiative to the Taiwan-Japan Fisheries Agreement: One international law and international relations' perspective). *Review of Global Politics*. **42**, 15–21.

Tseng, H. S. and Ou, C. H., (2010). Taiwan and China: A unique fisheries relationship. *Marine Policy* [online]. *34*(6), 1156–1162. [Viewed 2 January 2015]. Available from: doi: https://doi.org/10.1016/j.marpol.2010.03.015

Tucker, N. B. & Glaser, B., (2011). Should the United States Abandon Taiwan?. *The Washington Quarterly*. **34**(4), 23-37. [Viewed 2 October 2020]. Available from: doi: 10.1080/0163660X.2011.609128

Twimitv., (2015). Lǐdēnghuī rìběn wàiguó tèpài yuán xiéhuì yǎnjiǎng-jiànlì táiwān zhǔtǐ xìng zhī dào QA (Lee-Teng Hui's speech at the Foreign Correspondents' Club of Japan on establishing Taiwan's subjectivity QA, 23 July 2015) [online]. *YouTube*. [Viewed 25 July 2015]. Available from: https://www.youtube.com/watch?v=O7KhgF-QBv0&list=WL&index=10

Tzou, J.W., (2016). Interview: Kishi urges stronger Taiwan-US-Japan ties. *Taipei Times* [online]. 19 December. [Viewed 16 March 2018]. Available from: http://www.taipeitimes.com/News/taiwan/print/2016/12/19/2003661511

Uchida, K., (2006a). Dainanashou Soutousenkyoto Nihonseihuno [Moushiire] (Chapter 7 The presidential election and Japanese government's [request]). Daijyoubuka Nittaikankei [Taiwantaishi]no Honneroku (Is Japan–Taiwan relations all right? The record of the Honne of the Japanese ambassador in Taiwan). Tokyo Japan: Sankeishinbunshanohon. pp. 171–197

Uchida, K., (2006b). Soutou senkyo to nihon seihu no moushiire (The Japanese government's offer (of disagreement) and the (Taiwanese) presidential election). In Daijoubu ka? Nittai kankei (Is Japan–Taiwan relations alright?). *Sankei Shimbun*, pp. 171–197.

UK Defence Journal., (2016). Russian surveillance aircraft to fly over UK. *UK Defence Journal* [online]. 27 July. [Viewed 27 July 2016]. Available from: https://ukdefencejournal.org.uk/russian-surveillance-aircraft-fly-uk/

UK Reuters., (2015). Angry Chinese families seek answers about missing Malaysian plane. *UK Reuters* [online]. 7 August. [Viewed 7 August 2016]. Available from: http://uk.reuters.com/article/uk-malaysia-airlines-crash-idUKKCN0QC07X20150807

UK. RUSI, (2015). *International Day of United Nations Peacekeepers 13th Annual Conference*, 20 May 2015 [online]. London: Royal United Service Institute. [Viewed 20 May 2015]. Available from: https://rusi.org/event/international-day-united-nations-peacekeepers-13th-annual-conference

United States of America. American Institute Taiwan De Facto U.S. Embassy in Taiwan, (2009). De facto U.S. Embassy in Taiwan, United States of America: Wikileaks. [Viewed 25 March 2017]. Available from: http://cables.mrkva.eu/cable.php?id=222329

United States of America. United Nations Charter., (1945). Action with Respect to Threats to the Peace, Breaches of the Peace, and Acts of Aggression Article 51 Chapter VII 2016, 23 August [online]. New York: Repertory of Practice of United Nations Organs – Codification Division Publications. [Viewed 1 March 2017]. Available from: http://legal.un.org/repertory/art51.shtml

United States of America. US Department of State., (2020). *ICAO's Outrageous Practice of Blocking Twitter Users Who Reference Taiwan* [online]. Washington, D.C.: US Department of State. [Viewed 1 March 2020]. Available from: https://www.state.gov/icaos-outrageous-practice-of-blocking-twitter-users-who-reference-taiwan/

United States., (1949). *The North Atlantic Treaty 1949*, 4th of April [online]. Washington D.C. North Atlantic Treaty Organization. [Viewed 1 March 2017]. Available from: http://www.nato.int/cps/en/natohq/official texts 17120.htm

USA. CSIS., (2015). PacNet #4 — Cybersecurity cooperation with Taiwan: An opportunity for the US-Japan alliance. 14 January [online]. Washington D.C.: CSIS. [Viewed 17 January 2015]. Available from: https://www.csis.org/analysis/pacnet-4-cybersecurity-cooperation-taiwan-opportunity-us-japan-alliance

USA. Kan, S. and Morrison, W. at Congressional Research Service., (2014). *U.S.-Taiwan relationship: Overview of policy issues, 11 December* [online]. Washington D.C.: Congressional Research Service. [Viewed 27 September 2016]. Available from: https://www.fas.org/sgp/crs/row/R41952.pdf

USA. Office of the Press Secretary, President Obama, B. (2013). *Statement by the President on H.R. 1151, 12 July* [online]. Washington DC: The White House. [Viewed 16 August 2016]. Available from: https://www.whitehouse.gov/the-press-office/2013/07/12/statement-president-hr-1151

USA. Pilger, M., U.S. – China Economic and Security Review Commission., (2016). *ADIZ update: enforcement in the East China Sea, prospects for the South China Sea, and implications for the United States, 2 March* [online]. Washington DC: U.S. – China

Economic and Security Review Commission. [Viewed 15 May 2016]. Available from: https://seasresearch.files.wordpress.com/2016/03/adiz-update_0.pdf

USA. US Congress, (2013). *US-China Economic and Security Review Commission* – 2013 Section 2: Taiwan, Annual Report to Congress, 20 November 2013 [online]. Washington: US Congress. [Viewed 26 June 2015]. Available from: http://www.uscc.gov/Annual Reports/2013-annual-report-congress

Uruguay. Seventh International Conference of American States, (1933). *Convention on Rights and Duties of States Montevideo Convention Signed at Montevideo*, 26 *December 1933* [online]. Montevideo: Montevideo Convention on Rights and Duties of States. [Viewed 22 January 2019]. Available from: https://treaties.un.org/doc/Publication/UNTS/LON/Volume%20165/v165.pdf

Vatikiotis, M., (2016). Course correction fighting Islamic extremism in Southeast Asia requires new mindset. *Nikkei Asian Review*. 18–24 November. p. 30–31.

Vosse, W., (2018). Learning multilateral military and political cooperation in the counter-piracy missions: a step towards de-centering of Japan's security policy? *The Pacific Review* [online]. **31**(4),480–497. [Viewed 4 December 2018]. Available from: doi: 10.1080/09512748.2017.1371213

Wakabayashi, M., (2015). Kotonaru Rekishikan Ga Konzaisuru-[Shinnichi] TaiwanNo Shousou (The various situations of [pro-Japanese] Taiwan where different historical interpretations exist). *Gaikou* (Diplomacy). **32**, 50–55.

Wakefield, B.A., (2016). Abe's law: Domestic dimensions of Japan's collective self-defense debate [online]. In: S. Goto, (ed.). *Japan's vision for East Asia – Diplomacy amid geopolitical changes*. Netherlands, Leiden University: Woodrow Wilson International Center for Scholars. pp. 44–60. [Viewed 1 January 2017]. Available from: https://openaccess.leidenuniv.nl/handle/1887/37220

Waltz, K. N., (1979). *Theory of international politics*. Reading, Mass, Addison-Wesley Pub. Co.

Wang, Q., (2000). Taiwan in Japan's relations with China and the United States after the Cold War. *Pacific Affairs* [online]. **73**(3), 353-373. [Viewed 30 September 2018]. Available from: doi: 10.2307/2672024

Wan, M., (2007) Japanese strategic thinking toward Taiwan. In: Rozman G., Togo K., and Ferguson J.P. (eds.) *Japanese strategic thought toward Asia. Strategic thought in Northeast Asia.* Palgrave Macmillan, New York. Available from: https://doi.org/10.1057/9780230603158_7

Wang, C., (2013). Taiwan 'should be a part of security'. *Taipei Times* [online]. 6 February. [Viewed 18 March 2017]. Available from: http://www.taipeitimes.com/News/taiwan/archives/2013/02/06/2003554355

The Washington Times., (2016). Locking Taiwan out of ICAO unwise. Washington Times [online]. 28 September. [Viewed 29 September 2016], Available from: https://www.washingtontimes.com/news/2016/sep/28/letter-to-the-editor-locking-taiwan-out-of-icao-un/

Watanabe, T., (2015). Chutaikankeino Shinkawa Touitsuni Itaruka? Bouchousuru Chuugokuni Taiousemarareru Nichibeitai Toua 10gatugou (Will the integration of Taiwan-China relations lead to unification? Japan-US-Taiwan that faces the need to deal with growing China). *Toua*. **580**(October), 28–36.

Watson, A., (1984). European international society and its expansion. In: H. Bull and A. Watson (eds.). *The expansion of international society*. Michigan: Clarendon Press. pp. 13–32.

Watson, A., (1990). Systems of states. Review of International Studies. 16, 99–109.

Watson, A., (1992). The evolution of international society. London: Routledge.

Wendt, A., (1992). Anarchy is what states make of it: The social construction of power politics. *Cambridge University Press* [online]. **46** (2), 391-425. [Viewed 13 January 2020]. Available from: doi: https://www-cambridge-org.sheffield.idm.oclc.org/core/journals/international-organization/article/anarchy-is-what-states-make-of-it-the-social-construction-of-power-politics/B03BC7C9AAC5211B6DC319C077C1A854

Wendt, A., (1995). Constructing international politics. *The MIT Press* [online]. *International Security*, **20** (1), 71-81. [Viewed 11 January 2020]. Available from: https://www.jstor.org/stable/2539217

Wendt, A., (1999). A social theory of international relations. Cambridge: Cambridge University Press.

Welch, D., (2013). What's an ADIZ? *Foreign Affairs* [online]. 9 December. [Viewed 23 August 2016]. Available from: https://www.foreignaffairs.com/articles/east-asia/2013-12-09/whats-adiz

Wight, M., (1952). British colonial constitutions 1947. Oxford: Clarendon Press.

Wight, M., (1977). Systems of states. Leicester: Continuum International Publishing.

Wilkins, T. S., (2012). Taiwan-Japan relations in an era of uncertainty. *Asia Policy - Project MUSE* [online]. **13**(2012), 113-132. [Viewed 22 July 2020]. Available from: doi:10.1353/asp.2012.0005.

Wilson, P., (2016). English School, neo-neo-style. In: P. Wilson, ed. *The English School in retrospect and prospect: Barry Buzan introduction to the English School of International Relations: The Societal Approach*, September 2013, Warsaw, Poland. London: SAGE Journals, pp. 109–115. [Viewed 2 January 2019]. Available from: doi: 10.1177/0010836715610595

Wirth, C., (2015). Securing Japan in the 'West': The US-Japan alliance and identity politics in the Asian century. *Geopolitics* [online]. **20**(2), 287–307. [Viewed 27 December 2018]. Available from: doi: 10.1080/14650045.2014.999155

Wu, J., (2015). Taiwan looks to boost maritime cooperation with Philippines, Japan. *Reuters* [online]. 6 August. [Viewed 6 August 2015]. Available from: https://www.reuters.com/article/us-taiwan-coastguard/taiwan-looks-to-boost-maritime-cooperation-with-philippines-japan-idUSKCN0QB14Z20150806

- Wu, J. R., (2016). Taiwan aims to build ties with Japan on maritime, regional issues. In: R., Birsel (ed.). *Reuters*. [Viewed 23 December 2017]. Available from: https://af.reuters.com/article/worldNews/idAFKCN127086
- Wu, J. R., (2017). Amid tension, Taiwan says it wants 'new era' of peace with China. Reuters [online]. 20 January. [Viewed 20 January 2017]. Available from: https://in.reuters.com/article/taiwan-china-idINKBN1540TV
- Wu, S-S., (2018). Coast Guard build-up: Taiwan's approach to the South China Sea. *University of Nottingham Taiwan Studies Programme Taiwan Insight* [online]. 16 October. [Viewed 3 January 2019]. Available from: https://taiwaninsight.org/2018/10/16/coast-guard-build-up-taiwans-approach-to-the-south-china-sea/
- Yaeyama Mainichi Shimbun., (2013). Taiwangyosento Jimotogyosen Ga Shoutotsu Torikimego Hatsunojiko (Taiwanese fishing vessels and local fishing vessels clashed, the first after the conclusion of the agreement). Yaeyama Mainichi Shimbun [online]. [Viewed 10 May 2019]. Available from: http://www.y-mainichi.co.jp/news/23274/
- Yang, M., (2009). Dai Hasshou Higashi Ajia No Kouzou Hendou To Nittai Kankeino Saihen (The chapter 8th, the structural shift and the restructuring of Japan—Taiwan Relations). *Nittai Kankeishi (The History of Japan—Taiwan Relations)*. Tokyo: Toukyo Daigaku Shuppan [The Tokyo University Press]. pp. 197–230.
- Yang, M., (2016). Jouhankino Hounichigaikokujinsuu, Shijyoubetsude Taiwangasani Nihyakujiuugoman Kyou (The first half of the year, Taiwan secures the third largest share for foreign visitors in Japan, more than two million fifteen hundred thousand). *Focus Taiwan* [online]. 21 July. [Viewed 21 July 2016]. Available from: http://japan.cna.com.tw/news/atra/201607210002.aspx
- Yang, P., (2005). Japanese-Taiwanese relations and the role of China and the U.S. *NBR Analysis Japan–Taiwan Interaction: Implications for the United States* [online]. 16(1), pp.84–106. [Viewed 1 July 2017]. Available from: http://www.nbr.org/publications/nbranalysis/pdf/vol16no1.pdf
- Yang, Y., (2009). Dainanashou Anzenhoushou No Nijyuuno Sankakukankei (Chapter 7 Tow-layered triangular relations). In: M., Kawashima and Y., Matsuda, eds. *Nittaikankeishi 1945-2008*. Tokyo. Japan: Toukyoudaigakushuppansha. pp. 173–230.
- Yang, Y., (2009). Higashi Ajiano Kouzouhendou To Nittaikankei No Saihen-2000-06nen [The structural transformation in East Asia and the reorganization of Japan—Taiwan relation from the year 2000 to 2006]. In: Kawashima, S., Shimizu, R., Matsuta, Y. and Yang, Y., eds.. *Nittaikankeishi 1945—2008* (History of Japan—Taiwan relations 1945—2008). Tokyo: The Tokyo University. pp. 197—230.
- Yang, Z., (2016). Explaining national identity shift in Taiwan. Journal of Contemporary China, 25(99), 336–352, DOI: 10.1080/10670564.2015.1104866
- Yannish. S., (2015). Civilizations and global hierarchies: An English school approach. *E —International Relations*[online]. 28 May. [Viewed 24 December 2018]. Available from: http://www.e-ir.info/2015/05/28/civilization-and-global-hierarchies-an-english-

- Yasuda, J., (2016). Dairokushou Chuugokuno Kuuikitousei To Saihensei-Taiwanni Totteno Aratana Hidentou Anzenhoshou No Kyoui (Chapter Six: China's airspace control and the reorganization The new non-traditional security threats for Taiwan). *Taiwan Wo Meguru Anzenhoshou Security Around. Keio Institute of East Asian Studies at Keio University*. Tokyo: Keiou Gijyukudaigaku Shuppankai. pp. 181–216.
- Yeh, C., (2015). Lee Teng-hui and the battle for Taiwan's historic identity. *Ketagalan Media* [online]. 24 August. [Viewed 19 August 2015]. Available from: http://www.ketagalanmedia.com/2015/08/24/lee-teng-hui-battle-taiwans-historic-identity/
- Yeh, J., (2017). Taiwan, Japan sign marine emergency and rescue operations MOU. *Focus Taiwan* [online]. 20 December. [Viewed 21 December 2017]. Available from: http://focustaiwan.tw/news/aipl/201712200009.aspx
- Yeh, Y. H., Tseng, H. S., Su, D. T. and Ou, C. H., (2014). Taiwan and Japan: A complex fisheries relationship. *Marine Policy* [online]. *51* 293–301. [Viewed 25 December 2018]. Available from: doi: https://doi.org/10.1016/j.marpol.2014.09.017
- Yeh, Y., Tseng, H., Su, D. and Ou, C., (2015). Taiwan and Japan: A complex fisheries relationship. *Marine Policy* [online]. **51**, 293-301. [Viewed 5 October 2015]. Available from: https://doi.org/10.1016/j.marpol.2014.09.017.
- Yeo, M., (2019). Japan inks deal with Kongsberg for F-35 standoff missile. *Defense News* [online]. 13 March. [Viewed 10 March 2019]. Available from: https://www.defensenews.com/air/2019/03/13/japan-inks-deal-with-kongsberg-for-f-35-standoff-missile/
- Yi-hsuan, L. and Chung, J., (2018). Hualien earthquake: Japanese specialist team arrives in Hualien. *Taipei Times* [online]. 9 February. [Viewed 9 February 2018]. Available from: http://www.taipeitimes.com/News/taiwan/archives/2018/02/09/2003687355
- Yoji, K., (2015). Preserving order in Japan's EEZ: Toward a new paradigm of maritime security. *Nippon* [online]. [Viewed 8 June 2015]. Available from: http://www.nippon.com/en/currents/d00153/
- Yomiuri Shimbun Editorial., (2016). Chutai kankei kinchou Atsuryoku yori taiwa de antei hakari tai (Would like to stabilize the tense China-Taiwan relations via dialogue rather than pressure). Yomiuri Shimbun Editorial. 8 October.
- Yomiuri Shimbun., (2016). Higashi shina kai keibi satou shi ga keikoku (Mr, Sato makes a warning about the policing in the East China Sea). Yomiuri Shimbun. 9 October.p7
- Yomiuri Shimbun Interview., (2016). Chuugoku wa kensetsu teki taiwa wo, wakamono kouryuu no shourei yobikake (China should have a constructive dialogue, calling on the youth exchanges). Yomiuri Shimbun Interview. 7 October.p.6.
- Yomiuri Shimbun., (2016) Nittaikaiyoutaiwa Getsunainimo Saisoutou Gyougyoushigenkyouryokuni Kitai Hoshitono Kaiken (Japan–Taiwan maritime talks

will happen this month: President Tsai expects maritime resource cooperation). *Yomiuri Shimbun* [online]. 6 November.

Zarakol, A., (2010). Ontological (in)security and state denial of historical crimes: Turkey and Japan. *International Relations* [online]. **24**(1), 3–23. [Viewed 29 December 2018]. Available from: doi: 10.1177/0047117809359040

Zielonka, J., (2013). Europe's new civilizing missions: the EU's normative power discourse. *Journal of Political Ideologies* [online]. **18**(1), 35–55. [Viewed 27 December 2018]. Available from: doi: 10.1080/13569317.2013.750172

Zha, D., (2001). The Taiwan problem in Japan-China relations: An irritant or destroyer? *East Asia: An International Quarterly* [online]. **19**(1-2), 205-224. [Viewed 29 September 2018]. Available from: https://search-proquest-com.sheffield.idm.oclc.org/docview/60140711?accountid=13828

Zhang, X., (2011). China in the conception of international society: The English School's engagements with China. *Review of International Studies* [online]. **37**(2), 763–786. [Viewed 29 December 2018]. Available from: doi: 10.1017/S0260210510000422

Zhang, X., (2015). Regional international society in East Asia? A critical investigation. *Global Discourse* [online]. **5**(3), 360–373. [Viewed 29 December 2018]. Available from: doi: 10.1080/23269995.2015.1053190

Zhongyang Tongji She., (2016). ICAO Dàhuì zǔzhǐ rènhé guójiā wèi táiwān fāyán (The ICAO Assembly prevents any country from speaking for Taiwan) [online]. *CNA*. 30 September. [Viewed 7 October 2016]. Available from: http://www.cna.com.tw/news/ahel/201609300084-1.aspx